# Examining just and resilient adaptation in light of the Bacton-Walcott sandscaping scheme



Public information board on the sandscaping scheme overlooking Bacton beach and Bacton Gas Terminal, North Norfolk, Autumn 2022 (Photo: I Cotton)

Isabel Cotton 100006696 A thesis submitted for the degree of Doctor of Philosophy University of East Anglia School of Global Development

June 2024

This copy of the thesis has been supplied on condition that anyone who consults it is understood to recognise that its copyright rests with the author and that use of any information derived there-from must be in accordance with current UK Copyright Law. In addition, any quotation or extract must include full attribution.

### Abstract

England's coastline is highly vulnerable to coastal erosion, and an increasing number of settlements will be at risk as this century progresses. However, the country is underprepared for adapting to long-term coastal change (CCC, 2023). In 2019, the UK's first large-scale sandscaping project was completed on the North Norfolk coast to protect the nationally important Bacton Gas Terminal and nearby villages. While existing research has examined the direct environmental and social impacts of sandscaping, this study addresses a research gap on the broader implications of sandscaping for preparing for long-term coastal change. As a potentially transformative coastal strategy, this research uses an environmental justice lens to examine transformation and resilience, given scale-sensitivities and issues of equity attributed to resilient approaches.

An interdisciplinary, mixed methods approach is adopted to investigate local resident and policymaker perspectives of sandscaping and wider coastal change, using a survey and semi-structured interviews, combined with analysis of geomorphological change. Results reveal that sandscaping has physically transformed the coastal system, by dramatically increasing beach volume and width. Views on the effectiveness of sandscaping amongst local residents are wide-ranging, and coastal adaptation is disputed where it threatens the integrity of settlements, and if unaccompanied with sufficient practical and financial support. Residents have an increased sense of security of future coastal change through the protection afforded by sandscaping, but there is a risk of maladaptation if reduced concern of erosion risk in the future affects willingness to engage in coastal adaptation in the present.

Overall, the research demonstrates incorporating justice principles into coastal adaptation planning is fundamental for successful community engagement, both to overcome historic environmental justice issues and to facilitate community willingness to adapt. Through its interdisciplinary approach, this thesis also reveals that evaluations of nature-based solutions like sandscaping should go beyond geomorphological analysis, to also consider social dimensions.

#### **Access Condition and Agreement**

Each deposit in UEA Digital Repository is protected by copyright and other intellectual property rights, and duplication or sale of all or part of any of the Data Collections is not permitted, except that material may be duplicated by you for your research use or for educational purposes in electronic or print form. You must obtain permission from the copyright holder, usually the author, for any other use. Exceptions only apply where a deposit may be explicitly provided under a stated licence, such as a Creative Commons licence or Open Government licence.

Electronic or print copies may not be offered, whether for sale or otherwise to anyone, unless explicitly stated under a Creative Commons or Open Government license. Unauthorised reproduction, editing or reformatting for resale purposes is explicitly prohibited (except where approved by the copyright holder themselves) and UEA reserves the right to take immediate 'take down' action on behalf of the copyright and/or rights holder if this Access condition of the UEA Digital Repository is breached. Any material in this database has been supplied on the understanding that it is copyright material and that no quotation from the material may be published without proper acknowledgement.

## **List of Contents**

Preliminary material	
Abstract	2
List of Contents	3
List of Figures and Tables	7
List of Acronyms	10
Acknowledgements	11
Chapter 1. Introduction	12
1.1 Managing coastal change in a climate warming world – is transformational	al
adaptation required?	12
1.2 The rise of the 'resilience' narrative	15
1.3 What does 'just adaptation' to coastal change look like?	16
1.4 Trade-offs in coastal management	17
1.5 Study aims and research questions	19
1.6 Thesis outline	23
Chapter 2. Background context	25
2.1 Introduction	25
2.2 Policy overview	25
2.2.1 History and evolution of coastal management	25
2.2.2 Roles and responsibilities	31
2.3 Introduction to mega-nourishment	36
2.3.1 Different coastal management approaches	36
2.3.2 Mega-nourishment: a scaled-up form of beach nourishment	41
2.4 Introduction to case study	45
2.4.1 Geography and demography	45
2.4.2 History of coastal management	51
2.4.3 The Bacton-Walcott sandscaping scheme	58
2.5 Concluding remarks	64
Chapter 3. Literature review	65
3.1 Introduction	65
3.2 Resilience	65
3.2.1 Defining resilience.	66
3.2.2 Resilience in socio-ecological systems	69
3.2.3 Different 'types' of resilience	71
3.2.4 How is resilience gained or reduced?	77
3.2.5 Resilience and uncertainty	80
3.2.6 Criticisms of resilience.	81
3.3 Environmental justice	83
3.3.1 Environmental justice: definitions, origins and development	83
3.3.2 The different themes of environmental justice	85
3.3.3 What causes environmental justice issues?	87
3.3.4 Scales in environmental justice	90
	92
3.4.1 Defining adaptation	92
3.4.2 Evaluating adaptation: adaptation 'effectiveness'	95
3.4.3 Public perceptions of adaptation	96
3.4.4 Risk perceptions and anticipatory adaptation	.100
3.5 Conceptual tramework	.102
3.5.1 Summary of key gaps across chapter and theoretical insights	.102
3.5.2 Conceptual tramework diagram	.105
3.6 Concluding remarks	.106

Chapter 4. Methodology	108
4.1 Introduction	108
4.2 Overview of research design	108
4.2.1 Epistemological and ontological position	109
4.2.2 Mixed methods	111
4.2.3 Case study	114
4.3 Data collection	116
4.3.1 Geomorphological LiDAR data	116
4 3 2 Survey	119
4 3 3 Interviews	124
4.3.4 Summary and timeline of research methods	129
4 4 Data analysis	130
4 4 1 Geomorphological analysis	131
4 4 2 Closed-text survey questions	131
4 4 3 Open-text survey data and interview data	132
1.5 Research ethics and limitations	136
4.5.1 Ethice	136
4.5.1 Limitations	130
4.6. Concluding remarks	1/1
Chapter 5. Understanding perceived (offectiveness) of the Bacton-Walco	44
conduction of the standing perceived enectiveness of the Dacton-Walco	117
5.1 Introduction	1/12
5.2 Background context	142
5.2 Duckyround context	140
5.2.1 Public perceptions of beach nounstiment	143
5.2.2 Public communication on the Dacton-Walcoll Sandscaping	115
5 2 Meterials and Mathada	140
5.3 Materials and Methods.	140
5.3.1 Data collection.	140
5.3.2 Data analysis	140
5.4 A Derechived effectiveness of condecening	149
5.4.1 Perceived effectiveness of sandscaping	149
5.4.2 Geomorphological changes	100
5.4.3 Impacts of the sandscaping scheme	159
5.4.4 Trust in, and communication from, policymakers	162
5.5 Discussion	100
5.5.1 Perceived effectiveness of the sandscaping scheme	166
5.5.2 Trust in coastal management at Bacton and Walcott	170
5.5.3 Social impacts of sandscaping	172
5.5.4 Perceptions/ preferences of 'soft' versus 'hard' engineering	1/4
5.6 Conclusions	175
Chapter 6. Environmental justice in coastal adaptation: insights from	
Happisburgh	178
6.1 Introduction	178
6.2 Background context	179
6.2.1 Managed realignment and coastal adaptation: an environmental justi	ce
perspective	179
6.2.2 Coastwise	182
6.3 Materials and Methods	183
6.3.1 Data collection	183
6.3.2 Data analysis	184
6.4 Results	188
6.4.1 Perceptions of managed realignment	188
6.4.2 Lived experience of coastal change	192
6.4.3 Community consultations and involvement in coastal	
management	200

6.5 Discussion	205
6.5.1 Environmental justice issues of managed realignment	205
6.5.2 Psychological dimensions of living alongside a retreating coast	207
6.5.3 Community involvement in coastal decision-making	209
6.5.4 Communication channels and conflicting narratives.	211
6.6 Conclusions	213
Chapter 7. Challenges to anticipatory coastal adaptation in the use of	
transformative nature-based solutions: local resident and policymaker	
perspectives of a post-sandscaping frontage at Bacton and Walcott	215
7 1 Introduction	215
7 2 Background context	216
7.2.1 Anticipatory adaptation to coastal change and percentions of coastal	
risk	216
7 2 2 The future of the Bacton-Walcott frontage	210
7.3 Materials and Methods	220
7.3 1 Data collection	220
7.2.2 Data collection	220
7.3.2 Data dilalysis	22 I 222
7.4 Results	223
7.4.1 Feeling secure and the future of Dacton Gas Terminal	220
7.4.2 Sediment volume change Bacton to Happisburgh, 2015-2022	229
7.4.3 Perspectives on future managed realignment at Bacton and Walcott	233
7.4.4 Perceived role of local residents in coastal decision-making	.231
	241
7.5.1 Perceived sense of security from coastal change	241
7.5.2 Uncertainty.	244
7.5.3 Community involvement in coastal decision-making	247
7.6 Conclusions	248
Chapter 8. Discussion and synthesis	250
8.1 Introduction.	250
8.2 Revisiting the conceptual framework in light of key empirical insights	251
8.2.1 The influence of environmental change on social perceptions	252
8.2.2 The added complexity of uncertainty	253
8.2.3 Wide-ranging views within and between social groups	254
8.2.4 The relationship between time and resilience	.258
8.2.5 Key barriers to adaptation – information, communication and trust	259
8.2.6 Key barriers to transformation – lack of policy guidance and funding	for
coastal adaptation	262
8.2.7 Revised conceptual framework	265
8.3 Theoretical contributions of this thesis	.268
8.3.1 Transformation, coastal resilience, and maladaptation	.268
8.3.2 Environmental justice	.271
8.4 Thesis methodological contributions	.279
8.5 Limitations and areas of further research	.281
8.5.1 Social impacts of sandscaping and coastal resilience	.282
8.5.2 LiDAR availability and data extent	283
8.5.3 Sequential research methodology, and timing of policy	
developments	.284
8.5.4 Additional areas of future research	284
8.6 Concluding remarks to chapter	.285
Chapter 9. Conclusions	287
9.1 Introduction	287
9.2 Conclusions by research question	287
9.2.1 Research question 1 (Geomorphological changes and implications for	or
physical resilience)	287
· · · · · · · · · · · · · · · · · · ·	

change)	290
change)	
9.2.3 Research question 3 (Incorporating environmental justice into coasta	
adaptation policy)	293
9.3 Empirical insights from the research2	295
9.3.1 National-level empirical insights	296
9.3.2 Local-level empirical insights	298
Appendices	801
Appendix 1. Survey questions and accompanying information	
sheet	301
Appendix 2. Interview questions and information sheet – Bacton and Walcott	
residents	809
Appendix 3. Interview questions and information sheet – Happisburgh	
residents	312
Appendix 4. Interview questions and information sheet - policymakers	315
Appendix 5. Survey results	319
Appendix 6. Coding framework for interview data	332
Appendix 7. Ethics confirmation letter	337
Appendix 8. PhD Summary (for resident interviewees)	338
Appendix 9. PhD Summary (for policymakers)	343
References	348

## List of Figures

Figure 2.1 Timeline of key Acts of Parliament, strategy documents and non- statutory shoreline management plans
Figure 2.2 Summary of coastal management roles and responsibilities at a national and regional/local level
Figure 2.3 Different types of hard and soft engineering approaches to manage coastal risk
Figure 2.4 The initial extent of placed sediment in the Zandmotor scheme43
Figure 2.5. Case study location
Figure 2.6 Mosaic of images of Bacton48
Figure 2.7. Mosaic of images of Walcott49
Figure 2.8. Mosaic of images of Happisburgh50
Figure 2.9 Soft substrate cliffs at Happisburgh
<b>Figure 2.10</b> Existing or redundant hard defences at Bacton, Walcott and Happisburgh
Figure 2.11 Areas vulnerable to coastal erosion at Bacton and Walcott55
Figure 2.12 Area of initial placed sediment and protection of the sandscaping scheme
Figure 2.13 Before and after photos of Bacton Gas Terminal
Figure 3.1 Illustration of two different pathways for recovery
Figure 3.2 Illustration of the linkages between social and ecological processes     within a socio-ecological system
Figure 3.3 Factors that enable or constrain (individual/community) adaptive capacity.   78
Figure 3.4 Trade-offs in coastal management91
Figure 3.5 Factors affecting public perceptions of adaptation responses97
Figure 3.6 Conceptual framework diagram106
<b>Figure 4.1</b> Relationship between the geomorphological and social data in this thesis
Figure 4.2 Sampling strategy and target responses for survey121
Figure 4.3 Sampling strategy and target responses for resident     interviews   127
Figure 4.4 Timeline for research instruments used in this thesis

Figure 4.5 Steps involved in the coding process134
Figure 4.6 Map of data presented in each results chapter135
Figure 5.1 Beach profile elevation changes at Bacton and Walcott: -2m   depth 156
Figure 5.2 Beach profile elevation changes at Bacton and Walcott: -9m   depth 159
Figure 5.3 Respondent responses to survey question on trust in coastal management
Figure 6.1 Clifftop retreat at Happisburgh North and Happisburgh South 2015-     2022.   196
Figure 7.1 Sampling locations for sediment volume calculations222
Figure 7.2 Surface volume (gross, m <sup>3</sup> ) year to year (2015-2022) at Bacton and Walcott, and Happisburgh.   230
Figure 8.1 Proposed modification to conceptual framework – connection     between social and environmental components
<b>Figure 8.2</b> Proposed modification to conceptual framework – tensions and differences in views amongst social groups
Figure 8.3 Proposed modification to conceptual framework – relationship     between time and resilience
Figure 8.4 Proposed modification to conceptual framework – trust, information and communication
<b>Figure 8.5</b> Proposed modification to conceptual framework – lack of policy funding and guidance
Figure 8.6 Revised conceptual framework
Figure 8.7 Expanded scope for evaluating the effectiveness of mega- nourishment

## List of Tables

Table 3.1 Categorisations of resilience within literature	72
Table 4.1 Secondary data sources for LiDAR data	.119
<b>Table 5.1</b> Thematic analysis of survey open-ended questions on perceived effectiveness.	.150
Table 5.2 Thematic analysis of interview data on perceived effectiveness	.151
<b>Table 5.3</b> Reported positive and negative impacts of the sandscaping scheme.	.160

Table 8.1 Justice actions in adapting to coastal change	.275
<b>Table 7.2</b> Surface volume (gross, m <sup>3</sup> ) and percentage change (% against previous year).	.231
Table 7.1 Themes and codes on coastal management post-sandscaping	.224
Table 6.1 Themes and codes on environmental justice	.186
Table 5.4 Child codes relating to the overall theme of trust in policymakers	.164

## List of Acronyms

- ACM Anglian Coastal Monitoring (Programme)
- AOD Aerial Optical Depth
- CCC Committee on Climate Change
- CCAG Coastal Concern Action Group
- CCS Carbon Capture and Storage
- CLIFF Coastal Loss Innovative Funding and Finance
- CTAP Coastal Transition Accelerator Programme
- Defra Department for Environment, Food and Rural Affairs
- DLUHC Department for Levelling Up, Housing and Communities
- EA Environment Agency
- EEA European Environment Agency
- FCERM Flood and Coastal Erosion Risk Management
- GDPR General Data Protection Regulations
- IFCA Inshore Fisheries and Conservation Authority
- IPCC Intergovernmental Panel on Climate Change
- LiDAR Light Detection And Ranging
- MMO Marine Management Organisation
- NNDC North Norfolk District Council
- NSTA North Sea Transition Authority
- RH Royal HaskoningDHV
- SMP Shoreline Management Plan

## Acknowledgements

I would firstly like to thank the survey respondents and interviewees who gave up their time to participate in this research project. And to those who accompanied me on coastal walks of the frontage, shared materials about sandscaping or provided a background on local coastal history. I am indebted to you all in helping me to understand the local coastal context at Bacton, Walcott and Happisburgh.

Secondly I would like to thank my supervisory team – Dr Johanna Forster, Prof Irene Lorenzoni and Dr Trevor Tolhurst – for guiding me through this research and offering invaluable advice.

Thank you to both the SeNSS and ARIES Doctoral Training Programmes for funding this studentship. It was a pleasure to be involved in both training cohorts during my degree.

Many thanks to Dr Sophie Day and Rob Goodliffe, Lucy Goodman and other colleagues at North Norfolk District Council and Coastal Partnership East for all their support throughout the project and insights on coastal issues facing this part of the coast.

Thank you to Royal HaskoningDHV, particularly the help of Ruben Borsje, for facilitating research access to their SHORE LiDAR and bathymetry surveys and monitoring reports. Similarly, thank you to David Huddleston and colleagues at the Environment Agency for their support in accessing LiDAR data from the Anglian Coastal Monitoring Programme.

I am indebted to Dr Katy Appleton and Dr Amii Harwood for the ArcGIS training and helping me get to grips with the software for this project.

All my PhD colleagues – Kang, Gaurav, Matt, Hamida, Katie, Harriet, Fred, Rosie and Connie – thanks for all the coffee chats and keeping me sane with research during the pandemic.

Prof. Kate Sherren, and the rest of the 'lab' – thank you for the invaluable placement opportunity at Dalhousie University.

Mum, Dad, Nick, Seb and Oly, and friends, thank you for supporting and distracting me from PhD life. And to Grandpa, who caught my interest in East Anglia coastal issues all those years ago with the 'our beaches are vanishing' article.

Lastly Jack, from blustery North Norfolk winds to frosty Nova Scotia wind chill, sorry to put you through all the elements, and thank you for always being there.

### Introduction

This first chapter introduces the reader to the key themes of the research, and highlights the importance of the research topic. This is solely as a top-level overview to the study, with Chapters two, three, and four providing more in-depth context to the relevant policy and case study background, academic literature and methodology, respectively. This chapter ends with an outline of the research questions and the remaining chapters of the thesis.

## <u>1.1 Managing coastal change in a climate warming world – is</u> <u>transformational adaptation required?</u>

England's coastal communities face increasingly urgent environmental pressures in the 21<sup>st</sup> century (Committee on Climate Change, CCC, 2018). Coastal flooding and erosion affect over 500,000 properties a year, and cause annual damages of over £250 million (CCC, 2018). Such figures do not account for non-monetary loss such as mental and physical health impacts or disruption to local place for communities (Day, 2020). Further, if one considers the economic impact of loss of ecosystem services, damage to coastal environments costs the UK billions every year (and £15 billion, annually, by 2050) (Johnson et al., 2020a). During extreme weather events, coastal areas may also experience secondary impacts for a continued period after, such as power shortages, closed transport networks, and disrupted public or community services (CCC, 2018). Whilst certain regions are at higher risk, coastal change is a national issue, with the majority of England's coastal counties containing areas vulnerable to coastal flooding or erosion (Sayers et al., 2017).

Disconcertingly, impacts from coastal change are projected to become far more severe as the 21<sup>st</sup> century progresses. The extent of this severity depends in part on the degree to which global mitigation targets, of curtailing global warming to 2°C, are met, of which there remains considerable uncertainty. The UK has experienced consistent sea level rise in recent decades (Woodworth et al., 2009), and the observed 16.5cm increase since 1901 (Kendon et al., 2021) could, under 4°C global warming, rise to 1.12m by 2100 (Haigh et al., 2022). Whilst coastal erosion is a natural process that has always occurred, sea level rise, and more intense and frequent storms, are causing more extensive and accelerated erosion rates (Edwards, 2017). Sea level rise and storms expose coasts to greater wave height (Wong et al., 2014). This means there is a lower threshold with which storm surges will lead to flooding or erosion (Haigh et al., 2022). The UK is therefore likely to experience both more flooding and erosion, but also more severe flooding and erosion. Even if the Paris agreement is upheld and warming is contained to 2°C above pre-industrial levels, delay in the climate system entails that the UK will continue to experience sea level rise and extreme weather events far into the 21st century (CCC, 2018).

The number of properties at risk of flooding in England is expected to rise threefold, from 520,000 in present-day to 1.5 million by 2080 (CCC, 2018). For coastal erosion, there is a ten-fold increase in properties susceptible (residential and nonresidential), from 8,900 today to 100,000 by 2080 (CCC, 2018). These upperestimate projections, based on past erosion rates, are under a 'no active intervention' scenario of no further coastal defences, but do not account for future climate change accelerated risk (Jacobs, 2018). In comparison, analysis by Sayers et al., (2022) reveals that under high warming scenarios (2-4°C), it may no longer be economically viable to continue a 'hold the line' policy (i.e. continue with coastal defences) for nearly a third of England's coastline (affecting 160,000

13

properties). Therefore, the true scale of coastal properties at risk of erosion this century may yet to be realised, and there may be significant political decisions and pressure to change current coastal policy in some coastal areas. Although adaptation to coastal change has always taken place to some degree, it is the accelerated risk and scale of change, exacerbated by climate change, that is so challenging. And whilst coastal flooding affects more properties, the permanence of coastal erosion and the rapidly accelerating risk entail it is ranked as one of the most pressing climate impacts facing England and the rest of the UK (CCC, 2017), and therefore how we adapt to such rising risk requires urgent attention. Given the seriousness of this risk, there are increasing calls within academic literature and global climate policy for 'transformational adaptation' to coastal change, where it may not be possible to retain all coastal settlements and some relocation is required.

Amidst scientific uncertainty of coastal risk, it is also currently unclear as to how the UK government will respond to the increasing risk of coastal change. The CCC, the government's independent advisory body on climate adaptation, assessed England as underprepared in dealing with coastal change, with adaptation measures currently insufficient to address the projected impacts this century, according to the CCC<sup>1</sup>'s (2018) landmark assessment on preparedness to coastal change. A more recent assessment by the CCC (2023) continues to evaluate 'mixed progress' on adaptation in England to coastal erosion risk. As Chapter 2 will explore, there is currently a lack of policy guidance on adaptation to long-term coastal risk.

<sup>&</sup>lt;sup>1</sup> The Climate Change Committee is the independent body advising the government on climate change adaptation and mitigation policy.

#### 1.2 The rise of the 'resilience' narrative

The conceptualisation of adapting to climate change is becoming increasingly intertwined with the idea of 'resilience'. This can be seen at a global level, where the Intergovernmental Panel on Climate Change (IPCC) equates the objective of adaptation as one of increasing resilience (IPCC, 2022). Chapter 3 will explore how resilience can be defined and operationalised, but in essence resilience is about increasing one's capacity to withstand the impacts of a change (Folke et al., 2005). There is policy attention to resilience in numerous areas, not solely climate or coastal management policy: the built environment, mental and physical health, and disaster response sectors all use a framing of resilience (Matyas and Pelling, 2015).

Applying the concept of resilience in a climate or coastal policy context is challenging, because the concept is (amongst other aspects) dynamic, plural, and context-specific (Cutter, 2016; Bene and Doyen, 2018). The widespread use of the concept, in an ambiguous and subjective manner, is also a prominent criticism (Cutter, 2016). Coastal environments are made up of numerous components and feedbacks (species, ecosystems, social groups, networks and institutions), each of which will respond in different ways and on different timescales to a shock or intervention (Virapongse et al., 2016; Maclean et al., 2016). Furthermore, others have argued that changes to resilience is not a singular event, and levels of resilience will vary over time (Matyas and Pelling, 2015; Bene and Doyen, 2018). Whilst the current Flood and Coastal Erosion Risk Management Strategy (hereafter FCERM) states it will take time to increase social and environmental resilience to coastal risks, there is little discussion of what these temporal dynamics mean in practice (Environment Agency, hereafter EA, 2020). While there is ample research on the concept of resilience, Fisher et al., (2018) argue

that further work is needed that examines the complex nature of resilience in relation to time.

A key question raised by the previous section (1.1) is whether increasing coastal risk requires transformative coastal management solutions. As Chapter 3 will explore, different academic fields have broad agreement that transformation is a win-win strategy to build a high level of resilience within a system (e.g. Wong et al., 2014; Matyas and Pelling, 2015; Matin et al., 2018, Bene and Doyen, 2018; Milhorance et al., 2021). Furthermore, the IPCC's sixth assessment report (2022) on climate adaptation argues that a transformative approach has the ability to overcome the limits of other, less ambitious forms of adaptation, whilst also achieving other climate or societal goals. However, such assessments appear based on theoretical assumptions, with little research grounded in an empirical context (Brown, 2014). For example, does fundamental system change (i.e. transformation) always correspond to increased resilience? Could it lead to undesirable change (i.e. maladaptation)? Does whether transformation is forced or planned have a bearing on changes to resilience? These are key questions regarding transformation that will be explored through this research, using a case study of sandscaping on the Norfolk coast. Nature-based solutions are increasingly being used in climate adaptation (IPCC, 2022), and the use of novel nature-based solutions to increase resilience to coastal change is a clear example of where urgent climate risk and policy attention to resilience has led to calls for bigger, more innovative approaches. This research analyses the deployment of an innovative nature-based solution (sandscaping) in order to critically assess the concepts of resilience, transformation, and maladaptation.

#### 1.3 What does 'just adaptation' to coastal change look like?

16

The limitations of pursuing resilience approaches to adaptation have been raised by Leach (2008) and Eakin et al., (2009), who argue they prioritise healthy functioning of ecosystems over potentially disproportionate impacts on some societal groups. In England, some coastal communities are no longer protected by coastal defences, and may have to move in response to increased coastal erosion and flood risk. A coastal management policy switch in 2005 from 'hold the line' to 'managed realignment' in some coastal areas is still disputed to this day, as Chapter 2 will detail as background context, and the empirical chapters (Chapters 6 and 7) will explore. While in England 'resilience' is the framing with which the risk of coastal change is managed, this is not the case in other countries. The most striking example is the Netherlands, where the entire coastline is protected, and risk is controlled rather than an attempt to solely reduce its effects (Kaufmann et al., 2018). The Norfolk coastal villages analysed in this case study have had a changing and contested history of managing coastal change (outlined in Chapter 2). Furthermore, different coastal villages within the case study area, despite sharing similar high risks of erosion, have different levels of protection depending on whether the use of public funds for coastal management schemes can be justified. If only areas with significant populations or infrastructural assets are defended, this raises questions of environmental justice, on where the impacts of coastal change are felt, and how the perspectives of local communities are incorporated into decision-making. Environmental justice is the key analytical lens used by this research to explore the implications of coastal management approaches focussed on building resilience.

#### 1.4 Trade-offs in coastal management

As section 1.1. highlights, given the scale of the physical risk of coastal change that the UK will face this century, there will be difficult decisions in coastal management on where is and is not defended, for how long, and what outcomes are prioritised over others (Tompkins et al., 2008; Eakin et al., 2009). The different, and potentially conflicting choices in environmental decision-making can be described as 'trade-offs', which Foerster et al., 2015 (p.460) define as "*an explicit or implicit prioritisation of one set of values over another. Decisions about which values to prioritise involve complex temporal, spatial and sectoral dimensions*". There is ample work highlighting that adapting to climate impacts involves tradeoffs, such as unintended social, ecological, or economic impacts (Chaigneau et al., 2018; Papadimitriou et al., 2019), knock-on effects for different sectors (Sharifi, 2020) or conflicts between adaptation and mitigation objectives (Sharifi, 2020). Trade-offs manifest not only in the present-day but across different timescales, with future generations likely to experience the more severe impacts of climate change (Meyer, 2017).

As section 1.2 highlights, adaptation strategies that utilise green infrastructure or nature-based solutions are increasingly being adopted for their range of cobenefits, but research has nonetheless highlighted trade-offs in their use (Choi et al., 2021; Loon-Steensma and Vellinga, 2013). Furthermore, the assumption that coastal management strategies designed to serve both social and environmental needs, such as Marine Protected Areas, also avoid negative impacts for people and the environment, is contested (Chaigneau and Brown, 2016). Given the diversity of actors and complexity of issues in coastal settings, understanding the range of impacts of a coastal strategy across scales appears highly relevant. In a UK context, inherent in the decision-making process are choices about what degree of coastal risk is acceptable for society to live with, and what level of finance and resources is deemed appropriate to managing the issue (Tompkins et al., 2008). Conflicts can arise in what outcomes are prioritised, and at what scale (Eakin et al., 2009). For example, Cooper and McKenna (2008) argue it is increasingly difficult to justify allocation of funding for local schemes in the UK, if the decision-making parameters are viewed at a national level, and on longer time horizons.

There are also different approaches to valuing coastal environments, which goes beyond monetary forms of value to consider aspects such as intrinsic, instrumental and relational values (Himes and Muraca, 2018). For example, tangible (e.g. built monuments) and intangible (e.g. sense of place) forms of cultural heritage are not easily monetizable forms of value that can nevertheless be considered by local communities as significant attributes of where they live. As Fortnam et al., (2023) highlights, the impacts of a particular coastal management strategy may not be immediately obvious, and require a just consultation process with relevant stakeholders to identify. These non-monetary, or plural, forms of value may not be considered in decision-making on the costs and benefits of a certain coastal management approach. Where and when coastal impacts materialise, experienced by whom, and where and when resilience is built, are key themes of this research.

#### 1.5 Study aims and research questions

This thesis critically analyses a case study example of a transformative approach to managing and building resilience to coastal change in England – sandscaping, which was introduced on the Norfolk coast in 2019. This PhD uses an environmental justice lens to examine coastal change issues in the case study area, and the impacts of sandscaping. The aim is to critically unpack ideas of resilience, transformation, and maladaptation, by asking 'transformation for who or what, how, and on what scales?'. Resilience could be explored in infinitely different ways or discrete sub units within a system. This PhD focuses on differences between social and geomorphological resilience, and social resilience for different villages with different coastal management contexts. It uses mixed methods of qualitative and quantitative approaches (survey, interviews, and analysis of geomorphological data) to explore social and geomorphological dimensions to coastal change in the case study area. Chapter 4 will outline the methodology in detail.

Chapters 6 and 7 in particular will explore how adaptation to coastal change can be facilitated, and the different perceptions of local residents and policymakers on adaptation under a managed realignment policy scenario. It is important to research the social perceptions of sandscaping and managing coastal change; 1) to identify any unintended side-effects of sandscaping, that can be mitigated in the roll-out of future schemes, 2) to provide learning on how the public can be brought on board and engaged with novel coastal management strategies in their area, and 3) to design just coastal management approaches that are socially acceptable to relevant stakeholders.

A key question lies in how sandscaping might influence perceptions on long-term coastal risk (i.e. beyond the lifetime of the scheme): will sandscaping be repeated, and what is this dependent on? The implementation of sandscaping therefore raises important questions around the longevity of settlements on this stretch of coast. As Chapter 2 will outline, there is existing research analysing the geomorphological and ecological impact of mega-nourishment approaches such as sandscaping, but minimal research on social dimensions; how sandscaping is perceived, its social impacts, and how it might influence perceptions of managing future coastal change, which this PhD focuses on.

Since the use of sandscaping in England (the Bacton-Walcott sandscaping scheme), large-scale projects have been completed elsewhere internationally, including in Shippagan, Canada, in 2020 (Graham et al., 2023), and in Togo and

20

Benin in April 2023 (Dredging Today, 2023). The Crown Estate have identified twenty-two other coastal areas in England where sandscaping could be replicated (Flikweert, 2017), and a feasibility study is currently being conducted for another sandscaping scheme in Norfolk, in Gorleston (Eastern Daily Press, 2022). As of yet, no further schemes have been approved in the UK. Therefore, this research provides empirical findings on the social and geomorphological impacts of sandscaping that can support the design and implementation of future schemes. Research that identifies co-benefits or unintended social impacts (e.g. whether sandscaping increases or decreases resilience to coastal change) may either make it easier to justify funding for future schemes, or inform future schemes on potential social or justice issues associated with sandscaping's implementation.

#### Aims and research questions

The first aim of this research is to investigate the impacts of sandscaping, based on the first 4 years of the Bacton-Walcott sandscaping scheme (i.e. this PhD gathers data from 2019-2022). In particular, it seeks to understand in what contexts sandscaping can be considered a transformational adaptation strategy (for both physical and social dimensions). Geomorphological analysis is undertaken to explore how the coastal system is responding to sandscaping. Covering different case study villages, including defended and undefended coastlines, the second aim of this thesis is to explore public perceptions of coastal change and coastal adaptation, from an environmental justice perspective. The social dimensions considered here are how local residents perceive and are impacted by the sandscaping scheme and coastal change, across time and space. Local residents' perspectives are compared to coastal policymakers, to identify areas of similar or contrasting views that have implications for coastal management policy. This is with a view to considering the third aim of this research, which is to provide insight on how resilience, equity, and justice can be combined in coastal adaptation policy.

To answer these aims, the following research questions have been designed:

- 1. Has sandscaping transformed the physical resilience of the coastal system? Is the scheme working geomorphologically as expected, and what are the observed coastal changes in the first four years?
- a. How have beach profile and sediment volume budgets changed at Bacton and Walcott?
- b. What are current rates of cliff retreat at Happisburgh?
- c. Is there a significant difference in beach volume in the four years before and after sandscaping, and if so, where on the coastline is this the case?
- 2. What are local residents' perceptions and experiences of coastal change, over different timescales, and how do these compare to coastal policymakers, and between villages of differing proximity to sandscaping and differing coastal management plans (e.g. defended or undefended)?
- a. What are the social impacts and perceptions of sandscaping amongst Bacton and Walcott residents? (Chapter 5)
- b. For Happisburgh residents, what are the perceptions and lived experiences of past, present and future coastal change? (Chapter 6)
- c. For Bacton and Walcott residents, how do perceptions of sandscaping affect willingness to adapt and perceptions of coastal change beyond the lifetime of the scheme (ca. 20 years in the future)? (Chapter 7)
- 3. What are the implications of the above findings on incorporating environmental justice into coastal adaptation policy that focuses on building resilience?

#### 1.6 Thesis outline

This thesis is divided into nine chapters (eight subsequent to this introductory chapter). Chapter two provides background context for the research and introduces the case study. An overview is provided of coastal management policy, sandscaping, and the case study villages.

Chapter three contains the literature review, structured around the topics of resilience, environmental justice, and adaptation. Key concepts and research gaps from these topics are used to develop a conceptual framework, summarised at the end of the chapter.

Chapter four provides an overview of the methodology, as a mixed methods, interdisciplinary case study research project. The positionality of the researcher (an individual's many social identities that can be perceived by others; Folkes, 2022), and the ontological and epistemological perspective of the work, is reflected on, before outlining data collection and analysis. Ethical issues and methodological limitations are also explored.

Chapter five is the first results chapter, presenting local residents perceptions and geomorphological observations of the effectiveness of the sandscaping scheme. This chapter answers research question 2a, on the impacts of sandscaping and local resident perceptions on how the scheme has performed in the first few years, as well as future perceptions of sandscaping effectiveness. It also contributes to answering research question 1, where residents' perceptions are compared alongside an analysis of beach profile changes at Bacton and Walcott post-sandscaping.

Chapter six is the second results chapter, and presents empirical findings on the environmental justice issues put forward by residents in different coastal villages, of differing policy contexts, on managing coastal change. This includes Bacton and Walcott residents, on the implementation of the sandscaping scheme, and in answering research question 2b, on the lived experience of Happisburgh residents living alongside an undefended coastline. Residents' perceptions are considered alongside policymakers' perspectives, highlighting areas of agreement and disagreement. This analysis is supported by geomorphological analysis of cliff retreat at Happisburgh, which answers research question 1b.

Chapter seven is the third and final empirical chapter, presenting findings for research question 2c on Bacton and Walcott residents and policymaker perceptions of coastal management after the lifetime of the sandscaping scheme. Topics explored include the future of Bacton gas terminal, perceptions of coastal risk and of managed realignment. Social perspectives are presented alongside sediment volume calculations pre- and post- sandscaping from Bacton Gas Terminal to Happisburgh, tested for statistical significance, which forms the final element of the geomorphological analysis (research questions 1a and 1c).

The empirical chapters are succeeded by a discussion and synthesis chapter, which outlines the main theoretical and methodological contributions to the research, and potential modifications to the conceptual framework, from the crosscutting themes emerging across the three empirical chapters. The chapter ends with a consideration of limitations and potential areas of future research.

The final chapter of the thesis (Chapter nine) is the conclusions chapter. It begins by summarising the key findings of each results chapter in relation to the research questions. Subsequent to this overview, research findings are summarised in relation to relevant insights for practitioners, policymakers, academics and other stakeholders.

### 2. Background context

#### 2.1 Introduction

This chapter introduces the relevant policy background and local context for this thesis. An outline of the history and development of UK coastal management policy is presented first in section 2.2, and includes an overview of key policy actors and funding streams. The advantages and disadvantages of different strategies to managing coastal change, such as hard engineering and soft engineering, are summarised in section 2.3, with a particular focus given to beach nourishment and the Zandmotor scheme in the Netherlands, of which the sandscaping scheme was developed from. Lastly, this chapter introduces the case study area in section 2.4: the villages of Bacton, Walcott and Happisburgh. The history of coastal management in each of the villages is described, before an overview of the Bacton-Walcott sandscaping scheme.

#### 2.2 Policy overview

#### 2.2.1 History and evolution of coastal management

While much of the UK's coastal defences were originally built in the Victorian era (French, 2004), this section focuses on coastal management policies post-1945, given the significant policy changes at this time, for example the Coast Protection Act (1949), which remains pivotal legislation today. The following section is split into two epochs; 1945-2005 and 2005-present, to reflect the significant shift in approach following the Department for Environment, Food and Rural Affairs'

(Defra<sup>2</sup>) (2005) *Making Space For Water* strategy (outlined below). As UK coastal management is devolved, and implemented at a regional level, this section also focuses on policies relevant to England and the North Norfolk coast.

#### 1945-2005: risk-based approach

Today's governance structure for coastal management was set with the passing of the Coast Protection Act (1949) after the Second World War. The Act (ibid) gave local coastal authorities (and risk management authorities) the powers and responsibility to implement coastal management works, where previously this responsibility lay with national government. A few years later in 1953, the UK experienced a devastating flood event that sparked deep conversations on coastal management (Frew, 2012). The 1953 floods resulted in loss of life and infrastructural damage across the country, and it became clear that indefinite coastal protection for the entire UK coastline would be challenging (Frew, 2012). Despite this, the roll-out of coastal defense schemes intensified, contributing to a perception amongst the wider population that the coastline could remain static and unchanging (Nicholls et al., 2013).

Coastal management plans for specific areas began to be developed in the 1980s, but the publication of Shoreline Management Plans (SMPs) in 1996 (EA, 1996) heralded the first coordinated approach to coastal management across different coastal areas in the UK (French, 2004). Despite being non-statutory (not legally binding), SMPs remain the principal policy document that underpins coastal management decision-making in England and Wales (POST, 2021). The plans provide a strategic overview of how different areas of England's coastline are to be managed, in a coordinated way, by setting short (0-20 years), medium (20-50

<sup>&</sup>lt;sup>2</sup> The Department for Environment, Food and Rural Affairs (Defra) is the government department responsible for policy and strategy relating to flooding and coastal erosion, within its broad environment remit.

years) and long term (50-100 years) policy objectives (CCC, 2018). A second, revised version of SMPs was published in 2005, which was later updated in 2012 and 2019 with new policy information (Jacobs, 2019). Figure 2.1 illustrates the release of each SMP version alongside key strategy documents or Acts of Parliament relating to coastal management. There are four overarching policy designations within SMPs: 'hold the line', 'advance the line', 'managed realignment', and 'no active intervention' (POST, 2021). While being well-used, the plans have been criticised for not being statutory policy documents, or not accompanied with practical guidance (O'Riordan et al., 2014), funding or support (Vikolainen et al., 2017) to local authorities.



## Policy timeline

**Figure 2.1** Timeline of key Acts of Parliament (rectangle), strategy documents (circle), and non-statutory shoreline management plans (diamond), alongside the overall evolution from a risk-based (yellow) to a resilience-based (green) approach in England. (Figure: I Cotton)

#### 2005-Present: resilience-based approach

In 2005, the release of Defra's *Making Space for Water* strategy heralded a dramatic policy shift, by concluding that not all of England's coastline could be protected indefinitely from coastal erosion, as to do so would be economically

unviable. There was also increasing evidence that hard defences were exacerbating erosion risk in adjacent areas to which they were implemented (discussed further in section 2.3.1 below). Due to this, many SMPs with a 'hold the line' policy were revised. This was hugely unpopular with many coastal communities, and there is longstanding disagreement in many coastal areas between residents and their local SMP designation (the response within this thesis' case study villages are outlined in section 2.4.2). Making Space for Water was a significant policy change, because the strategy reflects a fundamentally different philosophy for managing coastal erosion risk. Where before the objective was to control risk, Making Space for Water conceded that the risk of coastal erosion could only be reduced. This had huge ramifications for coastal communities, used to a coastline that would forever be protected. Coastal protection is now only afforded to areas of significant strategic importance, infrastructure or assets, or availability of partnership funding. The implication is that some coastal areas subsequently face a risk of flooding and erosion where previously they may have been protected by coastal defences. Policymakers, Stallworthy (2006) argues, have deprioritised the risk of coastal erosion for certain communities since Making Space for Water, on the basis of ever increasing climate risk and cost. Social trade-offs, the author (ibid) argues, have been legitimised.

Consequently, policy documents after 2005 have adopted a resilience, rather than risk-based, approach to coastal management (Townend et al., 2021). A resilience-based approach is framed as one where the objective of coastal management is to increase the ability of coastlines and coastal areas to *respond* to erosion risk (Townend et al., 2021), rather than *reducing* that risk. A resilience-based framing can be seen in the 2010 Flood and Water Management Act, but much more significantly in the more recent 2020 FCERM . However, neither policy document substantiates how resilience to coastal change can be measured (Townend et al.,

2021). In the FCERM strategy, resilience is described as enhancing natural and social capacity to respond to flood or erosion events, and reduce economic damage and social harm (EA, 2020), but discusses property and infrastructure resilience in far greater detail than social resilience of people that live in these areas. This corroborates with earlier evidence submitted from the Local Government Association to the Environment, Food and Rural Affairs Committee's (EFRA) 2019 inquiry on coastal flooding and erosion, arguing that beyond property resilience, social resilience is not yet being fully considered in national policy documents (EFRA Committee, 2019). Soon after the release of the current FCERM strategy (EA, 2020), the government announced £200 million for flood and coastal projects in the Flood and Coastal Innovation Programmes (HM Government, 2023a). All of the announced projects focus on building physical or social resilience to coastal change as their principle aim, and/or trialling coastal adaptation (such as the £36 million Coastal Transition Accelerator Programme (CTAP), detailed below).

At a regional level, coastal management authorities in East Anglia (where the case study of this research is situated) have trialled, and are currently trialling, several coastal management approaches focused on building resilience to coastal erosion. The relevant local authorities are North Norfolk District Council (NNDC), Great Yarmouth Borough Council, and East Suffolk Council, who also work in partnership together on coastal management issues through Coastal Partnership East (2022). NNDC are currently running Coastwise, the principal coastal policy of relevance to this research, which is the North Norfolk project on coastal adaptation funded through CTAP. CTAP is a £36 million programme by Defra, running from 2022-2027, that focuses on planning and preparing for coastal adaptation (Defra, 2022). NNDC is one of two local authorities taking part; the other is East Riding of Yorkshire Council, running the Changing Coasts East

30

Riding project. The two local authorities will trial pilot projects on how at-risk coastal areas can best transition and adapt to rising erosion risk. CTAP is the first time since the 2012 Pathfinder programme (also involving projects in Norfolk) that significant national funding has been allocated specifically to supporting coastal adaptation. Local resident and policymaker perceptions of CTAP are explored in detail in Chapter 6. Other coastal adaptation projects currently underway in East Anglia include other local projects funded through the aforementioned Flood and Coastal Innovation Programme, such as the Resilient Coasts project in Great Yarmouth and Suffolk, trialling the development of adaptation "toolkits" (EA, 2023a, p.1) with local communities. A particular recent focus in East Anglia has been on trialling innovative funding schemes for relocating assets at risk of coastal erosion. This includes CLIFF, or Coastal Loss Innovative Funding and Finance (a local levy), the Local Authority Coastal Adaptation Fund (a funding scheme for atrisk properties), and the Coastal Accumulator Fund (a lifetime fund for properties) (NNDC, 2022a). Therefore, national policy at present is focussed on resiliencebased adaptation, with the North Norfolk coast at the forefront of trialling and testing policy developments.

#### 2.2.2 Roles and responsibilities

While policy for managing coastal flooding and erosion is generally set at a national level, and under the responsibility of the EA (POST, 2021), decision-making is led by local authorities, who approve and manage local coastal projects (NNDC, 2016). To do this, local authorities (be it unitary, district or county level) also work in partnership with utility and public infrastructure providers, but can also more widely work with private and third sector groups, and local community groups (Day et al., 2015). This can also involve partnership groups of different stakeholders, such as the Local Government Association Coast Special Interest

Group, and in the context of the Norfolk coast, Coastal Partnership East and the Norfolk Strategic Flooding Alliance. Figure 2.2 below illustrates the breadth of relevant stakeholders, working at different levels and with different remits, on coastal governance. Whereas the national EA has the responsibility for coordinating overall response (Milligan et al., 2009), it is local councils that principally enact policy (POST, 2021). The precise roles and responsibilities at national and local level have evolved over time. Most recently, after devastating flooding across the UK in summer 2007, the Pitt Review (2008) recommended the EA be responsible for a strategy to manage flooding and coastal change, and risk management authorities be responsible for implementing it. These roles and responsibilities were later enshrined in law through the Flood and Water Management Act (2010).



**Figure 2.2** Summary of coastal management roles and responsibilities at a national (yellow) and regional/local (green) level. Policy design refers to strategy set by the EA, but through this the government department Defra also has a remit (central sphere, yellow). Policy delivery is actioned by local authorities, of which there are three main types of local government (middle sphere, light green). Key stakeholders are wide-ranging and depend on context and location, but can include a combination of the above (outer sphere, dark green, not exhaustive) (Figure: I Cotton).

The differentiation of responsibilities amongst policy actors is less clear-cut for managing coastal erosion specifically, as opposed to coastal flooding. Although officially district and unitary authorities manage coastal erosion and statutory authorities (the EA) manage flood risk, management often overlaps or aligns, and an agreement on roles is established between different actors (Coastal Partnership East, 2019). Furthermore, and more crucially, no single risk management authority has the capacity for *adaptation* to coastal erosion risk, or to compensate communities and businesses whose properties succumb to erosion (Coastal Partnership East, 2019). As of March 2023, 54 Coastal Erosion Assistance Grants have been provided to local authorities nationally for removing at-risk properties (CCC, 2023), but this grant solely covers the demolition costs to remove houses, and does not provide property compensation. This is in contrast to other European countries, for example France and the Netherlands, where financial support and lease-back options are available (Vermaat et al., 2005; Milligan and O'Riordan, 2007). For example, an insurance scheme providing compensation for natural hazard damage is available in France, funding of which is provided by the government and a national levy on property insurance (French Government, 2022). In England, adapting to coastal erosion can be unreported, and the CCC's 2023 progress report could not provide a clear assessment of efforts across the UK to adapt to long-term erosion risk (the overall evaluation from the Committee was 'mixed progress' specifically because there is no standardised adaptation procedure). Therefore in England, the governance context for adapting to coastal erosion is currently ambiguous, and with a vacuum of designated legal responsibility.

Increasingly, as evidenced in the EA's (2020) FCERM strategy, local communities are framed as key actors in managing coastal change. Roles described in the strategy document (EA, ibid) include collaborating on coastal management plans

and future decisions about local areas at risk. In the document, adaptation is framed in terms of providing "support" to high-risk communities to facilitate "transition" and adaptation (EA, ibid, p.13). Adaptation and transition is also described as a "choice" taken by communities, with support from local authorities (EA, ibid, p.47). This language implies considerable agency on the part of communities on decisions about coastal change, such as decisions about relocating. This raises the question, if local authorities and other risk management authorities play a supportive role in the local delivery of coastal adaptation, who plays the leading role? There is therefore a lack of policy clarity on coastal adaptation in terms of how responsibilities overlap, what specifically communities are expected to do, and who has overall responsibility for coastal adaptation (which, as noted above, is no particular actor's responsibility at the moment). To date, there is less research on community perceptions of their role in coastal management, with previous research focusing on the perspectives of local authorities (Esteves and Thomas, 2014; Van der plank et al., 2020). Perspectives on community roles and responsibilities in coastal adaptation are explored in greater detail in Chapter 7.

Asides from roles and responsibilities, key questions remain on how adaptation to coastal erosion will be funded. The aforementioned adaptation projects in section 2.2.1 are all one-off programmes, with no permanent or designated funding stream for coastal adaptation (EFRA Committee, 2019). Comparing this to coastal improvement works or defences, there are multiple funding streams available at both the national and local level. Two UK government departments provide funding: Defra, and the Department for Levelling Up, Housing and Communities (DLUHC, formerly MHCLG). Of this, the biggest funding stream is from Defra to the EA through Grant-in-Aid (over £1 billion in 2020), which provides funding to local authorities or other relevant risk management authorities (Defra, 2021).

34

Alongside this centralized funding source, funding may be available through local taxes and levies, other local council budgets, and from third parties (e.g. partnership funding). With tight local authority budgets, local councils have minimal resources to divert such funding to coastal adaptation, and in 2023 the conservative-run Great Yarmouth Borough Council submitted a motion urging central government for more resources to address coastal adaptation needs (EDP, 2023a). At the time of writing, CTAP is the only central government funding stream for adaptation, which is only in two pilot locations (North Norfolk and East Yorkshire) and from 2022-2027.

In their March 2023 progress report, the CCC (2023, p.192) stated that local communities in coastal areas where protection is no longer viable "*should be supported to relocate*". Throughout the document, the only reference to this support is the need for "*subsidies*" for relocation (CCC, ibid, p.196), and it is not specified whom these subsidies should be for, and for what precisely, with regards to relocation. In practice, different coastal areas have different financial support available to them, because public funding streams such as Defra's Flood Defence Grant-in-Aid to the EA require the use of a cost-benefit ratio<sup>3</sup> to justify funds. This entails national-level public funding is awarded according to a utilitarian logic (the greatest good versus money spent) rather than in areas of high priority or vulnerability<sup>4</sup> (Johnson et al., 2007). The environmental justice implications of this approach to funding allocation determines the amount of national government grant awarded, which is combined with any local partnership funding (Defra,

<sup>&</sup>lt;sup>3</sup> 'Benefits' are understood as the economic cost of future damages avoided by introducing a coastal management or coastal defence scheme (HM Government, 2018).

<sup>&</sup>lt;sup>4</sup> Although there is some evidence of the Treasury taking a vulnerability-based approach to awarding funding, in the way different areas are assessed as 'valuable'. The characteristics of a neighbourhood are not valued 'like-for-like', for example households, and socially vulnerable households, are valued more greatly than businesses or general physical infrastructure (HM Government, 2018).
2021), such as local community groups, local enterprise partnerships, or private finance, to fully fund a scheme (EA, 2020).

To conclude for section 2.2, coastal management policy in the UK has evolved from a risk-based to a resilience-based approach, with significant social implications for coastal areas that will no longer receive funding for coastal defences. There is currently little practical guidance on how resilience in coastal management can be achieved or measured. Furthermore, there is little clarity around fostering and supporting coastal adaptation, with no clear roles, responsibilities and funding sources (other than the current CTAP in Norfolk and East Yorkshire). This includes little practical guidance on how communities are expected to be involved, despite policy rhetoric. Consequently, key questions surrounding the 'what' 'who' and 'how' for coastal adaptation are still very much unanswered. The next section in this chapter will outline different approaches to managing coastal risk, and introduce mega-nourishment, the coastal management approach used in this thesis' case study area.

# 2.3 Introduction to mega-nourishment

# 2.3.1 Different coastal management approaches

In terms of geology, beaches are a type of depositional coastal landform, formed and altered by the movement of sediment through wave and tidal dynamics. Deposition of sediment occurs when these dynamics exist at a low rate. Beaches take varying shapes (i.e. gradients) depending on these dynamics and (sediment) grain size, resulting in dissipative (low gradient) or reflective (high gradient) beaches. However, beach shape will vary considerably across the seasons, with a net migration of sediment onshore during summer, when waves have minimum swell and steepness, and offshore during winter. As a result, a summer beach has a steeper profile than during the winter months. This is significant for erosion risk, which is therefore greater during winter, without the protection of beach height and ample sediment. Sediment is transported to and from beaches, and cross-shore, through the process of longshore drift. A coastal system with a negative sediment budget (i.e. greater sediment material removed than deposited) will result in erosion. The opposite (i.e. a positive sediment budget) will result in accretion, but coastal systems can also be in equilibrium through balanced sediment budgets, with no significant change to the position of the coastline (Leeder, 2011).

Coastal management approaches work in different ways to affect these geological processes, targeting either the dissipation of wave energy and/or reducing tidal flow velocity. Hard defences, such as sea walls and revetments, do both, forming a physical barrier between waves and the land. However hard defences can be problematic for multiple reasons. Reducing cliff erosion in a particular area can affect sediment budgets for adjacent beaches (Ballinger and Dodds, 2020). Thus, this can actually increase the vulnerability to erosion for neighbouring coastlines, through reduced sediment accretion. As well as inhibiting the movement of sediment, hard engineering structures can be detrimental to marine flora and fauna (Hanley et al., 2014). Therefore, hard engineering approaches have been criticised as failing to account for long-term coastal processes, local ecology, wider spatial areas, and future climate risk (Ballinger and Dodds, 2020).

Conversely, the other main approach to coastal management, so-called 'soft engineering', is increasingly being adopted because it works with, rather than inhibits, natural coastal processes (Buser, 2020). As Figure 2.3 illustrates, most coastal management interventions fall into either a hard defence or soft engineering category (although hybrid or green-grey approaches also exist, for example, habitat tiles on sea walls) (Chapman and Blockley, 2009). Soft

engineering approaches, which can also include ecosystem-based approaches or nature-based solutions, are now generally considered more desirable than hard engineering because they are iterative and evolving, responding to coastal stresses rather than being static (Spalding et al., 2013). Where hard defences sought to resist coastal change, nature-based solutions reduce risk by mimicking or working alongside natural coastal processes, and through this, potentially transforming coastal areas into fundamentally different systems (Spalding et al., 2013; Buser, 2020). Nature-based solutions can both minimise the impacts of events, and increase adaptive capacity to respond to future events (Seddon et al., 2021). This is argued to lead to better societal and environmental outcomes (Chausson et al., 2020; Seddon et al., 2021) and can enhance ecosystem services (Vikolainen et al., 2017). In turn, greater ecosystem services can improve coastal processes, and thus coastal resilience, as increased coastal functioning can enhance a system's capacity to absorb shocks such as extreme storm events. While potential drawbacks to nature-based solutions can include requiring a bigger spatial area than certain hard defences, this is not the case for all types, with mega-nourishment requiring no extra (inland) space, and reinforcing a landscape already present (Spalding et al., 2013).



**Figure 2.3** Different types of hard and soft engineering approaches to manage coastal risk (according to literature). Figure 2.3 is not exhaustive and serves as an illustration only of the most common types (Figure: I Cotton).

Commonly deployed coastal nature-based solutions in the UK include wetland restoration (e.g. saltmarshes), and beach nourishment, however there are also active projects for marine (off-shore) nature-based solutions such as sea grass and kelp forests (POST, 2021). Saltmarshes and beach nourishment work by dissipating the energy of waves through vegetation and the enhanced deposition of sediment (in the case of salt marshes) or dunes/increased sediment supply and wider beaches (in the case of beach nourishment) (Moller et al., 2020). Although existing research has indicated saltmarshes may only be effective at certain water depths and wave energies, other studies have shown that wave energy is still dissipated by saltmarshes under high water depth and wave energy scenarios (Moller et al., 2014), suggesting there is still a benefit to saltmarshes in any environment. This provides coastal frontages with protection from waves and storm surges, lowering the risk of flooding or erosion (Hanley et al., 2014). Other

nature-based solutions include dune restoration, common across northern Europe, which induces sediment redistribution across the coast and increases biodiversity (Laporte-Fauret et al., 2021). For offshore nature-based solutions (kelp forests and seagrass meadows), coastal protection is provided indirectly through biological characteristics, where species reduce the impact of waves (Hanley et al., 2014). Overall, soft engineering approaches offer flexible, adaptive coasts, where beach profile varies year to year and seasonally (Hanley et al., 2014).

The use of soft engineering has increased in recent years, with the EA's current FCERM strategy (2020) doubling the funding allocated for nature-based solutions in England and Wales. This contrasts with the aftermath of the 1953 floods, where hard defences dominated (Nicholls et al., 2013) and the roll-out of hard engineering accelerated to maintain the 'hold the line' policy for much of England's coastline (O'Riordan et al., 2014). Therefore the move from a risk-based to resilience-based approach to coastal policy, as outlined in the previous section, has mirrored a move from hard defences to soft engineering approaches. Soft engineering approaches have become increasingly innovative in recent years, with mega-nourishment in Norfolk in 2019 (this thesis' case study, outlined in section 2.4.3) and coastal wetland restoration in Essex in 2018 becoming the largest-scale implementation of their kind in England (RSPB, 2023; Johnson et al., 2020b). Given their novel approach and scale, understanding whether innovative nature-based solutions are effective - or not - is now critical to informing the roll out of similar schemes in the future. The next section outlines the use of beach nourishment and mega-nourishment in greater detail, as the latter is the coastal management approach analysed in this study.

### 2.3.2 Mega-nourishment: a scaled-up form of beach nourishment

First introduced in the UK in the 1960s (Buser, 2020) beach nourishment was initially used to increase the longevity of existing hard defences (Hanson et al., 2002). Also referred to as 'sand replenishment' and 'beach fill' (de Schipper et al., 2021), early projects on the south coast redistributed shingle sediment between different areas of coastline, with the first dredging of sand taking place in 1972 (Hanson et al., 2002). Existing schemes in England include the Lincshore project in Lincolnshire, which began in 1994 (EA, 2019). Beach nourishment is the regular placement of additional sand on vulnerable beaches to increase beach volume. Sand is replenished typically every few years (Stive et al., 2013), or when monitoring indicates beach volume has fallen below a certain threshold (Walkden et al., 2015). Unlike hard defences, beach nourishment does not inhibit, and rather supports, the coastal processes that allow transportation of sediment (de Schipper et al., 2021), with sediment migrating updrift and downdrift from the location of its original placement.

While offering relatively fewer impacts on coastal environments than other approaches (Moreno and Munoz-Perez, 2021), the success of beach nourishment is sensitive to choice of grain size (Hanley et al., 2014). Too coarse (or too fine) grain size may be ineffective, and cause marine biological impacts (Hanley et al., 2014; Parkinson and Ogurcak, 2018), for example a decline in habitat suitability and abundance of some species as observed by Peterson et al., (2006). Further, beach nourishment schemes have been criticised for their overall cost, high sediment supply requirements (Parkinson and Ogurcak, 2018) and for considering too narrow spatial scales in their assessments (Lazarus et al., 2011), by failing to consider the effect of nourishment on neighbouring beaches and the broader coastal system (Lazarus et al., 2011; Gopalakrishnan et al., 2018). Parkinson and

Ogurcak (ibid) also criticise the use of beach nourishment as only delaying inevitable forced retreat for at-risk areas, and is therefore not a long-term solution. Over the last decade, a significantly scaled-up type of beach nourishment has been attempted to manage coastal change, known as 'one-off' or 'meganourishment'. The Zandmotor project in the Netherlands, which deposited 21.5 million m<sup>3</sup> sand near the Hague in 2011, is the first mega-nourishment approach implemented worldwide (Bontje and Slinger, 2017; Climate ADAPT, 2019), and inspired the design of mega-nourishment projects at Bacton and Walcott in the UK (the case study of this research) and Togo and Benin (The Maritime Executive, 2022). Beach nourishment has regularly occurred across the Dutch coast since the 1990s, but the Zandmotor scheme is a dramatic and innovative expansion of this process on a much larger scale (Mulder and Tonnon, 2010). For example, the volume of sediment used in the Zandmotor scheme (21.5 M m<sup>3</sup>) is greater than that used across the entire Dutch coast for beach nourishment annually (Bontje and Slinger, 2017), and corresponds to placing 10,000 m<sup>3</sup> sand for every metre of beach (Mulder and Tonnon, 2010). The difference between beach nourishment and mega-nourishment is not precisely defined, but considered to be a placement of greater than 500m<sup>3</sup> sediment per metre (de Schipper et al., 2021). In meganourishment, sediment is distributed over a much smaller lateral area, so that the beach extends further into the sea than traditional beach nourishment (Stive et al., 2013) (see Figure 2.4).



**Figure 2.4** The initial extent of placed sediment in the Zandmotor scheme © Rijkswaterstaat / Joop Van Houdt, subsequently reprinted in Stive et al., (2013).

Mega-nourishment is considered to have many advantages over smaller-scale beach nourishment. While both processes use the same mechanism (i.e. gradual redistribution of sediment over time through natural coastal processes), the bigger size and scale of mega-nourishments means it lasts much longer (up to 20 years, compared to a few years for beach nourishment (Stive et al., 2013)). Crucially, unlike beach nourishment, mega-nourishments do not require regular intervention (de Schipper et al., 2021). For this reason, they can be more cost-effective (Slobodan and Walvin, 2015), quicker to implement (Moreno and Muñoz-Perez (2021), and therefore more viable in contexts where beach nourishment is unfeasible (Brown et al., 2016). Environmental impacts can be minimised more quickly, whereas regular sand placement may exacerbate impacts because of an absence of recovery period for coastal systems (Moreno and Muñoz-Perez, 2021). The Zandmotor scheme is also expected to offer wider ecological benefits, such as supporting dune formation and new ecological communities (van Bergen and Nijhuis, 2020).

Research thus far on mega-nourishment schemes (principally the Dutch Zandmotor scheme, given it was the first scheme of its kind in 2011), has focused

on analysing how the coastal environment is responding geomorphologically or ecologically. This has been widely studied for the Zandmotor scheme, for example De Schipper et al., (2016) on movement of placed sediment; Hoonhout and de Vries, (2017) on the dynamics of wind-blown sand and dune formation; Luijendijk et al., (2017) on factors affecting placed sediment movement, including wave dynamics, and Roest et al., (2021) on a comparison of erosion and accretion and different beach depths. The last sub-section of this chapter (2.4.3) explores initial findings of the geomorphological development of the Zandmotor scheme in further detail, comparing this to the modelled expectations of the Bacton-Walcott sandscaping scheme. There has also been research on the ecological implications of the Zandmotor scheme (for example Van Puijenbroeck, 2019), in terms of impacts to the number of species living in the intertidal zone (Herman, 2019), the impact of placed sediment on juvenile fish nursery areas (Post, 2019), and the suitability for dune plant species (Pit et al., 2020). The Zandmotor scheme has also been used as a case study in technological developments for remote scanning and field measurements (Vandebroek et al., 2017).

What is comparatively lacking is research into the social impacts of meganourishment. One of the few existing studies from a social perspective has focussed on exploring mega-nourishment's unique (i.e. multi-stakeholder) governance arrangements (Bontje and Slinger, 2017; Flikweert, 2017; Vikolainen et al., 2017). Furthermore, the five year monitoring report of the Zandmotor scheme (Taal et al., 2016) solely assessed social impacts in terms of recreational changes (through public surveys), such as implications of geomorphological change for swimming (Radermacher, 2018) but this does not draw on social perspectives from beach users themselves. Similarly, the 10-year evaluation report of the Zandmotor scheme continued a geomorphological and ecological focus, with social dimensions limited to exploring recreational changes for local

44

users (Huisman et al., 2021). More recent research on mega-nourishment (Day, 2020; Day et al., 2023; Vreugdenhil and Slinger, 2023) has examined social perceptions, focusing on the social impacts and added value of the scheme. Social perceptions of mega-nourishment have therefore to date been explored specifically in the context of recreational preferences or social impacts, with unexplored dimensions on public acceptability, perceived effectiveness, or the impact of mega-nourishment on perceptions of wider coastal change. This mirrors a gap in the literature where there is also a lack of research on public perceptions of smaller-scale beach nourishment. As explored in greater detail in Chapter 5, while public perceptions of beach nourishment have been well studied (for example Lozoya et al., 2014; Prati et al., 2016; Cabezas-Rabadán et al., 2019), research with policy-makers, stakeholder organizations (Ariza et al., 2014; Bontje et al., 2019), and recreational beach users (Cabezaz-Rabadán et al., 2019; Usher, 2021) is more extensive compared with local residents. Therefore, more needs to be known about the wider social impacts of mega-nourishment, and how it is perceived by the public. The final section of this chapter introduces the case study of this thesis: the villages of Bacton, Walcott and Happisburgh, and the Bacton-Walcott sandscaping scheme.

# 2.4 Introduction to case study

# 2.4.1 Geography and demography

The case study area for this thesis is the villages of Bacton, Walcott and Happisburgh, on the North Norfolk coast of England. Bacton, along with the adjacent villages of Walcott and Happisburgh, are situated south of Cromer and approximately 20 miles north-east of Norwich (see Figure 2.5). Bacton Gas Terminal lies to the west of Bacton village, and has been operating since 1969 (Shell, 2021). The terminal is a significant piece of energy infrastructure, responsible for a third of the UK's gas consumption (Shell, 2021).



**Figure 2.5** Case study location on the East coast of England (starred in map a and b). The area sits within SMP 6 (Kelling Hard to Lowestoft Ness), which is denoted with a blue line in map b. The three case study villages are underlined in red in map  $c \otimes$  Edina Digimap

Bacton is the largest of the three villages, with an estimated population size of 1,194 residents, followed by Happisburgh (889) and Walcott (548) (ONS, 2011<sup>5</sup>). All three villages are sparsely populated (in the lowest 20 percentile nationally), with a population density below 1,000 people per square kilometre (ONS, 2021), The villages, like many other parts of Norfolk, are rural. The majority of land use in North Norfolk (72%) is agricultural, with only 5% land taken up for developed use (MHCLG, 2020).Tourism and agriculture are the main sources of revenue to the local area, with several holiday parks, hotels, and holiday cottages across the three villages. 14% of properties in Bacton are classified as second homes or holiday lets, a figure that has increased year on year since 2017 (NNDC, 2021).

The villages are ranked in the 3<sup>rd</sup> (Walcott and Happisburgh) and 4<sup>th</sup> (Bacton) most deprived deciles in England's Indices of Multiple Deprivation, with low crime rates (10<sup>th</sup> decile), average income and access to healthcare (5<sup>th</sup> decile), but low access to housing and other services (1<sup>st</sup> decile) (MHCLG, 2019). The villages have higher than average older populations, with 36% of residents in the surrounding Bacton area, and 27% of residents in the surrounding Walcott and Happisburgh area, aged 65 and above (ONS, 2021). Correspondingly, 74-85% of residents are not in employment (nor for the last 12 months) (ONS, 2021). Figures 2.6-2.8 overleaf show a series of images of the three villages.

<sup>&</sup>lt;sup>5</sup> Due to the way in which Lower Super Output Areas are presented in the 2021 ONS dataset, it is difficult to obtain population estimates solely for each village, and therefore update population estimates beyond the ONS (2011) census.

# Bacton



Figure 2.6 Mosaic of images of Bacton (Photos: I Cotton)

# Walcott



Figure 2.7 Mosaic of images of Walcott (Photos: I Cotton)





Figure 2.8 Mosaic of images of Happisburgh (Photos: I Cotton)

### 2.4.2 History of coastal management

The coastline from Bacton to Happisburgh is highly physically vulnerable to coastal erosion and extreme storm events. Unconsolidated cliffs, comprised of rock types such as clay and fine sands (Royal HaskoningDHV (RH), 2017), are highly susceptible to erosion, and have been in retreat for several thousand years (Wingfield and Evans, 1998), in part due to the East of England subsiding (Nicholls et al., 2013). See Figure 2.9 for images of the cliffs at Happisburgh. The Norfolk coast has seen one of the biggest UK rates of sea level rise in the late 20<sup>th</sup> century (Woodworth et al., 2009). Between 1885-1968, the long-term average cliff erosion rate at Bacton was estimated to be 0.52 metres per hectare (Brooks and Spencer, 2019). In the 1960s erosion rates slowed, due to the installation of local hard defences (Brooks and Spencer, 2019; Payo et al., 2020). Examples of historical hard defences (shown in Figure 2.10) includes timber groynes, revetments, and concrete walls (at Bacton and Walcott) (RH, 2018a), and revetments and rock armour (at Happisburgh) (Payo et al., 2020). Hard defences at Happisburgh failed in the 1990s, some of which were subsequently removed, which further accelerated erosion rates (Payo et al., 2020). As outlined in the introduction (section 1.1), environmental change is also intensifying due to the consequences of climate change (sea level rise and more frequent and intense storms). Consequently, erosion rates in the villages in the 21<sup>st</sup> century exceed historical averages, and were calculated to be 4m yr<sup>-1</sup> between 2013-2018 at Bacton and Walcott (Rumson et al., 2019) and, in some locations, 8-10m yr<sup>-1</sup> at Happisburgh (at the turn of the century, Payo et al., 2020).

Coastal erosion manifests in significant and permanent impacts for the local communities in each village. Farmland and residential buildings have frequently been lost in the past at Happisburgh (BGS, 2021), and 35 residential properties have been lost, as of 2019, due to erosion (EFRA Committee, 2019). A caravan

park was relocated away from the cliff edge in 2012 (Frew, 2012). All three villages were significantly affected by the 1953 and 2013 floods (Mott MacDonald, 2016; BGS, 2021), and during the winter 2013 storm alone, the cliffs at Bacton retreated 5-10m (Vikolainen et al., 2017).



**Figure 2.9** Soft substrate cliffs at Happisburgh (Figure a and b). Cliff vulnerability to erosion at Happisburgh is also exacerbated by surface runoff (observable in Figure b) (Photos: I Cotton).



**Figure 2.10** Existing or redundant hard defences at Bacton, Walcott and Happisburgh. This includes damaged structures (a, Happisburgh) rock armour (b, Happisburgh) revetments (c, Bacton, now partially submerged by sediment from sandscaping), and groynes (d, Walcott, now partially submerged by sediment from sandscaping) (Photos: I Cotton).

Bacton, Walcott, and Happisburgh are part of SMP 6, which encompasses Kelling Hard to Lowestoft Ness (AECOM, 2012a) (see map b, Figure 2.5). Contrary to the first iteration, the second SMP designates the coastline from Bacton Gas Terminal to Happisburgh as areas where continual protection to 2100 is not feasible. If hard defences continue to be strengthened, the SMP argues, areas past Winterton (south of Happisburgh) will become more vulnerable to coastal erosion. Several studies have highlighted the benefit of such an SMP approach to sediment budgets for the coastline as a whole. Hall et al's. (2015) modelling study demonstrates the benefit of allowing cliffs from Sheringham to Happisburgh to erode naturally, feeding low-lying cliffs past Happisburgh to Winterton, increasing beach volume and providing flood protection to the coast and the Norfolk Broads. Meanwhile, the implementation of the first phase breakwaters at Sea Palling in Norfolk was modelled by Thomalla and Vincent (2003) as responsible, through the creation of tidal tombolos (narrow stretches of sediment or spits), for cutting off the distribution of sand at Winterton. Consequently SMP 6 recommends a 'hold the line' policy for Bacton Gas Terminal, managed realignment from present day at Happisburgh, and from 2025 at Bacton and Walcott. Figure 2.11 illustrates the areas of Bacton and Walcott vulnerable to erosion (in red or orange shading), at different periods into the future. SMP 6 recommends maintaining defences no longer than is required to develop necessary plans for relocation or other options, expected to be required towards the end of this century (AECOM, 2012a).



**Figure 2.11** Areas vulnerable to coastal erosion at Bacton and Walcott, under a 'hold the line' scenario to 2025, and managed realignment scenario thereafter (annotated from Mott MacDonald, 2014) Orange shading indicates areas susceptible in 20-50 years, red shading 50-100 years.

The change from 'hold the line' to 'managed realignment', following the publication of the second Kelling Hard to Lowestoft Ness SMP in 2005, was met with widespread dismay by local communities (Milligan et al., 2009). A public consultation on the draft document received 99.6% objections (of approximately 2,400 local residents that responded) (CCAG, 2023a). 'Managed realignment' or 'managed retreat' refers to the deliberate process of allowing the shoreline to move inland, for areas at risk of erosion or flooding (AECOM, 2012a). It ensures natural processes control the rate of coastal change, without intervention or defences that may slow the process. Given the loss of land and property associated with the policy, it is a contentious SMP policy designation. As section 2.2.1 outlines, the change to managing coastal risk at Bacton, Walcott and Happisburgh under a 'managed realignment' policy is therefore an abrupt, opposing approach to the previous 'hold the line' policy, and the continual protection communities had for decades (Nicholson-Cole and O'Riordan, 2009; Nicholls et al., 2013).

After hard defences failed in the 1990s at Happisburgh, no further defences have been introduced (apart from piecemeal measures such as introducing rock armour in the 2000s) (Law et al., 2011). The coastline at Happisburgh has therefore been largely undefended over the last 30 years. This was despite a series of proposals for coastal schemes in the 1990s and 2000s, however, no proposal met the costbenefit ratio for the local authority to justify use of public funds (CCAG, 2023b). The local community have continued to run a long and at times high-profile campaign about the lack of coastal defences (particularly following the revised SMP publication in 2005), reaching national and even international news (Tebboth, 2014; CCAG, 2023c). This included forming a Coastal Concern Action Group (CCAG), which inspired and networked with many local coastal campaign groups across the country (CCAG, 2023b). Local residents articulated calls for social justice, in requesting financial support and greater involvement in coastal decision-making (Tebboth, 2014; Famuditi et al., 2018). After news in 2004 of the upcoming SMP update, local residents set up a charity to crowdfund for hard defences (of which £40,000 was raised in the first few years), including the 'buy a rock for Happisburgh' campaign (Law et al., 2011). Happisburgh residents refused to engage with local authorities if compensation for affected community members would not be considered, and social impacts relating to coastal erosion, such as property loss and forced relocation, were repeatedly discussed between NNDC and the community (Milligan and O'Riordan, 2007).

While community campaigning did not lead to a change in SMP policy, in 2012, Happisburgh was one of 15 locations across England to receive funding as part of the Pathfinder programme, a central government funded programme to trial innovative property roll-back and coastal adaptation options (Frew, 2012). The £3 million pathfinder project in Happisburgh involved the demolition of houses and a caravan park near the cliff edge, later using some of this land for a new car park, revenue of which goes to the parish council (Frew, 2012). Thus, although the Happisburgh Pathfinder programme did not strictly provide property compensation, for residents who lost homes to coastal erosion, affected individuals were offered a plot of land elsewhere, which could either be built on or sold. This was later introduced into local government planning policy (NNDC EN12 Policy – Relocation and Replacement of Development Affected by Coastal Erosion Risk) (NNDC, 2023a). Happisburgh residents continue to campaign for coastal defences and policy support, now under the banner of the 'Save Happisburgh Action Group'. On the Facebook page for the campaign group, its articulated aims are to provide physical protection to the village, financial support to at-risk homeowners, and preserve the many historical sites in Happisburgh (Save Happisburgh Action Group, 2023).

At the time of the revised SMP publication in 2005, Bacton and Walcott were in a slightly different position than at Happisburgh. The change from 'hold the line' to 'managed realignment' was due at 2025, rather than the then present-day, and there were pre-existing hard defences protecting the coastline (Mott MacDonald, 2014), including timber groynes, revetments, and a concrete sea wall (RH, 2018a) that runs along much of the seafront at Bacton and Walcott villages. Despite these defences, Bacton and Walcott are still vulnerable to flooding and erosion, and nearly 200 local businesses and homes were flooded in the two villages following the damaging December 2013 floods (Mott MacDonald, 2016). The damage caused by the 2013 floods caused deep concern with local residents that a similar event could reoccur in the future (Day, 2020). A flood support group was set up on Facebook in the aftermath of the 2013 floods, which continues to this day. At the time of writing, the group has over 3,000 members on Facebook, which is roughly the same size of the total population for all three villages (ONS, 2011) suggesting inland and neighbouring community support within the group.

In 2014, the pre-existing hard defences at Bacton and Walcott were assessed as not in their optimum state and in need of repair (Mott MacDonald, 2014). The residual lifetime was calculated to be 15 years for the timber revetments and sea walls, and 5-10 years for the groynes (AECOM, 2012b), putting over 200 properties at risk of coastal erosion (as illustrated in Figure 2.11, Mott MacDonald, 2014). Several feasibility studies of different options to manage coastal change were explored by the local authority, including a range of hard defence reinforcements, soft engineering, or no active intervention (Mott MacDonald, 2014: 2016). Of particular consequence was the high vulnerability of Bacton Gas Terminal to erosion, which is located north of Bacton village and was only several metres from the cliff edge. The local authority, with the support of significant partnership funding from Bacton Gas Terminal, subsequently chose to introduce a large-scale mega-nourishment project to protect the terminal and adjacent coastline (sandscaping), which will be outlined next, the final section to this chapter.

### 2.4.3 The Bacton-Walcott sandscaping scheme

In July 2019, a mega-nourishment scheme was implemented at Bacton to protect the nationally important Bacton Gas Terminal from flooding and coastal erosion (NNDC, 2022b). Known locally as sandscaping (rather than mega-nourishment), the Bacton-Walcott sandscaping scheme is modelled from the Zandmotor project in the Netherlands (Johnson et al., 2020b). While the sandscaping scheme at Bacton is much smaller than the Zandmotor scheme (1.8 million m<sup>3</sup> placed sediment, compared to 21.5 million m<sup>3</sup> for the Zandmotor) (De Schipper et al., 2016; Johnson et al., ibid), it is still on a vastly greater scale to traditional beach nourishment: the initial placed sediment stretched for 6km at Bacton (Johnson et al., 2020b), whereas existing beach nourishment schemes typically target a particular beach. Figure 2.12 indicates the extent of placed sediment in 2019. Note that Bacton and Walcott are in, but Happisburgh is outside, the area of initially placed sediment, and therefore the area that was intended for coastal protection from the scheme.



Figure 2.12. Red shading indicates the area of initial placed sediment and coastal protection of the sandscaping scheme, July 2019 (annotated from © Edina Digimap

The Bacton-Walcott sandscaping scheme is the first use of mega-nourishment in the UK. The shape of the sandscaping scheme differs from the Zandmotor scheme: placed sediment was spread further along the coastal frontage, resulting in no peninsula as in the Zandmotor scheme (Stive et al., 2013). However, the mechanism with which it provides coastal protection is the same: over time, the significant volume of additional placed sediment will gradually erode, be transported and deposited further up and down the coast through wave, tide, and inshore wind action. Sandscaping initially raised the height of Bacton beach by 7m, and the increased beach volume and height acts as a barrier between wave energy and erosion at the base of the cliffs (Hall et al., 2015). Figure 2.13 shows before and after photographs of Bacton gas terminal as a result of sandscaping, illustrating this transformation.



**Figure 2.13.** Before (top, July 2018) and after (bottom, Sept 2019) photos of Bacton Gas Terminal © Mike Page.

The Bacton-Walcott sandscaping scheme cost approximately £21 million (Johnson et al., 2020b), and was therefore a hugely costly coastal management scheme. The primary reason why sandscaping was implemented at Bacton was the need to protect the nationally important Bacton Gas Terminal against coastal risk (CCC, 2018). As previously mentioned, the terminal was only metres from the

cliff edge before sandscaping was introduced. Now that sandscaping has been implemented, the terminal is expected to be protected from flood and erosion for 15-20 years. Meanwhile, the adjacent villages of Bacton and Walcott are expected to be protected for at least that timespan.

A significant proportion of funding for the Bacton-Walcott sandscaping scheme was from the private sector: two-thirds of the £21 million came from the terminal operators (Shell and Perenco), with the remaining funding supplied by Defra (through FCERM Grant in Aid), the EA, NNDC, and funding contributions at a local level (such as crowdfunding) (Johnson et al., 2020b). The scheme was first conceptualised in the aftermath of the December 2013 floods, which highlighted the need for coastal protection at the terminal, and where several metres of erosion had occurred. At this time, RH undertook a feasibility study for NNDC and the Crown Estate in 2013, which highlighted that mega-nourishment would be a technically feasible coastal management option for this stretch of coast. (Johnson et al., 2020b). The placed sediment used in sandscaping was sourced from a licensed extraction site on the seabed at Great Yarmouth (Vikolainen et al., 2017). This placed sediment is of a similar grain size, or in certain cases coarser, than the pre-existing sediment on Bacton and Walcott beaches (Vikolainen et al., 2017).

Following the feasibility study, conversations about the viability of sandscaping continued for several years between the EA, NNDC, RH, the Crown Estate and terminal operators, evolving into a partnership (Johnson et al., 2020b). This included numerous consultations with key stakeholders from 2014-2017, such as Natural England, the Marine Management Organisation (MMO), and the Eastern Inshore Fisheries and Conservation Authority (IFCA) (Vikolainen et al., 2017). Public engagement also took place at this time, comprising of a consultation on the environmental assessment of the scheme, through press releases, and meetings with local groups (e.g. between NNDC and parish councils) (Vikolainen

61

et al., 2017). Further detail about public engagement on the sandscaping scheme is provided and focussed on in Chapter 5 (section 5.2.2), as it is particularly pertinent to empirical results presented in that chapter.

The Bacton-Walcott sandscaping scheme is described - by local and national policymakers and the engineers that designed the scheme (RH) - as providing additional time and opportunities for the villages of Bacton and Walcott to prepare for and adapt to the risk of flooding and erosion in the future (RH, 2018a; Johnson et al., 2020b). The scheme is therefore framed as an innovative coastal management strategy that builds social, as well as environmental, resilience to future coastal change, because it allows time to prepare for adaptation. Perceptions of future coastal change at Bacton and Walcott post-sandscaping is a key theme of this research, explored in Chapter 7.

# What are the modelled geomorphological changes of the sandscaping scheme, and is it working as expected?

The sandscaping scheme is being monitored by RH on behalf of NNDC. This includes regular bathymetric and topographic survey campaigns<sup>6</sup>, twice a year, of geomorphological changes above and below mean sea level. This secondary dataset is analysed in this thesis, to investigate beach profile and volume changes, alongside changes to the rate of cliff retreat. As of Autumn 2023, five survey campaigns have been conducted (August 2019, October 2019, February 2020, November 2020, June 2021 and September 2021) (RH, 2022a). Data from the monitoring campaigns show the sandscaping scheme is working broadly as it was expected to. Modelled expectations were that the nourished beaches would lose sediment over time, both seaward and alongshore, but for the majority of this placed sediment to remain in the coastal system over the scheme's lifetime, rather

<sup>&</sup>lt;sup>6</sup> The topographic surveys use LIDAR technology, and the bathymetric surveys use single beam jet skis. The resolution of both surveys is 0.5m x 0.5m.

than lost offshore (NNDC, 2022b). In the first two years post-sandscaping, an estimated 276,000m<sup>3</sup> placed sediment has migrated (representing 16% of the overall 1.8 M m<sup>3</sup> sand placed initially). 46,000m<sup>3</sup> of the 276,000m<sup>3</sup> migrated sand has moved in a south-easterly direction. There has therefore been a net erosion of sediment at Bacton Gas Terminal in the first two years, but net accretion at Bacton and Walcott village (NNDC, 2022b). This movement is expected, given the majority of sediment was placed at Bacton Gas Terminal, and therefore it is the terminal that was expected to see the greatest net sediment volume decrease over time, feeding areas updrift and downdrift.

These early geomorphological changes of the sandscaping scheme broadly mirrors the evolution of the Zandmotor scheme. The 4-year Zandmotor evaluation estimated that 95% of placed sediment still remained in the immediate area in the Delfland coast (Taal et al., 2016). This was higher than the modelling predicted, which suggests the Zandmotor scheme could exceed its 20 year predicted lifetime (Taal et al., 2016). For the sandscaping scheme, this figure was 84%, in the first two years (RH, 2022a). Unique local factors are expected to explain lower than expected sediment loss for the Zandmotor scheme, such as additional migrating sand from neighbouring beaches by prevailing winds, and placed sand from a nearby beach nourishment process in 2013 (Taal et al., 2016). Overall, placed sediment from the Zandmotor scheme has been transported 3.6km further along the coast than the 2.2km extent of its original placement (Taal et al., 2016). Geomorphological analysis by RH of the sandscaping scheme (2020: 2022a) found sediment accretion at Happisburgh, approximately 2 miles south of Walcott, the southern boundary of initially placed sediment. This is an unintended and potentially positive impact of the sandscaping scheme, in offering protection from erosion to a greater stretch of coastline than anticipated. Increased sediment volume and beach width at Happisburgh can be observed in the data (RH, 2022a).

Sediment accretion, in the upper intertidal zone and the shoreface, has also increased updrift of the sandscaping scheme, at Mundesley (RH, 2022b). However, crucially, it is unknown how much sediment will arrive at Happisburgh and how long it will remain.

# 2.5 Concluding remarks

In conclusion, this chapter has provided the reader with the necessary background to the research, providing an overview of coastal management policy and strategies to manage coastal risk, focusing in particular on mega-nourishment. The case study of the research, the Bacton-Walcott sandscaping scheme, has been introduced, with a history of coastal management at Bacton, Walcott, and Happisburgh. Key findings from this chapter include a lack of policy clarity on building social resilience to coastal change and facilitating coastal adaptation, despite the attention and use of resilience as a concept in recent coastal policy documents, and a research gap on exploring the social dimensions of sandscaping. The next chapter will outline the academic fields of study on resilience and adaptation, alongside environmental justice, in a literature review, drawing upon previous research to develop a conceptual framework for the thesis.

# 3. Literature review

# 3.1 Introduction

This literature review draws upon three main research topics to study coastal change – the resilience (section 3.2), environmental justice (3.3), and adaptation (3.4) literature. As much studied research fields, this review focuses on UK and European coastal contexts, given the case study of this thesis is the North Norfolk coast. The linkages between these topics form an important part of this thesis' conceptual framework, outlined in section 3.5. The academic and grey literature presented in this review were identified through keyword searching on Scopus, Web of Science, and snowball sampling from key papers. The aforementioned topics are chosen for their different insights on the implications of sandscaping, and what the objectives of managing coastal change should be (according to these perspectives). As argued in the introduction (Chapter 1), sandscaping has many attributes of a resilient approach, and Chapter 2 revealed that resilience is a key concept within which the objectives of current coastal policy (for example the current FCERM strategy, (EA (2020))) are framed. Meanwhile, environmental justice can be used to understand the social impacts of a particular coastal management approach, and the adaptation literature reveals how humans respond to and perceive a change to their environment.

# 3.2 Resilience

This section begins by presenting the definitions of resilience used in this review (section 3.2.1). An overview of socio-ecological systems theory is provided next, outlining the systems framing used in this thesis (3.2.2). Next the review considers

different types of resilience, with a particular focus on transformation (3.2.3). Subsequently this section explores how resilience is built or lost, focussing on the concept of adaptive capacity, and how this can be applied to a coastal context (3.2.4). The relationship between resilience, uncertainty and maladaptation is also considered (3.2.5) before ending with a discussion on the criticisms of resilience (3.2.6), and in particular, where the concept may raise tensions with environmental justice.

## 3.2.1 Defining resilience

There are many different definitions of resilience, and uses of the term across disciplines (Brown, 2014). For example, resilience is applied in engineering, environmental sciences, health, psychology and disaster studies literature, amongst others (Brown, 2014; Matyas and Pelling, 2015). Even within the same field, the definition of resilience can be contested (Twigger-Ross et al., 2015). For example, analysis of climate change research by Bahadur et al., (2011) finds 16 different uses of the term. Although there is relevant theory on resilience, and in particular transformation, from the sustainability transitions literature (IPCC, 2022), this literature review specifically draws upon definitions of resilience from the socio-ecological resilience literature, given that this research takes a systems lens to understanding social and environmental dimensions to coastal change.

The central idea to resilience, is that it is a measure of the ability to withstand and bounce back from shock events (Holling, 1973; Tompkins and Adger, 2004; Adger et al., 2005a). Work by Holling (1996) provides an early and prominent definition of ecological resilience<sup>7</sup>, which builds upon concepts from engineering resilience

<sup>&</sup>lt;sup>7</sup> Ecological resilience is a term used to describe resilience of the natural environment and its associated biophysical process, ecosystems and all species residing within it (Holling, 1996).

theory. As such, in Holling's (ibid) definition, resilience is recovery from a stress event back to an original system state, and the idea of social resilience is not included. Later work by Folke et al., (2005) does incorporate social resilience, (described as the resilience of people at an individual, community or population level) but similarly considers one system state: "*the capacity of a system to absorb disturbance and reorganize while undergoing change so as to still retain essentially the same function, structure, identity, and feedbacks*" (Walker et al., 2004, p.1). Where these definitions of resilience are prescriptive in stating resilience is returning to precisely how the system was before a shock, later definitions are more open as to whether a system can change in certain ways, as long as system functioning is not compromised. This is a key area of evolution amongst the literature, and Figure 3.1 highlights the two proposed theories.



**Figure 3.1** Illustration of two different pathways for recovery (bounce back resilience, solid line, and bounce forward resilience, dotted line) (annotated from Townend et al., 2021).

The solid black line on the diagram depicts the system returning to its original state (known as 'bounce back resilience') whereas the dotted line depicts the system entering a new state with a higher level of functionality ('bounce forward resilience'). The latter is a more recent theory (Greg-Lloyd et al., 2013), arguing resilience is the ability to bounce forward and morph into a different state, if the status quo cannot be maintained (Folke, 2006; Eakin et al., 2009; and Rozer and Surminski, 2020). In this regard, Folke (2006) argues, bounce forward resilience can be transformative, if the system has an ability to change in response to pressures. Transformative resilience is discussed further in section 3.2.3. The idea of 'bounce forward' resilience came to fruition by work on social resilience from the social sciences (Bene and Doyen, 2018), and therefore is a particularly relevant concept for analysing social systems. It also reflects an increasing complexity to theories on resilience, where rather than a pre-defined end state or equilibrium, there is the potential for multiple different future system states following a disturbance, each with their own associated resilience (Greg-Lloyd et al., 2013), that might involve learning and adaptation (Cutter, 2016).

This thesis draws upon the more recent definitions of resilience, which principally focus on retaining functionality and the avoidance of harm, and uses two definitions to consider both social and environmental resilience. Environmental resilience is defined in this thesis as "the larger the shock that the system can put up with and remain "viable" in the long run" (Bene and Doyen, 2018, p.980). For social resilience, Maclean et al's (2016, p. 523) definition is used, which includes the themes of learning and self-reflection: "the adaptive and learning capacity of individuals, groups and institutions to self-organise in a way that maintains system function...in response to a disturbance". Noteworthy is how Maclean et al., (2016) define resilience using the concept of adaptive capacity, which is also how resilience is defined in the EA's (2020, p.25) FCERM strategy: "the capacity of

people and places to plan for, better protect, respond to, and recover from flooding and coastal change". Adaptive capacity is therefore an important concept in understanding how resilience is built or lost, and is explored in section 3.2.4.

## 3.2.2. Resilience in socio-ecological systems

Figure 3.2 from Virapongse et al., (2016) illustrates the way in which social and ecological dimensions (left and right circle) influence each other (central circle) within a system. This relationship can be highly interdependent, for example early work by Adger (2000) demonstrates the dependency of societal use of mangrove ecosystem services<sup>8</sup>, and the negative societal impacts that stem from mangrove loss. This high dependency between the social and natural world shows the influence of each on the other, and therefore the possibility that change within one part of a system potentially impacts another. Greg-Lloyd et al., (2013) highlight the influence of land use planning and infrastructure development in the UK on increasing the vulnerability of the coastal zone to flooding and erosion. Therefore, an understanding of both social and environmental components, and how they interact, is necessary when assessing the impacts of a coastal policy or intervention. Within socio-ecological systems, resilience may vary between social and ecological components, and across spatial and temporal scales (Maclean et al., 2016). Howarth et al., (2020) argue that a single component within a system, or the entire system itself, has the capacity to cope or recover from a stress event (adaptive capacity). It is therefore important to consider resilience across multiple scales and dimensions within a system.

<sup>&</sup>lt;sup>8</sup> Goods and services provided by the natural world that is beneficial to society. There are many different types of ecosystem services, from regulating services (e.g. trees producing oxygen) to cultural services (e.g. mental health benefits from spending time outdoors). (UK National Ecosystem Assessment, 2021).



**Figure 3.2** Illustration of the linkages between social and ecological processes within a socio-ecological system, from Virapongse et al., (2016).

Although this thesis draws upon socio-ecological systems theory, an environmental, rather than strictly ecological, system is considered. Given that resilience literature evolved from ecological resilience theory, the terminology reflects this, despite much of socio-ecological systems literature using the terms 'environment' and 'ecology' interchangeably (Cote and Nightingale, 2011; Virapongse et al., 2016). Others have stressed that socio-ecological systems theory is not about implying separation of humans from nature, but allowing a focus on the connections between the two (as this research similarly explores) (Davidson-Hunt and Berkes, 2009). Therefore, although this review draws upon socio-ecological systems theory, a broader environmental lens is applied to the thesis. The system examined in this thesis is the interaction between the coast and the coastal residents of Bacton, Walcott and Happisburgh in North Norfolk (as outlined in Chapter 2). Contrary to the ecological processes displayed in Figure 3.2, this thesis solely considers coastal erosion, and explores geomorphological

components such as the cliff and beach dynamics of the coast. In terms of social components, this thesis considers the social processes (of individuals, communities, and institutions) living alongside the coast, their relationship to and perceptions of the coast, including how coastal change is managed.

### 3.2.3 Different 'types' of resilience

It has been suggested that the types of responses to shock events can be considered in terms of the level of resilience associated with that response. This section draws specifically upon the conceptual frameworks of Folke et al., (2010), Matyas and Pelling (2015), Bene and Doyen (2018) and Milhorance et al., (2021), who put forward different types of resilience, that represent different levels of resilience for socio-ecological systems. These authors were chosen for their explicit attention to different 'types' of resilience, considering resilience as a pluralistic concept rather than just one entity (Bene and Doyen, 2018). Furthermore, the authors (ibid) apply their resilience categories either to socioecological systems (Folke et al., 2010; Bene and Doyen, 2018) or from a disaster risk management perspective, which is still of relevance to this thesis, as it focuses on the connections and interdependencies between the social and natural world (Matyas and Pelling, 2015; Milhorance et al., 2021). The different categories of resilience presented by the authors are summarised in Table 3.1. The name of each category is presented verbatim in column 2 ('resilience categories'), with analysis of their work in columns 3 and 4.
Authors	Resilience categories	Category descriptions (paraphrased by author)	Relationships between categories
Folke et al., 2010	Resilience	Absorbing and potentially altering, but not beyond the parameters of the same system state.	There is not one category a system can be in, and over time different categories can be occupied.
	Adaptability	The ability to learn or alter responses, within the same system state.	Resilience is periods of persistence, interrupted by sudden or slow onset
	Transformability	Deep rooted change that shifts into another system state.	adjustment or transformational change, before the dormant resilience period begins again.
Matyas	Resistance	Not described in paper.	An approach could aim
and Pelling, 2015	Incremental adjustment	Temporary change, returning to the original system state. Surface level, rather than superficial, change.	for resistance in the short-term and more drastic change in the long-term. Multiple system
	Transformation	Challenging the way social systems operate. The structure and functioning of the system has changed so much to the effect that it can now be considered a different system.	at one time. E.g. resistance could happen at one part of the system and transformation at another, within the overall system.
Bene, and Doyen, 2018	Resistance	Withstanding or deflecting a shock or stress in the same system state.	The five responses sit on a continuum, rather than strict categories.
	Absorptive capacity	The maximum amount of stress a system can take (persist) while maintaining structure, functioning, and the same system state.	are not mutually exclusive.
	Incremental adaptation	Adaptive capacity, learning, and enacting change from that learning.	
	Adaptive preference	Altering perceptions by coping with a shock.	
	Transformation	New system created through significantly different structure and functioning.	
Milhorance	Persistence	Living with the shock,	The authors justify the
et al., 2021		system state.	use of three categories, given that

 Table 3.1 Categorisations of resilience within literature (Table: I Cotton).

Incremental adjustment	Ability to learn and do things differently, the changing actions result in a slightly different system.	it has been argued in literature there is considerable overlap in the five response model developed by Bene and Doyen
Transformational response	The system has changed significantly by actions enacted.	(2018).

As can be seen in Table 3.1, there is considerable overlap in the resilience categories presented by the authors (ibid), with similar incremental depths of change, and yet each author adopts slightly different terminology. For example, where Folke et al's., (ibid) earlier paper refers to simply 'resilience' as a first category, other authors refer to 'resistance' (Matyas and Pelling, 2015; Bene and Doyen, 2018) or 'persistence' (Milhorance et al., 2021). This choice reflects the evolution of the concept of resilience in literature beyond a bounce back definition, as discussed in section 3.2.1. All frameworks include both bounce back and bounce forward types of resilience, and all authors argue that resilience increases from one category to the next. Furthermore, all authors agree that categories are interchangeable, not mutually exclusive, and that a system can transition between categories (Folke et al., 2010; Bene and Doyen, 2018) or have multiple system responses at the same time (Bene and Doyen, 2018). It is important to note that similarity between the frameworks partially reflects the fact that authors draw upon each other's work. Both Matyas and Pelling (2015) and Milhorance et al., (2021) draw upon the work on Bene (citing for example Bene et al., 2013 and Bene et al., 2015; 2018, respectively), and Milhorance et al., (ibid) also cites the work of Folke et al., (2010) in developing their own framework.

Differences in terminology between the authors can also be seen with the median categories of resilience ('adaptability', 'incremental adjustment' and 'incremental adaptation' are all used). This reflects a contrast within literature on whether the

extent of system change in this category results in the same, or a slightly altered, system state. While all authors bar Milhorance et al., (2021) argue it involves the same system state, Folke et al., (2010, p.1) argues adaptability is change within a greater "*stability domain or basin of attraction*". Overall, differences in terminology are due to differences in scale and framing of the wider socio-ecological system state by each author. While further comparative analysis could be taken on the different types of resilience presented by different frameworks in literature, this is beyond the focus of resilience within this thesis, which is principally concerned with the concept of transformation. Therefore, the last type of resilience presented by the authors, 'transformation', is discussed next in greater detail.

As seen in Table 3.1, whereas categories with lower resilience relate to withstanding shocks without introducing any change to the system (e.g. 'resistance' or 'coping'), or altering part of a system, but not fundamentally so (e.g. 'incremental adaptation'), 'transformation' is framed as a change that significantly alters the structure and functioning of a system, so that effectively a different system has been created. The new system has significant length (i.e. permanence) and depth (i.e. deep-rooted) of change. This concept of transformation appears highly relevant to the definition of social resilience by Maclean et al., (2016), adopted by this thesis (section 3.2.1), so long as system functioning is retained (a key element of the definition). All of the authors in Table 3.1 argue transformation has the highest level of resilience, and of all the types of resilience, 'transformation' has the greatest agreement amongst the different frameworks. Bene and Doyen (ibid) argue resilience categories can be best understood as a continuum or scale, with 'transformation' at the top of that scale. However, this suggests an inherent assumption within the literature, that because 'transformation' provides the greatest opportunity for a system to change, it also provides the greatest opportunity to build resilience. Kates et al., (2012) similarly

74

refer to the change from transformation as innovative, large-scale, and significantly altering system structure and functioning. However, by this logic, transformational responses provide the greatest opportunity for both desirable and undesirable change.

Undesirable change is referred to in literature as 'maladaptation', which is associated with a reduction in resilience (Milhorance et al., 2021), and is a key concept linking the resilience and adaptation literature (maladaptation is also explored in section 3.4.1). There is therefore disagreement within the resilience literature on the relationship between transformation and maladaptation, simultaneously presenting transformation as the most desirable system response (Milhorance et al., 2021) and also a response with the greatest likelihood of maladaptation, given the amount of change that is being introduced to a system (Matyas and Pelling, 2015). This has led to attention on the different causes of transformation, with recent work highlighting that transformation can be 'deliberate' or 'forced' (IPCC, 2022). Deliberate transformation is presented as the more desirable, and more likely to lead to desired adaptation goals and higher resilience, whereas forced transformation may have unintended consequences (i.e. maladaptation) (Folke et al., 2010; IPCC, 2022).

In their definition of transformation, Folke et al., (2010, p.6) argue: "*transformations consist of three phases: being prepared or even preparing the social-ecological systems for change, navigating the transition by making use of a crisis as a window of opportunity for change, and building resilience of the new social-ecological regime*". In this definition, it is implied that having 'preparation time' with deliberate transformation leads to greater resilience. The above findings demonstrate there is a need for further empirical analysis on transformation, and its impact on overall system resilience and maladaptation. Furthermore, the conceptual frameworks analysed in Table 3.1 focus on categorising resilience of an overall system, and

there is less analysis by the authors on how individual components within a system can build resilience (with the exception of Milhorance et al., 2021, who apply their conceptual framework to consider the impact of drought for households in Brazil). There is therefore also a need for research that critically analyses resilience at different scales within a system.

Applying this theory on transformation to a coastal change context, both sandscaping and adaptation options under a managed realignment policy scenario could be viewed as examples of transformative coastal change. Sandscaping has dramatically increased the height and volume of Bacton and Walcott beach (Johnson et al., 2020b), and arguably fits all three of Kates et al's., (2012) categories of transformational adaptation, in that it is innovative, largescale (although by how much is not quantified in the paper), and significantly alters the geomorphological (and thus very likely the ecological) structure and functioning of the coastline. Meanwhile, adaptation options for managed realignment, such as property rollback, the closure and opening of old/new economies (and its impact on livelihoods), and demolition of coastal towns and relocation, are regarded in literature as transformative policies to deal with coastal risk (O'Brien, 2017; Coastal Partnership East, 2019; Haasnoot et al., 2021). Such policies reflect Folke et al's., (2010, p.5) definition of transformational change (covering similar themes to Bene and Doyen's (2018) in Table 3.1): "shifts in perception and meaning, social network configurations, patterns of interactions among actors including leadership and political and power relations, and associated organizational and institutional arrangements". This definition mirrors the future reality that some coastal communities face in moving, rebuilding, and potentially rearranging infrastructure and social networks away from their current location. The emphasis of changing social perceptions within Folke et al's (ibid) definition also highlights the role of perceptions (explored further in 3.4.3) in

76

building social resilience. This research seeks to critically analyse the transformative nature of sandscaping, and managed realignment, in further detail, and implications for coastal resilience.

## 3.2.4 How is resilience gained or reduced?

If resilience can be understood as a plural concept, with multiple types each with their own level of system change – how is resilience built, maintained, and lost? The frameworks presented in Table 3.1 argue that responses with greater resilience draw upon a greater number of adaptive capacities. For example, the *"ability to adapt"* (Matyas and Pelling, 2015, p.7) *"adjust responses"* (Folke et al., 2010, p.1), or make *"proactive and informed decisions"* (Milhorance et al., 2021, p.679). This highlights the importance of 'capacity' as a concept to measure resilience change. The literature presents three main capacities of relevance to building resilience; 'capacity to prepare', 'capacity to recover', and 'capacity to adapt' (Oriangi et al., 2020) (although there is variation, with other authors using 'capacity to adapt', or adaptive capacity, as an umbrella or all-encompassing term (Cinner and Barnes, 2019)). The use of a capacities framing stems from a US Army Corps definition of resilience (Townend et al., 2021), and is widely utilised in disaster studies literature (Oriangi et al., 2020).

'Capacity to prepare' refers to an ability to anticipate and lessen the impact of a shock (Oriangi et al., 2020). Factors found to contribute towards a 'capacity to prepare' include knowledge, awareness and risk perception of a shock (Lwin et al., 2020; Alizadeh and Sharifi, 2021) alongside previous experience (Lwin et al., 2020) and agency (Cinner and Barnes, 2019). Meanwhile, 'capacity to recover' refers to an ability to live alongside a shock or stress event (Oriangi et al., 2020). Factors found to enable recovery include heterogeneity (i.e., resource availability),

flexibility (e.g., of livelihoods), and access to social support networks (Tompkins and Adger, 2004; Lwin et al., 2020; Alizadeh and Sharifi, 2021). The last capacity of relevance, 'capacity to adapt', refers to an ability to move or change in response to a shock. Factors found to enable or constrain adaptive capacity include the strength of connection to place (sense of place is explored further in section 3.4.3) and relationships with others (such as community cohesion and trust in institutions) (Adger et al., 2005a; Lwin et al., 2020; Alizadeh and Sharifi, 2021). Figure 3.3 summarises the range of factors highlighted in the literature that enable or constrain the three different relevant adaptive capacities to building resilience.



**Figure 3.3** Summary diagram illustrating the factors that enable or constrain (individual/ community) adaptive capacity as found in this literature review (Figure: I Cotton).

Applying the concepts 'capacity to prepare', 'capacity to recover' and 'capacity to adapt' to a coastal context, increases or decreases of sediment (and sufficient sediment within a coastal system), and normal functioning of natural geomorphological processes such as longshore drift, are important factors contributing to the capacity to recover and withstand shocks from coastal erosion events (Hall et al., (2015). Another enabler of recovery capacity, for physical or environmental resilience, is heterogeneity, in terms of a diversity of species within a coastal system (Adger et al., 2005a). Heterogeneity facilitates resilience where there is more likely to be differential impact to external shocks within a system

(Adger et al., 2005a). This entails the structure and functioning of ecosystems is less likely to be depleted, and therefore the system is more likely to have a higher capacity to recover following a disturbance.

Turning to social resilience, the EA's FCERM strategy (2020) describes some of the actions that will be necessary for coastal communities to adapt to rising erosion risk. These include: "facilitating community engagement, finding alternative housing for those at risk and in some cases supporting businesses to find opportunities in neighbouring areas", and subsequently (p.58): "address safety and access challenges, while 'roll back' policy is still in operation" (EA, 2020, p.57). A focus on planning for the future is also present in the latest IPCC (2022) assessment report, where the objective of adaptation is defined in terms of capacity building, and in particular, "the capacity to anticipate and respond successfully to change" (IPCC, 2022, p.72). The above actions appear to focus on a framing of resilience as having the capacity to prepare for erosion events. However, many of the factors which build 'capacity to recover' and 'capacity to adapt', as shown in Figure 3.3, are equally argued as relevant for building social resilience. For example, Cinner and Barnes (2019) report an individual's capacity to learn and organise, flexibility and access to resources (capacity to recover factors) all increase resilience, meanwhile Tompkins and Adger (2004) found a diversity of social networks (capacity to recover and adapt factors) are beneficial for local communities whilst living alongside coastal change. Therefore, from a social resilience perspective, it appears there is currently greater policy attention to factors that support a capacity to prepare for coastal change, despite the relevance of factors that support the capacity to recover or adapt.

#### 3.2.5 Resilience and uncertainty

It is stressed both within peer-reviewed literature, grey literature, and policy documents that it takes time to build social and environmental resilience to coastal change (Mott MacDonald, 2014; EA, 2020; Day, 2020; Townend et al., 2021). However, this length of time is not specified, nor how it might vary from a geomorphological to a social perspective. This gap links to opposing arguments within literature and policy on whether resilient systems are inherently more uncertain (as advocated by Folke, 2006 and Bellamy, 2019). For example, the EA's (2020) FCERM strategy advocates that, by moving from reactive to more anticipatory adaptation measures, coastal management can become more resilient by being more ready to respond to flood or erosion events, which in turn reduces uncertainty. Leach (2008) however, argues that a resilient approach, in terms of being flexible for a range of long-term outcomes, demonstrates robustness but not necessarily reduced uncertainty. Meanwhile Bahadur et al., (2013) argue that an ability to deal with uncertainty is an attribute of being socially resilient to environmental change.

As highlighted previously in section 3.2.3, transformative resilient approaches can be both accidental and intentional (i.e. 'forced' or 'deliberate'), of which the former has higher associated uncertainty (IPCC, 2022). This raises an interesting question; do coastal management approaches focussed on increasing resilience also increase uncertainty, and where or how is this uncertainty felt in a system? The lived experience of uncertainty in relation to coastal change is not discussed explicitly in the EA's FCERM (2020) strategy, which acknowledges the economic cost of mental health impacts from flooding or erosion, but provides no indication of how communities can be supported to live alongside such uncertainty of coastal risk. Indeed, many criticisms of resilience stem from the practical challenges of operationalising the concept, which is discussed next.

#### 3.2.6 Criticisms of resilience

Several criticisms have been raised regarding the concept of resilience, many of which relate to its practical application. Resilience is often framed as an end goal or an aspiration, with an absence of concrete operational understanding (Brown, 2014; Bene and Doyen, 2018; Townend et al., 2021). In other words, it is often not clear how resilience, and progress towards it, can be measured, particularly for social dimensions (Cutter, 2016). Compounding this, the way resilience is defined and the parameters used to demonstrate it are varied, and selection choice shapes how resilience is perceived and understood (Cutter, 2016; Ensor et al., 2021). This is notwithstanding the potential inaccuracies of using an indicator to measure resilience, when the use of proxies is not an exact measurement of what it seeks to represent (Ensor et al., 2021). Greg-Lloyd et al., (2013) highlights how key concepts frequently discussed in resilience literature, such as adaptive capacity, resilience, and vulnerability, have different definitions depending on which branch of science or social science is considered.

The significance of this ambiguity, is that the objectives of resilience, and its intended recipients are unclear, or can be contested (Cutter, 2016). Some uses of resilience can therefore be normative, and will mean different things to different people with different priorities (Ensor et al., 2021). Further to this, Brown (2014) argues it is not typically specified 'who' and 'what' is building resilience, and in order to operationalise the concept, this needs to be made explicit (Townend et al., 2021). This is despite the limitations of resilience, in potentially causing disproportionate impacts on society, being well argued in literature (Leach, 2008; Eakin et al., 2009; Brown, 2012; Matyas and Pelling, 2015). Despite the policy attention afforded to resilience, the concept has been criticised for being blind to scale sensitivities (Brown, 2013; Matin et al., 2018). For example, research by

Lau et al., (2021) in Papua New Guinea found temporal trade-offs between ensuring community members could fish and self-sustain versus developing the healthy functioning of the ecosystem in the long-term and recovering stocks. This highlights the sensitivity of scale in pursuing resilience, and between social and environmental components of a socio-environmental system. This is therefore further evidence of a research gap in understanding scale-sensitive trade-offs of resilience.

Beyond issues of practical applicability, the ambiguous and subjective use of resilience in socio-environmental systems means some parts over others may be prioritised to promote (specific and potentially limited) resilience (Matyas and Pelling, 2015). Parts, rather than the whole, of a socio-ecological system builds resilience at any one time, and what elements are prioritised is a political choice (Matyas and Pelling, 2015). Resilience, argues Hayward (2013), is also a concept absent of political ecology questions on the causes of environmental damage or social inequality (as seen in the EA's FCERM (2020) strategy). Further, by not sufficiently capturing the unequal distribution of power and resources in society, Matin et al., (2018) argue resilient approaches put forward are by those not vulnerable or marginalised, calling into question issues of participative justice. The reduction of unevenly distributed policy consequences is a growing area of research examining what is termed 'inclusive', 'equitable' or 'just' resilience (Meerow et al., 2019). For example, Matin et al., (2018, p.203) define equitable resilience "as one which takes into account issues of power, subjection, and resistance; makes visible socially constructed limitations faced by groups and communities at all levels; and thinks about these issues in a joined-up way to avoid unsustainable interventions being made in the name of either disaster response or development'. This focus on issues of equity in following resilient approaches can be explored using the framings of environmental justice literature, and is the

82

next subject of this review. The environmental justice lens of this thesis can indicate where, when and by whom positive and negative impacts are felt, which, as Leichenko and O'Brien (2019, p.183) describe, can help in "*building equity into resilience*".

# 3.3 Environmental justice

This section firstly introduces the field of environmental justice (3.3.1), before focusing on the three main themes of justice (distribution, participation and recognition) (3.3.2). The causes of environmental justice issues is reviewed next (3.3.3), with a particular focus on a UK coastal policy context. Lastly, this section considers how a plurality of scales can help analyse environmental justice issues in coastal management (3.3.4).

#### 3.3.1 Environmental justice: definitions, origins and development

Environmental justice can be described as both a grassroots activist movement and an academic field of study, concerned with the differential experiences of environmental change on different societal groups (Coolsaet, 2020). It emerged in the United States in the 1980s amid growing realisation that non-white communities were disproportionately affected by environmental harms, for example by living closer to chemical waste plants and suffering lower levels of air quality (Bullard, 1990). However, research has identified other origins of the field, from other communities and other points in time (Murdock, 2020). While first focussing on the differential experiences societal groups face (distributional justice) (Kaswan, 2020), the field has expanded to consider the ability of individuals, particularly marginalised groups, to participate in environmental decision-making (participatory justice) (Marion Suiseeya, 2020) and the extent to which individuals are accepted as a legitimate voice for environmental concerns (recognition) (Coolsaet and Néron, 2020). Distribution, participation, and recognition reflect the three main and commonly referred to branches of environmental justice (Murdock, 2020).

The field of environmental justice has evolved considerably over the last few decades. Notable transitions include an increasing recognition of capabilities (i.e. rights and requirements to a healthy and satisfied life), and not just equity (i.e. uneven impacts across society) (Schlosberg, 2013), which reflects the expansion of the field away from solely considering distributional justice issues. Furthermore, research has widened from distinct spatial contexts to consider global issues such as climate change (leading to the field of climate justice), resource depletion, and the degrowth movement (leading to the field of just transitions) (Schlosberg, 2013; Schlosberg, 2020). In particular, there is a large body of literature examining the ethics and justice of global climate mitigation, in terms of the allocation of rights and responsibilities (Adger et al., 2011; Dryzek, 2013) and climate change governance (Dryzek and Tanasoca, 2021).

The application of environmental justice to climate adaptation, however, has been much less studied (Edwards, 2020), and is a research gap this thesis contributes to. Furthermore, much of the environmental justice literature focusses on international contexts. While the concept has been applied to the UK to consider different social inequalities and deprivation, including the differing socio-spatial experiences of flooding (Walker, 2009; Walker and Burningham, 2011; Fernandes-Bilbao et al., 2011; Sayers et al., 2017), urban air quality (Walker et al., 2005; Dietz and Atkinson, 2007; Briggs et al., 2008; Mitchell et al., 2015), and coastal management (Cooper and McKenna, 2008; Eakin et al., 2009; Thaler et al., 2017), there is less research in the UK more recently on coastal change and managed realignment, and not in the context of sandscaping. Particularly in a UK

context, the term 'environmental justice' may be used interchangeably with 'social justice' or 'social equity', and use a social disadvantage or social vulnerability framing to explore the unequal impact of environmental issues (for example Walker and Burningham, 2011; Preston et al., 2014; Sayers et al., 2018).

#### 3.3.2 The different themes of environmental justice

As mentioned in 3.3.1, environmental justice can be broken down into three main concepts or 'issues': distribution, participation, and recognition. Distributive justice can be defined as "how harms and benefits are distributed and experienced" (Kaswan, 2020, p.22), and was first conceptualised as the experiences different societal groups face to environmental harms (Bullard, 1990; Blais, 1996). Such experiences (hereafter impacts), are typically framed as 'physical harm' from conditions or environmental 'stigmatic harm' associated with certain neighbourhoods (Kaswan, 2020, p.29). Distributional justice initially focused on the spatial configuration of where impacts are felt (Holifield et al., 2009), but more recent research has stressed the relevance of temporal dimensions to impacts (Pellow, 2020; Bopp and Bercht, 2021) as delays to when impacts materialise, and linked to questions of intergenerational justice.

There is no clear definition of how distributional justice or injustice can be measured, which is subject to debate in literature (Kaswan, 2020). Several authors argue the concepts of 'equality', 'fairness' and 'justice' are used interchangeably (see for example Ikeme, 2003; Johnson et al., 2007), despite having different philosophical foundations in egalitarianism (Rawls, 1971), utilitarianism (Arthur and Shaw, 1978) and capabilities (Sen, 2009). Therefore, depending on which philosophical foundation of justice is put forward, the concepts of 'equality', 'fairness' and 'justice' are subjective and hard to measure (Thaler et al., 2017). For

example, 'fairness' for the greatest number of people may look differently to 'fairness' for the most vulnerable within society (Johnson et al., 2007), and equity in benefits and experiences may look differently to supporting the most vulnerable communities (Tyler, 2000). Consequently, it is important for a plurality of perspectives of distributional justice to be considered in coastal decision-making, recognising potentially different philosophical foundations.

The second key issue within environmental justice is participation, which is concerned with the degree to which all relevant stakeholders are included in decision-making (Marion Suiseeya, 2020). Stakeholders may not be consulted before a policy is introduced, or participation may be tokenistic rather than offering the opportunity for a particular group to meaningfully engage and contribute their views (Bell and Carrick, 2017). Participatory justice issues can also be caused by imbalances of power, where some voices are heard and respected to a greater extent than others (reflecting power imbalances both within and between different stakeholder groups) (Marion Suiseeya, 2020, Rodriguez and Inturias, 2018). Within the wider environmental justice field, and globally, there is ample research on the exclusion of minority or indigenous groups in the management of natural resources (for example Anguelovski and Corbera, 2023; Diep et al., 2022; Makey et al., 2022; Palmer et al., 2022; Reyes-Garcia et al., 2019). A key area of agreement in the participatory justice literature is the need for co-governance of natural resources, that involves all the relevant stakeholders from the outset, and where all stakeholders are involved in the design and implementation of a policy or project (Diep et al., 2022; Marion Suiseeva, 2020; Palmer et al., 2022; Reves-Garcia et al., 2019; Rodriguez and Inturias, 2018).

Participatory justice links closely to recognition, which considers whether all communities affected by a policy or environmental harm have been considered in policy making. When impacts are being considered, recognition asks questions of

otherness, reciprocal recognition, respect, and identity (Coolsaet and Neron, 2020). This could be in terms of whether a policy has considered different values or knowledge systems (the idea of epistemic justice, Miriti et al., 2022, Makey et al., 2022, Palmer et al, 2022), and ways of living for certain stakeholders, such as indigenous communities (Rodriguez and Inturias, 2018). Fraser and Honneth (2003) argue that if certain individuals are being disproportionately impacted within a population, this is the result of not being sufficiently valued, and a form of recognitional injustice. If individuals are impacted by a policy or harm where they weren't originally considered, Taylor (1994) argues this is another form of injustice, where being initially overlooked is a form of misrecognition. Although the three branches of environmental justice have typically been used in literature to separately analyse different injustices (for example economic issues with distributional justice, and cultural issues with recognition) (Coolsaet and Neron, 2020), considering both in tandem can provide a deeper analysis of environmental justice issues, and is applied here in the context of coastal management.

# 3.3.3 What causes environmental justice issues?

There are many reasons why an individual, and certain societal groups, are more impacted by environmental harms, which reflects both intrinsic (e.g. personal characteristics) and extrinsic factors (e.g. societal structures and actions). While age, gender, mental and physical fitness are all found to increase an individual's vulnerability to environmental harms (Bevacqua et al., 2018; Rizzo et al., 2020), early environmental justice work in the 1980s highlighted how communities most affected by an environmental issue reflected historical patterns of inequality (for example in the United States, housing demographics reflect economic income and historical patterns of racial segregation (Kaswan, 2020)). Income is widely argued to be a key determinant (Atteridge and Remling, 2018; Warner and Kuzdas, 2016;

Vasseur, 2021), where poorer residents have less resources at their disposal to buffer effects following an environmental harm, or may live in neighbourhoods less likely to be protected, or more likely to be displaced, by an environmental policy (Anguelovski et al., 2016). Indeed, the relative affluence or deprivation of an area (Cutter and Corendea, 2013) and related socioeconomic factors, is argued to affect an individuals' access to assets, services, skills and resources (Blaikie et al., 2005; Bevacqua et al., 2018; Rizzo et al., 2020). In this regard, justice issues can be seen from a political economy perspective, influenced by the distribution of power and capital.

Social determinants of vulnerability may intersect with physical climate vulnerability, where certain neighbourhoods have greater inherent vulnerability due to geography and climate variables (Ensor et al., 2021). In the UK, there has been considerable research exploring 'natural' and 'imposed' injustices to flood risk (Johnson et al., 2007; Walker, 2009) or climate change in general (Preston et al., 2014), highlighting 'multiple injustices' where high flood risk areas correlate with low income, low property price neighbourhoods (Rozer and Surminski, 2020). While there is broad research on the determinants of environmental justice issues both in the UK and globally, this review focuses on the impact of defended and undefended coastlines, given the case study context of the research. As highlighted in section 2.2.2, the unequal allocation of public funding for coastal management raises justice issues, where some areas of England's coastline are protected and some are not (or areas are protected to differing future timescales). Funding for coastal management in the UK requires a sufficient cost-benefit ratio to approve spend on schemes (Defra, 2021) (funding sources for coastal management are outlined in section 2.2.2). Schemes are therefore awarded according to a utilitarian logic (the greatest good versus money spent) rather than in areas of high priority or vulnerability (Johnson et al., 2007).

From an egalitarian perspective, this can be viewed as unjust (Rawls, 1971). Such a formula discriminates against sparsely populated areas and assets that can't be monetised, where it is more difficult to demonstrate a sufficient cost-benefit ratio (EFRA Committee, 2019). For example in Cornwall, many areas of coastline require funding to adhere to the Cornwall and Isles and Scilly SMP, but in reality, only areas with bids that demonstrate multiple or co-benefits will be recipients of funding (EFRA Committee, 2019). A high profile example is the City of London, which receives significant funding for flood defences and flood protection than more sparsely populated parts of the Lower Thames (Johnson et al., 2007). And in this thesis' case study area, the villages of Bacton and Walcott (high coastal erosion risk) are expected to receive multi-decadal protection from coastal erosion, and greater per capital spend, because of their close proximity to the nationally important Bacton Gas Terminal. Although parts of Europe follow a different policy logic (see 2.2.2), the United States (Louisiana) similarly allocates coastal defences in their 50-year Coastal Master Plan to areas with the greatest benefit per number of residents according to cost spent, rather than areas of highest physical or social vulnerability, and not all communities are protected (Hemmerling et al., 2019).

Differing levels of coastal protection lead to an uneven distribution of impacts from coastal erosion or flooding. Impacts include losing property or infrastructure to the sea (as has already occurred in Happisburgh, see section 2.4.2) (Frew, 2012), and disproportionately lower property prices, which can lead to economic impacts and demographic changes, such as a rise in second home ownership (Nicholson-Cole and O'Riordan, 2009). Analysis by Rozer and Surminski (2020) on property in flood risk areas argues that parts of England and Wales are at risk of mass migration out, and blight (economic decline and stigmatisation) of a local area, where many are unable to pay for insurance or get a mortgage. Clean-up after

storm events and the long-term impact of blight can therefore cause sustained financial loss for certain areas (Coastal Partnership East, 2019). Asides from economic impacts, storm events can cause ongoing mental health impacts, stress and trauma for local residents (Day, 2020).

## 3.3.4 Scales in environmental justice

Returning to the environmental justice literature at large, research typically assigns impacts to society in a binary way, for example at a national versus local level (Cooper and McKenna, 2008) or between 'haves' and 'have nots'; 'winners' or 'losers' (Walker and Burningham, 2011; Thaler et al., 2017; Ensor et al., 2021). However, a scale-sensitive lens is of relevance in coastal management, where there may be different social perspectives and different 'trade-offs' depending on a particular scale. For example, Cooper and McKenna (2008) highlight that it is more difficult to justify the allocation of public funding for local level coastal defences, if the issue is considered at a national level, in terms of a funding contribution for all citizens. Furthermore, Chapter 1 highlighted that climate change impacts will intensify into the future, making more and more coastal areas vulnerable (Sayers et al., 2022). Therefore, a wider temporal perspective may also make it harder to justify funding. Consequently, there are trade-offs according to scale in coastal management between social and ecological components, and across spatial and temporal scales (Cooper and McKenna, 2008; Walker and Burningham, 2011; Bopp and Berch, 2021). Figure 3.4 illustrates the different scales and perspectives of relevance.



**Figure 3.4** Trade-offs in coastal management, between different outcomes (red), different parts of the system (yellow) and different scales (blue). Figure 3.4 is a summary illustration of the literature reviewed in this section (Figure: I Cotton).

Figure 3.4 highlights there are many dimensions to an environmental policy or harm that interact. For example, coastal communities could experience positive impacts from coastal erosion policies in the present day, and negative impacts in the future (or vice versa and/or concurrently). It is therefore important to consider positive and negative social and environmental impacts across time and space. While previous literature states some societal groups are impacted by resilience or adaptation strategies, it is often not elaborated how impacts might vary over time or lead to knock-on impacts (Anguelovski et al., 2016). Furthermore, identification of spatial or temporal impacts is but a first step; how impacts are experienced, in an in-depth local context, is also vital for a more nuanced understanding of environmental justice (Bopp and Bercht, 2021). Recent empirical, local-scale research has taken place in international contexts, for example coastal communities in Norway and India (Bopp and Berch, 2021) and in the United States (Hemmerling et al., 2019), but not applied to date in a UK context, and specifically not in the context of sandscaping. The next section of this review turns to adaptation, the third and final concept of relevance in this research.

# 3.4 Adaptation

This section begins by defining adaptation and exploring its links between the resilience and environmental justice literature (3.4.1). Next this section considers how adaptation can be evaluated, exploring the theme of adaptation 'effectiveness' (3.4.2). Section 3.4 subsequently focuses on community perceptions of adaptation, and includes a review of factors that influence public perceptions of adaptation in the present-day (3.4.3) and of future coastal change (3.4.4), given that understanding and preparing for future coastal change is a key policy gap (CCC, 2018: 2021) (as highlighted in Chapter 1).

# 3.4.1 Defining adaptation

Adaptation is the term used to describe how we respond (i.e. the actions we take) to a change in environmental conditions, and is commonly applied in the context of climate change (IPCC, 2014). It is defined by the IPCC (ibid, p.5) as "*the process of adjustment to actual or expected climate and its effects*". Tompkins et al., (2010) similarly frame adaptation as a process rather than an endpoint, and argue adaptation is dynamic, iterative and cyclical. It can take place at multiple scales, be it at an individual, collective or institutional level, and can be indirect, e.g. that serve to build adaptive capacity (the means to adapt, section 3.2.4), or direct, for example an adaptation in itself (Tompkins et al., 2010). While the concept of adaptation is used outside of climate change literature and climate change is but one of numerous causes of stresses in socio-ecological coastal systems, this section largely draws upon such literature, given climate change is argued to be the principle driver intensifying shock events for England's coasts this century (EA, 2020). Secondly, while adaptation can be considered in an environmental sense,

in terms of how other species utilise, exploit or adapt to new conditions (Cooper and Pile, 2014), given this thesis solely explores environmental resilience in terms of geomorphological changes, this review is also bounded in scope to solely consider human, rather than ecological, adaptation.

Considerable crossover can be observed between the concept of adaptation and both resilience and environmental justice. The EA's (2020) FCERM strategy frames resilience as an integral part of adaptation, where the end goal of adaptation should be increased resilience. This can be seen in the strategy's definition of adaptation (described as being "*climate ready*" EA, ibid, p.25): "*changing our lifestyles, economy, infrastructure and local places to make us more resilient and adaptable to future consequences*" (EA, ibid, p.22). Furthermore, section 3.2.3 revealed that there are different types of adaptation responses within a system (for example 'incremental adaptation' and 'transformative adaptation', see Table 3.1), which vary in terms of their level of resilience and depth of change bought to a system (Cooper and Pile, 2014). Adaptation strategies can therefore be understood in terms of differing levels of resilience.

Meanwhile, crossover can be observed between the adaptation and environmental justice literature, with many adaptation frameworks including considerations of justice. For example, the IPCC's (2022, p.49) latest assessment report argues adaptation should be "*effective, feasible, and conforms to the principles of justice*". This is the first time that equity and justice have so closely been embedded in the IPCC's conceptualization of adaptation, where the previous (fifth) assessment report (IPCC, 2014) focused on metrics of vulnerability and resilience. In Adger et al's (2005b) highly adopted framework for successful adaptation, equity is one of four key principles (alongside effectiveness, efficiency and legitimacy), and social justice is one of eight pillars in Nicholson-Cole and O'Riordan's (2009) framework for successful coastal governance in Norfolk.

93

A key area where all three fields intersect can be seen with the concept of maladaptation, which can be defined (using Barnett and O'Neill's widely used definition, 2010, p.211) as "action taken ostensibly to avoid or reduce vulnerability to climate change that impacts adversely on or increases the vulnerability of other systems, sectors or social groups". Therefore, maladaptation refers to adaptation that has either not achieved its objective, or inadvertently caused negative side effects. This links to the environmental justice literature in where, when and whom experiences maladaptation. Leichenko and O'Brien (2019) argue that adaptation measures set with the goal of building resilience can reduce the risk of maladaptation, because the dual focus of resilience on both recovery/capacity building and risk reduction provides multiple co-benefits, citing for example the mitigation and adaptation co-benefits of utilising green infrastructure such as tree planting. However, whether resilience increases or decreases the risk of maladaptation is a key area of debate in the literature, highlighted in section 3.2.3, in the context of transformation.

More recently in the adaptation and resilience literature, there have been calls for 'just' adaptation, that upholds rights and identifies responsibilities for each societal group (Byskov et al., 2019). The term 'just resilience' has also be used in European Environment Agency (EEA) working papers (Breil et al., 2021; Lager et al., 2023), where justice principles are applied to the concept of resilience, but without a commentary on when, where and how these concepts may be compatible. Therefore, there is a growing use of hybridised concepts, such as 'transformative adaptation', 'just resilience' and 'just adaptation', that brings together the fields of resilience, environmental justice, and adaptation, and it is evident the three concepts are important in studying environmental change. However, there is little research examining the compatibility of these concepts, or potential assumptions that such concepts can be hybridised without contradictions. Therefore, further research that examines the relationship between these three concepts would be beneficial.

# 3.4.2 Evaluating adaptation: adaptation 'effectiveness'

As similarly found in this review with the concepts of resilience and environmental justice, there is ambiguity as to how adaptation can be measured. The IPCC's sixth assessment report (2022, p.49) uses a broad definition of adaptation effectiveness as "the extent to which an action reduces vulnerability and climaterelated risk, increases resilience, and avoids maladaptation". Meanwhile, the legally binding Paris Agreement (United Nations, 2015, p.1) calls on nation states to undertake "an effective and progressive response to the urgent threat of climate change", alongside a global goal for adaptation. However, there is no widely accepted and applied understanding of adaptation effectiveness because of the variety of scales, local context, and range of academic disciplines with which it is studied (Dilling et al., 2019; Owen, 2020; Singh et al., 2021). Furthermore, there are numerous potential goals of adaptation (Singh et al., 2021), and the decision of who or what to adapt, and where and why, is highly subjective according to individual judgement and context-specific factors, raising environmental justice implications on how adaptation strategies are designed and implemented (Dilling et al., 2019; Owen, 2020; Singh et al., 2021). Owen (2020)'s systematic review on adaptation practices globally points to six aspects of adaptation effectiveness: increased resilience, wellbeing, adaptive capacity and social/natural system functioning; or reduced vulnerability or climate impact (several of these aspects are present in the IPCC's (ibid) definition). There is ample work that broadly examines adaptation effectiveness (Doswald et al., 2014; Singh et al., 2021), but this has not been applied to the context of sandscaping.

As highlighted in Chapter 2 (section 2.3.2), existing research on beach nourishment and larger-scale mega-nourishment such as sandscaping has focused on the geomorphological response of the coastal system. It therefore considers effectiveness in terms of how such schemes reduce physical coastal flooding or erosion risk (Stive et al., 2013; de Schipper et al., 2016; Hoonhout and de Vries, 2017; Luijendijk et al., 2017; Martell et al., 2020; Bolle et al., 2020; Roest et al., 2021). This view of effectiveness does not consider social factors, as proposed by the IPCC (ibid). De Schipper et al. (2021) propose that beach nourishment evaluations need to go beyond geomorphology, to a more holistic consideration of social, economic and environmental impacts. The authors (ibid) highlight the social impacts of beach nourishment on recreational use of the coast, tourism, and property prices. Therefore, evaluating the coastal management strategy in this thesis' case study requires a holistic definition of effectiveness, that draws upon the experiences and perspectives of relevant stakeholders. To explore adaptation effectiveness, it is necessary to consider how adaptation and coastal change are perceived by the communities affected, which is reviewed next.

#### 3.4.3 Public perceptions of adaptation

Section 3.2.4 has already explored the factors that enable or constrain adaptive capacity to environmental change. This section compliments section 3.2.4, by focusing on intrinsic, cognitive-related factors that shape how people perceive an environmental policy or scheme. Folke et al., (2010) describe social perceptions as 'deep' and 'slow' variables that are significant in determining the eventual adaptation that takes place in socio-ecological systems. From a review of literature, factors affecting public perceptions of adaptation are summarised in Figure 3.5 according to three observable overarching themes; distributional, place-based, and values-based concepts and beliefs. Each is reviewed next in turn.



**Figure 3.5** Factors affecting public perceptions of adaptation responses, grouped into distributional (blue), place-based (green) and beliefs or values-based (red) concepts. Categorisations are not exhaustive, and are based on a review of academic literature to identify observable overarching themes (Figure: I Cotton).

# Distributional concepts

Turning first to distributional concepts (in blue in Figure 3.5), it has long been argued that public acceptability of a policy is influenced by perceptions of fairness, and determines legitimacy (Tyler, 2000). Adger et al., (2016) found legitimacy in adaptation governance, in terms of the extent to which authorities acted sufficiently and extensively, and equity in how flood impacts were spatially distributed, were two key factors affecting community willingness to adapt in two flood prone areas in Cumbria (England) and Galway (Ireland). Lau et al., (2021) similarly found perceptions of fairness, in terms of unequal access to fishing, impact an individuals' attitude, and actions, to coastal resources management in Papau New Guinea. Therefore, adaptation policies that are unjust in terms of equity of impact can be perceived unfavourably. Perceived inequity is found to vary within a

community: acceptability of air quality measures proposed in London were found by Dietz and Atkinson (2007) as determined in declining importance by how it personally affects the individual, their closest friends or relatives, and subsequently according to worldview and values (the latter, values and beliefs, is discussed in further detail below). Furthermore, the authors (ibid) argue that within communities, individuals with the most similar characteristics and experiences of the expected benefits and harms from an adaptation policy will coalesce as a "community of justice" around a set position (Dietz and Atkinson, 2007, p.446). Therefore, it is relevant to explore not only how individuals are differentially impacted, but how they perceive themselves and others are impacted by an adaptation policy, and the perceived effectiveness of the policy itself.

# Place-based concepts

Turning next to place-based concepts (shown in green in Figure 3.5) the influence of sense of place, which describes an individual or communities' impression, connections and emotions of where they live, has been well studied (Wolf et al., 2013; Quinn et al., 2015). Previous research has indicated that communities can be hesitant about changes to their local area which disrupt their sense of place (Quinn et al., 2015). In Clontarf, Northern Ireland, Clarke et al., (2018) report on a local community that recognised the need for, but nevertheless were against, the proposed flood mound and wall because it altered the physical landscape and seascape. Day et al., (2015) similarly argue that policies to manage coastal erosion, such as the demolition of pre-existing hard defences, equally interrupted sense of place for local communities in Norfolk. Sense of place is not necessarily homogenous for all individuals within a community, particularly when temporal questions arise on how to adapt to future climate change (Few et al., 2007). During the implementation of the 2012 Defra Pathfinder programme at Happisburgh, local opinion on the proposed adaptation measures (the retreat of the car park and

dereliction of seafront houses) was split between residents who wanted the local environment to stay the same and residents who felt that change, in order to protect the most vulnerable residents, was necessary (O'Riordan et al., 2014).

Sense of place has been explored in relation to, and appears to strongly correlate with, individual property. O'Donnell (2019) argues residents in coastal Australia exposed to flood risk were more concerned about damage to their property than their local environment. The author (ibid) also found a link between sense of place and climate risk, where residents sceptical of climate change justified their views through sense of place. Previous research in Happisburgh (Tebboth, 2014) revealed some local residents do not primarily attribute accelerated coastal change with climate change; dredging and the removal of coastal defences were instead seen as bigger contributors. This highlights that, alongside sense of place, beliefs about climate change, and the perceived source of the problem, is a key factor through which residents process perceptions of adapting to environmental change, and potential solutions. The following section continues to explore the role of beliefs and values-based concepts (shown in red in Figure 3.5) in influencing public perceptions.

# Values-based concepts

There are numerous studies examining the role of an individual's values in shaping perceptions (and acceptability) of adaptation measures (Adger et al., 2009; NCCARF, 2010; Graham et al., 2014). A study by Prati et al., (2016) found the local community and key stakeholder groups of Portonovo Bay (Eastern Italian coast) had different viewpoints on beach nourishment because of differing individual values and ideas of what needs to be protected. Those who viewed erosion more negatively were more likely to support the intervention, and place less weight on any negative impacts associated with beach nourishment (Prati et al., ibid). The authors (ibid) argue differences in perceptions stem from differing

values on protecting the environment; for some residents, protecting tourism and use of the coast was the most important concern (anthropocentrism), whereas for other residents, allowing coastal ecosystems to function naturally in response to coastal erosion was seen as critical (ecocentrism) (Prati et al., 2016). An individuals' values, and beliefs about climate change, can be influenced by social norms that in turn, influence perceived acceptability for adaptation measures (Tompkins et al., 2010). Bontje and Slinger's (2017) analysis of perceptions of the Dutch mega-nourishment scheme (the Zandmotor scheme) found differences in narrative of the scheme, as told by stakeholders, at the local and national level. Therefore, the voices of local actors and social networks in ones' own locality influence how an adaptation scheme is framed and perceived, and views may differ amongst different social groups.

#### 3.4.4 Risk perceptions and anticipatory adaptation

The CCC's (2018) review of coastal adaptation, and 2021 progress report (CCC, 2021) calls for greater consideration of longer-term adaptation to coastal change. Preparing for future adaptation ('anticipatory adaptation') has been explored within the literature to mixed findings on its advantages and challenges: whilst it is strongly argued in academic literature, grey literature on coastal management, and coastal policy that communities need time to prepare for coastal adaptation (Day et al., 2015; Mott MacDonald, 2016; NNDC, 2016; EA, 2020), other authors have argued an overemphasis on time, and an overemphasis on the future, affects the way the public adapts to climate risks in the present (Nobert and Pelling, 2020). The negative impact of risk reduction on behaviour and perceptions has been studied in international contexts, for example younger residents in Japan who are less concerned than older residents about the risk of tsunamis, which may be due to a lack of direct experience (Arias et al., 2017). Studies by Wolf et al., (2009) and

Nobert and Pelling (2020) on perceptions of extreme heat events for elderly residents in the UK found the majority of interviewees took little preparatory action (for example, retrofitting the house with air conditioning). Instead, the majority of adaptation was reactive during a heatwave event. Although anticipatory adaptation was stressed by local authorities, Nobert and Pelling (ibid) found the focus on future impacts did not resonate with interviewees, who perceived risk in the present moment and felt they could adapt at the start of a heatwave. Lack of anticipatory adaptation to climate change has similarly been found in coastal contexts (Few et al., 2007).

Other studies have shown a variety of factors contribute to social risk perceptions and form a barrier to anticipatory adaptation. Previous research has shown individuals consider society at large at greater risk of climate change than themselves, and therefore underestimate individual vulnerability (Sjoberg, 2000). and some individuals have a negative perception of adaptation, that the problem is too great to mitigate (Kollmuss and Agyeman, 2002). A correlation has also been found between risk perception and age with regards to flooding (Grothmann and Reusswig, 2006; Whitmarsh, 2008; Zaalberg et al., 2009). The relationship between direct experience of an environmental harm and risk perception has also been studied, of which there is a mixed picture. Whilst Whitmarsh's (2008) study on household perceptions of climate change adaptation in Hampshire did not find direct experience of flooding correlated with any greater concern of climate change, or behavioural response (i.e. taking adaptation actions), UK-wide research by Capstick et al., (2015) did find a link between flood experience and concern of climate change. Alongside age, an individual's environmental values were found to be a greater predictor of climate change concerns and adaptation behaviour than experience of flooding in Whitmarsh's (ibid) study. It is therefore relevant to explore social risk perceptions of future coastal change to understand implications for anticipatory coastal adaptation.

# 3.5 Conceptual framework

## 3.5.1 Summary of key gaps across chapter and theoretical insights

This thesis draws upon the fields of resilience, environmental justice, and adaptation to develop a conceptual framework that explores the impacts of sandscaping and implications for coastal resilience. Drawing upon the research reviewed above, and a systems framing as highlighted in section 3.2.2, this research examines in what ways sandscaping can be considered a transformative adaptation strategy (i.e. at what scales and parts within a socio-ecological coastal system). Transformation is the highest resilience category as presented by the authors in Table 3.1. However, sections 3.2.3 and 3.2.5 highlight that this appears based on theoretical assumptions that the greatest amount of systems change leads to increased resilience, with ambiguity on the link between transformation, maladaptation and uncertainty, and how this relates to whether adaptation is deliberate (i.e. anticipatory), or forced. This conceptual framework seeks to critically unpack the relationship between transformation and maladaptation, in three case study villages currently facing two different scenarios: Bacton and Walcott are temporarily protected from coastal risk by the nature-based solution sandscaping, while Happisburgh is not, and currently under a managed realignment SMP policy. As section 3.2.3 indicates, sandscaping is presented in the academic literature as a transformative adaptation strategy to managing the risk of coastal change.

102

This thesis takes an environmental justice lens, to not solely consider how resilience might be built at the system level, but how this varies across components (social, environmental) and scales (spatial, temporal). This research therefore moves from a typical analysis of 'winners' and 'losers' in the environmental justice literature, acknowledging the many trade-offs across scales that has been highlighted by literature in this review (sections 3.2.2 and 3.3.4). This is a research gap for the UK, with recent work on scale-sensitive distributional impacts based in international contexts (as seen in 3.3.4). Rather than 'winners' and 'losers' in where coastlines are defended, it is likely communities may experience a mix of positive and negative impacts by coastal change policies, and trade-offs on different spatial and temporal scales. By drawing upon the lived experience of communities, this research also explores justice issues of participation and recognition in coastal management. Key questions for exploring resilience from an environmental justice perspective, that draw upon the themes of research questions 2a, 2b and 2c (perceptions and experiences of coastal change), include;

- Where within the system is there evidence of increased resilience?
- Similarly, what are the impacts of sandscaping, and how are they distributed across spatial and temporal scales?
- Is there any evidence of (or risk of future) maladaptation?

An analysis of perceptions of sandscaping (research question 2a, Chapter 5), managed realignment (research question 2b, Chapter 6) and long-term coastal change (research question 2c, Chapter 7) allows this thesis to investigate social resilience, by examining factors that enable or constrain adaptive capacity. Adaptive capacity is focussed on, given that definitions of social resilience examined by this review (e.g. Folke et al., 2005; Maclean et al., 2016; EA, 2020; Howarth et al., 2020) describe social resilience using this framing. Therefore, adaptive capacity can be seen as central to understanding changes in social resilience. Section 3.2.4 has shown there are numerous social and institutional factors that affect adaptive capacity and in turn, resilience. While section 3.2.4 revealed that current coastal policy documents focus on factors that enable or constrain capacity to prepare, this review has shown that capacity to recover and capacity to adapt factors are also relevant. In addition, section 3.4.3 has shown there are a number of cognitive-related factors that influence public perceptions of a policy, of coastal change, and willingness to adapt. All of this is drawn upon here, ultimately to evaluate local resident perceptions of sandscaping, of wider coastal change, and connections between the two. This is conducted alongside investigating changes to physical resilience bought about by sandscaping, by looking at beach profile changes and sediment budgets at Bacton and Walcott before and after the implementation of the scheme, alongside cliff retreat estimates for Happisburgh.

This fills several gaps that have been identified in this chapter: a research gap on the social impacts of sandscaping (Chapter 2, section 2.3.2, and Chapter 3, section 3.4.2), an absence in coastal management policy on the specifics of building social resilience (Chapter 2, section 2.2.2, and Chapter 3, section 3.2.4), and a current policy gap in facilitating community preparedness for long-term coastal change (Chapter 1, section 1.1, and Chapter 3, section 3.4.4). In examining the above, this thesis considers the compatibility of merging the concepts of resilience and environmental justice into adaptation. Section 3.4.1 has shown it is commonplace to hybridise these concepts, with less research examining their compatibility. Furthermore, section 3.2.6 of this review revealed the concept of resilience is hard to measure, variably measured, and such ambiguity both leads to and masks disproportionate impacts, with resilience focusing on change for an overall system, rather than at finer scales (Mikulewicz,

104

2019). Considering both an individual, village and system scale in this thesis, by critically analysing social and environmental resilience from a scale-sensitive, justice perspective, allows this thesis to evaluate differences in scale or focus for resilience, environmental justice, and adaptation.

## 3.5.2 Conceptual framework diagram

Figure 3.6 illustrates the key concepts and interrelationships of the conceptual framework. The first yellow box in bold represents the central aim of the thesis, which is to explore the impacts of sandscaping and implications for coastal resilience. Transformation, and its relationship with maladaptation, is represented in the yellow boxes underneath, as a relationship that will be tested in this research. Given that positive and negative impacts of interventions can manifest at different parts of a socio-ecological system (i.e. social and environmental components) and on different scales (temporal and spatial), these four different facets are shown next, in the green boxes. The green boxes represent the scale-sensitive, environmental justice lens of this thesis. The row of red boxes considers how local residents both experience and perceive sandscaping, and in terms of changes to adaptive capacity. The influence of different adaptive capacities, and aspects of coastal management, are recognised in the white boxes at the end of the diagram.



**Figure 3.6** Conceptual framework diagram. Boxes in yellow refer to resilience, boxes in green refer to scalar elements of environmental justice, and boxes in red refer to perceptions of adapting to coastal change. Boxes in white refer to factors that influence adaptive capacity (bottom of diagram) or different subjects of coastal change (left of diagram) (Figure: I Cotton).

# 3.6 Concluding remarks

In conclusion, this literature review has explored the fields of resilience, environmental justice, and adaptation to develop a conceptual framework that examines the impacts of sandscaping and implications for coastal resilience. This literature review has highlighted key questions in the application of resilience to coastal management policy; namely that a systems framing could overlook impacts within the human sphere of a coastal system. Meanwhile, the reasoning that transformational approaches lead to the greatest system resilience, and reduce the risk of maladaptation, has been empirically underexplored. An environmental justice lens and scalar focus allows this thesis to consider how the impacts of sandscaping, and resilience, vary spatially, temporally, and for local communities and the environment. This provides evidence on how issues of justice can be better incorporated into adaptation strategies, such as novel nature-based solutions, aimed at building resilience. The following chapter outlines the methodological approach of the research.
# 4. Methodology

#### 4.1 Introduction

This chapter outlines the overall methodology and research design of the thesis. While an overview of data collection and data analysis is given in this chapter (4.3 and 4.4, respectively), further details about the analytical approach are provided in the relevant results chapters (Chapters 5-7). This structure is used because the results chapters all draw upon different data, and therefore have a different focus in data analysis. The intention is therefore that this chapter provides a top-level overview of methodology, and that this is complemented by the relevant detail provided in each of the methods sections of Chapters 5-7. This chapter opens with an overview of the epistemological position of the thesis (section 4.2), and the choice to use a case study and a mixed methods approach. The research instruments used in this thesis (geomorphological analysis, survey, interviews) are introduced next (4.3), providing justification for why each was selected. Section 4.4 details the overall analytical approach, ending with an overview of the main ethical issues and limitations of the research (4.5).

#### 4.2 Overview of research design

There are three key aspects of this thesis' methodology; 1) an epistemology grounded in critical realism, 2) the use of mixed methods to create an integrated, iterative research design, and 3) a case study approach. The following section outlines why each of these aspects were selected, and their importance to the overall research design.

#### 4.2.1 Epistemological and ontological position

There are numerous ontological phenomena considered in this thesis, which include (but are not limited to) people's perceptions, beliefs, observations, and their interpretations of these; perceptions of community, of others, of self, and of collective responsibility (Mason, 2002). The above are aspects of the social world (Mason, 2002), but this thesis also considers environmental phenomena, such as cliff and beach elevation, beach volume, and natural coastal processes that cause sediment redistribution and coastal change. It therefore considers multiple forms of knowledge informed by the social and natural sciences. To reflect this, the research draws upon a philosophical position of critical realism, and theories of social constructivism, that there is both an objective reality and our socially produced, subjective interpretation of it (Bhaskar, 1975), depending on the phenomena studied (Sumner and Tribe, 2008). Critical realism accepts the world, and 'knowledge', is seen through a human lens (Bryman, 2006), and social constructivism posits that aspects of society are socially produced. This research adopts a paradigm that is a combination of critical realism and social constructivism. It argues, like Wiltshire (2018) and Bogna et al., (2020), that a hybrid philosophical position offers unique broader findings and a richer interrogation of the subject matter than one position alone.

It is necessary for this thesis to have an epistemology that sits between positivism and interpretivism, because the social and natural sciences suit different epistemologies (Bryman, 2006): in the social and natural world, different aspects of reality are being considered. Each is governed by their own set of processes, and choosing a purely positivist or interpretivist standpoint would be inapplicable to one or the other. In other words, believing there is no objective reality (constructivism) would be problematic for studying physical laws of nature, in the same way that believing there is no socially produced reality (positivism) would be problematic for studying social systems (Maxwell and Mittapalli, 2010). Furthermore, how the study is presented to participants, and interactions with the researcher, will influence participant responses. The positionality of the researcher, and how they are perceived by participants, has a bearing on the qualitative results collected in this research.

Maxwell and Mittapalli (2010) highlight the advantage to a constructive middle ground epistemological position, arguing an openness to other epistemologies itself opens up not only a greater opportunity for research findings, but epistemologically-critical findings. Research methods tend to reflect epistemological positions (Bryman, 2006), however this is not a clear-cut distinction, as both quantitative and qualitative research methods can explore social perceptions and actions, and use numeric metrics (Bryman, 2006). It has been argued that critical realism is an epistemology that lends itself to a mixed methods approach, given it can be applied to both quantitative and qualitative stances, but can also highlight where each method has its shortcomings (Maxwell and Mittapalli, 2010). This thesis' epistemology of critical realism therefore lends itself to a suite of particular research methods, which will be outlined in the subsequent section. With regards to this study, LiDAR (Light Detection And Ranging) data is used to analyse coastal processes (quantitative analysis), and surveys and interviews are used to analyse public and policymaker perceptions (largely qualitative analysis). Analysis of coastal processes is considered to be limited by data accuracy, coverage, and the degree to which sampling at a particular point in the year is representative (e.g. of seasonal coastal change). For the latter, there is a 'true reality' of the rate of coastal erosion, limited by data collection and analysis choices, whereas for the former, how people feel about

110

sandscaping will be more changeable, socially produced, and therefore difficult to observe as a 'true reality'.

#### 4.2.2 Mixed methods

A mixed methods approach is adopted to answer the research questions (see section 1.4) on local resident and policymaker perceptions and experiences of coastal change, alongside geomorphological changes. As highlighted above, mixed methods can be deployed where research questions cut across disciplines (and therefore ontological and epistemological positions) (Bryman, 2006) and require multiple research instruments to be answered fully. Given that the interdisciplinary nature of the research requires drawing upon both social and geomorphological data, both quantitative and qualitative research instruments are used to study a coastal system. An often-cited critique of mixed methods research is that it deploys research methods from conflicting epistemological and ontological positions (Bryman, 2006). Drawing upon the arguments by Bryman (2006) and Maxwell and Mittapalli (2010), this thesis combines epistemological and ontological positions, considering they bring unique insights when used together and integrated, as similarly argued above. This thesis uses qualitative and quantitative data, to complement insights gained from each and answer different aspects of the issue of coastal change in the study area (Clark et al., 2023). In doing so, this thesis addresses another common critique of mixed methods, that qualitative and quantitative components of a research project are typically kept separate (Cresswell, 2011). The remainder of this sub-section on mixed methods outlines how the thesis integrates and triangulates mixed methods within the research design.

A key aspect to the research is not solely the use of mixed methods, but the *close integration* of qualitative and quantitative methods, to create a *highly iterative* research design. Figure 4.1 illustrates this process. Coastal geomorphology data is used to inform the study of social perceptions, that in turn situates and highlights the significance of the geomorphological findings. This creates an iterative, circular process between the data, where each informs the direction, or focus, of the other. Comparing findings from the two datasets also provides unique insights not possible by focusing on one alone. Geomorphology and social perceptions are interdependent in the coastal system, so it is therefore pertinent to deploy and integrate mixed methods for the phenomena studied. Furthermore, mixed methods approaches are often used in case study research projects (Bell and Walters, 2014) which is also a key aspect to this research.



**Figure 4.1** The close relationship between geomorphological and social data in this thesis (Figure: I Cotton).

Both Morgan (1998) and Cresswell and Plano Clark (2011) argue mixed methods research can be categorised into various approaches, depending on the degree and order to which research methods are applied. Figure 4.1 illustrates a similar method to Cresswell and Plano Clark's (ibid) Exploratory Sequential Design, where quantitative research methods are applied to inform qualitative research methods. Crucially however, this thesis adopts an iterative process, where each research method informs the direction of future geomorphological and social science analysis. Whereas Figure 4.1 emphasises this iterative nature, this is not as explicitly articulated in either Morgan's (ibid) or Cresswell and Plano Clark's (ibid) mixed methods archetypes. As well as adopting an iterative approach, this thesis also uses triangulation, comparing findings on a particular topic area using a survey and interviews (outlined next). It therefore attempts to do what Morgan (1998, p.372) refers to as "*true triangulation*" where quantitative and qualitative approaches are given equal weight and evolve in the research study at the same time.

#### Triangulation

Drawing findings from multiple research methods, potentially for the purpose of identifying whether both lead to similarly supportive findings, is referred to as triangulation (Bryman, 2006). In this study, both a survey and interviews were conducted to gather social perceptions of coastal change. By applying more than one method and triangulating the social data, unique limitations of a research method can be overcome with the strengths of another (Morgan, 1998), and findings can be consolidated (Bell and Waters, 2014). Another advantage of triangulation is it identifies findings that may be a product of specific biases with a research method (Morgan, 1998).

In this project, a survey was used to capture a range of opinions, and interviews were used to capture depth of opinion. This is a common method in human geography, to allow coverage of both the breadth and depth of an issue (Secor, 2010). Furthermore, understanding the range of public perceptions first (i.e. undertaking the survey before the interviews) allowed for a reflection on the most pertinent issues for local residents, and to design interview questions that investigated aspects arising from the survey. Deploying qualitative research

methods (interviews) after quantitative methods (a survey) is a typical triangulation approach to further explore findings within social data (Bryman, 2006).

#### 4.2.3 Case study

For this research, a case study approach is chosen, consisting of an in-depth analysis of coastal issues at Bacton and Walcott (protected by the sandscaping scheme), but also the adjacent village of Happisburgh, so as to elicit, explore and examine perceptions of locations at a (currently) defended and undefended coast (Mikkelsen, 2005). This is paired with an analysis of geomorphological change at the different locations within the case study area. The major advantage of a case study approach is its depth (Kitchin and Tate, 2000; Flyvbjerg, 2006; Cresswell, 2009), and is a key reason why it is adopted here. As revealed in section 2.3.2, there is minimal research to date on the social impacts of sandscaping. A case study is a suitable approach in this context, where sandscaping is a distinctive example of coastal management (Guest et al., 2013), and has been introduced to a specific stretch of coast (Swanborn, 2010; Guest et al., 2013).

One of the major criticisms of a case study approach is that it is contextual and cannot be considered representative of other situations or locations (Flyvbjerg, 2011); it cannot serve to draw transferrable findings (Flyvbjerg, 2006). Flyvbjerg (2011) argues that the focus on such aspects by critics reflects a wider preference within the social sciences towards theoretical rather than empirical insights, and the emphasis of gaining insights that may be transferable into other contexts. Such a position overlooks the value of case study research, of particular relevance here, which is findings are derived from the context within which they were studied. To date, there is only one sandscaping scheme in the UK, and very few schemes elsewhere in the world (as listed in section 1.4). Comparing the Bacton-Walcott

114

sandscaping scheme to international schemes would be of little relevance to this thesis' research questions, because coastal policy can vary from country to country (for example, coastlines are defended indefinitely in the Netherlands, which is where the first mega-nourishment scheme is located). Hence the rationale for a one case study research design for this project.

As Flyvbjerg (2006: 2011) argues, a case study can be illuminating, if it has original aspects that mean it is insightful. In regards to the research in this thesis, there is an urgent policy need for research on how best to support communities in coastal areas at high vulnerability to coastal change, given the increasing coastal risk facing the UK (as outlined in Chapter 1), and a lack of guidance currently within coastal policy on this topic (as highlighted in section 3.5). Although case studies are geographically limited, this narrow breadth allows a detailed exploration of a subject area (Kitchin and Tate, 2000), which in turns makes it more likely that potential assumptions about a topic area (in this case public perceptions on coastal change) can be empirically tested (Flyvbjerg, 2006). The third research question of this study seeks to understand how the concepts of resilience and environmental justice might align (or not) in an empirical setting (in this case, coastal management policy). This requires the depth offered by a case study approach, which often generates process-orientated findings, in terms of how socio-ecological system components might relate to one another (Cresswell, 2009). Finally, as mentioned in the previous section, case study research often uses mixed methods (Kitchin and Tate, 2011), another key aspect of this thesis' research design.

#### 4.3 Data collection

Geomorphological LiDAR data, a survey, and interviews were the three research instruments used in this research. The following section outlines why each was chosen, the sampling strategy, and the process for data collection.

#### 4.3.1 Geomorphological LiDAR data

There are several methodological options to examine geomorphological coastal change, from manual methods such as walking surveys or photography (aerial and ground) to various different remote sensing methods, which can be groundbased, airborne, or use earth satellites. This study uses a type of airborne remote sensing, LiDAR, which measures elevation of the earth's surface by the time taken for radiation (in the form of visible light) to be re-emitted. Coupled with location data, this produces an output in the form of point cloud data (NOAA, 2023). LiDAR data was chosen, given it is a highly accurate data collection method to monitor changes in the profile of the coast (Gonçalves et al., 2019), and particularly applicable to fine substrate types such as sand, and gentle slopes (EA, 2019), which is the case on the North Norfolk coast where this research is situated. In contrast, alternative methods such as manual surveys and aerial photography can sample a wide area of the coast with accuracy, but are time consuming (Zhou and Xie, 2009). Where historically geomorphological changes were traditionally captured using manual field measurements, LiDAR data is increasingly being adopted as the technology improves (Rumson et al., 2019).

The advantage of remote sensing is the ability to capture a greater range of Aerial Optical Depth (AOD) accurately, and previous research indicates that coastal geomorphological processes at Bacton and Walcott take place up to a depth of - 12 AOD (i.e. below tide) (RH, 2020). Furthermore, LiDAR data has specifically

been applied to look at the effect of beach nourishment on coastal systems (Zhou and Xie, 2009), and is used by RH (who designed the sandscaping scheme) in their post-sandscaping monitoring work (RH, 2020). There is also an open access LiDAR data set collected annually at a national level by the EA, through the Anglian Coastal Monitoring (ACM) programme. Therefore, data to model geomorphological changes from the Bacton-Walcott sandscaping scheme were already available in the form of LiDAR data, which justifies the use of this particular remote sensing method.

#### Sampling strategy

To answer research question 1a, secondary LiDAR data is used to identify how beach profile and sediment volume have changed in the first 4 years of the Bacton-Walcott sandscaping scheme (2019-2022). Analysis of beach profile (Chapter 5) focuses on the villages of Bacton and Walcott, to compare observations with resident's experiences of the scheme, but analysis of sediment patterns (Chapter 7) covers the entire case study area (Bacton Gas Terminal to Happisburgh), to explore where placed sediment is migrating along the coast. Although attribution of geomorphological change to the sandscaping scheme is not possible, a beforeafter-control-impact analysis of sediment changes in this research can nevertheless reveal how the coastal system is responding. A sampling timeframe of 2015-2022 is chosen to account for year-to-year sediment differences, which can vary considerably due to variation in wave, climate and meteorological conditions (Leeder, 2011). Research question 1b analyses cliff retreat at Happisburgh from 2015-2022, similarly to compare observations with resident's experiences of coastal change. For the last part of research question 1 (1c), sediment volume for the case study area during pre-sandscaping years (2015-2018) is compared with the post-sandscaping years (2019-2022), to investigate if there is a statistically significant difference in beach volume that suggests the coastal system has been physically transformed by sandscaping, and to identify where this significant change has occurred.

This research uses two secondary datasets: annual LiDAR surveys produced by the ACM programme, and ad-hoc LiDAR surveys produced by RH. The ACM programme collects regular field surveys of various coastline environmental data including topographic and bathymetric LiDAR, and photography (ACM, 2022). Annual LiDAR datasets began in 2011, with field topographic surveys preceding them since 1987 (ACM, 2022). This is freely available to download (https://coastalmonitoring.org/cco/) as yearly digital elevation models. Data cells were imported into ArcGIS Pro, and combined into a uniform mosaic of the coastline at Bacton and Walcott. The ACM programme LiDAR data have 1 band, 1x1m resolution (cell size) and are 32 bit (pixel depth) floating point data type, extending to a depth of approximately -2m AOD. The data are watermasked (to avoid false elevation readings from the waters' surface). Voids, being generally less than 2m size, were left without interpolation in the digital elevation model, to avoid introducing inaccuracy. The second dataset used in this research is LiDAR and bathymetry surveys (combined into a single digital elevation model) collected biannually from 2019 by SHORE Monitoring & Research, on behalf of RH. Table 4.1 summarises key characteristics of each dataset, indicating why both are required in this research. Crucially, although RH data covers a greater vertical range, it does not extend to Happisburgh, one of this research's case study villages, and data before sandscaping is not available. This thesis requires four years of LiDAR data pre- and post- sandscaping to conduct a before-after-controlimpact analysis. Therefore, this research extends the geomorphological analysis currently being undertaken by RH to identify changes in cliff retreat (Chapter 6) and sediment volume (Chapter 7) from Bacton to further south at Happisburgh.

	Anglian Coastal Monitoring Programme (ACM)	SHORE data by Royal HaskoningDHV (RH)
Time period	Since 2011 – ongoing	2019- ongoing
Access	Open-access, available through the Channel Coastal Observatory website <sup>9</sup> , and the Defra Data Services Platform <sup>10</sup>	Not yet publicly available. Obtained for this thesis through a data sharing agreement
Vertical extent	Topographic and bathymetric (to -2m AOD)	Topographic and bathymetric (to -15m AOD)
Coverage	Entire North Norfolk coastline	Bacton Gas Terminal to 1.2km South of Ostend. Data does not extend past Happisburgh car park

 Table 4.1 Secondary data sources for LiDAR data (Table: I Cotton)

## 4.3.2 Survey

A survey was chosen as the first social research instrument of this thesis, to give an initial, broad feeling of residents' views and experiences of the sandscaping scheme and of coastal change. Key advantages of surveys are the greater number of possible responses, compared to interviews or focus groups, in a shorter period of time (Cresswell, 2009; Secor, 2010). Furthermore, a survey can highlight the main issues for a population from a sub-sample (Cresswell, 2009). This reflects the purpose of the survey in this research, which does not attempt to derive representative findings of public perceptions at the population level . Rather, the aim of the survey was to elicit views on sandscaping, and the main issues of coastal change pertinent to local residents. Whilst the survey contained mostly open ended questions, some were also closed with a series of choice items.

<sup>&</sup>lt;sup>9</sup> Channel Coastal Observatory <u>https://www.channelcoast.org/cco/</u>

<sup>&</sup>lt;sup>10</sup> Defra Data Services Platform https://environment.data.gov.uk/

Another key advantage of a survey, relevant for this case study context, is that it was designed to be anonymous. An anonymous survey can give participants greater confidence in sharing negative perceptions on a topic matter, knowing their opinions are not personally identifiable (Lavrakas, 2008). Coastal erosion and coastal flooding are sensitive topics, with several residents having prior experience of coastal damage (as outlined in section 2.4.2, the 2013 floods inundated 200 properties at Bacton and Walcott). Coastal change could also be a divisive issue, in terms of differing public preferences for how coastal risk should be managed.

One of the major challenges of using a survey is it is highly sensitive to question wording, and unless questions are asked directly by a researcher, one cannot check if a respondent understood the question before results are collected. The implication is that different respondents may interpret questions differently, making it difficult to group and compare answers (Fowler, 2009). However, a trial run (piloting) of a survey can test for understanding, survey length, and allow an opportunity to adjust wording (Fowler, 2009). A pilot survey was arranged with 5 individuals (residents and local councillors from neighbouring villages), all of whom had pre-existing knowledge of the area or the sandscaping scheme, but did not live in Bacton, Walcott or Happisburgh. Care was taken to avoid subjective or vague wording in the design of the survey questions, and questions were tweaked after the pilot test, to improve clarity in wording. Survey questions were initially developed by reflecting on key topics from the conceptual framework, and research gaps identified in the literature review.

#### Sampling strategy

This thesis' case study has a definable sample population, where approximately 2,631 people live in the three villages (population figures shown in Figure 4.2 (ONS, 2011)). The population size of the case study area and a confidence interval

120

were used to select the sample size of the survey. It is typical to adopt a 95% confidence interval in survey research (Lavrakas, 2008), which is a measure of the degree to which a survey sample is representative of the wider population. Therefore, given a population size of approximately 2,631 people and a 95% confidence interval, a sample size of 400-500 surveys would represent an approximate margin of error of 5%, also a typical standard in survey research (Lavrakas, 2008).



Figure 4.2 Sampling strategy and target responses for survey (Figure: I Cotton)

As the number of households is lower than the population estimates (single-person households in Bacton, Walcott and Happisburgh vary from 30-56% (ONS, 2021)) not all households were contacted. This thesis estimated 50% of households in each village would reach 400-500 completed surveys. This was also judged sufficient to capture the range of resident's opinion, where it has been shown new

insights become increasingly limited after a certain threshold, alongside trade-offs with time and resources (Guest et al., 2013). Given previous survey research on public perceptions of beach nourishment yielded a response rate of 22% (Ariza et al., 2014), this survey anticipated a 20% response rate (80-100 completed surveys).

Households were selected through systematic sampling in each village. Flyers advertising the survey were put up in a few locations around the villages (the village hall and village shop), so that residents may already be aware of the research when surveys were delivered. High resolution Ordinance Survey maps from Edina Digimap (https://digimap.edina.ac.uk/) were downloaded and printed, and used to identify every other house where a survey would be delivered. There were a few exceptions which 'reset' the systematic sampling pattern. As this study sought perceptions from local residents only, holiday homes and holiday parks were excluded from the analysis, and houses where it was clear someone did not live there. It was decided that residents would have a stake on coastal change issues through primary residency (either property ownership or rental) that visitors would not. Where such holiday properties were identified, they were omitted from the survey, and the house next door was targeted. Systematic sampling of every other house was chosen to obtain a geographically-distributed range of potential responses within the villages, including residents immediately on the seawall or cliff edge, and those further inland. Previous research (Day, 2020) had adopted a similar sampling strategy in Bacton and Walcott, surveying for residents' views in 2019 prior to the sandscaping scheme, which proved effective.

The survey ran in a paper format, with the option to complete online, through a weblink provided in the information sheet, if preferred by the participant. The online survey was hosted through SurveyMonkey. The researcher delivered all surveys across Bacton, Walcott and Happisburgh over two days (Thursday 27<sup>th</sup> and Friday

28<sup>th</sup> January 2022). The researcher had a large print version of the survey on hand, should it be requested by any resident. Surveys were distributed directly through letterboxes, and participants were asked to leave completed surveys on their doorstop for collection the following week. Where residents were home, the researcher knocked on doors, aiming to introduce the research directly to as many households as possible, and provide an opportunity to ask questions. Care was taken to avoid double-counting respondents by limiting surveys to one household, and affixing a unique identifier number to each survey (if paper version), or to one IP address (if online version) (Gillespie et al., 2015)., following the method of Day (2020).

The survey was designed with neutral response answers, in case of disagreement with other answer options (Gillespie et al., 2015), and question design avoided ambiguity, leading, and double-meaning questions (Bell and Waters, 2014). The survey had 24 questions in total (see Appendix 1 for the survey and accompanying information sheet). The results chapters each draw upon different questions from the survey, topics of which are specified in sections 5.3, 6.3 and 7.3. Residents had a weekend to complete the survey, if they were returning the paper version. If residents were not around, the online survey remained open for a further month. Completed paper surveys were collected from residents' doorsteps (as per instructions in the information sheet) on Monday 31st January and Tuesday 1st February 2022. Given the researcher, on foot, could only be in one village at once, the researcher enlisted support of an assistant during collection, to lower the risk that paper surveys were missed if participants put out their survey later on in the day. A total of 485 surveys were delivered to residents, of which 100 were completed, 86 of which were paper surveys, and the remaining 14 surveys were completed online. This yielded a response rate of 21% overall (20% Bacton village, 22% Walcott village and 20% Happisburgh village), similar to previous surveys of beach nourishment (for example Ariza et al. (2014), who had a response rate of 22.4%). 96% of residents surveyed were primary residents of the area, with 4% second-home owners who live in the area for part of the year. Some surveys were completed and returned for every street in which they were distributed.

#### 4.3.3 Interviews

In this study, interviews were used as an opportunity to explore further issues or aspects raised by the survey, or aspects not covered in the survey questions. Where survey answers indicate an individual's overall position on sandscaping and managing coastal change, interviews provide an opportunity to understand the reasons, and significance of, such perceptions, and provide a richness in understanding resident's lived experience of coastal change that would not be possible to capture to the same depth with an open-text survey question (Kitchin and Tate, 2011; Guest et al., 2013). For policymakers, coastal change is a multifaceted issue, involving many stakeholders and many different policies across government departments, and this complexity would not be grasped without an in-depth conversation on the topic. In-depth qualitative methods are appropriate for exploring views and perceptions (Cresswell, 2009; Lindlof and Taylor, 2011; Kitchin and Tate, 2011; Guest et al., 2013). Interviews are suited to research questions investigating nuances in public views of an issue and are commonly applied when seeking to further understand participant perceptions and underlying motivations (Lindlof and Taylor, 2011). Although one disadvantage of interviews, in contrast to an anonymous survey, is it requires participants to 'present themselves' with their opinions, interview questions in this study were designed sensitive to question wording on coastal change issues. Moreover, questions were sent beforehand to participants, allowing time for participants to become familiar with the research.

124

Given this research is primarily concerned with what people think, and why, as opposed to how they live, participant observation, participatory-based, and other ethnographic methods were less appropriate for this study. While focus groups can be used to explore public perceptions, interviews are a particularly useful research method to explore sensitive topics (Guest et al., 2013). Some aspects of the topic of coastal change (e.g. policy support) is contentious, political, and will be perceived differently by different interviewees. It was felt policymakers and residents would not be as forthcoming in a focus group setting, or may serve 'diplomatic' views they perceive to be more socially agreeable (Kitchin and Tate, 2011; Guest et al., 2013). By contrast, an individual interview provides a space for participants to discuss the issue without potentially triggering different or upsetting views.

A semi-structured interview technique, for both local residents and policymakers, is used in this research, with a pre-designed list of questions. A semi-structured (rather than unstructured or fully structured) approach was deemed the most suitable in this research context, where there are relevant aspects to ask all interviewees as common questions, but space for unscripted questions, suitable for understanding individual experiences of coastal change (Guest et al., 2013). Semi-structured interviews therefore allow the advantages of both a structured and unstructured interviewing approach (Kitchin and Tate, 2011). Another key advantage is the method allows for some degree of participant framing of the issues discussed, in recognition that the positionality of the interviewer and the framing of questions will dictate what is being said to a certain degree (Lindlof and Taylor, 2011) (positionality is further discussed in section 4.5). It is challenging to predict all relevant questions prior to an interview, and spontaneous, follow-up questions can often reveal unexpected insights, as the participant leads the interviewer through their perspective on the issue (Guest et al., 2013)

Furthermore, time in the interview dedicated to free-flowing conversation can lead to the richest qualitative data (Lindlof and Taylor, 2011).

Interviews with local residents comprised of 10-12 initial questions, split into two parts. The questions and information sheet for Happisburgh residents were slightly different, with some questions about sandscaping omitted for lack of relevance, given the village is outside the area of the scheme. Interviews with policymakers had an initial 18 questions, split into three parts, but similarly some questions were omitted depending on if the policymaker had local, regional or national expertise. Overall, interviews varied in length from 20 minutes to 90 minutes. See the appendices for the interview schedules and information sheets for Bacton and Walcott residents (Appendix 2), Happisburgh residents (Appendix 3), and policymakers (Appendix 4).

The interview schedule was designed with neutral probing (Lavrakas, 2008) to minimise leading the participant. All questions were open-ended, and interviewees were sent a copy of the questions in advance (with the email noting the interview would not follow a rigid format, questions were a guide only). Given the sensitivity of the issue, sending a copy of questions in advance was important so resident interviewees could understand the research better and what topics the interview would cover, and therefore would feel more comfortable. For policymakers, pre-existing questions helped to focus the interview, given the vast possible questions on coastal policy. While no questions were removed after the first few interviews were conducted, question wording was edited sightly to improve clarity. As part of the iterative approach of the research design, policymaker interviews were conducted after resident interviews, to identify the main themes to which it would be beneficial to have contrasting resident and policymaker perspectives. Interviews with local residents were undertaken in Autumn 2022, and interviews

with policymakers early the following year (2023) (see section 4.3.4 for more information on the timing of each research method in this study).

## Sampling strategy

An option to leave contact details for further information about the interviews at the end of the survey yielded 37 contacts. Survey responses for these contacts were analysed, and participants were contacted in priority order to allow for a mix of residents from each village, with different views and experiences of sandscaping and coastal change alongside varying age, gender and length of residency in the villages. As illustrated in Figure 4.3, further participants were recruited through; snowball sampling with interviewees, community gatekeepers (introduced via interviewees), advertising on community pages, and from speaking to residents around the villages.



**Figure 4.3** Sampling strategy and target responses for resident interviews (Figure: I Cotton).

In total, 30 residents were interviewed. At the participants' choice, 19 interviews took place in-person at a local village hall or café, eight interviews took place over the phone, and one online. The remaining two interviews were conducted outside on the seafront, as a walking interview, at the participant's preference. It was not possible to record (and therefore transcribe) these two interviews due to weather conditions, and so resident interview data (i.e. coding totals) reported in the empirical chapters do not include these two walking interviews. The resident interviews includes one instance where two residents (of the same family) preferred to be interviewed at the same time for logistical reasons, but each interviewe equally answered the interview questions and was coded separately. Interview data collection stopped at the point of thematic saturation and no further leads on participants.

In contrast, policymaker interviewees were identified through purposive sampling. Purposive sampling is suitable in this context, where there is a small and finite pool of relevant policymakers with a remit on sandscaping or coastal change issues in this case study context. Relevant policymakers were identified in collaboration with NNDC, the main gatekeeper of this project, who could provide introductions to the research via email. Given policymakers' role in the civil service or public sector, contact details were publicly available. In total, six policymakers were interviewed, comprising of two local, two regional and two at a national level, to get perspectives across policy jurisdictions and scales. At the participant's permission, policymaker interviews were conducted online, for logistical reasons (unlike residents, policymakers were based at different parts of the country, making inperson interviews challenging). Furthermore, at the time of the interviews (Winter 2022/2023), remote meetings were still commonplace following the Covid-19 pandemic. The 6 policymaker interviews includes one instance where two policymakers (of the same organisation) preferred to be interviewed at the same time, but each interviewee equally answered the interview questions and was coded separately.

## 4.3.4 Summary and timeline of research methods

Figure 4.4 shows the order and timing of data collection, as well as how each research method informed subsequent stages of data collection. Figure 4.4 demonstrates there is an iterative logic to the order of the geomorphological analysis and social research instruments (black arrows). But the diagonal arrows and italics free text illustrates some of the key themes that arose at each stage that also informed *across* the natural and social science analysis.



Figure 4.4 Timeline for research instruments used in this thesis (Figure: I Cotton).

## 4.4 Data analyses

This section describes the overall scope of the analysis, and justifies the analytical approach and the software used. More detail on the specific steps to geomorphological analysis (and for the survey and interview data, the topics analysed) can be found in the empirical chapters (sections 5.3, 6.3 and 7.3).

#### 4.4.1 Geomorphological analysis

ArcGIS software was chosen to analyse LiDAR data, given it is a common software choice to analyse digital elevation models (Adhikari et al., 2016). This thesis predominantly used ArcGISPro (Esri, 2023), given it is a more sophisticated analysis software than ArcMap, with greater tools and functionality. While analysis was conducted in ArcGIS, Microsoft Excel was used to create graphs of geomorphological change, and to carry out statistical tests. This research identified three key geomorphological dynamics of the coast that local residents would observe and experience, but also would be impacted by sandscaping, and therefore could be investigated to explore whether sandscaping has physically transformed the coastal system. These were: cross-shore beach profile (Chapter 5), cliff rate of retreat (Chapter 6) and beach sediment volume (Chapter 7). See 5.3, 6.3 and 7.3 for an outline of how each geomorphological analysis was conducted in ArcGIS.

#### 4.4.2 Closed-text survey questions

Survey questions and answers were manually inputted into a password protected Microsoft Excel sheet from the collected paper surveys and downloaded online surveys. Descriptive statistics were calculated in Excel for close-ended survey questions (i.e. questions that had predefined answer categories rather than a comment box), and used to create either a summary table or graph (pie chart or bar graph). See Appendix 5 for a breakdown of survey results by question. For open-text survey questions, responses were imported into NVivo. NVivo was chosen to store and analyse open-text survey questions, to allow this research to more easily compare codes and themes arising from the survey and interview data, as interview transcripts were also imported into NVivo. Both open-ended survey data and interview data were analysed using the same coding process and qualitative analysis (thematic analysis), which is detailed next.

#### 4.4.3 Open-text survey data and interview data

Interviews were recorded using a dictaphone and transcribed verbatim using Microsoft Word's transcription function (at the interviewees permission) and imported into NVivo. The transcription was manually checked and edited by the researcher for any inaccuracies to the audio recording. For policymaker interviews conducted over Microsoft Teams, an automatic transcript of the meeting was generated, and corrected manually for any errors. A separate folder was created in NVivo (release 1.6.1-1.7.1) (QSR, 2022) for interviews with Bacton, Walcott and Happisburgh residents, alongside policymakers, so that responses could be compared between villages and between residents and policymakers. The survey data was kept in a separate folder, with an anonymous survey ID number allowed for grouping answers by village.

Qualitative survey data and interview data were analysed using thematic analysis. Thematic analysis is a widely used analytical approach in the social sciences that identifies the main themes, or key ideas of significance, within a dataset (Braun and Clarke, 2006). Other approaches exist to analyse qualitative data, such as interpretative phenomenological analysis and narrative analysis, but these approaches are principally focussed on how data is communicated by participants, rather than the subject matter of what is being discussed (Braun and Clarke, 2006). Thematic analysis is therefore the more appropriate analytical approach for this thesis, given the research questions of this study aim to identify key issues in coastal management in the case study area (and therefore a focus on subject matter, rather than how it is told). Nevertheless, given this thesis takes a social

constructivist position to epistemology, coding was mindful of both what is being described by participants, and how.

A three-step coding technique was used (i- initial, ii- focused, and iii- theoretical) advocated by Charmaz (2006), and developed from a grounded theory analysis approach by Glaser & Strauss (1967). Initially responses were read to simply identify what was being said. Next, this was used to develop groups of similar content, hereafter referred to as 'codes'. In comparison to step 1, step 2 moved from simply describing data to grouping data according to shared meanings or significance. The researcher returned to survey responses or interview transcripts and coded text according to step 2 codes. This process involved iterations, through repeated reflection of the data and discussions with the supervisory team. The step 2 codes are the principal codes analysed in this thesis (and for the interview data, form the developed coding framework, see appendix 6), but are considered alongside step 3 codes. Step 3 codes were the final step of coding, and theoretically based, where the researcher returned to the thesis conceptual framework and literature review to identify codes that connect the data to the conceptual framework, alongside other pre-existing theories or frameworks. Following coding, similar step 2 codes were grouped (referred to as 'parent codes'), and according to overarching themes.

These steps to coding were taken so that the researcher could evolve from more descriptive to interpretive coding, whilst staying 'close' to the qualitative data (Coffey and Atkinson, 1996). This allowed for both inductive (empirically driven) and deductive (pre-determined) codes, which allowed the research to consider unexpected themes that may not be included in the thesis framework or pre-existing research (an advantage of inductive coding), but also to connect and situate research findings to the research framework and existing research (an advantage of deductive coding) (Braun and Clarke, 2006). Figure 4.5 illustrates

this approach to coding. Although qualitative survey data and interview transcripts were coded and analysed using the same approach, separate codes were developed for each dataset, given the research methods were conducted sequentially and focussed on different topics.



**Figure 4.5** Steps involved in the coding process, leading to the development of a coding framework (Figure: I Cotton).

## Overview of data presentation in this thesis

Figure 4.6 is an illustrated map of the data presented in this thesis. Each results chapter draws upon a different mix of data, across the different research methods. The methods section in each results chapter (5.3, 6.3 and 7.3) details the topics covered and analysis steps for social and geomorphological data in that chapter.



Survey

**Figure 4.6** Map of data presented in each results chapter. The different research methods (survey, interviews, and LiDAR analysis) are sub-divided into data groups. For the social data, these groupings are Bacton & Walcott residents (yellow), Happisburgh residents (green) and policymakers (red). For the geomorphological data, these groupings are according to the topic of analysis: beach profile (light blue), beach volume (medium blue), and cliff rate of retreat (dark blue) (Figure: I Cotton).

#### 4.5 Research ethics and limitations

#### 4.5.1 Ethics

This research obtained ethical clearance (application number ETH2122-0192) from UEA DEV S-REC ethics committee in December 2021 prior to fieldwork, which as part of this process included completed risk assessments and contingency methodologies in case of disruption due to Covid-19 (See Appendix 7 for confirmation of the approved ethics submission). The following section details the main ethical issues pertinent to the research methodology.

#### Risk to participants

As described in chapter 2 (section 2.4.2) the villages of Bacton, Walcott and Happisburgh have previous experience of flooding, previous properties lost to erosion (BGS, 2021), and some residents reporting trauma from previous storms (Day, 2020). This research therefore explores a very sensitive personal topic, and the survey and interview schedule were designed sensitive to this, with no unnecessary questions asked. Although parts of the interview discussion were expected to be positive (i.e. the coastal protection benefits of the sandscaping scheme), previous research on Happisburgh highlights the emotive views of coastal erosion (Day et al., 2015). Therefore interview questions on future coastal erosion risk (the most sensitive topic) were asked towards the end of the interview, after more straightforward questions on perceptions of sandscaping. Research training was also completed, prior to fieldwork, on how to engage with communities on sensitive research topics. It was explicitly stated in the survey and interview information sheets that taking part is voluntary, and that participants could terminate the interview at any stage, should they wish to do so.

#### Informed consent

Participants require free, prior and informed consent in order to ethically engage with research (Sieber and Tolich, 2012). Separate information sheets for the survey and interviews were created, to explain the purpose of the research, topics covered, time asked of the participant, and to confirm anonymity and voluntary participation (Gillespie et al., 2015). Information sheets were checked against the ethical research guidelines outlined in DEV at UEA. Participants taking part in the interview were asked to sign, date and return consent forms to the researcher. Where written consent was not possible (in rare cases, where an interviewe couldn't sign for a practical reason), verbal consent to take part in the interview was given by the interviewee. Participants had an opportunity to ask further questions about the study at any time, with the researcher's email address provided. Participants were able to withdraw from the research at any point until submitting the survey or taking part in the interview.

#### Data use, storage and retention

Participants have the right to anonymity in analysis and write-up (Sieber and Tolich, 2012), with only anonymised excerpts used in the thesis. For interview participants, permission to record the conversation was explicitly asked at the beginning of every interview, and was explicitly stated on the information sheet. Interview transcripts were kept on the researcher's personal computer, and not shared with anyone else. Survey questions did not ask for participants name, address or contact details (a contact detail was optional at the end should participants want to participate in the interviews). The only personal data collected from the survey were a participant's age, in a closed-ended age bracket question format. Once collected, the survey responses were therefore not personally identifiable, as this was not necessary for the research. Contact details were stored securely on a password protected file in accordance with the General Data

Protection Regulations (GDPR), with the minimum amount of personal data necessary (ICO, 2018). It was stated in the survey and information sheet that contact details, should they be provided, were kept by the researcher alone, and not viewable or shared with anyone. Transcripts and survey data will be kept no later than necessary for the submission of the thesis and associated publications. The data policy for SurveyMonkey, which hosted the online survey, was checked beforehand in terms of complying to GDPR laws for personal data.

#### 4.5.2 Limitations

The last sub-section of this chapter outlines the main limitations to the research design. Chapter 8 (section 8.5) returns to many of the limitations outlined here, discussing these in light of the research findings, and in relation to future work. The principal limitation to the study is that it analyses a coastal management scheme (sandscaping) in its first 4 years (2019-2022), rather than its full lifetime (expected to be 20 years). This simply reflects the timeframe of doctoral research (3.5 years) and the year in which sandscaping was implemented (2019). Therefore, this thesis can only provide early indications on the impact of sandscaping and changes to resilience. There is nonetheless still a rationale to researching sandscaping in the early years of the scheme, given it is being replicated in other parts of the world (as noted in Chapter 1, section 1.4). Furthermore, some social and geomorphological impacts as a result of sandscaping are likely to be instantaneous or apparent within the first four years of the scheme, and therefore an early study is still beneficial.

This doctoral project took place (2020 - 2024) during an unprecedented time for research, where there were significant policy restrictions due to the Covid-19 pandemic. This delayed the researcher travelling to the case study area, and the

start of fieldwork. Fieldwork took place shortly after restrictions were lifted (January 2022). However, during the first year of the PhD, it was uncertain what policy restrictions there would be during fieldwork, therefore both an online and paper version of the survey and interviews were prepared, in case of tightening restrictions. This turned out to be advantageous, as the survey was administered in both formats and used by respondents, so it is likely the number of responses was greater than if solely a paper or online version had been used.

There are limiting factors to the research instruments used in this thesis, as also highlighted earlier in this chapter (section 4.3). Findings from the survey and interview data will be limited by the sample size and number of willing participants, and the timeframe available for fieldwork. Respondents who chose to take part may not disclose certain opinions, emotions, or experiences, given sensitivities of the topic area, or may express what they believe is the 'correct' view (Gulbrium and Holstein, 2000). It is therefore important to be mindful of potential biases, inaccuracies, or undisclosed perceptions associated with the social data. This was not notable during the fieldwork, but it is possible that survey and interview participants may have chosen not to disclose certain perceptions (for example, about the sandscaping scheme and coastal change). There is also a risk, particularly for the survey data, that participants may have misinterpreted the question. Triangulation of the survey and interview data allows the researcher to compensate for this, if it occurs, by gathering multiple data on a research topic and exploring themes using multiple research instruments.

Limitations with the geomorphology data generally relate to accuracy errors, with the associated errors in generating LiDAR data forming the largest total error in LiDAR analysis (Zhou and Xie, 2009). However, for the research questions in this thesis, understanding general trends on how beaches are changing, rather than absolute accuracy on beach profile, beach volume or cliff retreat values, is sufficient. Therefore, accounting for variation in how LiDAR data is generated by the ACM programme and by RH across years, and the influence of past meteorological conditions and storm surges on sedimentation patterns, while noteworthy from a geomorphological perspective, are less relevant in the context and scope of this research. Of relevance to comment on is the LiDAR data availability, which is one dataset per year for the ACM data, at varying times during the Winter season each year. This inhibits any analysis of seasonal changes, and furthermore the data is of a limited vertical extent to –2m AOD. The implication is that ACM LiDAR data may reveal sediment loss on the upper beach, when sediment gain may be undetected below the foreshore (beyond –2m AOD). Using the ACM LiDAR data in combination with the RH data overcomes this limitation, with the latter capturing elevation change to a depth of approximately –15m AOD.

Lastly, it is important to note the influence of the researcher's positionality in the overall research. Every researcher will have biases in how research is analysed and conducted depending on their academic background (Cresswell, 2009). As highlighted above in section 4.2.1, an interdisciplinary research study that draws upon a critical realism and social constructivism epistemological position allowed the researcher to consider findings across disciplines. Furthermore, a researcher's personal characteristics will have a bearing on how they, and the research, is perceived by participants (Folkes, 2022). A field diary was kept during fieldwork, to introduce a reflexive exercise to the data collection. By recording reflections with how the fieldwork went, the researcher was able to externalise reflections during data collection that may have a bearing on research findings (Folkes, 2022, Smith et al., 2021). The fieldwork diary was returned to and considered during the data analysis stage.

## 4.6 Concluding remarks

This chapter has provided a background to the methodology and research instruments used in this research. It has summarised the overall approach and research design of the work, in relation to data collection and analysis, as well as highlighting the main methodological constraints to the study. The following chapter (Chapter 5), is the first empirical chapter of the thesis, and presents findings in relation to research questions 1a and 2a.

# 5. Understanding perceived 'effectiveness' of the Bacton-Walcott sandscaping scheme

This chapter is based off the publication; Cotton, I., Forster, J., Lorenzoni, I. and Tolhurst, T.J. 2022. Understanding perceived effectiveness of a novel coastal management project: The case of the Bacton-Walcott sandscaping scheme, UK. Frontiers in Marine Science, **9**, Article 1028819. <u>doi.org/10.3389/fmars.2022.1028819</u>. IC's role in the paper: conceptualisation, methodology, data collection and analysis, writing (lead), review and editing

## 5.1 Introduction

This is the first empirical chapter of the thesis, and presents results relating to the perceived effectiveness and impacts of sandscaping for Bacton and Walcott residents. As highlighted in Chapter 2, while nature-based solutions are increasingly being promoted over hard defences, there is a lack of empirical research on the effectiveness of novel approaches such as sandscaping, which are deployed at an innovative (larger) scale. This chapter explores research question 1a, in terms of analysing changes to beach profile, and research question 2a, by seeking to understand the range of social impacts residents may experience, and how 'effectiveness' of sandscaping is understood locally. This chapter therefore considers effectiveness from multiple perspectives (as advocated by the IPCC's (2022) definition of adaptation effectiveness<sup>11</sup>, section 3.4.2); both from a geomorphological perspective, in terms of the ability of sandscaping to reduce flood and erosion risk, and from a social perspective, drawing on the social experiences, impacts and perceptions of local residents over

<sup>&</sup>lt;sup>11</sup> The IPCC (2022, p.49) defines adaptation effectiveness as "the extent to which an action reduces vulnerability and climate-related risk, increases resilience, and avoids maladaptation".

time. Section 5.2 provides a brief overview of existing research on public perceptions of beach nourishment, and background to public communication of the sandscaping scheme in 2018 and 2019. Section 5.3 outlines the data presented in this chapter, and in particular the methodological approach to calculate beach profile changes. Sections 5.4 and 5.5 present and discuss the main findings, respectively, before a concluding summary of the chapter (section 5.6).

#### 5.2 Background context

#### 5.2.1 Public perceptions of beach nourishment

This subsection provides additional detail to the literature review (section 3.4.2-3.4.3) on public perceptions and perceived effectiveness of adaptation, by focusing here specifically on perceptions of beach nourishment and coastal nature-based solutions. As highlighted in section 2.3.2, there are few studies to date on local residents' perceptions of larger-scale mega-nourishment schemes such as sandscaping, with existing social research focusing on its unique governance and partnership approach (Vikolainen et al., 2017; Clipsham et al., 2018; Johnson et al., 2020b) or from the perspective of specific recreational users, such as wild swimmers (Radermacher, 2018).

Previous research on smaller-scale beach nourishment and nature-based solutions reveals a divergent range of views amongst stakeholders (Ariza et al., 2014; Prati et al., 2016), influenced by lived experience (Prati et al., 2016; Usher, 2021), and in some contexts, low levels of awareness of the scheme (Marin et al., 2009; Roca and Villares, 2012; McKinley et al., 2020). The range of views amongst stakeholders particularly relates to the perceived objectives of beach nourishment
(Ariza et al., 2014; Prati et al., 2016). Prati et al. (ibid) found that local stakeholders (that use, work near or study the nourished beaches of Portonovo Bay, Italy) who perceive erosion as a strongly negative process were more likely to support beach nourishment and were less concerned about any negative impacts. Further, Usher's (2021) survey of recreational surfers in Virginia, US, found varying perceptions in different areas, with more negative perceptions for beaches that have been more heavily nourished, suggesting beach nourishment has a greater impact on wave quality for surfers over time. Marin et al. (2009) explored local residents' perceptions of locally nourished beaches in northern Italy, finding low levels of awareness, with over half of respondents having not previously heard of beach nourishment. Of those aware of the scheme, 56% had negative perceptions of it, due to changes in sand grain size, and concerns over project cost, effectiveness (in achieving its objectives), and water quality. Other negative perceptions of beach nourishment by local residents include disliking wider beaches, despite the overall benefit of increased beach size (in Gold Coast, Australia) (Todd and Bowa, 2016).

Asides from beach nourishment, there have been several studies on public perceptions of other coastal nature-based solutions, which similarly show both a diversity of public views and low public awareness in certain contexts. McKinley et al's. (2020) national public survey of saltmarshes in Wales found uncertainty about their function and purpose: 15-40% of respondents selected 'unsure', when asked a series of questions on different benefits of saltmarshes. Roca and Villares (2012) also found low levels of knowledge on managed realignment amongst stakeholders who work or use the Ebra Delta in Spain, alongside a diversity of opinions, and low trust in policymakers. Indeed, other studies have found that trust in policymakers, in particular, is a key barrier in public acceptance of coastal nature conservation initiatives (Milligan et al., 2009) and of nature-based solutions

144

generally (Anderson and Renaud, 2021). Methods of public communication was found by Schernewski et al. (2018) to be a key factor in whether coastal residents in Germany trust adaptation strategies.

Furthermore, existing research has found lower levels of perceived effectiveness of nature-based solutions when compared to hard defences. Roca and Villares (ibid) found that over 70% of stakeholders surveyed perceive a strategy of managed realignment as less effective (in reducing coastal erosion risk) than hard defences. Studies on nature-based solutions typically consider 'effectiveness' largely in terms of reduced coastal risk or increased coastal protection (Roca and Villares, ibid; Gray et al., 2017; Anderson and Renaud, 2021). Gray et al. (2017)'s interviews with coastal residents in the US state of New Jersey similarly revealed a perception that hard defences are more effective than natural infrastructure (i.e., nature-based solutions such as dunes), which suggests hard defences may be the preferred option to manage coastal change. Anderson and Renaud (2021) argue that nature-based solutions are 'judged' to a higher standard (for example in terms of effectiveness, value for money, and impacts) than hard defences, and argue that policymakers need to sell the benefits of nature-based solutions more persuasively. This is particularly crucial in a coastal context, where hard defences can increase the risk of erosion for adjacent coastal areas, and therefore are not always a viable option (French, 2004).

## 5.2.2 Public communication on the Bacton-Walcott sandscaping scheme

Section 2.4.3 outlines the Bacton-Walcott sandscaping scheme in detail, including the modelled expectations of the scheme, that placed sediment would migrate cross-shore and alongshore over time, but largely remain within the coastal system rather than migrating offshore (NNDC, 2022b). This subsection briefly provides additional context on what was communicated publicly in the years before sandscaping was implemented (2019) about how Bacton and Walcott beaches were expected to change. Several public engagement events took place before and during the implementation of sandscaping, which included a local liaison group of community members, village drop-in events, information letters to Bacton and Walcott residents, and temporary public information stands in the area (NNDC, 2017), alongside reporting in local and national news (for example BBC, 2017). The communicated objectives of the scheme include protecting the nationally important gas terminal and the adjacent villages of Bacton and Walcott from coastal erosion and flooding (RH, 2018a), and to "*provide time to the communities to adapt to coastal change*" (Johnson et al., 2020b, p.39). In this regard, sandscaping is framed as a nature-based coastal management strategy, that can also facilitate adaptation to future coastal change, given its predicted 20 year timeframe of coastal protection (Johnson et al., 2020b) (themes explored further in Chapter 7).

### 5.3 Materials and Methods

#### 5.3.1 Data collection

#### Social data

This chapter draws exclusively upon the survey responses (77) and interviews (22) of Bacton and Walcott residents, and compares these perspectives with geomorphological observations of beach profile change at the Bacton and Walcott frontage. Bacton and Walcott residents are the villages closest to the sandscaping scheme, and unlike Happisburgh residents, are within the area of intended coastal protection from the scheme (and so are focussed on in this chapter). Chapter 4

details the sampling strategy for the survey (section 4.3.2) and interview (4.3.3) data collection, and a copy of the survey and interview questions can be found in the appendix (Appendices 1 and 2). Themes and topics reported on in this chapter, which form the basis of the structure of the results section (5.4), include:

- Do local residents perceive the Bacton-Walcott sandscaping scheme to be performing effectively (in the present day and the future) (section 5.4.1)?
- How do residents' perspectives compare with geomorphological observations (section 5.4.2)?
- How have residents been (positively or negatively) impacted by the sandscaping scheme (section 5.4.3)?
- What is the level of trust in coastal management decision-making for Bacton and Walcott (section 5.4.4)?

Subsequent to the survey data collection and analysis, semi-structured interviews were conducted in October and November 2022. The 22 interviewees included both participants that did and did not take part in the survey. Interview questions reported on in this chapter include further questions on the sandscaping scheme, such as how the beach profile has changed, perceived effectiveness of sandscaping as a coastal management strategy, official communication about the scheme, and communication preferences for the future.

## Coastal data

To examine how the local coastline and geomorphology has changed at Bacton and Walcott since sandscaping was implemented, changes in the elevation and profile of sediment on the beach were analysed. Two sets of secondary data were used, covering areas of the coast to varying depths from the upper (dry) beach to the subtidal zone, over the period 2018-2022 (for ACM programme data) and 2020-2021 (for RH data, the more limited timeframe of which reflects data availability). Section 4.3.1 outlines the characteristics of both datasets, and justifies why both are needed to analyse changes to beach profile at Bacton and Walcott: crucially, the RH data samples elevation to a greater vertical extent (-15m elevation, sampled to up to -9m here) than the ACM programme data, revealing geomorphological changes below the foreshore. In addition, the ACM programme data, which has archives from 2011, is used to illustrate beach profile in the year before sandscaping was implemented (2018).

#### 5.3.2 Data analysis

#### Social data

As detailed in Chapter 4 (4.4.2 and 4.4.3) survey responses were inputted into Microsoft Excel to produce descriptive statistics and summary charts for quantitative, closed-ended questions, while open-ended survey questions and interview transcripts were thematically analysed. Details on the 3-step coding technique to thematic analysis (i- initial, ii- focused, and iii- theoretical, advocated by Charmaz (2006), and developed from a grounded theory analysis approach by Glaser & Strauss (1967)) in particular are detailed in section 4.4.3. Triangulation of the survey responses and interview transcripts was used to gather data about perceived effectiveness from multiple research instruments. In particular, the interview data allowed greater depth of insight into perceived effectiveness and the ability to clarify any ambiguity in certain survey responses.

## Coastal data

Digital elevation models of the secondary LiDAR data were viewed and analysed in the mapping software ArcGIS Pro. ArcGIS Pro was chosen over ArcMap, because it offers greater functionality to draw line transects in ArcGIS that was needed for the analysis. To calculate changes in beach profile before and after the implementation of sandscaping, aerial photography (available as an ArcGIS Pro basemap layer) was used to identify the location of the sea wall at Bacton and Walcott, which was used as the starting point for line transects, in a seawards direction. The most popular areas of Bacton and Walcott beach, as identified by aerial photography, were sampled (starting at Bacton car park in Bacton and opposite the amenities on Coast Road in Walcott). Perpendicular line transects to the sea wall were generated, spaced 100m apart. In total, 9 line transects were used to sample the beach in each village. After trialling different sampling frequencies, elevation was sampled at points every 1m along each line transect. Data was exported to Excel and an average for each set of 9 transects sampled at Bacton and Walcott was used to create a line graph of cross-shore beach elevation changes. This analysis was conducted for both the ACM programme and RH LiDAR data.

## 5.4 Results

#### 5.4.1 Perceived effectiveness of sandscaping

This section draws upon findings from multiple open-text survey questions and interview transcripts, that reveals the range of opinions amongst Bacton and Walcott residents on the effectiveness of sandscaping. 'Effectiveness' is considered by local residents in terms of the ability of the scheme to protect from the risk of coastal erosion and flooding, but also includes residents' observations of the scheme (how it is functioning, whether it is functioning as expected, and effects over time). Table 5.1 lists the themes of perceived effectiveness expressed in the survey and the number of respondents that mentioned each theme. Table 5.2 presents thematic codes relating to perceived effectiveness expressed in

interviews. Themes are not mutually exclusive in both sets of data, with multiple

themes referred to by a single participant.

**Table 5.1** Thematic analysis of answers across the survey's open-ended questions, summarising themes which relate to perceived effectiveness of the sandscaping scheme. The different initial codes (column 2) are grouped together according to theme (column 1). Themes are ordered by frequency of appearance amongst survey answers. Questions asked include; 'Why do you think the sandscaping scheme will impact you and your village positively or negatively in the future?', 'Has the sandscaping scheme altered your views on how coastal change (i.e. coastal erosion and flooding) could be managed in your village?' and 'What coastal management, if any, do you think should happen in your village in 15-20 years, which is after the projected lifetime of the sandscaping scheme?' (Table: I Cotton).

Perceived effectiveness – survey findings		
<u>Theme and</u> response rate	<u>Codes/ theme</u> <u>description (similar</u> <u>codes grouped</u> <u>together)</u>	Example quotes from survey
Evidence it works (31% of respondents expressed this theme)	Sand building up, altered beach, feel protected, not experienced any flooding or erosion, less storm damage, technology working, can monitor effectively, better solution to hard defences	"There have been a similar number of storms since, but none have resulted in flood damage" "It appears to be doing what was intended" "sandscaping has been a successful (so far) way to manage erosion, because it works with nature rather than causing problems elsewhere as other methods have done"
Hard defences would have been better (26%)	Doesn't fully protect coast, stop cliff erosion, hard defences sturdier, sandscaping not fully effective, need further defences, not implemented successfully, only partially works	"Still believe rock barriers like Sea Palling are the best solution" "The only way to fully protect Bacton/Walcott and the gas terminals is to build proper reefs. Pumping sand onto beach is a pointless task" "sandscaping did not work in my view - although saving flooding maybe twice. Reefs like Sea Palling has seem to work better."
Doubt sandscaping will last full 20 years (23%)	Lots of sand gone, won't last	"We have not had any flooding/ extreme high tides so it has not been proven" "Not sure how it will look in 5-10 yrs + how it will effect erosion as I was told it only lasts 10 yrs!" "Not sure it will last as long as it is supposed to, high tide is already splashing over the top" "I need a lot more convincing as two years on I am already concerned and we were told it would last 20/25 years"

Observed drop in sand on beaches (16%)	Sand gone from beach / washed away / sand disappearing / might not come back / reduced protection /	"Now that all sand which was pumped ashore has goneIf they continue with this pointless project, then all the same issues will return" "Unsure if the recent loss of sand will return or if it is just offshore" "Most of the sand put to protect Bacton/Walcott has now gone after a short time of protecting"
Sand needs topping up (14%)	Needs topping up / needs funds to maintain	<i>"I think the sand will need to be topped up as we have lost a large amount since it was completed (2019)" "It helps and should be topped up" <i>"It will need to be maintained e.g. Topping up"</i></i>
Changed opinion on sandscaping (5%)	Not aware of sandscaping previously, didn't know would work on this scale, initially sceptical, changed opinion, increased knowledge	"Always felt there was little could be done, showed me solutions are possible" "Didn't think sandscaping would work at all, but works in the short term as wave break further out to sea" "I was sceptical about scheme because sand shifts all the time but it seems to be working".

**Table 5.2** Thematic analysis of interview data on perceived effectiveness. The table lists parent codes (column 1) and child codes (column 2) relating to the theme of perceived effectiveness expressed by interviewees. Column 4 provides a brief description of each child code, and column 3 details the number of interviewees which expressed that theme (Table: I Cotton). As outlined in section 4.3.3, it was not possible to record (and therefore transcribe) one Happisburgh resident and one Bacton resident walking interview due to weather conditions, and so coding totals reported in this chapter do not include the walking interview from Bacton (i.e. coding percentages in Table 5.2 are out of 21 rather than 22) (Happisburgh interviews are not reported on in this chapter).

Perceived effectiveness – interview findings			
Parent code	Child codes	<u>%</u>	Code description
Perceived effectiveness	Present day	95%	<b>Positive</b> : Working well / created sand bar / breaking waves / sand will come back <b>Negative</b> : Loss of sand / groyne's resurfacing / needs maintaining/topping up to be effective
	Future effectiveness	57%	Not had big storms yet / untested / unsure it will last 20 years/ sand loss rate too fast to last 20 years
	Hard defences	29%	Perceived effectiveness of hard defences / need hard defences to support sandscaping
Communication	Communication mis- understandings	29%	3.1.1 General sandscaping misunderstandings (Beach access and safety, sand martins, wind-blown sand) weren't

	told wind-blown sand would happen / not told when beach access closed / affects tourism/trade
43%	3.1.2 About beach profile Confusion/ surprise about how beach profile changed post- implementation in 2019, and what was expected. Lack of information on this.

The most frequently mentioned view by survey respondents (31%, Table 5.1) is that sandscaping is working effectively, which predominantly relates to the fact that there has been no observed flooding or erosion in the two years since the scheme was implemented, despite storms occurring during this period. In interviews, the fact that waves are breaking further out to sea, a sand bar has formed, and water is subsequently shallower, were all commonly discussed amongst those with a positive perception on the effectiveness of sandscaping:

"When there's a northerly gale, a lot of the energy is taken out of the power of the sea before it gets to the seawall. And so it's quite gentle by, in respect to how it used to be, you know, it would come up against the wall and slam against the wall all over the top, whereas now it's breaking on that sandbar that's out there. And I think that's what it's supposed to do" ~ Bacton resident

On the other hand, there are some residents that are sceptical about the effectiveness of sandscaping, which principally relates to the observed decline in the amount of sand on the (dry) beach at Bacton and Walcott. For example, 16% of survey respondents explicitly mentioned seeing a reduction in sand, some commenting that the amount is "*most*", "*so much*" or "*all gone*". Meanwhile 14% of survey respondents called for the amount of "*lost*" sand to be "*topped up*" on the beach, in order to last its expected lifetime and to restore the beach profile level to the volume of its initial placement in 2019. Interview data confirms that the reduction in sand on the beach – observable through objects such as groynes

resurfacing over time – is the primary reason for doubting sandscaping's effectiveness, with some expressing surprise at the loss of sand. This suggests that some residents expected the placed sand to remain on the beach and perceive a reduction in sand as evidence sandscaping is not working to prevent coastal flooding or erosion.

"So you're back down to the walkway. That was when it was originally, that was 3-4 foot under sand, the groynes are all showing again now. They weren't supposed, you were never supposed to see them again" ~ Walcott interviewee

Interestingly, while some residents are reassured when witnessing sand 'reappearing' after a drop in beach sediment volume, others remain sceptical about the scheme. This highlights the same observation is being used by different residents to reaffirm people's understandings of the scheme:

"So in the meantime, obviously I took a lot of interest in what was going on in Walcott, and of course there as well, the water took the sand away and it brought it back, bought away. I thought it's working, brilliant" ~ Walcott interviewee

"You know local people say 'oh, but it'll all come back again', you know, 'that's what the tides do'. And it, you know, but it always did do that. And so when they put so much there, I mean before when the tides were out, you could see all the groynes and everything. So for it to have covered the groynes up a huge amount, where has it all gone again? (laughs) I mean has it all gone back down to Yarmouth where they brought it from?" ~ Bacton interviewee

The observed drop in sand on Bacton and Walcott beaches also appears linked to the opinion held by 26% of survey respondents that hard engineering defences would have been more effective. One respondent conveyed the drop in placed sediment as evidence that sandscaping is a "*pointless task*", meanwhile another stated that sand would always get "*washed away*". There is a perception by some residents that on its own, sandscaping is not a fully effective coastal management

strategy, and that "*further measures*" such as hard defences are needed, together with sandscaping. Interview data additionally suggests that hard defences are trusted partly given they have been used on the coastline for a long time. This indicates that the novelty of sandscaping could contribute to doubt over its effectiveness:

"But obviously seawalls are clearly very effective because they've been there for many, many decades and, so although they are the most effective, yeah, they aren't, they are a bit unsightly (laughs)" ~ Bacton resident

"I remember it used to knock blocks out the seawall at the top, but now it doesn't, because they double skinned it. But whether or not, that's effective to a point isn't it? And sandscaping is not the answer, it's like to build sandcastles on the beach (laughs) and sit and watch it fall away" ~ Bacton resident

Nearly a quarter of survey respondents (23%) are sceptical that sandscaping will continue to work for 15-20 years (which is the official expected lifetime of the scheme as communicated by the local council (NNDC) and RH (2018a), who designed the scheme). Table 5.1 reveals the varying timescales of effectiveness perceived by survey respondents (e.g., anywhere between 10 and 25 years). Interestingly, this perception is not limited to survey responses with an overall negative perception of sandscaping. For example, some residents who were highly positive about the scheme and convinced it is working in the present day, remain doubtful that sandscaping will continue to work into the future, because of the reduction in volume of sand. Interview data indicates more widespread doubt (57% of interviewees) about whether sandscaping will continue to be as effective in 20 years, than was apparent in the survey data (23% survey responses). For both survey and interview data, this view is expressed both by those positive and negative about the scheme. There is a perception that the sandscaping scheme, 2-3 years in at the time of the survey and interviews, is as of yet untested by extreme weather:

154

"I don't think it's really been. I mean, I think as a protective, as extending the life of the seawall, it's absolutely effective. And as improving the amenities of the beach, it's been effective. In terms of flood risk, I don't know it's been properly tested yet because we haven't had a big surge (laughs)" ~ Walcott resident

"But these guys who were seeing this sand as it is now thinking 'That's superb, brilliant, job done', little do they know that if the wind and tide clicks, that's gone. And if you look now to when they first put it in, its disappearing and there's only one way It can go pretty much, and it goes back to where they got it from, it just goes up the coast" ~ Bacton resident

Much of this concern about the effectiveness of sandscaping in the future relates to extrapolating the rate of sand loss seen in first 3 years over to the next 15-20 years.

"No I don't, really, because how, it's supposed to last for 20 years, and if you look at how much that has already been washed away and we haven't really had spectacularly high tides or any storms to speak of" ~ Bacton resident

"And so the way it's going at the moment, when it's all gone, it's hard to believe that it's still going to all be there in 15 years' time". ~ Bacton resident

Overall, as shown in Table 5.1 and 5.2, the data reveal a divergence between residents that perceive sandscaping as effective, and those that do not, alongside a range of observations that residents draw upon to justify their opinions, and perspectives of sandscaping into the future.

## 5.4.2 Geomorphological changes

This section explores the geomorphological changes post-sandscaping that residents report in the survey and interview data. Analysis of coastal LiDAR data at Bacton and Walcott, in terms of the movement of sediment between 2018-2022, reveals a similar trend as observed by local residents. Panel (A) (Bacton) and

panel (B) (Walcott) in Figure 5.1 shows the cross-shore winter beach profiles in the year before sandscaping (2018, dotted black line) compared to the first (2019, orange line), second (2020, blue line), third (2021, purple line) and fourth (2022, green line) year since implementation.



**Figure 5.1** Beach profile elevation changes at Bacton (panel (A)) and Walcott (panel (B)) from 2018-2022, comparing changes on the upper shore and foreshore (to a depth of approximately -2m) before (dotted black line) and after sandscaping (year 1 in orange, year 2 in blue, year 3 in purple and year 4 in green). Transects begin at the sea wall at Bacton and Walcott, and transect length increases seaward, as plotted on the x-axis. Panel (A) and panel (B) show secondary data from the ACM Programme (2022) (Figure: I Cotton).

The foreshore is the area between the average high water mark at neap tide (MHWN) and at low tide (MLWN) i.e. what is often considered to be the beach – the EA (2011) calculated this to be +1.05m – -0.75m for Bacton, in their 2011 coastal flood boundary conditions dataset.

Panel A and B in Figure 5.1 illustrate the initial, dramatic effect of sandscaping increasing beach elevation and width. Where in 2018 pre-sandscaping, there was very little upper 'dry' beach (the area above average high water mark at neap tide (MHWN)) at Bacton and Walcott, after sandscaping was completed in 2019 the beach was several meters higher and much wider. Elevation dropped slightly in 2020, compared to 2019. From 2020 to 2022 the elevation in the upper 'dry' beach (first ~50m of transect length) continued to decrease i.e. the volume of sediment on the upper 'dry' beach continued to decline. Therefore, as similarly observed by residents, there has been a decrease in placed sediment on the upper 'dry' beach, in the first four years since sandscaping was implemented. However, in 2022 a slight increase in elevation can be observed, largely at Bacton but also at Walcott, revealing that sediment levels are fluctuating year to year, and not consistently declining post-sandscaping. In 2019, sandscaping moved the location of the foreshore approximately 50m offshore (panel A Bacton and panel B Walcott) and there has been little change in elevation at this point post sandscaping (approximately 75m along the transect).

Meanwhile, Figure 5.2 uses secondary data from RH and reveals sub-tidal changes in beach profile after sand placement (to a depth of approximately -9m) in the first (2020, blue line) and second (2021, yellow line) year of the sandscaping scheme. Overall, changes between 2020 and 2021 reveal a decrease in elevation/sediment in the upper subtidal zone just below the foreshore (of approximately -1m to -4m elevation) occurring approximately 100m to 200m along

157

the transect, and an increase in elevation/sediment between approximately 200m to 350m along the transect (Figure 5.2, panel A Bacton and panel B Walcott). Therefore, while Bacton and Walcott have overall seen a decrease in sediment on the upper shore and subtidally just below the foreshore from 2019-2022, in the first two years since sandscaping was implemented there was little change in the new location of the foreshore (i.e. where the 0m sea level is located), and an increase in elevation/sediment in the lower subtidal zone. Whilst the source of the accreting sediment in the lower subtidal zone is unknown, it may in part be migrated sediment from the upper 'dry' shore and/or upper subtidal. Both villages show a similar trend, although changes in sediment for the area sampled at Walcott (panel B) are larger than those at Bacton (panel A).



**Figure 5.2** Panel (A) and panel (B) reveal beach profile elevation changes at Bacton (panel (A)) and Walcott (panel (B)) for the timeseries 2020-2021, which corresponds to the two years after implementation of the scheme. Panel (A) and panel (B) sample to a depth of approximately -9m, revealing changes in the subtidal zone. Transects begin at the sea wall at Bacton and Walcott, and transect length increases seaward, as plotted on the x-axis.; panel (A) and panel (B) are secondary data from RH (2022c) (Figure: I Cotton).

## 5.4.3 Impacts of the sandscaping scheme

Analysis of survey questions on the impacts of sandscaping reveal that for many

residents, sandscaping provides several positive impacts, beyond the principal

positive impact of (thus far) no flooding or coastal erosion. Table 5.3 lists the different impacts reported by respondents, grouped into main themes. Impacts were not mutually exclusive; that is, respondents reported positive and/or negative impacts, or combinations of these; and some respondents reported no impacts.

**Table 5.3** Reported positive and negative impacts of the sandscaping scheme from residents in the survey. Impacts are grouped into codes of similar themes. Impacts reported as positive are presented first (in blue shading); negative impacts are shown in orange shading. Impacts reported both positively and negatively (for example increased visitors to the village) are listed twice. Questions asked include; 'Have there been any positive impacts of the sandscaping scheme to you and/or the village you live in? If yes, please specify what these positive impacts are' and 'Have there been any negative impacts of the sandscaping scheme to you and/or the village you live in? If yes, please specify what these negative impacts are' (Table: I Cotton).

Impacts of the sandscaping scheme		
	Main theme (impact)	Codes
	No flooding/ erosion	-No flooding or erosion events and associated physical impacts (property damage, inundation, house shaking, sea spray, overtopping of sea wall, cliff collapses).
- <u>\-</u>	Bigger/ sandier beach	<ul> <li>-Restorative benefits of having a wider, bigger, sandier beach.</li> <li>-More attractive beach and coastal scenery.</li> <li>-Change (reverting back) to how beach used to look in the past.</li> </ul>
<u>.</u>	Recreational opportunities	-Recreational benefits, with calmer sea for swimming, kayaking, sunbathing, bird watching, and new shallow areas in the sea. -Cleaner beach /less rubbish washed up.
Ĩij	Physical access/ safety getting on/off beach	<ul> <li>Permanence of access for different parts of beach at all times of day (e.g. including during high tide).</li> <li>Now possible to walk between villages along the coast.</li> <li>Improved physical access and safety in getting on/off beach for wheelchair users/ users with reduced mobility.</li> </ul>
**	Reassurance/ peace of mind	-Mental health benefits of greater reassurance, peace of mind, and reduced anxiety about flood or erosion risk and impacts.
£	Increased property value	<ul> <li>Perceived financial benefits from increased property value and the village being a more desirable place to live.</li> <li>Not incurring financial expense from flood or erosion property damage.</li> </ul>
	Coast road stays open	-The main road connecting the villages of Bacton and Walcott to other parts of the coast does not flood, providing reliability for transport and access.

<b>Å</b> h	More visitors and trade	-More visitors to beach, more trade to shops, cafes, pubs. Village thrives and has financial viability.
<b>Å</b> ħ	More people using beach	Greater numbers of tourists and visitors to the beach.
Ŭ	Impact of more visitors	<ul> <li>Impact of more visitors:</li> <li>Cars (traffic, inconsiderate parking blocking houses and roads, visitors not using car parking provided, nowhere to park in village).</li> <li>Litter (more dog waste and other litter).</li> <li>Antisocial behaviour, petty crime.</li> </ul>
ချီ	Wind-blown sand	<ul> <li>-Wind-blown sand into gardens, open windows, car screens, blocking gutters/ drains, damage to outdoor equipment, and depositing around the village.</li> <li>-Required locks to be taped over and financial cost and stress to clean-up houses and gardens.</li> </ul>
Ϊÿ	Change in physical access to beach	<ul> <li>-Access to beach is harder due to slopes, the loss of concrete path along the beach by sea wall makes it harder to walk for some (prams, wheelchairs).</li> <li>Some areas now closed.</li> <li>-Change in aesthetic to beach (appears scruffier).</li> <li>-Safety issues of groynes partially or fully submerged by sandscaping.</li> </ul>

Survey respondents spoke about the restorative benefits of having a superior sandy beach and 'coastscape', as well as better beach access for recreational activities such as walking and swimming at all times of the day, rather than just at low tide. A few survey respondents conveyed enthusiasm that the coast is "*back to its 'old' sandscape beach*", the village is "*surviving*" again, and the coast road remains open, strengthening local residents' sense of place of their locality. Respondents also reported psychological benefits (such as reduced anxiety) from the reduction in coastal flood or erosion risk – the significance of this is explored further in Chapter 7. Lastly, residents reported financial benefits of no property damage from storms, higher house prices, and more visitors (and thus trade) to the village.

Although reported impacts were largely positive, and 32% of survey respondents gave no negative impacts, the remaining respondents did report negative impacts. These included wind-blown sand, the practical impact of more visitors, and beach safety/access issues. For some wind-blown sand was an inconvenience (for

example depositing around village, blowing through windows, requiring locks to be taped over), but for others it has led to damage and financial expense to clean up properties and gardens. This demonstrates how impacts from the sandscaping scheme have been experienced to differing extents amongst local residents. Some impacts, for example the impact of more visitors to an area, was reported positively by some (e.g., more trade, more popular village) but negatively by others (e.g., traffic, accessing beach). This indicates that the experience of impacts, and whether they are perceived as positive or negative, vary at an individual level.

### 5.4.4 Trust in, and communication from, policymakers

Survey respondents were asked about trust in local coastal management decisionmaking (n=73). Figure 5.3 shows a clear range of opinion amongst respondents on whether coastal change is managed appropriately for their village: 25% respondents either strongly or partially disagree with this statement (and of this, 14% strongly disagree with this statement), and 29% neither agree or disagree. Meanwhile, 47% of residents strongly or partially agree that they 'trust that coastal change is managed appropriately in my village'. This indicates that while views are mainly neutral or positive, there are a sizeable number of negative views on local coastal management decision-making.



**Figure 5.3** Responses to survey question on trust in coastal management; 'I trust that coastal change is managed appropriately in my village' (n=73) (Figure: I Cotton).

The interview data with local residents provides further insight into perceived trust with different types of policymakers (Table 5.4). For some Bacton and Walcott residents, there is a degree of scepticism regarding what policymakers say in terms of motive. Residents spoke in interviewees that local and national policymakers are driven by funding and finance considerations, rather than what coastal management is effective or needs maintaining. Residents also raised doubt on how truthful, genuine, and accountable policymakers are – this was in particular about policymakers at a national level. The theme of trust is further explored in the context of Happisburgh in Chapter 6, bringing in the perspectives of Happisburgh residents and coastal policymakers on managing coastal change, and the relationship between local residents and coastal policymakers.

**Table 5.4** Child codes (column 1) relating to the overall theme of trust in policymakers expressed by resident interviewees. Column 3 provides a brief description of each code, and column 2 details the number of interviewees which mentioned that theme (Table: I Cotton). As outlined for Table 5.2, coding percentages are out of 21 rather than 22, due to one walking interview where it was not possible to record or transcribe.

Trust in policymakers – interview findings		
Child code	% interviewees	Code description
4.1 Motive of policymakers	52%	Scepticism of motive/ Vested interests/ official timeframe / actions / genuine words
4.2 Strained relations with policymakers	67%	Scepticism of sandscaping due to mistakes of past coastal interventions / strained relations / engagement going nowhere
4.3 Good relationships with policymakers	24%	Anecdotes of good relations/ experiences between residents and policymakers, good levels of trust

Only a minority of interviewees (24%) expressed positive perceptions of relationships with policymakers on coastal management (such as trusting their judgement and the information provided). At a local level, trust in policymakers was often described by resident interviewees with regards to strained relations in the past (for example the extent to which coastal defences are maintained), a lack of meaningful consultation with local residents on the choice of sandscaping as a coastal management strategy (public participation in coastal decision-making is explored in greater detail in Chapter 6), and a lack of maintenance on safety or access issues to the beach after sandscaping:

"Sometimes it's like going round in circles because nobody will take responsibility. And they were saying it was the sandscaping scheme, which was obviously provided by the gas works and stuff, but it's trying to find someone actually say 'yeah, OK we'll be responsible for aftercare'. Just doesn't happen" ~ Bacton interviewee

"One of the things that I thought they were very disingenuous about was, erm, some people like the RSPB, giving them guidance on the netting they should put up. And they, either they did it deliberately or they didn't supervise it properly, but the whole of the Cliff got netted. People were kind of ripping it down because they were so upset about it, and when you spoke to them and they said, 'well, we were never given that advice' and I said 'well, you were because I read the report!'" ~ Bacton interviewee

The interview data also revealed there are ranging views on the overall quality (e.g. availability) of information from local policymakers whilst sandscaping was being implemented. This suggests that residents' access to the public communication outlined in section 5.2.2 varied, in the months and years prior to the implementation of sandscaping in 2019. However, where there was strong agreement amongst interviewees is an interest in receiving more information about sandscaping and how its evolving, as it has implications for what coastal management follows sandscaping in 20 years' time. Perceptions of coastal management post-sandscaping are explored in greater detail in Chapter 7.

"I think generally local people need reassurances because they can see what appears to be a massive amount of money being spent, and whilst that's not being taken from them, you still want to believe that it was a valid investment. I think it kind of needs explaining as a process, like this is what we expect to happen, This is as expected, do not be alarmed. it seems like quite a long term plan, but that will go in a flash, and we need to know what happens next!" ~ Bacton interviewee

"It would be nice to be getting some kind of feedback about you know, how successful, those that had planned the scheme as it were, feel that it's been, is it doing what they expected, is the engineering of it kind of worked how they expected and the like. Because I think just recently there's been a little bit of doubt, you know?...there's quite a bit of sand that suddenly gone up at Ostend, and people were a little bit worried about, is the lifespan of the scheme going to be what it originally was so, so I think there's still. There's still room for a bit more communication there" ~ Walcott interviewee

Local residents conveyed being very open to what form information about

sandscaping and coastal management takes, but are more concerned that the source of information is independent and evidence-based. There are some differences of opinion on what types of institutions are perceived as credible:

"But again, yeah you want something that's independent, and not, that's the key thing these days is that, I think people want to trust a source of information". ~ Bacton resident

"I think people trust it more from the Council. And because obviously we receive communications from them about other things and. As much as we can trust anybody in authority at present! Then we have to, we have to work In partnership with council, don't we, because there's certain things we need from them, so I think If it's got this sort of NNDC stamp on it then it has a level of authenticity that most people would would buy into". ~ Bacton interviewee

*"I would like that to come from some group that doesn't have an affiliation to whoever decided on the sandscaping".* ~ Walcott interviewee

### 5.5 Discussion

#### 5.5.1 Perceived effectiveness of the sandscaping scheme

Research question 2a seeks to explore resident perceptions of sandscaping, and this chapter finds that different views of effectiveness are apparent amongst Bacton and Walcott residents. Reflecting on the first few years of the scheme (two years at the time of the survey and three years for the interview data), some perceive it to be working (largely due to no flooding or erosion and a longer beach/ presence of a sand bar) and some do not (due to a decrease in placed sediment on the beaches and objects such a groynes now resurfacing). Diverging views amongst local stakeholders and concerns over effectiveness have been found elsewhere in other coastal contexts, e.g., of smaller-scale beach nourishment (Prati et al., 2016) and of managed realignment (Roca and Villares, 2012). The contrasting perceptions at Bacton and Walcott indicate different understandings

amongst residents on how sandscaping is expected to evolve and change the local coastline over time. In answer to research question 1a (investigating beach profile changes), the LiDAR (elevation) data of the coast shows that although there has been a decrease of sediment on the upper (dry) beach and the first ~100m of the subtidal in the first two years of the scheme, sediment stayed at similar levels on the foreshore (between high and low tide) and increased in the lower subtidal, where it will still contribute to reducing erosion risk. That local residents observed the drop in beach volume on the upper beach, and used this observation to inform perceived effectiveness of sandscaping, highlights the link between geomorphological change and social perceptions in this empirical context, which is illustrated in the conceptual framework diagram (section 3.5) by the connected 'social' and 'environmental' boxes.

The geomorphological observations of beach profile explored in this chapter match the modelling of the scheme (NNDC, 2022b) indicating that sand would decrease on the beaches over time, and that sand would migrate seawards after storm events. Modelled expectations of cross-shore sediment movement were for placed sediment to migrate from the upper 'dry' shore to the foreshore (RH, 2022a), which is observable in Figure 5.1 from 2019-2022. Panels A and B of Figure 5.2, and residents observations in section 5.4.1, confirm the reappearance of subtidal sand bars on the frontage, similarly found by RH in their geomorphological analysis, who argue that sediment which has migrated to form subtidal sand bars is likely to remain rather than be lost offshore (RH, 2022a). This suggests that sandscaping is working as expected in the first few years of its lifetime, as cross-shore sediment migration matches the modelled expectations of waves further out to sea, rather than at the sea wall, offering greater protection during storms. Identifying the source and direction of sediment migration cross-

shore and alongshore (in other words, attributing beach profile changes to sandscaping), is very difficult given the limited LiDAR data, and is not possible without analysis of the sediment, which is beyond the scope of this study. For local residents, some of the profile changes reported here would be difficult or even impossible to observe from land, given the foreshore zone is under the mean high-water mark (i.e. below high tide and only periodically observable) and the subtidal is covered by water and largely unobservable. This highlights how incomplete knowledge and understanding of how sandscaping works and the changes that have occurred post sandscaping can lead to an opinion that the sandscaping scheme is not working.

The survey and interview data revealed doubt among some residents on the longterm effectiveness of sandscaping (i.e. for a full 15-20 year lifespan). Even those residents who feel the scheme has been effective to protect from erosion and flooding in the first 2-3 years expressed some scepticism that it will continue to be effective in 20 years. Observations of a drop in sand volume on the upper (dry) beach appear to contribute to this perception. This raises a question on local residents' understanding about how sandscaping would change the local beach at Bacton and Walcott over time. As mentioned in section 5.2.2, the local council (NNDC) organised several community engagement events before and during the implementation of the scheme, which included a Local Liaison Group, village dropin events, and public information boards at Bacton and Walcott beaches (NNDC, 2017), alongside reporting in the media (e.g. BBC, 2017). Despite this, the scale of doubt (almost 1 in 4 survey respondents and over half of interviewees) about the long-term effectiveness of sandscaping over the next 20 years suggests that diversity of opinions may be influenced by several factors, as highlighted in literature. This includes different perceptions of nature-based solutions (Anderson and Renaud, 2021) and on the goal of adaptation (Dilling et al., 2019). Further,

studies specifically in the context of coastal management have shown different risk perceptions (Prati et al., 2016; O'Donnell, 2019) and perceived responsibility (Clément et al., 2015) of coastal change amongst coastal populations. More broadly, there is also long-standing research on public scepticism of expert knowledge, both within and outside of environmental policy and issues (Sjöberg, 1999; Fairbrother, 2017). This illustrates a challenge for policy-makers in local stakeholder engagement where there are many diverse opinions relating to coastal management, and numerous potential factors contributing to held opinions.

Presenting data and keeping local residents regularly updated with how beaches will change, could be one way for coastal managers to answer questions and engage residents on the changing profile of sandscaping, which will evolve over time. This could be further explored in future research, to consider how and why different aspects of information and forms of communication are trusted or not, are how they are interpreted. In particular, findings in section 5.4 suggest that policymakers could be more explicit in public communication about how beaches will change visually, and that changes in beach profile while sandscaping is in place, such as fluctuating sand levels and groynes resurfacing, is to be expected and will not reduce the overall effectiveness of the scheme. Figures 5.1 and 5.2 reveal that the greatest cross-shore sediment changes have occurred around mean sea level, as similarly found by RH (2022a), and the Zandmotor scheme, where analysis of the first five years (Roest et al., 2021) saw minimum movement beyond -8m below mean sea level, and greatest activity in the intertidal zone.

Interview data reveals that nearly all residents would like to be kept updated about how sandscaping is performing and evolving. Figure 5.1 reveals that the greatest change in the volume of sediment occurred in the first year (Winter 2019), which concurs with analysis by RH (2022a) that cross-shore sediment movement, particularly at Bacton Gas Terminal, is slowing (in 2022), and similarly in the Netherlands, that the most active phase of sediment movement for the Zandmotor scheme was the first six months after implementation (De Schipper et al., 2016; 2021). This dynamic changing profile of the coast highlights that community engagement on sandscaping could continue after implementation, and also presents an opportunity to reinforce communication channels and improve trust (explored in next section, 5.5.2) between residents and local policymakers on coastal management. This appears particularly relevant given the current policy context in the UK, where at-risk coastal villages, such as Bacton and Walcott, have a change in SMP policy to managed realignment in the medium term (from 2025), which will require time and dialogue between policymakers and local residents (themes discussed further in Chapters 6 and 7).

#### 5.5.2 Trust in coastal management at Bacton and Walcott

This chapter's findings also show a divergence in opinion amongst residents on the extent to which they trust how coastal change is managed in their village. A quarter of survey respondents stated that they felt some or strong distrust (with 29% stating neutral responses to the question of trust). Whilst survey data highlighted almost half of respondents partly or strongly trusted coastal management for their village, a notable minority of residents have low levels of trust. The interview data highlighted there is greater distrust for national rather than local coastal policymakers, but also revealed negative perceptions at a local level on the financing of sandscaping, ongoing maintenance of the scheme, and prior consultation with local residents (views on consultation and coastal management are discussed in greater detail in Chapter 6). Community distrust of coastal management has long been highlighted, such as Myatt et al., 2003's study on local community perceptions of proposed managed realignment in Freiston shore, Lincolnshire. Meanwhile, recent research on other coastal communities that face high erosion risk, such as Hemsby (also on the Norfolk coast), similarly found that issues of trust between residents and policymakers were a challenge for, and high trust an important precursor to, community engagement on coastal adaptation (EA, 2021a). Low trust in how coastal change is managed can have implications for future coastal decisions, potentially acting as a barrier to public engagement and to the participation in, and acceptance of, longer-term adaptation strategies (Milligan et al., 2009; O'Riordan et al., 2014; Anderson and Renaud, 2021). Furthermore, trust between communities and policymakers can be damaged where there is high staff turnover, indicating that trust is built at an individual level but can be difficult to maintain (EA, 2021a). Taken together, all of the above highlights the significance of building trust and relationships between policymakers and local communities, and that it is still a relevant objective in managing coastal change (EA, 2021a). Trust in institutions is included in Figure 3.3 as a factor influencing 'capacity to adapt', which is one of three adaptive capacities present in the conceptual framework. In this empirical context, trust appears to be a significant factor shaping residents' perceptions of coastal change.

For Bacton and Walcott residents, it is unknown what coastal management will happen after the lifetime of the sandscaping scheme. As a novel, untested, naturebased solution, the actual lifetime of the sandscaping scheme is uncertain, with its 15-20 year estimate only serving as a guide. Survey responses showed a range of resident perceptions on how long the scheme could last (from anywhere between a few years to 25 years), with no geomorphological evidence available yet to confirm the lifetime of the scheme. Meanwhile, a 10-year evaluation of the first mega-nourishment scheme (the Zandmotor scheme in Holland), implemented in 2011, suggests it is likely to last longer than its official 20 year estimate, because the loss of sediment observed in the first decade of the scheme is lower than

expected (Huisman et al., 2021). This further highlights the uncertainty of the longevity of sandscaping as a coastal management strategy; it is likely to be unique to each coastal context, which raises an important question for coastal managers on when and how to prepare for future coastal risk. The perspectives of Bacton and Walcott residents and policymakers on coastal management postsandscaping is returned to and explored in detail in Chapter 7. This chapter has revealed varying expectations amongst residents across Bacton and Walcott of when sandscaping will end, and differing levels of trust in the coastal management process for their villages. The current flood and coastal erosion strategy for England (EA, 2020, p.49) calls for "local leadership and support from the local community" in building resilience to future coastal change, but an agreement on what this looks like in practice may be extremely difficult. The launch of the £36 million CTAP project by Defra (2022) may be an opportunity to work through multiple perspectives and rethink how future options of coastal change can be considered and planned for by at-risk coastal areas. Perspectives of, and implications for, the CTAP project, are explored next in Chapter 6.

### 5.5.3 Social impacts of sandscaping

Research question 2a also seeks to understand the social impacts of sandscaping, and survey responses from residents highlight a wide range of both positive and negative impacts, in the first 2-3 years of the scheme. Sandscaping has provided multiple positive impacts to the local villages, but a uniquely positive experience has not been felt or reported by all. Across the social data presented in this chapter, residents think differently about the scheme despite drawing upon the same observation or impact. For example, the contrast between residents who felt sandscaping is not working because of the reduction in sand and those who felt it is working despite it, or residents who felt sandscaping was positively impacting their village because of more tourists and those who perceived this as a negative impact. Longitudinal work, or more in-depth qualitative work, such as interviews or focus groups, could be conducted later in the lifetime of the sandscaping scheme, to compare how and why residents' perceptions and experiences change over time. This could be analysed in conjunction with a study on mineralogy, to explore where placed sediment from the scheme has migrated along the coastline.

The role of an individual's values in influencing different perceptions of beach nourishment has been found elsewhere (Prati et al., 2016), and the findings presented in this chapter suggest there may be deeply engrained beliefs, which alongside personal experience, shape an individual's perception of a coastal management strategy and its effectiveness. This concurs with findings from this thesis literature review (Figure 3.5), that there are a range of factors shaping public perceptions of adaptation and coastal management strategies. For example, D'Souza et al., (2021) found a link between support for hard defences and more conservative political values, while Jones and Clark (2014) highlight the level of trust in institutions, and perceived acceptability of coastal management strategies, correlates with social trust and social capital (further underlying the role of trust already discussed above in section 5.5.2).

The range of social impacts and experiences of sandscaping also further underline the argument by de Schipper et al. (2021) that evaluations of beach nourishment's overall 'effectiveness' needs to consider both social and physical impacts. This appears a relevant learning not just for the implementation of sandscaping, but evaluations of other nature-based solutions, to draw upon social perceptions and experiences, and therefore broaden evaluations beyond effectiveness that also incorporates principles of environmental justice. This is a relevant finding for research question 3, on the relevance of incorporating environmental justice principles into coastal management policy that focuses on resilience. In the context of mega-nourishment at Bacton and Walcott, the survey and interview data suggest that the range of social impacts and perceptions on sandscaping have implications for building the social resilience of communities to coastal change after the lifetime of the scheme (research question 2c, explored in Chapter 7). This is in terms of potentially conflicting preferences for future coastal management, and which is an overlooked component of system resilience in the beach nourishment literature (section 2.3.2).

#### 5.5.4 Perceptions/preferences of 'soft' versus 'hard' engineering

Over a quarter of surveyed Bacton and Walcott residents perceive hard defences to be a more effective coastal management strategy to protect from flooding and erosion compared to sandscaping. This correlates with findings from existing literature (section 5.2.1) on scepticism of soft engineering over traditional hard defences (Roca and Vilares, 2012; Anderson and Renaud, 2021). Examples of hard defences mentioned in the survey are those historically used at this area of the coastline, or those used successfully elsewhere (such as rock barriers, reefs, breakwaters, and a sea wall). Numerous factors could be contributing to this perception, including that sandscaping is relatively novel and therefore seen as uncertain or untested, or, as highlighted in interview data, that hard defences have for decades historically been used at the villages of Bacton and Walcott to manage the risk of coastal change. Therefore in this context, there is a perceived familiarity with hard defences, compared to the novelty of sandscaping. Sandscaping works very differently to hard defences, in that it is constantly changing, partially invisible or submerged by tides, and of a much larger spatial and temporal scale. The issue remains that if sandscaping is being compared like-for-like to how hard defences are engineered (as similarly argued by Anderson and Renaud, 2021), indeed this soft engineering approach could be perceived as less effective. Sand will migrate,

as noted by residents in the survey, while hard defences stay put, and thus may be perceived to better protect from coastal risk. This is despite evidence that hard defences can themselves exacerbate erosion, and by suppressing the movement of sediment, can increase the risk of erosion elsewhere along the coast (French, 2004; AECOM, 2012b: Nicholls et al., 2013).

Even residents who perceive sandscaping to be effective also suggest hard defences should be further installed on the Bacton and Walcott frontage, to further protect from coastal flood and erosion risk. Amidst the uncertainty on the perceived lifetime of sandscaping, this indicates that hard defences (either in combination or replacing sandscaping) are seen by some respondents as a more reliable coastal management strategy. It is therefore important that the advantages and disadvantages of nature-based solutions, or any novel strategy, particularly where implemented in places with historically very different management strategies, are included in communication to and discussions with key stakeholders, and as already highlighted, for this engagement to be ongoing (Anderson and Renaud, 2021). For the case of the Bacton-Walcott sandscaping scheme, further exploration of the merit of different approaches to communicating nature-based solutions, from the perspective of key stakeholders, would be an important next step.

## 5.6 Conclusions

This chapter has found wide differences in perceptions, and notable levels of doubt, on the 'effectiveness' of sandscaping at present and in the future, alongside different lived experiences of the scheme and prevailing distrust by some residents about coastal management. This highlights the need for further engagement with Bacton and Walcott residents on how sandscaping is evolving on the coast. This could potentially be through citizen science, that includes residents in ongoing monitoring and reporting change. Keeping residents updated on changes to sandscaping with environmental data and communicating the advantages of nature-based solutions appear relevant in this context, but the diversity and contrast of resident perceptions illustrates deeper challenges for future coastal management planning. There is a need to think through how future coastal change can be planned for, drawing upon multiple social perspectives. In particular, there is a need to consider how residents may engage in varying ways, and how residents' differing views may inform future coastal management. Facilitating this requires further attention, given the greater role envisaged by policymakers (for example the EA, 2020) on UK coastal communities contributing to coastal decisions made in their area.

From a geomorphological point of view, this chapter reveals the coastal system is functioning differently in response to increased sediment from the sandscaping scheme, with the formation of subtidal sand bars and the beach profile taking a more natural shape. Geomorphologically, sandscaping appears to match Kates et al's (2012) description of transformation (section 3.2.3), in that it is innovative, large-scale, and has significantly altered the coastal system. However, in a coastal management policy context where strategies are becoming increasingly innovative or with multiple win-win objectives (such as the use of sandscaping), there is a need to expand how schemes are monitored and evaluated to encompass the broad range of physical, social, and other relevant objectives, in its unique context. Findings here of residents' perceptions and geomorphological observations of the Bacton-Walcott sandscaping scheme indicate that considering nourishment strategies in terms of physical risk alone is not sufficient in measuring overall 'effectiveness'. There are some differences between residents' perceptions of the local

176

context of Bacton and Walcott, further research could be conducted elsewhere to understand how perceived effectiveness of adaptation compares in different contexts, either where mega-nourishment is implemented or otherwise.

Therefore, whilst this chapter has focused on perceived effectiveness, it has several findings on trust and engagement between residents and local authorities that have implications for coastal management in the future. This is explored further in the following chapter, in terms of residents perceptions of coastal adaptation, and lived experience of coastal change. Chapter 6 explores this using an environment justice lens, prominent in this thesis' conceptual framework, and does so by expanding scope to include the perspective of residents from Happisburgh, which are considered in relation to the perspectives of Bacton and Walcott residents, and policymakers. This is explored in the context of CTAP, the current policy programme on coastal adaptation on the North Norfolk coast (2023-2027), and which will involve discussions between local residents and policymakers on coastal adaptation.

# Chapter 6. Environmental justice in coastal adaptation: insights from Happisburgh

### 6.1 Introduction

As highlighted in section 2.2.2, flood and coastal risk management policy in England increasingly promotes coastal adaptation plans that are co-developed by local communities themselves. Focusing on the village of Happisburgh, this chapter adopts an environmental justice lens to explore local resident and policymaker perspectives of managed realignment, that may have a bearing on the implementation of local coastal adaptation plans. This chapter primarily answers the second part of research question 2 (2b), which explores past, present and future perceptions and lived experiences of coastal change at Happisburgh. Whilst focusing on the views of Happisburgh residents, views from adjacent coastal villages (Bacton and Walcott), alongside coastal policymakers, are also reported on here, to examine the variety of perspectives on managed realignment. These perspectives are presented alongside maps and calculations of recent (2015-2022) cliff retreat at Happisburgh (research question 1b), to illustrate the extent of geomorphological change that local residents are adapting to.

It begins by briefly outlining previous research on managed realignment from an environmental justice perspective (section 6.2) alongside relevant background context on coastal policy (i.e. Coastwise) for this chapter. The materials and methods section (6.3) outlines the subset of thesis data presented in this chapter, and the method adopted for calculating cliff rate of retreat at Happisburgh. Resident and policymaker perspectives and cliff retreat calculations are presented in the results section (6.4), which is structured according to different environmental

justice harms. Section 6.5 discusses the main themes arising across the data, before a conclusion on the chapter's main findings (6.6).

#### 6.2 Background context

## 6.2.1 Managed realignment and coastal adaptation: an environmental justice perspective

As outlined in the introduction to case study (2.4), the coast at Bacton and Walcott is currently defended until approximately 2040 by sandscaping, whereas at neighbouring Happisburgh, the coast is undefended and the SMP policy is currently managed realignment. Managed realignment is a policy option adopted in coastal areas that allows the shoreline to change, rather than defending its current position (AECOM, 2012). This can be facilitated through adaptation measures such as rollback, that moves infrastructure or whole settlements away from the coastline. For this reason it can be highly contentious for local communities and is commonly disputed (Tebboth, 2014; Nordstorm et al., 2015; Famuditi et al., 2018; Buser, 2020) (see section 2.4 for further case study background). Although relocation is presented as a transformative adaptation action to managing coastal risk in section 3.2.4 (O'Brien, 2017; Coastal Partnership East, 2019; Sayers et al., 2022; Huggel et al., 2022), academic research has also highlighted a risk of maladaptation from transformative responses, and a link between reactive adaptation, lower levels of resilience, and greater risk of maladaptation (section 3.2.3). The nature of the relationship between these concepts forms an important part of this thesis' conceptual framework, and are tested empirically with the results presented in this chapter, alongside the other empirical chapters.
As highlighted in the literature review (section 3.3.3), policy decisions to protect coastal areas at risk, or to not defend them, raise environmental justice issues (Cooper and McKenna, 2008). Longer-term impacts of at risk coastal areas which become undefended can include lower house prices, blight, demographic changes (such as rise in second home ownership) and uncertainty for the future (Nicholson-Cole and O'Riordan, 2009; Famuditi et al., 2018; Buser, 2020; Coastal Partnership East, 2019; Day, 2020; Brown et al., 2023). Extant research on the justice dimensions of managed realignment is predominantly in international contexts, focussed on risks to preserving cultural integrity for areas forced to retreat, particularly for vulnerable groups, and exploring how managed realignment can correct historical socio-economic injustices, such as land dispossession and discrimination (Siders and Ajibade, 2021; Maldonado et al., 2021; Perez and Tomaselli, 2021). There is increasing recognition that multiple dimensions of cultural heritage (from built infrastructure to social traditions and values, Vecco, 2010) are at risk through coastal change (Sesana et al., 2021).

As outlined in section 3.3.3, existing research on environmental justice issues in coastal management within a UK context has focussed on the social impacts of the change in coastal policy (SMPs) in 2005 away from indefinite coastal protection. Justice issues include the disproportionate financial and wellbeing impacts of coastal communities without coastal defences (distributional justice), and the extent to which communities were included in the process of the SMP policy change (participatory justice) (Milligan and O'Riordan, 2007; Nicholson-Cole and O'Riordan, 2009; Adger et al., 2011; Day et al., 2015; Famuditi et al., 2018). Research has also highlighted that vulnerability to flood risk in the UK can correlate with higher socio-economic vulnerability or deprivation (Walker and Burningham, 2011; Sayers et al., 2018). There is comparatively less research to

180

date on the implications of environmental justice issues of managed realignment on community willingness to adapt to future coastal change.

In England, there is increasing policy attention on future adaptation plans of coastal settlements to be co-produced by local communities: it is described within the current FCERM strategy (EA, 2020) and is a key objective of CTAP, which provides funding to two local authorities in England (Norfolk and East Riding of Yorkshire) to trial options for coastal adaptation (Defra, 2022). This mirrors increasing attention and adoption of co-production in environmental governance worldwide (Turnhout et al., 2020). From an environmental justice perspective, community participation is argued to lead to more just outcomes, because local perspectives are incorporated into decision-making (Marion Suiseeya, 2020). Although previous research has cautioned the extent to which participatory coastal management in the UK has meaningfully incorporated local views (Few et al., 2007), other authors have highlighted the potential of coproduction within coastal adaptation policy to address the environmental justice issues associated with managed realignment (Tubridy et al., 2022).

Furthermore, although significant questions remain about the lack of funding and clearly defined roles on coastal adaptation (Brown et al., 2023), alongside the viability of reduced state and increased citizen responsibility in flood and coastal erosion risk management (McGinlay et al., 2021), a policy shift towards coproduction nonetheless represents an opportunity for stakeholders to input their views on adaptation to future coastal change. Empirical research that examines in-depth community perspectives of involvement in coastal decision-making, under a managed realignment context, is therefore timely to understand how to maximise community engagement in adaptation. To date, the co-production literature has focussed on issues of knowledge inclusion, particularly regarding

181

indigenous groups or key stakeholders, (Muhl et al., 2023; Wamsler, 2017) or in urban environments (Wamsler, 2017), rather than coastal contexts.

## 6.2.2 Coastwise

This research coincided with the start of CTAP, which was launched in March 2022, after the survey took place (January 2022). This subsection briefly details what was known publicly about the scheme at the time of the research: a general overview of CTAP can be found in section 2.2.1. CTAP was announced by Defra in a press release in March 2022, and Members of Defra and the local council (NNDC) travelled to Happisburgh to formally announce the scheme (Defra, 2022). As detailed by interviewees, this was not a publicised event, however interviewees remarked seeing Rebecca Pow MP (Floods Minister and Parliamentary Under Secretary of State for Defra), alongside other policymakers at Happisburgh. Resident interviews were conducted in Autumn 2022, before the local project 'Coastwise' was formally announced (in summer 2023) (NNDC, 2023d), but at a time where there was local and national media reporting about the programme, alongside a letter to Happisburgh interviewees (residents) from the local MP Duncan Baker that referred to coastal issues and CTAP. Interviews with policymakers were conducted in Spring (2023), at which time a draft business case for CTAP was being prepared to the EA. At the time of writing (Spring 2024), Coastwise has begun stakeholder engagement in Bacton, Walcott and Happisburgh through public drop-in events at village halls ('Coastwise cafes', which ran in February 2024) (NNDC, 2023e).

## 6.3 Materials and Methods

#### 6.3.1 Data collection

This section details the data selection and analysis steps for data presented in this chapter. General information on the sampling strategy for the geomorphological (section 4.3.1) survey (4.3.2) and interview (4.3.3) data collection, and a copy of the survey and interview questions can be found in the appendix (Appendices 1-4).

#### Interview and survey data

A total of 36 semi-structured interviews were completed; 22 of which were Bacton (n=15) and Walcott (7) residents, a further 8 were interviews with Happisburgh residents, and lastly, 6 interviews with local, regional and national policymakers. Due to the particular focus within this chapter on the coastal context in Happisburgh, survey findings from Happisburgh residents (n=23) are also presented here. Given the survey focussed on residents' views of coastal management in their own village, as opposed to neighbouring villages, survey responses from Bacton and Walcott are not included in this chapter. Interviews with local residents across the three villages and policymakers consisted of different questions but on similar themes. Themes discussed and reported on in this chapter include;

- Perceptions of a managed realignment coastal policy at Happisburgh (6.4.1)
- (For residents only) the lived experience of coastal change (6.4.2)
- Experiences of past community consultations on local coastal policy (6.4.3)

#### LiDAR data

To situate local resident and policymaker perspectives and experiences of environmental change at Happisburgh, secondary LiDAR data from the ACM Programme was analysed in ArcGIS Pro to calculate annual cliff retreat at Happisburgh between 2015-2022. Cliff retreat was calculated at Happisburgh only (rather than the entire case study area), given the focus on the coastal context at Happisburgh in this chapter, but also given that the villages of Bacton and Walcott are low lying (fronted by a continuous sea wall, and therefore not cliff frontages). The LiDAR data vary in date each year, but were all collected between October and January, giving a winter beach profile. The data have a minimum accuracy of +/- 0.10m RMSE (root mean square error). Similarly to Chapter 5, data were downloaded from the ACM programme's website (https://coastalmonitoring.org/cco/) as yearly digital elevation models, imported into ArcGIS Pro, and data cells were combined into a continuous raster dataset.

#### 6.3.2 Data analysis

#### Interview and survey data

Section 4.4.3 outlines the approach to analysing qualitative data from the surveys and interviews. Interviews were thematically analysed using both an inductive and deductive coding approach, using the 3-step coding technique (i- initial, ii- focused, and iii- theoretical) advocated by Charmaz (2006), and developed from a grounded theory analysis approach by Glaser & Strauss (1967). Codes analysed in this chapter (Table 6.1) are a subset of the overall coding framework (appendix 6). Codes in this chapter specifically relate to different aspects of environmental justice (distributional justice, participatory justice and recognition, given these categories of harm are commonly used to analyse environmental justice issues

(see Coolsaet, 2020)). As outlined further in section 3.3.1, these relate to the differential experiences societal groups face (distributional justice) (Kaswan, 2020), the ability of individuals to participate in decision-making (participatory justice) (Marion Suiseeya, 2020) and the extent to which individuals are accepted as a legitimate voice for environmental concerns (recognition) (Coolsaet and Néron, 2020). The three categories (hereafter themes) were applied, in the third step of qualitative coding, to organise codes of resident and policymaker perspectives according to environmental justice themes. Consequently, Table 6.1 and the results section is structured to firstly present views on managed realignment and the impacts of coastal change that relate to distributional justice (sections 6.4.1 and 6.4.2) followed by views on public consultations and coastal decision-making that relate to procedural justice and recognition (6.4.3). Whilst attention is given to the most common themes from the interviews and survey (explicitly stated in text), this is not at the exclusion of less common themes that nonetheless provide interesting insights into justice and/or this thesis' research questions.

**Table 6.1.** Themes (column 1) and codes<sup>12</sup> (column 2 and 3) on environmental justice arising from interviews with Bacton, Walcott and Happisburgh residents, and policymakers. As outlined in section 4.3.3, it was not possible to record (and therefore transcribe) two walking interviews with local residents (due to weather conditions), and therefore counts for Column 4 in Table 6.1 are out of 34 rather than 36. Light grey shading = expressed by residents only, dark grey shading = expressed by policymakers only, no shading = expressed by both interview groups (no. or % references by policymakers in relation to total references in brackets) (Table: I Cotton).

1. Theme	2. Parent codes	3. Child codes	4. No. interviews expressing code	5. No. references expressing code
Distributional justice	Distributional justice	Fairness in where gets protection	20 (4)	35 (29%)
		Impact of no defences	6	14
		SMP policy	5 (1)	9 (11%)
		Unjust decision- making	4 (1)	9 (11%)
		Impact of policies on risk	12 (1)	14 (7%)
		Poor implementation of policies	3	7
	Infrastructural intervention	n/a	23	48
	Capabilities	Witnessing coastal change	7 (1)	13 (8%)
		Worry for future of village and livelihoods	13	22
	Intergenerational justice	n/a	5 (2)	5 (40%)
	Relations of place	Cultural heritage and cultural value	11	28
		Sense of place	13 (2)	32 (6%)
	Inevitability	n/a	17 (2)	25 (8%)
	Scalar	Jurisdictional	17 (5)	62 (48%)
	mismatches	Spatial	17 (2)	26 (8%)
		Temporal	20 (6)	31 (48%)
	Uncertainty	n/a	10 (5)	24 (67%)
Procedural justice and recognition	Procedural justice	Meaningful consultation	14	35
		Limits of consultation	14	21
		Everybody has a right to know	15	22
	Knowledge	Lack of knowledge	8	15
	-	Sources (who/what) of knowledge	16 (3)	29 (10%)
		Themes or subject of knowledge	14 (1)	26 (4%)

<sup>&</sup>lt;sup>12</sup>References were counted for each separate point made by an interviewee, but purely repeated points under a particular child code were only counted once (to avoid double-counting).

	Trust	Motive of policymakers	15 (1)	35 (6%)
		Strained relations with policymakers	21 (1)	61 (5%)
		Good relationships with policymakers	9 (2)	12 (33%)
	Recognition	n/a	8	16
	Community relations	Sandscaping disagreements	15	23
		Car park disagreements	6	20
		General village tensions on future options	14	27
		Differences of opinion on coastal management	9	21
		Parish councils	10 (3)	12 (25%)
	CTAP	Language style	10 (1)	18 (6%)
		Perceptions on purpose	18 (6)	40 (38%)
	Adaptation	n/a	(6)	37

# Use of key terms by interviewees

Local residents and policymakers commonly use the terms 'managed realignment' and (managed) retreat' interchangeably. While used synonymously, in theory 'managed realignment' refers solely to shoreline change, and therefore could refer to accretion and not just erosion. Furthermore, the term 'adaptation' is typically used by interviewees to refer to an undefended coastal policy scenario (i.e. managed realignment), although adaptation could also take place on a defended coastline. In particular, policymaker interviewees use the terms 'adaptation' or 'transition' to refer to specific actions that might fall under a managed realignment scenario, such as property rollback or relocation. This may be because 'managed realignment' or 'retreat' can be viewed unfavourably by local stakeholders (Nicholson-Cole and O'Riordan, 2009).

# Coastal data

Two sampling areas were chosen to investigate rates of cliff retreat at Happisburgh, given the frontage contains two cliff areas of differing shape and

erosion rates. A sampling area of 850m length was chosen at two sections to provide sufficient coverage of the overall frontage. In ArcGIS Pro, polygons of the sampling areas were created, and LiDAR data from 2015-2022 was extracted and clipped. The cliff line at each sampling area was identified using the slope function, and the top of the cliff line was traced for each year's LiDAR data. A line transect that represented a desired straight line of the cliff was created, to generate line transects every 5m along the frontage, at a 250m sampling length. This was used to calculate the distance between the youngest (2022) and oldest (2015) cliff lines. At Happisburgh South, gullying resulted in irregular erosion into and behind the general cliff line. To include this erosion in the rates of retreat, where it occurred the distance between the 2015 and 2022 cliff line transects was calculated solely using the areas of eroded land (as opposed to both eroded and intact areas), i.e. the distance across the gulleys was included, but not any intervening land. This provided an estimate of cliff retreat across an 8-year period, which was used to calculate an annual rate of cliff retreat, both for each transect and an overall average for each sampling area (Happisburgh North and Happisburgh South).

## 6.4 Results

#### 6.4.1 Perceptions of managed realignment

Almost 20 years after a policy of managed realignment was introduced at Happisburgh, interview and survey data confirmed it is still disputed by Happisburgh residents, some of whom are vocal that Happisburgh should be defended. 78% of completed surveys from Happisburgh explicitly called for new or reinstated coastal defences in open-text answers. Meanwhile all Happisburgh interviewees highlighted the built cultural heritage or beauty of the village, with many listed buildings that are at risk from erosion. This highlights a sentiment of feeling let down by a national coastal policy formula that uses cost-benefit analysis, which only prioritises defences in areas of significant assets or population, rather than aspects of non-monetary value strongly felt by local residents. It also reveals the strong sense of place Happisburgh residents feel to their village:

"It's frustrating when you're basically told that, you're not worth it, you're not important. There's not enough show. A church founded in 9 something (century) or other isn't important. The lighthouse, the iconic things in all of North Norfolk, no...it's all going to go in the sea" ~ Happisburgh resident

"I don't know how much it costs to reinstate some sort of sea defences that would prolong the life of the village, but compared with what is at loss, and what potentially could just go" ~ Happisburgh resident

Even amongst Happisburgh residents who perceive management realignment in the long-term is inevitable nevertheless feel coastal defences are needed in the short-term, because the rate of coastal erosion is happening too quickly. The perceived risk is that the entire village of Happisburgh is under threat from erosion. This raises a critical question on how much time is required for adapting to managed realignment, potentially through relocation (the adaptation option described by interviewees below), and whether there is a role for coastal defences in the short-term to support a longer-term scenario of managed realignment.

"We need to buy some time, 30-40 years, so that the mechanism by which you move back an ancient village with old buildings, listed buildings, etc can be properly formulated. For the government and for the minister to come here and say 'adaptation is the only way forward and we, you, you're going to have to move', is facile in the extreme" ~ Happisburgh resident

"More natural ideas just to slow things down, like planting things that can root in to hold the cliff in a bit more... not the rate it's going at the moment, it's just absolutely crazy, it's a whole village that we're going to lose...I know I'm going to lose my house, but to save the whole rest of the village would be nice" ~ Happisburgh resident

But not all Happisburgh residents think coastal defences should be used, or are practically feasible or attainable. A support for defences at Happisburgh is also less prominent amongst interviewees from Bacton and Walcott (currently protected from erosion and flood risk by the Bacton-Walcott sandscaping scheme) for economic reasons, but it was also reflected that views on managed realignment would be different if their own village were to face a similar scenario:

"We are not going to get defences. It will never happen. So there's no point in trying to kick people by saying, 'oh, we're going to continue campaigning for defences'...that causes more grief, for people already under pressure" ~ Happisburgh resident

"I think it should be left to go 'cause you know, there are all sorts of things for the birds to eat, the seed heads on the so-called weeds. And let's let it go. But, so there are different views" ~ Happisburgh resident

"When you think about those places like Happisburgh where they've just gone 'managed retreat ok, it'll just, your house will just fall into the sea'. And so sort of conceptually, and economically, you can say ok, well, that's the only option, but actually, when it's your home, and your children's future, you think no". ~ Bacton resident

Local policymaker interviewees highlighted that coastal defences are being used in Happisburgh currently (rock armour) to little effect, and that managed realignment would be required with or without coastal defences in place. Policymaker interviews confirmed CTAP funding would not be used on coastal defences:

*"We've moved it (the rock armour) three times. You know, it's just not practical. There is absolutely no way you can stop erosion, whether defended or not".* ~ Local policymaker interviewee "CTAP isn't going to buy another 9,000 tons of rock, you know, that wouldn't be an appropriate sort of thing" ~ Local policymaker interviewee

Policymakers at all levels of government highlighted that the repeated use of defences for high-risk, fast eroding coasts needs to fundamentally change, referring to it as an unsustainable policy cycle that needs to be 'broken':

"You have to be careful that they give us some time for adaptation, and then when that time is out, they say, can you give us some more time? So that's the challenge" ~ Regional policymaker interviewee

"Getting a community to understand the need to adapt, getting them on board is at the moment very, very difficult. We have a cycle where we put defences in, and the community, they then lobby for further defences. " ~ Regional policymaker interviewee

In summary, this section has explored the different perspectives of local residents and coastal policymakers on the use of coastal defences and a managed realignment policy at Happisburgh. Happisburgh residents raise distributional justice concerns in the differential protection allocated to different sections of the coast in England, and argue for more pluralistic decision-making in how coasts are valued for coastal defence schemes, going beyond economic assessments to also consider non-monetary forms of value, such as the cultural heritage of a settlement. Different 'subjects' of justice are referred to by interviewees, in terms of whether the issue of coastal defences and managed realignment at Happisburgh is spoken about at an individual, community or national (population) level. Further, interviewees raise different arguments for or against the use of defences at Happisburgh according to different spatial and temporal scales.

#### 6.4.2 Lived experience of coastal change

Happisburgh residents spoke about the impact of living alongside an actively eroding coast, in both individual and community terms. At an individual level, residents highlighted the disproportionate impact of low house prices and difficulty selling or developing a local business, and the eventual cost to demolish their homes. At a community level, residents raised concerns about blight and a disappearing community:

"We treat people on the coast, like we would never treat anybody anywhere. No help when you lose everything you own. A bill for the demolition of your own property. You know, it's absolutely beyond the pale in the 21st century and the mother of democracies that you would treat people like that, but they do". ~ Happisburgh resident

In particular, the lack of financial compensation for properties lost to erosion was seen as a significant injustice. A few residents, largely from Happisburgh but also from Bacton and Walcott, explicitly describe managed realignment as an unjust coastal policy if it is without financial compensation for properties that are lost to erosion. This has been a longstanding argument by local residents of Happisburgh, that if property was purchased pre-2005 (when coastal policy was 'hold the line' nationwide), central government cannot change coastal policy without compensation. Managed realignment without financial compensation and/or practical support is seen as both an abdication of responsibility by national government, and removing rights of citizens on the coast. One interviewee from Bacton raised an example of citizen compensation in other policy contexts, such as the (then) current development of the high speed railway HS2, implying unfair treatment of citizens at a national level.

*"If you're going to compulsory purchase houses, knock them down because you're going to build a new railway...well, then those people are compensated* 

at fair market value... You can't just suddenly change a policy, for the greater good, and then do nothing and try and get away with it on the cheap. That's just not right, and I mean Happisburgh is the most ridiculous example" ~Bacton resident

"We have a bit of a legal case because it's kind of like going, 'really I think you should house us should we become, you know, homeless, because your policy decisions have made us homeless, or changed from when we bought it'" ~ Happisburgh resident

Local policymakers referred to the Coastal Erosion Assistance Grant (Defra, 2020), which provides local authorities with £6,000 to support the financial costs of demolishing at risk properties, but it was evident this is not on the scale of compensation envisioned by resident interviewees. Other financial support discussed by policymakers include a similar mechanism of planning rights that was offered in specific cases with the 2012 Happisburgh Pathfinder programme through the EN12 local planning policy (NNDC, 2024), or a national scale property reinsurance model for erosion or climate change (no such product currently exists in England asides from flood reinsurance (FloodRe), although a proposal for CoastRe has been circulated to Defra (EFRA Committee, 2019)). Regional and local policymakers highlighted the lack of centralised policy funding and guidance on adaptation to managed realignment is a key challenge at a local level. There is therefore still considerable uncertainty on financial mechanisms to support a policy of managed realignment, such as financial support discussed here specifically in relation to relocation of homes:

"Going back to Pathfinder, one of the things we did was when we got properties at risk. We actually worked with a reputable surveyor to actually give us a present day value so that in some cases, we might be able to give them that value so that they can move on, and they have a planning right to redevelop anywhere in North Norfolk" ~ Local policymaker interviewee

193

"(On reinsurance) We could look at that for coastal change or, you know, perhaps a wider climate change type reinsurance product, we could look at a levy funded type of model" ~ Local policymaker interviewee

"Because of the funding, political and social issues, we can't yet have a plan that definitively says we will help those people roll back and relocate because it's not currently supported...we'd love to have a plan, but unless the plan is underpinned by appropriate funding and legislation, we can't really have the plan" ~ Regional policymaker

Happisburgh residents expressed mixed feelings towards the adaptation options trialled during the 2012 Pathfinder project. While interviewees were generally supportive of the programme in principle because it offered a form of financial support for at-risk properties, it was felt the financial support offered for demolished houses should have been higher. Furthermore, residents expressed disappointment in the price of the new houses built inland in the village through the programme, and the delays to completing construction works:

"They've had very expensive houses built just on the outside of here, near the school. And they're not houses for people that can afford to live here." ~ Happisburgh interviewee

"And that went very badly in that three of them still haven't been built. The site itself isn't finished. There are people that have bought, and are living, but the road has never been finished in its tarmacking...and what they replaced them with, is ludicrously expensive, executive seaside homes with fancy kitchens, they didn't replace them with anything remotely affordable or accessible" ~ Happisburgh interviewee

Happisburgh residents (and Bacton and Walcott residents speaking about Happisburgh) articulated the disorientation they feel from the speed of loss of land at Happisburgh to coastal erosion. Processing loss of buildings and changes to one's local surroundings was described as surreal: "But then all of a sudden, you can't see those houses now...I can visualise the road in my head, but when I'm stood on the end of that road, you're just like 'Now, where is such and such's house'. Because you're trying to work it out. But it's just. It's been swallowed up like it's never existed". ~ Happisburgh resident

"It's got to mean, something to get your head around, the fact that when the coast disappears, I mean, it's not just sort of being knocked down, the whole landmass is gone isn't it? The whole shape of the country has changed. So I was looking at the Ordnance survey map, the new one that had just come out, and for the first time ever I noticed they've actually sort of noted the bay that's forming on us" ~ Happisburgh resident

## Cliff retreat (2015-2022)

The speed of environmental change at Happisburgh spoken about by interviewees is illustrated in Figure 6.1, which comprises of two graphs of 8-year cliff retreat across an 850m transect at Happisburgh North (panel a) and Happisburgh South (panel b), from 2015-2022. Below this are maps of Happisburgh North (panel c) and Happisburgh South (panel d) overlayed with the 2015 and 2022 clifftop line, to illustrate the extent of cliff retreat over the 8-year period. Cliff retreat is calculated as the distance (in metres, m) between the 2015 and 2022 clifftop line.



**Figure 6.1** Cliff retreat (metres, m) at Happisburgh North (panel a, panel c) and Happisburgh South (panel b, panel d) from 2015-2022 (Data: Anglian Coastal Monitoring Programme). Satellite imagery from ArcGIS Pro © ESRI, 2023 (Figure: I Cotton).

As can be seen from Figure 6.1, both the North and South frontages have actively receded in recent years, but cliff erosion is highly variable. The extent of erosion at Happisburgh North varies from 0.4-53.6m, and 0.0-33.3m at Happisburgh South, over the 8-year period analysed. Despite being so close geographically, Happisburgh North and Happisburgh South show different patterns of erosion and different average erosion rates across the whole transect. Cliff retreat is not linear, and can occur after a period of stability, affected by a transition to different geological/geomorphological properties (e.g. grain size), and rainfall intensity (BGS, 2021), causing highly localised erosion patterns. Erosion rates were 15.0m (+/- 11.2m) on average across 2015-2022 at Happisburgh North compared to 8.6m (+/- 8.3m) at Happisburgh South. This corresponds to an average annual cliff retreat of 1.9m yr<sup>-1</sup> (+/- 1.4m) at Happisburgh North and 1.1m yr<sup>-1</sup> (+/- 1.0m) at Happisburgh South. These estimates are larger than long-term average annual estimates for Happisburgh of approximately 0.4m yr<sup>-1</sup> (1907-1950) (Dickson et al., 2007) highlighting that erosion rates are currently greater than they were historically, and vary both naturally and due to anthropogenic intervention. For example, erosion rates at Happisburgh in 2001-2003 were 8-10m yr<sup>-1</sup> (Poulton et al., 2006), following the demolition of hard defences (Payo et al., 2020).

Furthermore, cliff retreat estimates in Figure 6.1 reflect an average over an 850m cliff sampling area. Annual cliff retreat specifically at Beach Road (which is located at the end of Happisburgh North transect, and where many houses have historically been demolished) is notably higher. For example, panel c (bottom picture inset) highlights that 54m of cliff has retreated near Happisburgh car park from 2015-2022. This indicates that some sections of the cliff are particularly vulnerable, and having retreated several tens of metres inland, whereas elsewhere along the cliff there is less discernible change. At Happisburgh South, an embayment (area of fast eroding coast) can be seen towards the mid-way point

of the transect (points 52-63 on the graph in panel b in Figure 6.1, corresponding to the bottom picture inset of 27m cliff change in panel d), suggesting a highly localised cause for erosion at this particular location. Given landslides and groundwater (water below the land surface) are notable contributors to erosion at the clifftop at Happisburgh (alongside erosion at the base of the cliffs due to waves) (BGS, 2021), the location of a historic channel refill could be one reason why the cliff is more easily eroded at this particular location. One Happisburgh interviewee articulated the impact of this localised and sporadic pattern of geomorphological change:

"Two years ago we were thinking 'oh that's getting close', but then nothing happened... we don't know where we are, it's, we're completely at the mercy of the tides and time" ~ Happisburgh resident

Happisburgh interviewees reflected on the impact of witnessing this coastal change, which causes significant concern for some, and is a detriment to their mental health:

"I occasionally I have nightmares about trying to yell at people, 'You just need to leave the house now! It's about to fall!' So every now and then I have sort of like bad dreams about it, and then wake up going 'Oh no, it's fine. It's not happening tonight' " ~ Happisburgh resident

"And I don't know whether I could bear to live here, looking at the church as I do, and obviously that goes first. That and the lighthouse kind of go at the same time, but I don't know whether I could bear the heartbreak of watching that building fall into the sea" Happisburgh resident

"It's like the elephant in the room, it's there, but no one wants to talk about it, and people that are trying to sell their house and things like that don't want others talking about it" ~ Happisburgh resident Bacton and Walcott residents, commenting on Happisburgh, raise similar fears for their villages in the future if Bacton Gas Terminal is not further protected after sandscaping. A few interviewees who live on the coastal frontage recall the experience of their house shaking during storms, before sandscaping was introduced in 2019:

"We can see from villages up the coast what will happen. And it is a very, it's a slow death of, you know, a particular village and the particular community" ~ Bacton resident

"We were relieved! about the sand (sandscaping scheme), that we might stand to be here a bit longer because of that. The house doesn't shake with the waves now, and the waves don't land on the front windows, like they used to" ~ Walcott interviewee

The wellbeing impacts from erosion were also raised by policymakers, arguing that conversations on adaptation and the long-term plan for at-risk settlements are needed now to create a counteracting, positive vision of the future:

"The whole point of coastal transition and adaptation is that coastal change doesn't take you by any surprise, that it is managed and we want to see not only resilient communities, but we want to see thriving communities". ~ National policymaker interviewee

In summary for section 6.4.2, Happisburgh residents raise many different impacts of coastal change (financial loss, blight, sense of loss, anxiety, and uncertainty for the future), that the village is disproportionately affected by through living alongside an undefended coast. Figure 6.1 highlights the fast rates of cliff retreat that the village has experienced in a recent 8 year period (2015-2022). This again highlights issues of distributional justice, in that certain settlements such as Happisburgh experience the impacts of coastal change more than areas which might be defended, such as Cromer (also on the Norfolk coast).

## 6.4.3 Community consultations and involvement in coastal management

Across all three villages, there is mixed opinion by residents on whether previous community consultations on coastal management have valued and incorporated local views. Views were particularly negative in Happisburgh, and interviewees felt their views were sought too late in the decision-making process, and thus not valued or listened to:

"I would like residents to be consulted prior to any policy being formed...government would set out its policies to us. Open up a 12 week consultation. And then just go ahead with what they said anyway, because people were only ever open to any formal consultation after the policy had been set". ~ Happisburgh resident

(Referring to a previous consultation on offshore wind farm development) "Parish council said no, District Council said no, central government went, 'you're having it. We've set aside as sort of a period of consultation, but the decisions already been made for it'." ~ Happisburgh resident

This was a particularly strong theme from Happisburgh residents who completed the survey, where nearly a third commented on feeling ignored or overlooked, despite the open-text questions of the survey not specifically eliciting views on past community engagement:

"Constantly ignored by higher powers" ~ Happisburgh survey respondent

"No one listens to the people of Happisburgh" ~ Happisburgh survey respondent

In the case of Bacton and Walcott, there are both residents who are deeply satisfied and deeply dissatisfied with how the decision for sandscaping was made

and publicly communicated, prior to its implementation in summer 2019. Residents who objected to sandscaping felt the decision had already been made to implement the scheme before community consultation. Because of this, some residents felt that local knowledge and lived experience of the area was overlooked and ignored. On the other hand, there are some Bacton residents who were satisfied with how local residents were involved and informed about the sandscaping scheme. As discussed in Chapter 5 (section 5.5.1) different resident perceptions on public consultation may link to an individual's overall trust in coastal management, perceptions of expert knowledge and views on sandscaping:

"What they could have done was bought in people who have been here, again, all their lives. And put them onto the steering committee and the decisions that were made. And just included them so that they could hear them out. But there wasn't the, locals weren't given anything like that. They were told 'this is what's happening, what do you think?' 'Well we don't like it'. 'Tough, it's going ahead'. That was the perception." ~ Walcott resident

I think they have to involve, you know, local people, and not just go ahead, the powers that be to say 'this is what we're doing'. I think they'd do it anyway, but at least they should, you know, say, 'right guys, what do you think?' and hear what people have got to say. They'll do it anyway, probably, but I don't think they've done that, and that's what they should do going forward" ~ Bacton resident

"I actually thought it was really good, how they, how the County Council actually organized it all...we had letters, we had flyers come through the post on a regular basis, there was a lot of interaction, and then it was, all the signs went up, and there was a picture of how it's going to work, and explaining how it works" ~ Bacton resident

Consultation, and information about the coast, was framed by residents across the three villages as a citizen right (and lack of information as an injustice). All three villages spoke about the right to information on how their local coast is changing,

as it affects livelihoods. Whether in the case of Bacton and Walcott on sandscaping, or of Happisburgh on managed realignment, a theme cross-cutting interviewees from different villages is a desire for more information and communication from coastal managers on coastal change:

"If there's going to be anything done which would inhibit my access to the sea, or anything which would have an impact on my enjoyment of it, I feel like I should be consulted because I bought there for that reason" ~ Bacton resident

Policymakers expressed the aim to get as many residents as possible involved in Coastwise, which aims to develop community adaptation plans for high risk areas on the Norfolk coast (perceived roles and responsibilities of local residents in local coastal adaptation is explored in the following chapter (section 7.4.4), with this chapter focusing on perspectives of public consultation and community engagement on coastal change). Multiple engagement approaches were mentioned to achieve this, but the most common route highlighted by interviewees is through parish councils:

"Well, we will be employing a Community Liaison Officer. And obviously the first step is to introduce him or herself to parish councils. We've got to encourage parish councils to hold public meetings so we can explain what we're likely to be doing". ~ Local policymaker interviewee

"So parish councils, I think are going to be absolutely critical. They provide that real grassroots connection to community.". ~ Regional policymaker interviewee

However, some resident interviewees expressed concern about parish councils acting as a community voice on coastal management issues for their village. Interviewees highlighted examples of differences of opinion between parish council members and other local residents on coastal management issues, such as the future location of Happisburgh car park, once access on Beach Road is lost due to erosion. Some local residents object to Parish Council proposals to create a new road, and thus alternative access route to the car park, at the back of Beach Road properties, and a few interviewees remarked they see no way in which differences of opinion can be solved:

"The entrance to the car park is, it's going to be threatened by the sea in a year or two. So we need to re-route. But it means putting a roadway down behind people's gardens, and now some of them are really up in arms against it. So whereas they'll tolerate the erosion or the erosion is some sort of theoretical thing, the idea of having a road down the back of their gardens, it just appals them and they're fighting it tooth and nail". ~ Happisburgh resident

*"I think you can't assume that the parish council speaks for everyone. Because there, always it's a small number of individuals and they, yeah. Not necessarily representative of everyone" ~* Walcott resident

Meanwhile, policymaker interviewees expressed the challenge in how coastal management issues are covered by the media, in terms of sensationalising the issue of erosion and reporting inaccurate details on coastal policy. One local policymaker recounted an article in the Eastern Daily Press in January 2023 (which inaccurately claimed at-risk properties would be offered property compensation through the CTAP project):<sup>13</sup>

"That was kind of a classic one where they just wanted to write a nice headline, a big headline, and they did. And then it went in the EDP. And actually the BBC surprisingly copied it...that's not what it was about. And it then raises people's expectations and they'll come back to us 'well, what about this' like, well, we never said that, you know, we said lots of other good things, but we didn't say that" ~ Local policymaker interviewee

<sup>&</sup>lt;sup>13</sup> Eastern Daily Press, 2023. North Norfolk council set to buy homes at risk of erosion <u>https://www.edp24.co.uk/news/23226258.north-norfolk-council-set-buy-homes-risk-erosion/</u>

"We anthropomorphize the sea, you know, we instil this feeling of grief and some of that is the media. I understand, you know, places lost to the sea. The lighthouse lost its battle with the sea". ~ Regional policymaker interviewee

Resident interviewees who had heard of CTAP at the time of interviews remarked about the Press release by Defra, and communication from their local MP, as being vague and offering limited information of what CTAP would involve. This further highlights themes of distrust in policymakers that was discussed in Chapter 5:

"Our MP sent round a nice little letter, kind of 'Oh! you know, I've been to Happisburgh. Heard you're a bit concerned. Don't worry, there's Pathfinder 2' sort of thing. And I'm going, what does that practically mean for anyone? And I couldn't find any practical anything in that document. There are a lot of woolly meaningless phrases and words" ~ Happisburgh resident

"Well, the first thing I think we'd like is for them to talk to us about what it should achieve because including 'transition' in the title immediately raises concerns for people". ~ Walcott resident

In summary, there is scepticism amongst local residents, although not expressed by all, that past processes to seek community views on coastal management strategies have not meaningfully incorporated the views of local residents. This is a sentiment expressed most strongly by survey respondents and interviewees from Happisburgh, reflecting the different history of coastal management than at Bacton and Walcott. Overall, there is a widespread desire across all three villages for more knowledge and a greater say in how the coast is changing, and how it is managed. At the time of the research, local residents raise concerns on the extent and transparency of information about CTAP that is available, and the methods of community engagement that the project might use. This section has also highlighted once again, through the example of Happisburgh car park, the wide differences of opinion on practical aspects of managing coastal change at a local level, which raises key challenges for coastal managers on how differences of opinion can be accommodated.

## 6.5 Discussion

#### 6.5.1 Environmental justice issues of managed realignment

The perspectives presented here highlight managed realignment at Happisburgh is still being disputed by local residents - almost 20 years on from the SMP policy change in 2005 – and may remain contested without a form of financial compensation and/or further short-term use of hard or soft engineering to slow the rate of coastal change. These findings correspond with past research on the SMP policy change at Happisburgh (Tebboth, 2014) and across Norfolk (Adger et al., 2011) that a lack of compensation for properties lost to erosion, or a lack of coastal defences, is perceived as removing rights and altering the 'social contract'<sup>14</sup> that existed between citizens and government pre-2006. Other studies elsewhere in the UK (Fairbourne, Wales) (Arnall and Hilson, 2023) and other countries (Gibbs, 2016; Lawrence et al., 2020; Hanna et al., 2020: 2021) similarly argue that managed realignment is being contested by communities at high risk, that the scientific basis of erosion projections or removal of defences are disputed, and that there are calls for property-level financial support. In this study, a frequently mentioned theme by Happisburgh interviewees is that the rate of coastal change experienced in recent years is too fast, ultimately threatening the cultural heritage of the village. It therefore appears that issues regarding managed realignment at

<sup>&</sup>lt;sup>14</sup> The 'social contract' was theorised by philosopher Jean-Jacques Rousseau (1762) as a delicate balance in the relationship between individual and state in terms of the rights and responsibilities of each other.

Happisburgh have evolved over time from solely debates on continuing to defend (Adger et al., 2011; Tebboth, 2014) - to which opposing justice arguments can be made depending on different philosophical foundations of justice and different scales (Cooper and McKenna, 2008; Adger et al., 2011) - to how managed realignment in a just manner can be facilitated. Local Happisburgh residents put forward their perspective of an environmentally just approach as one that controls the rate of change at a pace that preserves the identity and integrity of the village, with adequate practical and financial support.

Existing research has argued acceptance of managed realignment takes considerable time and engagement with local communities (Moore, 2012; Sayers et al., 2022), perhaps even decades (Haasnoot et al., 2021), an imperative for initiating conversations now, to ensure adaptation is planned rather than reactive (Haasnoot et al., 2021), and less uncertain (Gibbs, 2016). This sentiment was reflected by many policymaker interviewees in this chapter, and Haasnoot et al., (ibid) report examples internationally of reactive retreat that create negative socioeconomic impacts, such as not enough jobs where communities relocated. Of significance to research question 3, it can therefore be argued that proactive coastal adaptation is likely to be a more resilient and a more just approach to managing coastal change. However, a lack of funding and guidance on coastal adaptation entails there is still ambiguity on the practical dimensions to supporting communities to proactively adapt. The call for funding and national policy guidance for coastal adaptation by local and regional policymaker interviewees reflects longstanding arguments (O'Riordan et al., 2006; Day et al., 2015; EFRA Committee, 2019; Brown et al., 2023) that there is insufficient funding and guidance to practically support managed realignment at a local level for high risk coastlines such as Happisburgh. While the 2019 SMP refresh did lead to supplementary guidance documents (such as Natural Resources Wales, 2021),

there is little detail on overcoming challenges to community engagement in coastal change issues, and studies in other national contexts reveal there is a similar lack of policy guidance worldwide (Lawrence et al., 2020, Hanna et al., 2020: 2021).

While the results reveal differences of opinion within and between local villages on the use of coastal defences, there is a clear distinction amongst those arguing for defences and the perspectives of coastal policymakers, who stressed no hard defences or market-value property compensation is possible under current Coastwise funding. This raises a relevant question, argued by Few et al., (2007) on whether participation in coastal adaptation can ever be fully participatory, if it is concerned with the how rather than fundamentally if managed realignment should be the SMP designation, and this reflects the position of some Happisburgh interviewees in this chapter. Although adaptation options under managed realignment, such as property and infrastructure rollback, have already been trialled at Happisburgh (2012 Pathfinder Programme), it may be challenging for local policymakers to present a compelling vision for local residents of managed realignment in the future, without further defences or property compensation. Furthermore, it may be difficult to sustain coastal adaptation planning at Happisburgh after the end of Coastwise funding, if an equivalent funding source for coastal adaptation is not financed at a national level. This could endanger community trust and limit progress towards effective, accepted and long-term community coastal adaptation planning.

## 6.5.2 Psychological dimensions of living alongside a retreating coast

Local residents across all three villages express concern about the future of Happisburgh and the impact of blight on the village. Furthermore, recounts of past experiences of coastal storms highlight the disproportionate financial and wellbeing impacts experienced by undefended communities in such events, when compared to other coastal settlements whose SMP policy remains 'hold the line'. The impact of coastal change for undefended communities in terms of blight and storm damage has long been highlighted (Adger et al., 2011; Brown et al., 2023). However, that Happisburgh interviewees also reported impacts at a community level from the 2012 Pathfinder programme, in terms of the affordability and delay to the construction of relocated housing in the village, highlights that adaptation options associated with managed realignment (in this case property rollback) need to also consider longer-term, secondary impacts to a settlement, and how potential solutions impact the characteristics and sense of place of a community.

A strong theme emerging from resident interviews are the wellbeing impacts of coastal change, and in particular, processing the rate of cliff loss from erosion. At 1.1m–1.9m yr<sup>-1</sup> (+/- 1.0-1.4m), average annual retreat at Happisburgh North and South over a recent eight year period (2015-2022) is higher than historic annual average rates of cliff retreat (approximately 0.4m yr<sup>-1</sup>, 1907-1950) (Dickson et al., 2007). Alongside the speed of cliff loss, cliff retreat calculations reveal erosion at Happisburgh is highly localised and variable year-to-year, and therefore unpredictable how much of the coast will retreat, and where. The interrelationships between the social and natural components of the coastal system are evident from the empirical data presented in this chapter, and appear to relate prominently to uncertainty, which will be discussed further in Chapter 8. Disorientation associated with witnessing coastal change, where loss of one's local environment can render parts almost unrecognisable to before, is reported by Moore (2012) in their study of the 2012 Pathfinder programme in Dorset and East Devon.

Given that climate change is accelerating sea level rise and intensifying coastal storms across the UK (MCCIP, 2020), the psychological dimensions of coastal

change may become increasingly significant as this century progresses. Residents from all three villages in this study express concern in how climate impacts may worsen coastal change, a sentiment echoed by residents at risk of flooding in Surrey, and erosion in Hemsby (also in Norfolk) (EA, 2021a). A recent account of a Hemsby resident who has lost their home to erosion (interviewed by the UK newspaper, The Guardian, 2023) reveals they did not have access to counselling at the time, and mental health support for flooding is also argued to be inadequate (EFRA Committee, 2019). Wider provision of mental health support for affected residents could therefore be beneficial, particularly given that concern of flood and erosion risk could act as a barrier to community engagement. While the EA (2021b) have quantified the mental health impacts of flooding, no such figure exists for erosion (although a project is underway) (EA, 2023b). Considering how coastal adaptation policies can simultaneously support the mental health impacts associated with coastal change therefore appears a relevant objective, and there may be relevant learning from research on eco-anxiety and climate anxiety (see for example Clayton (2020)).

# 6.5.3 Community involvement in coastal decision-making

There is widespread desire from local residents across all three villages for more information and communication from policymakers on managing coastal change. This is a theme that also strongly emerged from the empirical data explored in Chapter 5, and relates here to the timing of community consultations on coastal management. Happisburgh residents in particular argue their views were either not considered, or considered too late to have meaningful input into decisionmaking, when past coastal management proposals were shared for public consultation by the local council or the EA (the latter with respect to flood risk). Previous research on community involvement in SMP policy in East Anglia

similarly concludes engagement was not fully participatory, or lacked sufficient and regular communication to communities (O'Riordan et al., 2006; Brennan, 2007; Adger et al., 2011). Issues of meaningful community consultation are not resigned to East Anglia or solely the SMP consultation process: other research on coastal management in Lincolnshire (Myatt et al., 2003), East Devon and Dorset (Moore, 2012), nationwide (Famuditi et al., 2018; EFRA Committee, 2019) and other locations (Costas et al., 2015; Domingues et al., 2017) all report community perceptions of being unable to fully contribute views in local decision-making. The significance of this, as evident in the perspectives of Happisburgh interviewees in this study, is there are high levels of residual scepticism in coastal decisionmaking, partly due to past experience. More positively, policymaker interviewees describe the role of communities in Coastwise as co-collaborating on transition plans about the future of local settlements. This suggests a higher and more collaborative approach to coastal governance, according to the different facets of coproduction outlined by Galende-Sánchez and Sorman (2021). Coastwise therefore has the opportunity to address past issues of participatory justice in coastal decision-making, with more equal power sharing of ideas on managing coastal change.

However, this research indicates there may be several challenges to community engagement in coastal adaptation discussions. The years-long dispute between Happisburgh Parish Council and local residents on rollback of the village car park exemplifies that a wide range of opinions will exist within a community specifically on *how* and *where to* with regards to relocating infrastructure. This has been found elsewhere, for example the rollback of a community café from the coast in Studland, Dorset, which took three years to reach consensus within the community (EFRA Committee, 2019). This has implications for how policymakers seek to engage local communities on coastal adaptation plans. While policymakers propose using multiple engagement routes as part of Coastwise, interviewees also highlighted the importance of parish councils in community engagement. The varying perspectives on managed realignment raised by different Happisburgh residents however, suggests that utilising one community group to express local views would be incomplete.

This study finds the views of Happisburgh residents coalesce around core just principles, such as financial support and community participation in coastal management, but diverge around the practical dimensions of adaptation planning (relocating infrastructure). A cluster of different positions, rather than one fixed 'community' view, is more apparent in the data. Wide-ranging engagement, using adept community facilitators, supported by time and resources, will be needed to ensure all voices are heard, as similarly argued by Moore (2012). Paradoxically, community engagement in coastal adaptation also needs to be mindful that not all residents may want to be involved in a topic as sensitive as coastal change, or involved to the extent envisaged by policymakers. Crucially, this requires high levels of trust from the outset between policymakers and local residents. Although later work in a Norfolk context has found relations between local authorities and communities have improved over time, and since the revised SMP publication (O'Riordan et al., 2014; Famuditi et al., 2018), that themes of tokenistic engagement are still expressed by interviewees here suggests building trust remains a key objective (as previously advocated by O'Riordan et al., 2006; Adger et al., 2011; Famuditi et al., 2018).

## 6.5.4 Communication channels and conflicting narratives

Resident interviewees recount previous public consultations, such as the 2005 SMP update and a previous Natural England proposal to allow flooding at Potter Heigham (a nearby village on the Norfolk broads), were leaked by the local press before the consultation was officially announced by local authorities (CCAG, 2023b). Meanwhile, policymaker interviewees reflected on the challenge of sensationalised or inaccurate media reporting of coastal change, including local and national news articles in 2023 that falsely claimed property compensation would be offered as part of Coastwise. The role of the media in sensationalising and influencing narratives on climate or coastal change has been highlighted (Moore, 2012; Harcourt and Dessai, 2023), and policymaker interviewees argue it falsely raises public expectations of coastal policy that cannot be matched. This could potentially undermine public trust in coastal management, and at the time interviews with local residents were conducted in Autumn 2022, there was some scepticism of CTAP based on the limited information available since Defra's announcement in Spring 2022. It may therefore be practical to delay public announcements of future adaptation funding or policies, until more of a business case has been prepared by local government, so that the absence of information after a policy press release does not feed doubt and scepticism by local residents (for example, that local community views are being overlooked), or misreporting in local or national media.

However, these findings also relate to a more fundamental issue about how the future of a settlement, and the issue of coastal change, is framed in society. Recognitional injustices could arise where local residents feel the future of where they live, and its links to personal identity is misrepresented by media, local authorities, or the general public. One Happisburgh interviewee remarked there are tensions within the village on the extent to which Happisburgh is perceived to be defined by coastal erosion, and discussed, which is feared to impact housing sales. In Fairbourne, Wales, a village similarly at high risk of erosion and under managed realignment, local residents have spoken publicly against media

labelling of being Britain's first 'climate refugees' (Arnall and Hilson, 2023). Furthermore, the decision by Fairbourne residents to set up a rival community group (Fairbourne Facing Change) due to disagreements with the local council initiated group (Fairbourne Moving Forward) (Arnall and Hilson, 2023) reflects not just disagreements in practicalities of managing coastal risk, but a battle of narratives about a settlement and how it is changing, and a desire for communities to frame the narrative themselves. At Happisburgh the campaign 'Save Happisburgh Action Group' (2023) has recently been set up, and which continues longstanding community activism on coastal issues within the village. There may therefore be relevant learning in using an established, local authority and local resident communities *provide rather than receive* information on coastal change to the media, and ultimately have greater ownership over the narrative of local coastal change.

#### 6.6 Conclusions

This chapter has highlighted a range of perspectives and environmental justice considerations on managed realignment at Happisburgh, that may act as barriers to facilitating community engagement in conversations on future adaptation plans for the village. Managed realignment, and adaptation options associated with the policy, are unlikely to be accepted by local residents without sufficient financial and practical support to adapt (distributional justice). Furthermore, a call for coastal defences or soft engineering may continue if residents perceive the rate of environmental change as so fast that it threatens the identity and future existence of their village. Residual scepticism and distrust in past community consultations present obstacles in persuading communities to engage in Coastwise, and to trust it as a process that will meaningfully incorporate their views (participatory justice).

Differences of opinion within local communities on the practical details of rollback should not be underestimated in terms of the length of time and engagement required to reach a future adaptation plan that the village collectively has a sense of shared ownership in (recognition).

Overall, the findings highlight that a policy of managed realignment, and willingness to engage in coastal adaptation planning, is only likely to be accepted locally if it addresses the aforementioned justice issues of distribution, participation and recognition. If not, coastal adaptation planning may not be perceived by communities as a fair and legitimate process to manage coastal change. This chapter therefore demonstrates that incorporating justice principles into coastal adaptation planning is fundamental for successful community engagement, both to overcome historic environmental justice issues and to facilitate community willingness to adapt in future coastal decision-making. The next chapter of this thesis continues to explore community perspectives of managed realignment and long-term coastal change, but switching from the context at Happisburgh to that of Bacton and Walcott. Chapter 7 presents views of Bacton and Walcott residents on coastal management after sandscaping. It does so ultimately to explore whether, and on what scales, sandscaping has impacted social resilience to managing coastal change at Bacton and Walcott.

# Chapter 7. Challenges to anticipatory coastal adaptation for transformative nature-based solutions: local resident and policymaker perspectives of a post-sandscaping frontage at Bacton and Walcott

This chapter is based off the publication; Cotton, I., Forster, J., Lorenzoni, I. and Tolhurst, T.J., 2024. Challenges to anticipatory coastal adaptation for transformative nature-based solutions. Global Environmental Change, **88**, Article 102893. Available at: <u>doi.org/10.1016/j.gloenvcha.2024.102893.</u> IC's role in the paper: conceptualisation, methodology, data collection and analysis, writing (lead), review and editing

# 7.1 Introduction

This chapter explores the perspectives of Bacton and Walcott residents and coastal policymakers on managing the risk of coastal change for the Bacton-Walcott frontage after the current sandscaping scheme. This principally relates to answering research question 2c, and draws upon the survey and interview data of Bacton and Walcott residents, and interviews with local, regional, and national policymakers. These perspectives are presented alongside sediment volume calculations, to explore changes at the Bacton to Walcott frontage in the first four years of sandscaping (research question 1a and 1c). A brief background section (7.2) summarises previous research on adaptation to future climate risk ('anticipatory adaptation'), with a particular focus on coastal contexts. Section 7.2 also outlines what has been said publicly to date by local government and other key stakeholders on coastal management post-sandscaping at Bacton and Walcott, and the future of Bacton Gas Terminal. Section 7.3 provides an overview of the data presented in this chapter, and the method used to calculate sediment volume estimates for the coast from 2015-2022. The results section presents resident and policymaker reflections on managing future coastal risk for Bacton Gas Terminal (7.4.1), perceptions of coastal adaptation under a future policy
scenario of managed realignment (7.4.3) and perspectives on the role of local residents in coastal adaptation (7.4.4), alongside sediment volume calculations (7.4.2). The chapter ends with a discussion section (7.5) focusing on the implications of these perspectives on long-term social resilience to coastal change.

## 7.2 Background context

### 7.2.1 Anticipatory adaptation to coastal change, and perceptions of coastal risk

This sub-section briefly summarises notable research in the UK to date on public risk perceptions of climate change or coastal change, alongside perceived citizen responsibility of adaptation. Whilst it is strongly argued that communities need ample time to prepare for coastal adaptation (Day et al., 2015; Mott MacDonald, 2016; NNDC, 2016; EA, 2020), other authors have argued an overemphasis on time, and an overemphasis on the future, affects the way in which the public adapts to climate risks in the present (Nobert and Pelling, 2020). Studies on adaptation to extreme heat (Wolf et al., 2009; Porter et al., 2014; Nobert and Pelling, 2020) find households typically opt for reactive adaptive measures (e.g. clothing choices) rather than proactive or anticipatory measures with long lead-in times (e.g. retrofitting air conditioning). In a coastal context, international studies highlight coastal defences can undermine risk perceptions for vulnerable communities (Luis et al., 2015; Arias et al., 2017; Nunn et al., 2021) where over time experience of storms and strong attachment to the coast combine to increase risk normalisation (i.e. acceptance of coastal risk) (Luis et al., 2015; Domingues et al., 2017: 2021; Bertoldo et al., 2021). Crucially, Luis et al., (ibid) argue the role of risk perception in motivating coastal adaptation may be time-sensitive, and weaken if risk is perceived to be far into the future. A large number of factors have been found to contribute to risk perceptions or concern of climate change, and include perceptions of personal vulnerability (Sjöberg, 2000), the scale of the issue (Kollmuss and Agyeman, 2002), direct experience of flooding or extreme weather (Spence et al., 2012; Capstick et al., 2015; Demski et al., 2016; Arias et al., 2017; Bruine de Bruin and Dugan, 2022), political values (Whitmarsh, 2008; Ogunbode et al., 2017) and age (Grothmann and Reusswig, 2006; Whitmarsh, 2008; Zaalberg et al., 2009).

Previous research on coastal adaptation has focused on property-level flood adaptation measures rather than erosion risk, or is typically from the perspective of key stakeholders rather than local residents (Esteves and Thomas, 2014; Van der Plank et al., 2020; Brown et al., 2023). Furthermore, nature-based solutions (the use of natural features and processes rather than man-made infrastructure (Seddon et al., 2020)) are increasingly framed as an alternative to the maladaptive, repeated use of hard defences in coastal management (Nunn et al., 2021). While the role of nature-based solutions in buying time and providing a stepping stone from 'hold the line' to 'managed realignment' has been acknowledged (Brown et al., 2023), there is little research directly evaluating this and the practical dimensions of the policy transition (Brown et al., 2023), alongside a lack of case study research at a local level on adaptation to coastal change (Harcourt et al., 2023; Magnan et al., 2023). Anticipatory adaptation in contexts where naturebased solutions are introduced has been understudied, with previous research on nature-based solutions instead exploring public perceptions on their effectiveness (Gray et al., 2017; Joseph and Humphries, 2018; Anderson and Renaud, 2021), uptake (Moraes et al., 2022) or underlying policymaker or stakeholder motivations for their introduction (Santoro et al., 2019; Ferreira et al., 2022).

217

As outlined in section 2.2.2, local authorities and risk management authorities have a responsibility to manage erosion and flood risk, respectively, but roles and responsibilities for adaptation is more ambiguous in the current FCERM policy (EA, 2020). Work by Van der Plank et al., (2022, p.14) argues responsibility for adapting to coastal flood risk can be broken down into five typologies: "personal responsibility to be aware and prepared...financial responsibility to bear the costs...citizen responsibility to be engaged in decision-making...legal responsibility to act within the scope of the law...and state responsibility to the welfare of its citizens". Therefore, the authors (ibid) argue responsibility for adaptation cuts across different actors, and includes the state but also at an individual level. Van der Plank et al., (2020) investigated perceived public responsibility for flood risk management in the North West (Lancashire and Cumbria) and South (Hampshire, Dorset, Isle of Wight) of England, from the viewpoint of local authorities and stakeholders. Interviews revealed a wide range of opinions on the role of communities, from solely being aware of flood risk, to being equally responsible in writing local flood risk management plans (Van der plank et al., ibid). There is increasing emphasis in FCERM policy (EA, 2020) on local community involvement in coastal adaptation plans, and a key objective of CTAP, as similarly discussed in Chapter 6 in the context of Happisburgh, is for local authorities to co-develop transition plans with local communities in high risk areas. While there has been ample research on public participation in SMPs (for example Few et al., 2007; Nursey-Bray et al., 2017), as similarly highlighted in Chapter 6, research is now needed to understand local resident perspectives on co-developing coastal adaptation plans, and how to facilitate a willingness to adapt proactively to coastal change amongst affected communities.

#### 7.2.2 The future of the Bacton-Walcott frontage

This sub-section outlines public announcements relating to the future of the sandscaping scheme and Bacton Gas Terminal, as it is pertinent to findings discussed in this chapter. General background context on the sandscaping scheme and coastal management at Bacton and Walcott can be found in section 2.4. There has been no public communication by either NNDC or the Bacton Gas Terminal operators that the sandscaping scheme will be repeated. There has, however, been frequent media coverage speculating that Bacton Gas Terminal will repurpose into a Bacton Energy Hub in the future and supply renewable energy (for example EDP, 2023b). In December 2022, the Bacton Energy Hub Special Interest Groups (2022) produced a report outlining the business case for renewable energy development at the site (hydrogen, offshore wind, and Carbon Capture and Storage (CCS)). Soon after the Government's announcement in summer 2023 of continued annual gas and oil licenses in the North Sea (HM Government, 2023b), Bacton Gas Terminal was officially granted a license for CCS from the North Sea Transition Authority (NSTA) in September 2023, the first time such permits were authorised (NSTA, 2023). During a visit to Bacton Gas Terminal in Autumn 2023, then Prime Minister Rishi Sunak described it as a "hidden hero" for UK domestic energy production (BBC, 2023c, p.1), signalling strong political interest at a national level in continuing to protect the gas site from the risk of coastal change, and the role it could play to meet the UK's 2050 Net Zero target.

NNDC's website for Coastwise (a £36M project in North Norfolk funded by Defra to trial different initiatives for coastal adaptation, see section 2.2.1) includes information about "*the plan for the Bacton to Walcott coastline once the sandscaping has depleted*" (NNDC, 2023d, p.1). It states there is a possibility "*further coastal work*" (NNDC, ibid, p.1) may take place by Bacton Gas Terminal,

but it does not specifically refer to a repeat of sandscaping, nor that any future work would protect the villages of Bacton and Walcott again. Instead, it stresses that the frontage should prepare to adapt to coastal change (under managed realignment): *"The project* (sandscaping) *looks to provide up to 15 years of protection for the Bacton Gas Terminal, creating time to plan and prepare for coastal change in this area. The terminal may complete further coastal work to protect the site in the future. This may benefit the Bacton to Walcott coastline, but this is not certain. We will need to work with the communities to plan for future coastal change, and Coastwise allows us to begin this planning process" (NNDC, <i>ibid, p.1*).

### 7.3 Materials and Methods

#### 7.3.1 Data collection

#### Social data

This chapter draws upon the survey responses (77) and interviews (22) of Bacton and Walcott residents, and compares these perspectives with the views of local, national and regional policymakers (6) working within a coastal flooding or erosion context. All of the policymakers interviewed were familiar with the sandscaping scheme and SMP policy context at Bacton and Walcott. As noted in Chapter 6 (section 6.2.2), CTAP was officially announced by Defra in spring 2022, subsequent to the survey (January 2022) but before resident (Autumn 2022) and policymaker interviews (Early 2023). Although CTAP had been officially announced at the time of the interviews, there was no official communication on where within Norfolk projects would take place, and NNDC had not announced information about the local project, subsequently named Coastwise. Chapter 4 details the sampling strategy for the survey (section 4.3.2) and interview (4.3.3) data collection, and a copy of the survey and interview questions can be found in the appendix (appendices 1,2, and 4). Section 4.3.2 details the survey response rate and response count by village, while section 4.3.3 details counts of in-person, phone or online interviews. Themes and topics reported on in this chapter include:

- Local resident and policymaker perspectives on the future of Bacton Gas Terminal, and the risk of coastal change during and after the sandscaping scheme;
- Policymaker perspectives on community involvement in Coastwise, and a policy of managed realignment;
- Resident perspectives on perceived responsibility in participating in coastal adaptation decision-making.

## Coastal data

This chapter analyses the same LiDAR dataset from the ACM Programme as investigated in Chapter 6, which is annual LiDAR (elevation) estimates from Bacton Gas Terminal to Happisburgh, from 2015 – 2022. Further information on the dataset can be found in section 4.3.1.

## 7.3.2 Data analysis

### Social data

Closed-text survey questions were analysed using descriptive statistics, and opentext survey questions and interview data were thematically analysed using an inductive and deductive coding approach that is detailed in Chapter 4 (4.4.2 and 4.4.3). Table 7.1 in section 7.4 details the codes and themes of the interview data explored in this chapter. The full coding framework can be found in Appendix 6.

## Coastal data

This analysis sought to compare changes in sediment volume from Bacton Gas Terminal to Happisburgh in the four years prior to sandscaping (2015-2018, inclusive) with the four years after sandscaping (2019-2022, inclusive). Five sampling areas of 850m length were chosen at Bacton Gas Terminal, Bacton, Walcott, Happisburgh North and Happisburgh South (see sampling areas in Figure 7.1 below). An 850m sampling length was used, given it covers the majority of the frontage at Bacton Gas Terminal and the three villages. As reported in Chapter 6, given the coastline at Happisburgh contains two cliff areas of differing shape and erosion rates, two sampling areas were chosen for Happisburgh (Happisburgh North and Happisburgh South) to account for these differences. The Happisburgh sampling areas (4 and 5 in Figure 7.1) are the same as in Chapter 6.



**Figure 7.1** Sampling locations for sediment volume calculations (from top to bottom): 1) Bacton Gas Terminal (green) 2) Bacton (yellow) 3) Walcott (orange) 4) Happisburgh North (Blue) and 5) Happisburgh South (Pink). Sampling locations were all a fixed 850m in length, and contained the maximum LiDAR data width across the years 2015-2022 (Figure: I Cotton).

In ArcGIS Pro, polygons of the five sampling areas were created, and LiDAR data from 2015-2022 was extracted and clipped. The cliff line or sea wall at each sampling area was identified using the slope function. Where the sampling area contained a cliff (for example at Bacton Gas Terminal, Happisburgh North and Happisburgh South), the top of the cliff line was traced for each year's LiDAR data (2015-2022). Conversely, where the sampling area contained a fixed sea wall (at Bacton and Walcott), the sea wall line was used for all LiDAR years, given the position of the sea wall is fixed. Therefore, beach volume calculations included all sediment at each sampling area to a depth of approximately -2m (which equates to the maximum LiDAR width in the ACM programme data), and to the top of the cliff or sea wall. The LiDAR data was clipped a second time, to remove any elevation data inland of the cliff/sea wall in each sampling area. Surface volume was calculated for each sampling area from 2015-2022.

## 7.4 Results

Table 7.1 summarises the codes and themes emerging from the interview data reported on in this chapter. The results section is ordered according to these three themes, and first presents views of sandscaping and Bacton Gas Terminal (section 7.4.1, which corresponds to codes under the first theme 'coastal management') followed by views on managed realignment in general and maladaptation (section 7.4.3, which corresponds to codes under the themes 'scale' and 'adaptation'), and lastly perspectives of community responsibility in coastal adaptation (7.4.4, codes under the theme 'coastal management' and 'adaptation'). The table reflects counts of Bacton and Walcott residents only (i.e. tallies have removed references from Happisburgh residents for this chapter). The table is shaded in grey-scale to illustrate themes discussed solely by residents

(light grey shading) and policymakers (dark grey shading). Themes expressed by

both interview groups are not shaded, and the number/percentage split of

references from policymakers is noted in brackets.

**Table 7.1** Themes (column 1) and codes<sup>15</sup> (columns 2 and 3) arising from interviews with Bacton and Walcott residents, and policymakers. Some parent codes (column 2) are divided further into relevant child codes (column 3) (parent codes without child codes are listed as 'n/a' (not applicable) in column 3). As outlined in section 4.3.3, it was not possible to record (and therefore transcribe) one Happisburgh resident and one Bacton resident walking interview due to weather conditions, and so coding totals reported in this chapter do not include the walking interview from Bacton (i.e. counts for Column 4 in Table 7.1 are out of 21 rather than 22) (Happisburgh interviews are not reported on in this chapter). In columns 4 and 5, codes with light grey shading = expressed by residents only, codes with dark grey shading = expressed by policymakers only, codes with no shading = expressed by both interview groups (the number or % references by policymakers are in brackets). Codes are arranged within each theme group (column 1) according to most common codes (see number of interviews that expressed a particular code, column 4) (Table: I Cotton).

1. Themes	2. Parent Codes	3. Child Codes	4. No. interviews expressing code	No. references expressing code
Coastal	Bacton Gas Terminal	n/a	17 (4)	34 (29%)
management	Feeling secure	n/a	17	47
	Perceived	Policymaker responsibility	13	26
	responsibility	Community responsibility	(2)	2 (100%)
Scale	Scalar	Temporal	18 (6)	29 (52%)
	mismatches	Spatial	13 (2)	19 (11%)
		Jurisdictional	13 (5)	43 (70%)
	Environment/	Climate change	11 (3)	21 (24%)
	social tensions	Need to consider environment	8	16
	Uncertainty	n/a	8 (5)	22 (73%)
Adaptation	Infrastructural intervention	n/a	19	37
	Coastal Transition Accelerator Programme (CTAP)	Perceptions on purpose	13 (6)	24 (63%)
	Adaptation	n/a	(6)	37 (100%)

<sup>&</sup>lt;sup>15</sup> References were counted for each separate point made by an interviewee, but purely repeated points under a particular child code were only counted once (to avoid double-counting).

## 7.4.1 Feeling secure and the future of Bacton Gas Terminal

Over three-quarters of the Bacton and Walcott residents interviewed expressed a sense of security from the risk of coastal erosion or coastal flooding by living adjacent to Bacton Gas Terminal. The terminal was described as an infrastructure that will be protected at any cost. Although the villages were vulnerable to flooding and erosion before sandscaping, some residents who had previous direct experience of storms commented that the introduction of sandscaping has assuaged any previous concerns about property damage.

"I said 'Look. Look out of our window, you've got two aerials there, and you've got Bacton gas site. They are not going to let that drop into the sea, because it supplies a third of the country's gas'. So, as long as that is there, this little house will be fine" ~ Bacton resident

"I think that we're very lucky 'cause we're one of the sort of safest places on the North coast for erosion because, we have got the gas site and they will never let it fall into the sea" ~ Bacton resident

"Our house had been trashed, and we had to move out for 15 months...We weren't overconcerned, unless we heard that the wind the tides and what have you were going to get that surge again... while that sandscaping's there, I don't believe it will. I believe we are perfectly safe" Bacton resident

Contributing to this perceived sense of security is the observation that the implementation of sandscaping- at a cost of £21M (Johnson et al., 2020b)occurred at a time where elsewhere along the Norfolk coast there has been no further intervention of defences, despite high erosion risk:

"I should imagine that Happisburgh, Hemsby, Scratby and that area, they'll be the areas that will be in need of moving, because already properties have fallen into the sea, so again, it may not look the prettiest thing, but the gas works in Bacton are really significant to Walcott, because they've got to protect that in the short term" ~ Walcott resident

"Bacton feels like a really secure location because so much has been committed to ensuring we stay buffered...when everything else is so uncertain at the moment, I mean the fact that we can be fairly sure that we won't fall into the sea like Happisburgh residents" ~ Bacton resident

Furthermore, residents reflected that Bacton Gas Terminal will only increase in importance in the coming decades: socioeconomic and geopolitical issues at the time of the interviews (Autumn 2022), such as the cost of living crisis and war in Ukraine, were both mentioned by resident interviewees as evidence that the terminal is of great importance to the UK's energy security. While several residents debated whether this would change through the pursuit of green energy policies, it was more commonly thought that the terminal will repurpose as a hydrogen and CCS site in the future, and therefore remain a critical energy infrastructure for the UK in the decades ahead.

"I did read an article. I need to reread it, which I think said something about using the gas field, the empty gas field, for carbon capture? ... Yeah, so they'd have to maintain that, I suppose, wouldn't they? ~ Bacton resident

"The money that the North Sea is going to generate, to put that sand there again by the Dutch, it's peanuts...they earn so much money out of those gas licences now, 'cause we're not going to buy a Russian gas again. Nor is Europe" ~ Walcott resident

There is evidence that the sandscaping scheme has impacted the property market, with a few new residents confirming they would not have moved to the area without the scheme:

"I hadn't really considered the coast because of coastal erosion...and then when we came here and we saw the sandscaping, and I think it sort of gave that, sort of, dream that I could live by the sea, that it would be safe for a while" ~ Bacton resident

*"I thought 'sod it, I'm going to go for it' 'cause it's going to be safe now for Bacton Gas Station, the terminal. And so they're going to protect that"* ~ Walcott resident

Drawing on the observations above, there is a widespread hope amongst residents that sandscaping, or some form of coastal protection, will continue beyond the current project. 51% of survey respondents (n=68) called for sandscaping to be repeated after the current scheme's projected lifetime<sup>16</sup>, and a further 28% of respondents called for more hard defences or some form of alternative sea defences. In interviews, while some residents expressed a hope for further intervention, for others this was more strongly felt as an expectation that the coastline at Bacton and Walcott will remain defended:

"I would hope for that by the time that this needs to be done again, I'm hoping that Bacton Gas Terminal is still important enough that needs to happen again" ~ Bacton resident

"That's a lot of money's worth there. They're not going to let that be undermined by the sea. They're going to save that" ~ Walcott resident

"Bacton gas site, it supplies a third of the country's gas, well, if that isn't in the national interest, I don't know what is...it's a national thing like the Channel Tunnel. And it needs to be looked after appropriately, and that needs national funding" ~ Bacton resident

Despite this belief, there is not currently, nor has there ever been, any public communication indicating that sandscaping will be repeated. Interviews with

<sup>&</sup>lt;sup>16</sup> In response to survey question; "What coastal management, if any, do you think should happen in your village in 15-20 years, which is after the projected lifetime of the Sandscaping scheme?" (total responses to question= 68)

policymakers confirmed this is merely speculation at present. There is a lot of uncertainty, currently, about the future of the terminal, the future decision-making of terminal operators, and its implications for the coastal management policy of the frontage.

"There are going to be those discussions sort of with, I imagine with government, with planners, about those possibilities and those companies. And then if it is re-used like that, it's like well within that mix needs to be the 'well, OK, how's the site going to be managed? Is it going to roll back from the coast or is there the potential for future, for the sandscaping, or other activities like that?' So, but I wouldn't want to say that would be the case because it's a complete unknown" ~ Local policymaker

*"I think, so that is the big question as to whether they* (the terminal operators) *have future, you know, aspirations to repeat a similar sandscaping exercise in 15 years' time"* ~ Regional policymaker

"The sandscaping project has certainly been successful in terms of buying time to have conversations, particularly with the community in that area, around long-term adaptation and that need to live with rising tides. And that's not to say that a future way of improving resilience wouldn't be another sandscaping. So I think it's too early to say" ~ National policymaker

In summary, the implementation of sandscaping appears to have contributed to confidence within the local community that the Bacton and Walcott frontage will remain defended. This confidence has led to anecdotal evidence of increased buying and selling of local property, and reflects a hope or expectation for many residents that sandscaping will be repeated at the end of its lifetime. There is evidence that this perceived sense of security is becoming more strongly felt by local residents since sandscaping was completed in 2019, due to emerging socioeconomic and geopolitical contexts that further underline the importance of Bacton Gas Terminal. However, a second sandscaping scheme is currently purely

speculation, and there has been no official communication, by local government or the terminal operators, that the current project would be repeated.

## 7.4.2 Sediment volume change Bacton to Happisburgh, 2015-2022

One policymaker reflected that any decision on whether to repeat sandscaping would be taken further into the scheme's lifetime, once it has become clear how the scheme is performing geomorphologically:

"You know, if it performs as we hope it will do, you know they (Bacton Gas Terminal operators) will get to a point where they'll need to start thinking about it, but that won't be at the end of the scheme, say 15 or 20 years. I personally would expect them to be looking at it probably 10 years if it's performing as we hope it is" ~ Local policymaker

This section analyses sediment volume change in the first four years of the scheme. Figure 7.2 and Table 7.2 detail changes to beach sediment volume at Bacton Gas Terminal and the three villages before sandscaping (2015-2018) and after sandscaping (2019-2022). Volume estimates in Table 7.2 encompass all sediment present from the top of the cliffs (at Happisburgh) or sea wall (at Bacton and Walcott) to the entire beach area, including the intertidal zone (to approximately -2m elevation). Percentage change in sediment volume compared to the previous year is also calculated and listed next to sediment volume estimates (in red or green font in Table 7.2, to indicate percentage decrease or increase, respectively).



**Figure 7.2** Surface volume estimates (gross, m<sup>3</sup>) year to year (2015-2022) at Bacton and Walcott (panel a) and Happisburgh (panel b), with respect to Bacton Gas Terminal (BGT, grey line in panel a and b, where the majority of sediment was placed from sandscaping in 2019) (Figure: I Cotton).

**Table 7.2** Surface volume estimates (gross, m<sup>3</sup>) and percentage change (% against previous year) at Bacton Gas Terminal, Bacton, Walcott, Happisburgh North and Happisburgh South from 2015-2022 (Table: I Cotton).

Surface volume (m <sup>3</sup> ), and percentage change (%) from previous year							
Year	Bacton Gas Terminal	Bacton	Walcott	Happisburgh North	Happisburgh South		
2015	453,879	180,424	119,436	253,669	373,726		
2016	465,606	154,949	118,735	241,437	404,196		
	(+3%)	<mark>(-14%)</mark>	(-1%)	(-5%)	(+8%)		
2017	449,106	156,393	118,427	215,417	343,845		
	(-4%)	(+1%)	(-0.3%)	(-11%)	(-15%)		
2018	456,980	183,725	146,903	246,317	334,239		
	(+2%)	(+17%)	(+24%)	(+14%)	(-3%)		
Sandscaping – Summer 2019							
2019	936,830	274,999	259,298	196,676	387,996		
	(+105%)	(+50%)	(+77%)	(-20%)	(+16%)		
2020	832,796	264,420	227,371	242,984	334,633		
	(-11%)	(-4%)	(-12%)	(+24%)	(-14%)		
2021	749,019	279,071	219,259	257,119	372,069		
	(-10%)	(+6%)	(-4%)	(+6%)	(+11%)		
2022	701,276	281,639	188,577	221,865	372,900		
	(-6%)	(-0.9%)	(-14%)	(-14%)	(+0.2%)		

Figure 7.2 and Table 7.2 reveal the dramatic impact of sandscaping in 2019 on beach sediment volume. This is most pronounced at Bacton Gas Terminal (+105% increase), where sediment more than doubled in the case study area (and where the majority of placed sediment was introduced), but also observable further down the frontage at Bacton (50% increase) and Walcott villages (77% increase). The 2019 changes to sediment volume in these three sampling areas are noticeably higher than interannual variability pre-sandscaping, and are statistically significantly different (independent t-test assuming unequal variance, to a 95% confidence interval, 2015-2018 vs 2019-2022) (Bacton Gas Terminal; p=0.007, Bacton; p<0.001, Walcott p=0.004). In contrast, changes to sediment volume at Happisburgh North and Happisburgh South after 2019 are not dissimilar to the

interannual variability pre-sandscaping, and are not statistically significant (to a 95% confidence interval, 2015-2018 vs 2019-2022) (Happisburgh North; p=0.57, Happisburgh South; p=0.89). Given Happisburgh is outside of the area of intended coastal protection from the sandscaping scheme, such findings are not surprising, but nonetheless highlight the distributional impact of the scheme in terms of the protection afforded to Bacton and Walcott, in comparison to Happisburgh.

With the exception of 2021 at Bacton (a 6% increase on previous year), the sampling areas at Bacton Gas Terminal, Bacton and Walcott have seen a decrease in sediment year on year since sandscaping was completed in 2019. Average yearly decline is -9% at Bacton Gas Terminal and -10% at Walcott, but a greater volume of sediment is estimated in the Bacton sampling area in 2022 (281,639m<sup>3</sup>) than 2019 (274,999m<sup>3</sup>). Although this suggests there has been a greater decline in sediment volume at Walcott than Bacton following sandscaping, estimates are based on LiDAR data sampled at different dates each year during the Winter season, and within a particular sampling area. In other words, a similar trend may be the case elsewhere at Bacton and Walcott, or at different times of the year. The considerable variability in sediment volume between Happisburgh North and Happisburgh South, which often see opposing changes year on year despite being close geographically, further highlight the uncertainty and variability in coastal change year to year. In conclusion, there has been a dramatic change in beach volume that has instantaneously built resilience to coastal storms at Bacton Gas Terminal, Bacton and Walcott. Sediment volumes are fluctuating, and placed sediment is gradually relocating along the coast over time, highlighting that the scheme, and associated physical resilience, is only temporary.

## 7.4.3 Perspectives on future managed realignment at Bacton and Walcott

The strongest cross-cutting theme from policymaker interviews, discussed at length and on multiple instances by all interviewees (regardless of whether they work in a local, regional, or national context) is the need to prepare now for a future managed realignment policy at Bacton and Walcott (for example code 'adaptation' in Table 7.1). Policymakers unanimously discussed the purpose of the sandscaping scheme as ultimately an opportunity to 'buy time' to prepare for adaptation:

"What was sort of woven throughout the business case that was submitted for the funding to Bacton and Walcott is the need to adapt and the need to start thinking about adaptation and that's really what the frontage needs to be considering" ~ Regional policymaker

*"It* (sandscaping) was about buying time, and we kind of said it's like turning the clock back. But when we get, we'll get back to a situation where we're going to need to change and adapt. So we need to use this time wisely" ~ Local policymaker

Policymakers stressed it was important to begin engaging with local residents on adapting to a managed realignment policy, and referred to the current Coastwise project in Norfolk (then known nationally as CTAP) as an opportunity to do this, which will involve developing adaptation or 'transition' plans for an area:

"We've reduced that risk, we've now given ourselves time, bought time, to really need to adapt the frontage and this is where I think CTAP can start to come in. And because one of the deliverables from CTAP is going to be transition plans amongst other things...So that's what we need to start thinking about how we are going to adapt this community" ~ Regional policymaker While local policymakers described discussions about adapting to managed realignment in Bacton and Walcott as in an early stage, policymakers also felt a sense of urgency is recognised by local residents on the need to plan for adapting to managed realignment:

"I think at the outset and in the early community engagement before the sandscaping was built, it was quite clearly highlighted...that this was a 20 year timeframe sort of project, and it was about buying time to allow communities to adapt. And so that message was clearly delivered" ~ Regional policymaker

"They (Walcott residents, parish council meeting in 2022) wanted to talk about adaptation and the Parish Council...they were talking about the need to plan to adapt. You know, what do we need to do next. Which was music to my ears, to be perfectly honest, that they're thinking like that. Rather than just thinking 'oh, well, we're safe now' or 'when's the next Coast Protection Scheme' ~ Local policymaker

However, these policymaker perceptions do not appear to reflect the perspectives of the majority of residents surveyed or interviewed here. The need to prepare now and plan for erosion risk after the sandscaping scheme's lifetime was not explicitly expressed by a single resident interviewee. Furthermore, an open-text survey question (n=58) asking residents if any actions should be taken now (other than sandscaping) to prepare for future coastal change found a third of responses (33%) called for more, or more maintained, hard defences. In addition, residents called for improved drainage or some other form of coastal management intervention. Only a handful of survey answers referred to community discussions or consultation, and no answers referred to adapting to managed realignment. For more elderly residents, should any property rollback or wider relocation form part of a managed realignment plan at Bacton and Walcott, there is a general perception this will be outside their lifetime, regardless of whether sandscaping is repeated or not. Interview data indicates that this group of residents is not unduly

concerned about, or sees the personal relevance of, future coastal management post-sandscaping.

"If, once and if, they ever do decide that we won't be getting gas anymore and they don't change what they do there, then any kind of protection we get, oh, I can see it will just go, but again I'm thinking it's not going to be in my lifetime" ~ Bacton resident

"People were 'oh ok, the sand's here, the sea's gonna last 20 years'. Lot of old people were saying, 'well, that will see me out. I'd be happy with that" ~ Walcott resident

One resident expressed a strong desire not to move, regardless of the future situation of erosion risk or an offer of property planning rights elsewhere (as was the case in the 2012 Happisburgh pathfinder programme) (Frew, 2012). Their reflections underline the strong sense of place many residents feel to Bacton and Walcott:

"In 10 years time if they turn around and say 'oh your house is right on the edge of the cliff, and you've got to be moved', I'd rather they spent the money to keep the cliff farther out to sea. Or they do the seawalls in good maintenance, and for me to see my time out here, 'cause I'm settled here and I'm happy here. And again, as you get older, being uprooted is even harder. To turn around to a young couple with a couple of kids or just a young couple or even a middleaged couple and say, 'well, we're gonna offer you your full right to your house and then you can go and move to somewhere else'. 'Yeah, OK fine'. They might be able to do that, by try moving a 70/80 year old person out of their property, they'll turn around and say 'I'll drown here'. They will" ~ Walcott resident

Even amongst younger residents, a common perception was that erosion risk would not return immediately post-sandscaping, and that the residual life of the sea wall and other currently redundant hard defences at Bacton and Walcott would buy an additional amount of time so that this group of residents will also not be directly affected. This was a view expressed both by interviewees living a few streets inland and on the immediate coastal frontage:

"Even if it wasn't replenished, and the seawall becomes more exposed again, then it's still probably another 10-15 years before the seawall would be breached" ~ Walcott resident

"There's a government website about flooding risk... and it was something like 50 years before where our house might be affected. Well, that's out of our lifetime. That's somebody else's problem, you know" ~ Bacton resident

"We'd have to move. Yeah, and I certainly know at Happisburgh people have been assisted in moving. So I imagine that if it came to that, there would be something like that, but we don't expect it in our lifetime, really" ~ Walcott resident

In summary, there is an urgent need, according to policymakers, to prepare for managed realignment at Bacton and Walcott, and sandscaping is framed as a vital project that has enabled a space for anticipatory adaptation to future erosion risk. However, across different generations at Bacton and Walcott, it appears this sense of urgency expressed by policymakers is not similarly felt by many residents, and that sandscaping has taken the question of coastal management outside of local residents' direct concerns, at the time interviews were conducted (Autumn 2022, year 3-4 of the sandscaping scheme). While largely confident that the frontage will remain protected, residents are at the same time fearful of what a managed realignment policy might entail, and refer to the rollback that can be witnessed at present at Happisburgh (explored in Chapter 6). There is a hope by many local residents that sandscaping has bought enough time that residents do not have to be directly confronted by a managed realignment policy.

### 7.4.4 Perceived role of local residents in coastal decision-making

Policymakers and local residents also appear to have different perspectives on the role of local residents in managing coastal change, as the Bacton-Walcott frontage moves to a managed realignment epoch. Policymaker interviewees, particularly those working at a local level, expressed a desire for local residents to be heavily involved in Coastwise, describing transition plans as something that should be 'co-created' with local communities. Furthermore, it was conveyed that the local community, alongside local businesses, share a certain level of responsibility in coastal adaptation

"I'm hoping with our stakeholder engagement theme, that's kind of running throughout all these different actions, communities are going to be involved from the start. We want co-creation and collaboration when it comes to these transition plans. So yeah, involved throughout in every aspect" ~ Regional policymaker

"A huge emphasis is going to be placed on North Norfolk and East Riding of Yorkshire (Local councils receiving CTAP funding) to actually work with and engage communities, that the whole aim is to look at actually co-creating the future, the future of that coast" ~ National policymaker

*"It's about everybody playing a role, and people taking and businesses taking certain levels of responsibility. So it's not local government doing it to people, but it's actually everybody"* ~ Local policymaker

When asked what the key challenges were to facilitating coastal adaptation under a managed realignment SMP policy, interviewees at different levels of government reflected on challenges to facilitating community acceptance of such a policy, and challenges to practically implementing a nationwide, replicable, policy: time, trust and place attachment were all highlighted as key barriers. "In short, it's not easy and the time it takes cannot be underestimated. And so what goes with co-creation is that joint respect and understanding, and that takes a certain amount of time because that requires trust across not only the coastal managers but also the communities that they're working with" ~ National policymaker

*"I think it comes back to the community engagement and the co-creation. And I think ultimately it's about the acceptance of change"* ~ National policymaker

"Probably one of the biggest ones to overcome is the human element and sort of the attachment to place" ~ Regional policymaker

But regional and local policymakers also highlighted the lack of centralised policy funding and guidance on coastal adaptation is a significant challenge:

"The difficulty is because of the funding, political and social issues, we can't yet have a plan that definitively says we will help those people roll back and relocate because it's not currently supported, so it's a bit cart before the horse or chicken and egg – whatever metaphor you want to use. We'd love to have a plan, but unless the plan is underpinned by appropriate funding and legislation, we can't really have the plan. So that's the difficulty that we need to solve" ~ Regional policymaker

"Which is always a challenge to do that sort of thing, but sort of co creating what? What might we need to do. And we've been sort of tasked with creating transition plans, but we've, there's no rulebook to say what, that is not defined in any way." ~ Local policymaker

Meanwhile, policymakers described the work Coastwise would support at a local level, which focused on a need to increase community knowledge and awareness of the risks of coastal change:

"One is around sort of improving coastal literacy, coastal understanding about the coast, people understand what's going on, why, that side of it. There's understanding the baseline. So that's people's perceptions, thoughts, hopes and dreams, you know. But it's also about the demographics and all of that element and also the geographical. So actually understanding the place better and it's then about really it's co-creating sort of a way forward with the community" ~ Local policymaker

"We probably got a piece of work on people in place and they're preparedness, which is obviously going to have a really strong stakeholder engagement theme running through it, whether it's guidance and a real educational element when it comes to living on a dynamic coast, that kind of thing" ~ Regional policymaker

Although engagement in Coastwise had not begun at the time of interviews with local residents, interviewees described a more limited role in coastal decisionmaking than policymaker interviewees. Picking up litter, acting as flood wardens, or supporting hard defence schemes at a property level (if directly living on the frontage) were some example roles suggested by resident interviewees.

*"Well, I think I think, You know, well, we're always picking up litter off the beach.* ~ Walcott resident

"So that's the sort of individual responsibility that you, you can choose to pick it, or just leave it to get worse. So that's one thing. I mean, there's not. We keep doing our little bank haven't we, you know, in our house. We've been sort of redoing our garden and creating a little bank, Which should we be flooded, might help keep the water away from the house, and help it go down the road, rather than, in a drain process" ~ Walcott resident

When prompted on whether residents could (or should) be involved in decisionmaking about managing coastal change at Bacton and Walcott, local residents described providing input to coastal management proposals, but reflections did not describe the level of responsibility (co-creation of adaptation plans) as envisaged by policymakers.

"You know, at the level that you've seen today, you know I'm not, not particularly technical, but I've got a keen interest. So yeah, I would certainly be part of discussions or would be willing to do it" ~ Bacton resident "I think in terms of leading it has to be the Council, and the local Council, because if local people can't live, it's going to be their issue, to have to sort out isn't? It if they've suddenly got a population - and I don't know how many people living in Bacton, 3,000 or whatever - who are not able to live there. Then we've got an issue" ~ Bacton resident

There is some scepticism amongst resident interviewees of public consultation (which relates to findings in Chapter 6). Furthermore, in response to a question about managing coastal change while sandscaping is in place<sup>17</sup>: 14% of survey answers articulated not knowing or feeling unqualified to answer, suggesting that residents may feel there is a limit to which they can contribute to decisions about coastal management on their local frontage.

*"I think there will always be people who will have an opinion on it and will want to be involved, although how effective that kind of thing is I don't know."* ~ Bacton resident

"I think that design of local endeavours to secure this sand and the coast and all of that should be definitely designed by specialists who know the impact of those things or who at least have done the research, you know... I think it does need to be like coastal specialists, you know, geography specialists" ~ Bacton resident

*'Not my speciality – leave it to the expert'* (Survey response to 'Other than sandscaping, do you think any actions could be taken now to prepare for future coastal change in your village (and are you aware of anything happening in your village at present?)

In conclusion, there is a widespread desire amongst coastal policymakers working at all levels of government for community involvement in preparing for a managed realignment policy at Bacton and Walcott. Survey and interview data of Bacton

<sup>&</sup>lt;sup>17</sup> Question 17: Other than Sandscaping, do you think any actions could be taken now to prepare for future coastal change in your village (and are you aware of anything happening in your village at present?) (total responses to question= 58)

and Walcott residents does not suggest that residents feel this collective responsibility, however, at this moment in time. Furthermore, a few residents expressed doubt on the effectiveness of public consultations in coastal management. Policymaker interviewees described increasing knowledge and awareness of coastal risk, in particular, as important objectives of community engagement through Coastwise.

### 7.5 Discussion

### 7.5.1 Perceived sense of security from coastal change

Approximately three years on from the introduction of sandscaping at Bacton and Walcott, the results highlight three significant, and related, perspectives of local residents: an expectation that Bacton Gas Terminal will continue to be protected, a hope that sandscaping will be repeated (or other coastal defences reinstated), and no strong sense of urgency (in contrast to coastal policymakers), to prepare now for coastal change post-sandscaping. Although there are different views amongst local residents about how quickly redundant defences would deteriorate, and how soon a significant risk of erosion would return, a contrast can nonetheless be seen between policymakers and local residents on the need for anticipatory adaptation to managed realignment. For policymakers, sandscaping has bought time to prepare for managed realignment, whereas for residents, sandscaping has bought time to postpone it. Meanwhile, Table 7.2 and Figure 7.2 shows the immediate, dramatic change in sediment volume at Bacton Gas Terminal and the villages of Bacton and Walcott, found to be statistically significant (2015-2018 vs 2019-2022), but this placed sediment is gradually redistributing over time, and the protection from flood and erosion risk afforded by sandscaping is only temporary.

Research on coastal risk management worldwide has demonstrated a link between enhanced physical protection to flooding and reduced social risk perceptions or resilience (Logan et al., 2018; Brown et al., 2023). This has been termed the 'levee effect', where lack of exposure to flood events increases social vulnerability to future events (Di Baldassarre et al., 2015). The implementation of higher sea walls and greater flood defences in New Orleans, United States and in Japan was argued to decrease societal risk perception, and contribute to a false sense of security for extreme events (Di Baldassarre et al., 2015; Plumper et al., 2017; Boret and Gerster, 2021; Rafliana et al., 2022). Furthermore, communities protected by infrastructural engineering are less aware of flood risk (Ludy and Kondolf, 2012) and have lost an important feedback loop for learning, through exposure to flood events, which is argued to be critical for building social resilience (Colten and Sumpter, 2009; Logan et al., 2018, Plumper et al., 2017; Martinez et al., 2020). Indeed, the concept of learning appears prominently in this thesis' definition of social resilience (section 3.2.1, Maclean et al's (2016, p. 523), and in the categories of resilience explored in the literature review (Table 3.1). Other authors argue the relationship between awareness of coastal change and risk perceptions is not clear cut, and the observed maintenance of defences can be used as evidence by communities - already highly aware of coastal issues - to reinforce a desired sense of security from the risk of coastal change (Luis et al., 2015). The latter findings appear to have relevance in this context, where Bacton and Walcott residents show high awareness of issues of coastal change and rollback occurring elsewhere on the Norfolk coast. Therefore in this case study context, the impact of sandscaping on coastal risk perceptions appears to be not reduced awareness and learning, but reduced prioritisation and personal relevance of risk.

Recent research on UK public attitudes to climate change adaptation (Harcourt and Dessai, 2023; Harcourt et al., 2023) argues that whilst concern for climate change is increasing, it can be superseded by what is perceived to be more immediate societal impacts. In this case study, the immediate relevance of the war in Ukraine and the cost of living crisis, explicitly mentioned by multiple resident interviewees, similarly appears to demote residents' concerns about future coastal change. Furthermore, some residents reflect that the risk of coastal erosion will only return to significant levels outside of their lifetime, a perception similarly observed in other high-risk, undefended coastal communities in South Portugal (Costas et al., 2015; Domingues et al., 2017). This relates to longstanding research highlighting the challenge of psychological distancing in relation to climate change (Lorenzoni et al., 2007; Harcourt and Dessai, 2023) where individuals express lower concern for impacts they perceive will occur in the future (Spence et al., 2012). Previous research on public perceptions of coastal adaptation (Few et al., 2007) argue that public willingness to engage in participatory decision-making is typically more challenging in coastal contexts, where coastal risk may not be immediate. Additionally, although perceived longevity of coastal nature-based solutions (and implications for adaptation) has not been explored, psychological distancing has been observed in coastal communities in relation to the risk of coastal erosion, (Domingues et al., 2017), suggesting that like coastal defences, the deployment of nature-based solutions may inadvertently work in a similar way to reduce concern of future coastal change.

These findings highlight a potential risk of transformative coastal management approaches such as sandscaping on maladaptation, and the tensions between transformation and maladaptation, a link made in the conceptual framework. Despite the significant socio-economic benefits of sandscaping in the present day

243

and an opportunity to proactively prepare for adaptation to managed realignment, the results also highlight a risk of maladaptation if sandscaping results in less community buy-in for future adaptation. Naylor et al., (2019) and Sayers et al., (2022) argue there is currently a strong risk of maladaptation in England where 'hold the line' policies are continued in areas it can't be sustained indefinitely, and that a desire to avoid blight in coastal towns is leading to delays in adapting to managed realignment (Brown et al., 2023). This is notwithstanding many other 'lock-ins' that have been identified as entrenching 'business as usual' in coastal management in England, such as lack of national funding and clearly defined roles for coastal adaptation (Groen et al., 2022). Therefore, of particular relevance to research question 1 on geomorphological change and research question 2c on perceptions of future coastal change, sandscaping could be both transformative and maladaptive, depending on the dimension (e.g. physical or social) and timescale studied. It is likely that the findings in this case study are rooted in the time period with which they were analysed (year 3 of the sandscaping scheme in 2022, amidst the backdrop of the war in Ukraine and emerging news stories on the terminal repurposing to hydrogen). Further research is required to explore perceived responsibility and urgency of coastal adaptation later on during the sandscaping scheme, and once discussions between policymakers and local residents on adapting to managed realignment at Bacton and Walcott are formally underway, through Coastwise or otherwise.

### 7.5.2 Uncertainty

The resident and policymaker perspectives reveal sandscaping has bought additional uncertainty to coastal management at Bacton and Walcott. This includes geomorphological uncertainty, as shown in Figure 7.2 and Table 7.2, in terms of how long placed sediment will remain at the Bacton and Walcott frontage,

but also uncertainty about the future of Bacton Gas Terminal. Community engagement in Coastwise is likely to be set against a cycle of media speculation about Bacton Gas Terminal repurposing as a hydrogen or CCS site. This may form a distraction to policymaker-initiated conversations on managed realignment, and highlights the challenge of introducing innovative nature-based solutions like sandscaping in the short-term, for coastal frontages that will require managed realignment in the long-term. Whether a second sandscaping scheme is funded by terminal operators, and crucially what coastal area this may cover, has much wider national and international-scale considerations, given the presence of the nationally important Bacton Gas Terminal within this frontage, and highlights the spatial and temporal trade-offs in coastal management that has similarly been highlighted in previous research (Cooper and McKenna, 2008; Brown et al., 2023). As speculated by interviewees, this includes the role of Bacton Gas Terminal in the UK's future energy portfolio, and international geopolitical factors that have a bearing on UK energy security. This raises issues of power and justice, in how, by whom, and at what scale, coastal management decisions are made, and who decides, who causes, and who feels the effects of climate change.

Studies internationally of communities facing coastal retreat similarly emphasise the relevance of uncertainty as a barrier to adaptation (Costas et al., 2015). In New Zealand, Hanna et al., (2020) argue an 'uncertainty contagion' can be seen, whereby numerous forms of uncertainty interplay and spiral into more uncertainty for where, when, and how coastal adaptation to managed retreat (a term used interchangeably with managed realignment) can be facilitated. In this case study, the introduction of sandscaping has added further uncertainty, of both geomorphological and social dimensions, to future adaptation under a managed realignment scenario. Hanna et al., (ibid) argue manifestations of uncertainty in managed retreat ultimately all stem from political inertia on a lack of funding and

245

clarity on how adaptation can be financed and supported, a point similarly raised by local and regional policymaker interviewees in this chapter. The challenge, Hanna et al., (ibid) argue, is where uncertainty on coastal retreat builds up to such an extent that it begins to cause paralysis. This can be seen at an individual level in Fairbourne, Wales, where the media report of some residents not spending money beyond essentials, given the unknown on when and where they may need to relocate, and how much this will cost (BBC, 2022).

Therefore, the implementation of sandscaping has increased physical resilience but also added numerous forms of uncertainty, suggesting that innovative naturebased solutions like sandscaping can be both resilient and uncertain. This chapter has also highlighted the inter-connected nature of uncertainty in relation to sandscaping, with many feedbacks between social perspectives and observable geomorphological change. Although debates within literature on the link between resilience and uncertainty are explored in this thesis' literature review (section 3.2.5), the prominence of uncertainty as a theme in the empirical findings of this chapter suggest this could be represented in the conceptual framework (a finding revisited in Chapter 8).

Across the interview data, a divergence can be seen in how policymakers and local residents respond to such uncertainty. With a responsibility to manage erosion risk for the frontage, policymakers articulate it is impossible to give a date on when Bacton and Walcott will transition from 'hold the line' to 'managed realignment', and that using a trigger-based scenario in community engagement and community transition plans is more practical (such as when levels of placed sediment from sandscaping fall below a certain threshold, or when overtopping of the seawall occurs to a significant extent). The challenge for local policymakers is that local residents use a personal lifetime, rather than a trigger-based scenario, in their perception of coastal risk. Whether a sea wall is decommissioned in 10 or

30 years' time makes a sizeable difference at an individual level to a local resident's future, in terms of whether they will be directly affected by managed realignment or not. Going beyond an individual framing, and using an intergenerational justice narrative to convey the importance of a community transition plan for future generations living in the area, may be useful for policymakers to persuade local residents on the need to collectively prepare transition plans for future coastal change. In relation to this, although this study elicits views of adults living in Bacton and Walcott, further research that examines the views of children and the sandscaping scheme may be insightful, given that campaigns such as the Schools Strikes for Climate shows that younger generations show high levels of concern about future climate impacts (Hickman et al., 2021).

## 7.5.3 Community involvement in coastal decision-making

This chapter highlights potential challenges to involving local residents in coastal adaptation planning, with residents expressing little perceived responsibility in coastal management decision-making, alongside doubt in public consultations and laymen's knowledge of coastal issues as found in Chapter 6. Residents may not want to be engaged in coastal issues to the extent envisaged by policymakers; cocreating adaptation plans for the Bacton and Walcott frontage post-sandscaping. To address this, policymaker interviewees articulated the need through CTAP to improve coastal literacy (knowledge and awareness of coastal change) across Norfolk coastal communities. The conceptual framework (section 3.5) presents three key dimensions to building adaptive capacity; the policymaker interview data in this chapter appears to suggest that there will be a greater focus through Coastwise on building capacity to prepare, which is one dimension of building capacity in the conceptual framework, rather than other dimensions (a finding revisited in Chapter 8).

## 7.6 Conclusions

An intervention (sandscaping) to bring about transformative change in managing the coast (a temporary respite from immediate erosion risk to a later requirement for managed realignment) may inadvertently create additional challenges in the longer-term transition from 'hold the line'. This includes decreased risk perceptions of future coastal change, through reduced urgency and perceived personal relevance of risk, alongside increased uncertainty and complexity in managing risk for Bacton Gas Terminal, with competing national and local-level, short-term and long-term coastal policy interests. This chapter has therefore provided a case study example of additional challenges to the use of nature-based solutions in the short-term, despite its immense benefits of coastal protection. While sandscaping has provided time to prepare for managed realignment, there is a risk of maladaptation on longer timescales (i.e. 20+ years) through decreased societal risk perceptions and relevance of coastal change. There is a need for further community engagement now, so that short-term physical resilience is not achieved at the expense of building longer-term social resilience to coastal change. Widereaching, sustained community engagement on future scenario planning, that adopts an intergenerational justice framing, could be one way in which Coastwise can facilitate community buy-in for discussions on managing future coastal change.

As the last empirical chapter of the thesis, this chapter provides insights on the challenges to anticipatory adaptation to managed realignment, which will be of relevance to other coastal areas and climate impacts preparing for future climate risk. It can be seen from the empirical data that a 'winners' and 'losers' framing,

typical in distributional justice literature (section 3.3.4) is a limited framing to explain the scale-sensitive dynamics in this context. The following Discussion and synthesis chapter (Chapter 8) examines cross-cutting themes emerging from the empirical chapters (5-7), for the purpose of exploring the main empirical findings of the research, alongside a review of the thesis theoretical and methodological contributions.

# Chapter 8: Discussion and synthesis

## 8.1 Introduction

This thesis examines the social and geomorphological impacts of the Bacton-Walcott sandscaping scheme, and social perceptions of wider coastal change. It uses an environmental justice lens, to assess the implications of sandscaping on building longer-term resilience to coastal change (i.e. beyond sandscaping's 15-20 year lifespan). Coastal adaptation and a managed realignment SMP policy is explored in this thesis through two very different case study contexts, one with time (Bacton and Walcott, with a temporary respite from erosion risk due to sandscaping, Chapters 5 and 7) and one without (Happisburgh, Chapter 6). Comparing the two contexts provides insights on transformation, maladaptation and anticipatory adaptation to coastal change, which is reflected on in this chapter.

This discussion chapter firstly identifies and presents the cross-cutting themes that arise across the three empirical chapters (Chapters 5-7) in section 8.2. It also maps visually the empirical findings and key barriers onto the conceptual framework via 'zoomed in' annotations to the conceptual framework diagram (Figures 8.1-8.5), that are subsequently presented altogether in a modified conceptual framework in Figure 8.6. Further insights on the theoretical contribution of the work is explored in section 8.3, in relation to the concepts of transformation, resilience, maladaptation (8.3.1) and environmental justice (8.3.2). Methodological contributions are explored in section 8.4. Lastly, this chapter considers the limitations of the thesis and areas of future work (section 8.5).

### 8.2 Revisiting the conceptual framework in light of key empirical insights

The conceptual framework (detailed in section 3.5) draws upon the fields of resilience, environmental justice and adaptation to explore the impacts of sandscaping and implications for coastal resilience. Specifically, Figure 3.6 presents the critical influence of social, environmental, spatial and temporal components (shown as green boxes) on transformation and maladaptation. While the connections between social, environmental, spatial and temporal components are visually illustrated in Figure 3.6 through connecting lines, findings from the empirical chapters suggest greater nuance could be added to the framework on the complexity of the relationship between concepts. This section draws out some of the key findings across the empirical chapters, and uses the conceptual framework to highlight the main complexities, trade-offs and scalar conflicts emerging from the research. This is done by zooming in on specific parts of the conceptual framework to illustrate where these connections and complexities arise.

In summary, five modifications to the conceptual framework are proposed. These are;

- A feedback loop to denote the influence on social perceptions of witnessing environmental change, and the role of geomorphological uncertainty (Figure 8.1);
- Greater nuance in the denotion of social perceptions, that vary within and between groups (Figure 8.2);
- More specific representation of the complexity of the relationship between time and resilience, that both too much and too little time risk undermining social resilience (Figure 8.3);
- A negative feedback loop of the interconnected barriers to adaptation of lack of information, communication and trust (Figure 8.4);
- The significance of lack of policy funding and guidance in increasing risk of maladaptation, at the expense of transformation (Figure 8.5).

These proposed modifications are explored in turn, and subsequently mapped all together onto a modified conceptual framework in Figure 8.6.

### 8.2.1 The influence of environmental change on social perceptions

The influence of observed changes in the physical environment on public perceptions is prominent across the empirical chapters. Residents draw upon physical observations of how the Bacton-Walcott frontage is changing to inform their perceived effectiveness of the sandscaping scheme (Chapter 5) and perceptions of future coastal risk (Chapter 7). Meanwhile in Chapter 6, Happisburgh residents describe the lived experience of coastal change, witnessing the loss of several metres per year in areas of the village such as Beach Road. It is also apparent in Chapter 6 how the lived experience of witnessing erosion rates shapes residents' perceptions of adaptation options for the village. The link between social and environmental components is therefore now represented in more detail in the conceptual framework as a feedback loop (shown in Figure 8.1), rather than a linear relationship (as previously). While much research exists within socio-ecological systems literature on the interdependency between the social and natural world, as reviewed in section 3.2.2 (for example Adger (2000); Greg-Lloyd et al., 2013; Virapongse et al., (2016)), research specifically on feedbacks between environmental change and social perceptions in a flood or erosion context are typically quantitative modelling assessments (Di Baldassarre et al., 2015; Logan et al., 2018), with a paucity of qualitative research, to which this thesis contributes.

# 8.2.2 The added complexity of uncertainty

The results chapters reveal that geomorphological uncertainty in particular has a bearing on the views of local residents and policymakers on managing coastal change. In the empirical chapters, this manifests in terms of uncertainty on the lifetime of the sandscaping scheme (Chapters 5 and 7) or uncertainty in where, when, and by how much, coastal erosion will affect Happisburgh (Chapter 6). The average rates of coastal retreat (1.9 m yr<sup>-1</sup> (+/- 1.4m) at Happisburgh North and 1.1m yr<sup>-1</sup> (+/- 1.0m) at Happisburgh South, over an 850m sampling area from 2015-2022) is fast but also highly variable, e.g. great change in one year or at one location, but little discernible change the next. Uncertainty is not represented in the original conceptual framework diagram, and solely acknowledged in the description of the conceptual framework (section 3.5). Therefore, uncertainty is now also incorporated in the above proposed feedback loop between the social and natural world. Figure 8.1 illustrates this proposed modification in blue, with a feedback loop between the green 'social' and 'environmental' boxes, that also links to a new blue box denoting the influence of uncertainty.



**Figure 8.1** Proposed modification (inset, top right) to the conceptual framework (Figure 3.6) to represent the connection between social and environmental components as a feedback loop rather than a linear relationship, influenced by uncertainty associated with geomorphological change. Proposed modifications are shown in blue. Boxes in yellow refer to resilience, boxes in green refer to scalar elements of environmental justice, and boxes in red refer to perceptions of adapting to coastal change. Boxes in white refer to factors that influence adaptive capacity (bottom of diagram) or different subjects of coastal change (left of diagram).

# 8.2.3 Wide-ranging views within and between social groups

One of the most apparent findings across the empirical chapters, and relating to research questions 2a, 2b, and 2c on social perceptions, is the range of opinion within local communities on managing coastal change. As outlined in Chapter 5,

residents of Bacton and Walcott had mixed views about the sandscaping scheme, how it has impacted the village, and how effective it is. Meanwhile Chapter 6 reveals differing views of Happisburgh residents on rollback of the village car park, and whether the village should continue to campaign for coastal defences. While there appears to be greater similarity of views amongst Bacton and Walcott residents in Chapter 7 on the future of the frontage post-sandscaping, residents demonstrate different strengths of opinion on the likelihood the frontage will continue to be defended. These chapters therefore demonstrate the breadth of opinion on coastal issues. Furthermore, views on sandscaping and managing coastal change vary greatly, in that there are individuals within the same village that hold strongly opposing opinions. For example, there are residents in Chapter 5 deeply satisfied and others dissatisfied with public engagement leading up to the sandscaping scheme, and residents of Happisburgh in Chapter 6 that feel the coast should be left to erode versus those that strongly argue further intervention is needed. This complexity in the range of diverging social perspectives is not a nuance currently presented in the conceptual framework.

This is significant because building a community consensus to adaptation plans for a village under a managed realignment policy, which is a key objective of the current Coastwise project, is therefore likely to be very challenging. In Chapter 7, policymakers describe an ideal future process in Bacton and Walcott where communities develop a shared vision of the frontage, but Chapter 6 shows, with the example of Happisburgh car park, the difficulties of achieving community agreement on rollback of a community asset. At the time of writing (Spring 2024), plans to reroute and eventually relocate Happisburgh village car park have been approved by NNDC (EDP, 2024), after years of discussion and objections by some local residents. This suggests sufficient time will be needed for discussions as part of Coastwise, where views may differ on how to repurpose at-risk land. As Chapter 6 concludes, Happisburgh residents are still disputing a policy of managed realignment, nearly 20 years after the SMP change; this demonstrates the challenges of enacting a disputed policy, which currently is being discussed by residents in the context of greater clarity at a national level, including finance and support (discussed later in this section as key barriers).

A particular challenge in the variety of social perceptions evident across the empirical chapters lies in how perceptions of impact shift over time as management approaches are introduced. Chapter 5 reveals the implementation of sandscaping has shifted perceptions of impact for local residents, where some Bacton and Walcott residents have moved from fearing direct flood and coastal erosion events (such as inundation, house shaking, property damage), to fearing the inconvenience of wind-blown sand (as a result of the protection afforded by sandscaping). There has therefore been a transition, for some residents, in the scale or severity of impact that they perceive or experience. However, how impacts and risk are (subjectively) perceived by individuals, and what people perceive as tolerable, can create challenges for coastal managers if potential solutions such as sandscaping are perceived in a negative manner, despite flood or erosion protection. It can equally be seen, in the perceptions of other residents, that the negative impact of wind-blown sand is temporary, given the longer-term benefit of peace of mind by living alongside a defended frontage. Chapter 6 outlines that the mental health impacts of Happisburgh residents living alongside an undefended frontage was a prominent theme.

Chapters 6 and 7 also highlight that views vary between residents and policymakers, and amongst policymaker interviewees (most notably between national and regional/local policymakers), in terms of current challenges to coastal management policy. This is encapsulated by two additional boxes in the conceptual framework, to highlight that views and lived experiences are not

256

homogenous, and will differ across and within different groups. While the thesis literature review documents the variety of factors that shape public perceptions on adaptation (section 3.4.3, summarised in Figure 3.5), the empirical results show perspectives can be opposing even within a particular village. Two additional blue boxes highlighting this nuance are therefore proposed underneath the 'social' green box in Figure 8.2.



**Figure 8.2** Proposed modification (two additional boxes, inset, top right) to the conceptual framework to represent the tensions and differences in views within and amongst social groups included in this research (residents and policymakers). Proposed modifications are shown in blue. Boxes in yellow refer to resilience, boxes in green refer to scalar elements of environmental justice, and boxes in red refer to perceptions of adapting to coastal change. Boxes in white refer to factors that influence adaptive capacity (bottom of diagram) or different subjects of coastal change (left of diagram).

#### 8.2.4 The relationship between time and resilience

Empirical results show the importance and consequence of time, with regards to building social resilience to coastal change. Findings in Chapter 6 show the importance of sufficient time to discuss adaptation options and build trust within communities. On the other hand, Chapter 7 shows the challenge of a long leadin time, in terms of persuading Bacton and Walcott residents of the importance and urgency of anticipatory adaptation to future coastal risk post-sandscaping. There is therefore a conflict between relevance and urgency: if urgency to adapt to coastal risk is removed, does it remove relevance of adaptation to coastal change? This finding, that both sufficient and insufficient time risks undermining social resilience, and risks maladaptation, adds nuance to the relationship between time and resilience, which is now mapped in blue on the conceptual framework diagram (see Figure 8.3).

Undefended coastal areas across the UK have a declining window of opportunity for anticipatory adaptation, where accelerating climate risk is being met with national policy inertia, as highlighted by local and regional policymakers in Chapters 6 and 7 (discussed later in this section as a key barrier). This relationship between time and social resilience was found across the adult residents interviewed and surveyed in this research. However, different generations (for example young people) may be more concerned with future coastal change impacts, and perceive to a greater extent the relevance of discussing a postsandscaping frontage far in advance (i.e. before the sandscaping scheme reaches the end of its lifetime) than the participants in this study. Further research, drawing upon the perspectives of younger generations (that was outside the scope of this study), could further unpick the complexities of the relationship between time and social resilience, across generations.



**Figure 8.3** The empirical chapters contrastingly show that too much time and too little time risk undermining building social resilience, through different mechanisms. This finding adds nuance to the relationship between time and resilience, on the conceptual framework diagram. It is important to highlight the complexity in relation to time and resilience, in that there are two opposing trajectories that nonetheless lead to a similar outcome. Proposed modifications are shown in the blue boxes (inset top right). Boxes in yellow refer to resilience, boxes in green refer to scalar elements of environmental justice, and boxes in red refer to perceptions of adapting to coastal change. Boxes in white refer to factors that influence adaptive capacity (bottom of diagram) or different subjects of coastal change (left of diagram).

# 8.2.5 Key barriers to adaptation - information, communication and trust

The empirical chapters show a clear link, and significance of, a perceived lack of

information and communication, and trust, among some of the study participants.

Lack of information on coastal change, and how coastal change is communicated to local residents, are related recurring findings across all three empirical chapters. In Chapters 5 and 6, Bacton, Walcott and Happisburgh residents unanimously call for greater information about how their coast is changing, e.g. how sandscaping is performing geomorphologically (Bacton and Walcott study participants), or when and how rollback, the Coastwise project, and other initiatives relating to managed realignment, would occur (Happisburgh participants). The data suggests a perceived lack of information on coastal change contributes to doubt or distrust, among local residents, in how coastal change is managed by local government, national government, or other key stakeholders. For example, a perceived absence of prominent public communication by NNDC or the terminal operators on how sandscaping is performing may have contributed to doubt amongst residents that the scheme will last for 20 years (Chapter 5). Of significance is whether other communication sources, particularly the media, fill this perceived paucity of information. Chapters 5 and 7 report of regular media speculation, mentioned by local residents, about how much sand has been 'lost' at the sandscaped frontage, or the future of Bacton Gas Terminal. In Chapter 6, policymakers commented on media articles in January 2023 that erroneously inferred from NNDC council meeting minutes that at-risk residents would be offered a form of property compensation through CTAP.

A further complexity is that the relationship between information, communication and trust varies from one resident to another. Contrasting perspectives are apparent in each chapter. For example in Chapter 5, where some local residents and policymakers recounted numerous community engagement events about the sandscaping scheme (before the works began), that were not recognised in the reflections by other local residents. Chapters 5 and 6 show that even where communication and information on coastal change exist, an individual's values

260

and their perceived trust in local authorities has a bearing to the extent to which available information is accepted. Therefore, communication sources and pathways are also found to be a key barrier to adaptation in this context. This also raises a question of 'who' within the community is reached by communication or engagement channels. The data appears to suggest that elected members within the community, or those within established groups, such as parish councils, are more engaged in communication on sandscaping or coastal issues from local authorities. Therefore, differences of views on managing coastal change, which is notable across the empirical chapters, may relate to 'who' receives 'what' communication sources. That parish councils are engaged in discussions on coastal change with local authorities to a greater extent than residents is significant, given that Chapter 6 indicates parish councils are perceived by some residents as not necessarily representative of wider community views. This is important, given that high levels of trust between local residents and local government are required for the Coastwise project, in terms of developing community coastal adaptation plans.

To what extent does the conceptual framework represent these emergent barriers, and the complexities in how they manifest and interrelate? At the bottom of the conceptual framework diagram (depicted in white boxes) the framework considers factors affecting social resilience through three different adaptive capacities; 'capacity to prepare', 'recover', and 'adapt'. 'Trust in institutions' is a capacity to adapt factor reviewed in section 3.2.4, alongside 'knowledge' (a 'capacity to prepare' factor). However, the reviewed resilience literature did not discuss the role of knowledge in relation to how information is communicated, and therefore the relationship between trust, information and communication is not represented in the original conceptual framework (Figure 3.6). This can now be explicitly mapped onto the diagram, as Figure 8.4 illustrates. Originally the conceptual

261

framework presented all adaptive capacity factors as equally influential. In light of the empirical results, Figure 8.4 now includes the relationship between trust, information and communication onto the diagram, to highlight the importance of this feedback loop.



**Figure 8.4** The interconnected barriers (depicted as two headed arrows) of perceived or actual lack of information, communication, and trust (inset top right). Proposed modifications are shown in blue. Boxes in yellow refer to resilience, boxes in green refer to scalar elements of environmental justice, and boxes in red refer to perceptions of adapting to coastal change. Boxes in white refer to factors that influence adaptive capacity (bottom of diagram) or different subjects of coastal change (left of diagram).

8.2.6 Key barriers to transformation - lack of policy guidance and funding for

# coastal adaptation

Two key barriers that appear repeatedly across the empirical chapters include the lack of funding for coastal adaptation (beyond CTAP), and a lack of clear policy guidance to support coastal adaptation at a local level. Local and regional policymaker interviewees explicitly describe this in Chapter 7 as hindering progress in coastal adaptation, describing their role as "*chicken before the egg*" (Regional policymaker interviewee) – difficult to achieve without funding and policy clarity from central government. A tension can therefore be seen at different policy scales (i.e. at different levels of government), and that national level inaction on managing coastal risk is leading to local level inertia. To date, villages such as the 2012 Pathfinder Programme, that trial options for how the village can adapt to high erosion risk (e.g. property rollback), but there is no certainty or support beyond the lifetime of an individual project. A national policy on coastal adaptation, with clarity on funding streams and support available to at-risk communities, would ameliorate this.

The arguments posed by participants to the study reflect similar findings in the literature. Gibbs (2016) argues managed realignment has not, and is not, being outlined by central government because it carries the greatest political risk (in terms of being potentially unacceptable to the electorate). Naylor et al., (2019) put forward a similar argument, that longer-term decision-making on adaptation in the UK is compounded by what the authors (ibid) refer to as organisation risk. In other words, political pressure at a local level to maintain defences, which is reiterated by central government after a storm event, in order to maintain the reputation of a local authority. Clarke and Murphy (2023) similarly argue that transformative adaptation to coastal change (adaptation options such as relocation) are compounded by objections in the present-day, which lock-in the risk of maladaptation. This is notwithstanding other policy challenges that policymakers

and local residents highlight (see the empirical chapters) as also constraining adaptation efforts on the ground, such as different local authorities and stakeholders having responsibility for different infrastructure on the coast, with different funding streams and planning policies that delay or constrain local action. This amounts to what Hanna et al., (2020, p.1, see Chapter 7) refer to as the *"uncertainty contagion"*, where a multitude of barriers of a political dimension inhibits action on coastal realignment.

These findings highlight the link between policy barriers and the risk of maladaptation, which is not currently represented in the conceptual framework diagram. In Figure 8.5, 'Lack of funding' and 'Lack of policy guidance' are now mapped as two key barriers that may lead to increased risk of maladaptation, and hinder transformation (two concepts already present on the diagram). Because of these policy barriers, there is a risk of continual reactive adaptation, such as property rollback after storm events (as has already occurred in Happisburgh, and notably Hemsby in 2023) rather than transformative, proactive adaptation, which policymakers are united (see Chapter 7) in calling for. This national policy vacuum is not solely the case for coastal adaptation, and can be seen in adaptation to other climate risks such as extreme heat, where current infrastructure and societal resilience falls short of long-term adaptation needs (CCC, 2023). A similar trend can be seen internationally, where work by Groen et al., (2022) finds an adaptation gap for coastal risk in the UK and Germany. Across sectors such as health, transport, and infrastructure, UK climate adaptation policy is falling to consider long-term climate risks (CCC, 2023), which is exacerbated by short-term political cycles that reinforce policies of the status quo (Groen et al., 2022). Greater clarity from central government on adaptation policy and funding to alleviate uncertainty of adaptation delivery at a local level is therefore a finding recurrent in multiple climate impacts and countries.



**Figure 8.5** Lack of policy funding and/or guidance (blue box shown in inset in topright corner) are two key barriers that appear as cross-cutting themes across the empirical chapters. These barriers risk perpetuating reactive approaches to coastal adaptation ('maladaptation' box) as opposed to more proactive approaches ('transformation' box). The barriers therefore are important factors in the relationship between the concepts of transformation and maladaptation, in this case study context. Boxes in yellow refer to resilience, boxes in green refer to scalar elements of environmental justice, and boxes in red refer to perceptions of adapting to coastal change. Boxes in white refer to factors that influence adaptive capacity (bottom of diagram) or different subjects of coastal change (left of diagram).

# 8.2.7 Revised conceptual framework

In summary, the conceptual framework offers an interdisciplinary assessment of understanding changes to coastal resilience bought about by sandscaping, across social and environmental dimensions. It could be applied to other socioenvironmental systems where an environmental policy or intervention is being introduced, to understand where and how changes to resilience occurs. The conceptual framework provides a schema to examine the scale-sensitive nature of transformation and resilience, and its links to maladaptation. This section has explored the complexities and scalar tensions that appear across the empirical chapters, using the conceptual framework as an illustration to map these 'insights'. This includes feedback loops between environmental change and social perceptions, tensions in the range of social views within and across social groups, complexities in the relationship between time and resilience across scales, and the notable barriers of a lack of information, communication, and trust, and a lack of funding or policy guidance. Figure 8.6 brings together all of the proposed modifications to the conceptual framework (shown in blue) that were individually illustrated across Figures 8.1-8.5, drawing upon the empirical insights of the thesis' case study.



**Figure 8.6** Revised conceptual framework (proposed modifications in blue). Boxes in yellow refer to resilience, boxes in green refer to scalar elements of environmental justice, and boxes in red refer to perceptions of adapting to coastal change. Boxes in white refer to factors that influence adaptive capacity (bottom of diagram) or different subjects of coastal change (left of diagram).

### 8.3 Theoretical contributions of this thesis

This section considers the theoretical contributions of the thesis. Section 8.3.1 discusses findings on the key thesis concepts of transformation, resilience, and maladaptation, and section 8.3.2 explores findings on environmental justice.

### 8.3.1 Transformation, coastal resilience, and maladaptation

This research utilises the Bacton-Walcott sandscaping scheme as a case study to investigate the impact of a transformative coastal management strategy on coastal resilience. Chapters 5 and 7 highlight the transformative nature of sandscaping in terms of altering geomorphological characteristics of the coast (beach width and volume). As such, the study offers several theoretical findings about the nature of transformation and its associated impact on the resilience of a socio-ecological system. The thesis conceptual framework (section 3.5) presents sandscaping as a transformative intervention that brings a high level of resilience to socioecological systems. The theory underpinning this assertion, that transformative strategies are associated with the highest levels of resilience, is discussed in section 3.2.3. As Table 3.1 explores, transformative responses are argued to bring the highest level of resilience because they open up the greatest opportunity for deep-rooted change (Folke et al., 2010, Matyas and Pelling, 2015, Bene and Doyen, 2018; Milhorance et al., 2021). At first glance, the empirical results confirm that sandscaping has been transformative to the coastal system at Bacton and Walcott, in both social and geomorphological dimensions, because the study reveals that beach profile and beach volume are operating in a new equilibrium, Bacton and Walcott residents have not been impacted by coastal storms as they once were, and are learning to live alongside a very different coast. However,

these findings on transformation are rooted in empirical results from the period in which the study was undertaken. If the impact of sandscaping on coastal resilience is considered further, on different temporal scales, evidence of transformation is less conclusive.

Firstly, the geomorphological changes brought about by sandscaping are temporary. While academic literature on transformation does not define the length of time with which an intervention in socio-ecological systems can be considered transformative, section 3.2.3 argues there has to be a permanence to such change. Although there is uncertainty on how long sandscaping will last, it is expected to provide protection from coastal storms for up to 20 years. Therefore, the intervention is finite. Secondly, Chapter 7 reveals Bacton and Walcott residents have increased confidence in the future of their village, and a hope that sandscaping will be repeated, which appears to have influenced risk perceptions with regards to future coastal change. If a managed realignment policy after sandscaping is met with increased community apathy to coastal risk, sandscaping may inadvertently be maladaptive. This is because it may introduce additional barriers to preparing for managed realignment for the Bacton-Walcott frontage, which is the longer-term transformative scenario for this part of the coast. Revisiting Maclean et al's (2016) definition of social resilience (section 3.2.1), Chapters 6 and 7 suggest that policymaker interviewees articulate what Maclean et al., (ibid) refer to as "the adaptive and learning capacity" (Maclean et al., 2016, p.523) to prepare for future coastal change, but this has not yet transpired amongst the local community, at the time of this research. Therefore, it is the influence of sandscaping on long term coastal adaptation that determines whether it can be considered a transformative coastal management strategy, and one that requires further, longitudinal research on sandscaping towards the end of its 20 year lifetime.

To conclude, the theoretical contribution of this research on transformation, evidenced in the empirical data, is that the relationships between transformation and resilience, and transformation and maladaptation, are highly temporally and spatially scale-sensitive. The conceptual framework explicitly depicts this dependency, as explored above in section 8.2. Sandscaping has temporarily built physical resilience through the addition of placed sediment, but may inadvertently endanger building longer-term social resilience to coastal change by influencing perceptions of coastal risk. Findings also demonstrate the different timescales with which physical and social resilience are built, and therefore the tension, for transformative strategies such as sandscaping, between resilience objectives of a physical nature and a social nature. However, it is important to note that these empirical findings reflect solely the early years of the sandscaping scheme (2019-2022) during which this study was undertaken, and that further work is needed to explore the impact of sandscaping on social resilience across longer timescales (discussed further in 8.5, areas of future work).

This research has also had the opportunity to explore several theoretical debates within literature about resilience. Firstly, it is debated whether greater levels of resilience in a socio-ecological system are concurrent with greater levels of uncertainty (section 3.2.5, Leach, 2008). The empirical results highlight that increased coastal resilience at Bacton and Walcott through sandscaping has also led to increased uncertainty, in terms of how long sandscaping will last and what will succeed the scheme (Chapter 7). Therefore, this study finds a link between increased resilience and increased uncertainty, but this is likely to be context-specific to the nature-based solution in this case study. Secondly, it is debated within literature (section 3.2.3) whether greater levels of resilience in a socio-ecological system also lead to a higher risk of maladaptation. Adaptation measures set with the goal of building resilience can reduce the risk of

maladaptation (Leichenko and O'Brien, 2019) because resilient-based policy approaches focus on longer timescales (Eakin et al., 2009). While this study finds a link between a high resilience strategy and added risks of maladaptation, this finding is limited by the timescales of the research. Finally, the research has also provided insights into different capacities of resilience (capacity to prepare, recover and adapt, illustrated in Figure 3.3, section 3.2.4) and the extent to which current policy rhetoric on coastal adaptation to managed realignment (Chapters 6 and 7) focuses on these different aspects of social resilience.

# 8.3.2 Environmental justice

# Theoretical insights on environmental justice

The theoretical insights discussed in 8.3.1 above, namely that the relationship between transformation, resilience and maladaptation is highly scale-sensitive, also bring relevant findings to understandings of distributional justice and the environmental justice field. The literature on distributional justice (section 3.3.4 of the literature review) typically frames subjects of environmental harms as either 'winners' or 'losers', but this thesis adds to work in the environmental justice field, by highlighting that there are many trade-offs in the introduction of a novel naturebased solution that make such a purely binary assessment of distributional impacts less relevant. For example, a binary framing works on a cursory level, that there are villages protected (Bacton and Walcott) and unprotected (Happisburgh) by coastal defences. But on closer inspection, Chapter 5 also reveals that Bacton and Walcott residents report mixed (i.e. positive and negative) impacts at a variety of levels from the sandscaping scheme. Furthermore, Chapter 7 highlights that although Bacton and Walcott residents are protected in the present day by sandscaping, the scheme is temporary. Bacton and Walcott residents may therefore be at coastal risk in the future, as Happisburgh residents are in the

present day (Chapter 6). Therefore, applications of environmental justice – especially distributional justice - in a coastal management context, particularly regarding the introduction and implementation of novel nature-based solutions, requires further nuance than a binary classification of benefits and harms and whom they befall. This research shows that temporal and spatial aspects affect such considerations, as well as the values and features of places and people's connection with these.

This research also highlights the need for a plurality of justice perspectives in coastal decision-making, and a recognition that environmental justice principles for coastal management can be built from different philosophical foundations. This study predominantly focuses on egalitarian principles of justice, by investigating where the impacts and perspectives of sandscaping and managed realignment are different amongst local communities. However it also considers a capabilities approach to justice, (in other words, a valued life as defined by participants (Sen, 2009)). A focus on egalitarianism in this research is adopted because current coastal management policy uses a utilitarian logic in determining SMP policy for coastal areas (that therefore lends itself to be explored with an egalitarian framing). However, other theories of justice such as the capabilities approach would also be relevant for coastal management policy, that considers what individuals perceive as of value in living and working in coastal areas. The way in which different justice perspectives can be applied to support a specific point of view is particularly evident in Chapter 6, where arguments using utilitarian, egalitarian and a capabilities framing are all made by residents within and across different villages, and by policymakers. This creates opposing arguments by different stakeholders, for example in Chapter 6 where points are made on coastal funding from a national, local, and non-economic perspective. It is therefore necessary for any application of environmental justice principles to coastal policy to be mindful of these different justice perspectives. In particular, Chapter 6 highlights why coastal policy pursued through a utilitarian logic alone raises several egalitarian justice issues in the way coastal impacts are experienced across coastal communities.

## The value of an environmental justice lens in the research

This thesis adopts an environmental justice lens to examine the impacts of sandscaping, and where resilience is built within the coastal system, across different scales. While the distribution of the impacts of sandscaping and wider coastal change could have been explored without an environmental justice lens, environmental justice was crucial in this research project to explore other justice elements, not just relating to distribution, which are also pertinent. For example, Chapter 6 highlights there are multiple forms of value of a coastal environment to the community of Happisburgh, such as the restorative benefits and unique coastal heritage of the coastal frontage. Presently, these non-monetary forms of value are not accounted for in coastal decision-making. This is a recognitional injustice that could otherwise have been omitted from the analysis of this study if an environmental justice framing had not been adopted.

Furthermore, Chapter 7 reveals the material value of Bacton Gas Terminal at a national level, supplying a third of the UK's gas supply, is expected by policymakers to be the crucial factor in deciding the future management approach at Bacton and Walcott. This could potentially interact with current policies, and affect policy aspirations at a local level to enable opportunities towards managed realignment sooner. An environmental justice lens, examining issues of recognitional justice, therefore allowed the research to explore 'who' (which actors) is driving coastal management policy, for this particular stretch of coast, and why this novel nature-based solution was introduced.

In addition, the use of an environmental justice lens was particularly relevant for exploring coastal adaptation during the time of the study, and in the democratic context in the UK, given that Coastwise, and coastal adaptation policy narrative at a national level, is increasingly promoting civic participation. All three empirical chapters raise questions of participatory justice in the level of trust, information, and willingness of coastal residents to engage in conversations around adaptation they have historically perceived as overlooking local concerns (and relations with policymakers). For example, Chapter 6 demonstrates that implementation of the principles of participatory justice (the degree to which local residents can meaningfully contribute to local decision-making on coastal management) and of recognition (the degree to which local residents can frame the future narrative of their settlement) are pivotal to the shaping and evolution of coastal management strategies and trust in coastal governance. An environmental justice lens therefore allowed the research to examine pertinent barriers for coastal adaptation policy and civic engagement in the deployment of nature-based solutions going forwards.

There are several implications, however, to the environmental justice issues that this thesis identifies across Chapters 5-7. Firstly, what can practitioners and local policymakers do to alleviate these aforementioned issues and enable more transparent and inclusive engagement? Chapter 6 highlights there are limits to cocreation of adaptation plans amongst residents and policymakers, if strong preferences, such as coastal defences, will not be an option 'on the table' to start off with. Similarly, the sandscaping scheme was implemented in spite of some objections by local residents. These are complex challenges with no straightforward answers. Nonetheless, this thesis can point to a set of justice actions in coastal management, which are summarised in Table 8.1. These are drawn from the justice issues reported across Chapters 5-7. By adhering to the justice actions in Table 8.1, local policymakers and coastal practitioners can enable transparent and inclusive engagement, and use co-production to work through the boundaries of co-creation (in other words, 'red lines' such as no further coastal defences). This will require time and opportunities for communities to meaningfully engage, so that an alternative future has local support. Table 8.1 can also be used as a guide to be reinterpreted by practitioners working in other at-risk settlements, to understand the starting point for communities to engaging on coastal adaptation issues.

**Table 8.1** Justice actions in adapting to coastal change, summarising insights from Chapters 5-7. Table 8.1 extends the analysis of environmental justice issues identified in Chapters 5-7 to identify the key aspects of an environmentally just approach to coastal management for at-risk areas.

Dimension of environmental justice	Actions for coastal management
Distributional justice	There is adequate practical and financial support for residents facing property or asset loss due to flooding / erosion.
	There is wider provision of mental health support when facing flood or erosion impacts. This could be as standard (for example during coastal management consultations).
	Wellbeing – 'thriving communities' – is a central objective of coastal adaptation.
Participatory justice	Wide-reaching, sustained community engagement, with adept facilitators supported by adequate resources, to identify and accommodate differences of opinion within communities on practical dimensions of rollback, and to build trust.
	There is regular information and communication of coastal policy provided to communities in accessible, non-technical language. This reaches all members within a community, rather than representative groups.
	A community and local policymaker group is set-up to regularly discuss coastal management issues.
Recognitional justice	There is a counteracting, positive vision for the future of a settlement, that overcomes the impacts of sense of loss and place attachment.
	Pace of coastal change does not threaten the integrity, viability, or cultural heritage of a settlement.
	Community group is recognised by wider public and media as the legitimate voice

# The viability of combining environmental justice with resilience and adaptation

A research gap identified in the literature review (section 3.4.1) is the growing use of hybridised concepts such as 'just adaptation' and 'just resilience' (for example in EEA working/technical papers (Breil et al., 2021; Lager et al., 2023)), without a discussion on the compatibility of such concepts. This is in spite of longstanding criticisms of resilience as a concept, for example, that it fails to consider social trade-offs and issues of power and equity (section 3.2.6). This research has considered the compatibility and crossover between components from literatures on resilience, environmental justice and adaptation, which are reflected in the conceptual framework of this thesis. This provides an opportunity to also examine these literatures with regards to transformation and maladaptation. While transformation and maladaptation have distinct literatures and are not always examined together, some studies have brought the concepts (and respective literatures) together (for example Clarke and Murphy, 2023).

The empirical findings of this research demonstrate several fundamental differences between resilience and environmental justice, that make hybridisation challenging (i.e. to achieve the objectives of both resilience and justice as defined in sections 3.2.1 and 3.3.1, respectively). In this case study, the two concepts can be seen to prioritise different objectives: through a resilience lens, the aim of sandscaping is to have geomorphological capacity to withstand coastal storms (Chapter 5). In contrast, a justice perspective is principally aimed at alleviating environmental harms on social groups (Chapter 6). The two concepts therefore have very different units of analysis; a more focused look at certain social groups (environmental justice) versus the entire socio-ecological system (resilience). This

research also shows that considerations of resilience and environmental justice operate on different timeframes. The justice issues of coastal management raised by local residents in Chapter 6 are situated in the present-day period of the study, and past lived experience, where implications for coastal resilience, explored in Chapter 7, emerge mainly in relation to future timescales. Interviewees applied different ways of thinking about coastal management to different timescales, depending on when impacts were perceived to occur. Lastly, the two concepts are rooted in different epistemological positions. Although later work on resilience has applied it to social systems, resilience (section 3.2.1) originates from the disciplines of ecology and engineering, whereas environmental justice (section 3.3.1) is rooted in social science epistemologies.

Despite these findings, which concur with academic literature that resilience and environmental justice have different framings, focuses and epistemological roots (Eakin et al., 2009), the empirical results of this thesis nevertheless also highlight the merit of a hybridised 'just resilience' approach to coastal policy (as similarly argued by Doorn et al., 2019). A principal finding from this research is that incorporating justice principles into a 'just resilience' framing overcomes some of the limitations of resilience, because a justice focus serves to illuminate scalesensitive trade-offs among components of a system that the latter does not consider (given its emphasis on a systems perspective). An example of this is where the environmental justice lens of the research has demonstrated how residents are differently impacted and perceive environmental change in different ways (Chapter 5 with regards to sandscaping, Chapter 6 with regards to coastal change at Happisburgh). However, the adopted definition of social resilience in this study (Maclean et al., 2016; section 3.2.1) does not acknowledge potentially differential impacts beyond observable change at the system level. Therefore, incorporating environmental justice principles into the concept of resilience can

help identify where changes to resilience occur at different scales, and therefore can add further nuance to Maclean et al's (ibid) conceptualisation of social resilience.

Secondly, an integral area where justice and resilience principles align, in a coastal management context, can be seen with regards to anticipatory adaptation. The empirical chapters indicate that anticipatory rather than reactive coastal adaptation may address perceived injustices and build higher levels of resilience in a socioecological system. For example, Chapter 6 highlights it is a question of procedural justice that managed realignment is proactively planned, because it allows time to facilitate adaptation that is developed and recognised by the communities themselves, through an engagement process that is deemed meaningful and fair. Similarly, Chapters 6 and 7 find that time is needed to build community adaptive capacity for coastal change, particularly with regards to 'capacity to adapt' factors such as building trust in policymakers, overcoming barriers associated with place attachment, or accommodating wide differences of opinion within a community. Bacton and Walcott residents (Chapter 7) have greater time to prepare for adaptation options associated with a managed realignment policy, than Happisburgh residents (Chapter 6), due to the existence of the sandscaping scheme. This study therefore supports research from the resilience literature (section 3.2.3) that deliberate (i.e., anticipatory) transformation is more resilient than forced (i.e. reactive) transformation responses, and that it is particularly relevant to combine the resilience and justice literatures in contexts where anticipatory adaptation and the avoidance of maladaptation are key objectives.

Lastly, a hybridised 'just resilience' concept also strengthens the conceptualisation of resilience, because it contributes to making the risk of maladaptation more explicit. Chapters 5 and 7 highlight that sandscaping is physically transformative but may enhance the risk of maladaptation through its effect on people's

278

perceptions of managing future coastal change. Meanwhile Chapter 6 highlights low acceptance of adaptation options for managed realignment where present and past coastal policy is perceived to be unjust. The definition of social resilience in this research (Maclean et al., 2016, p.523; section 3.2.1) does not mention the risk of maladaptation, solely stating the objective as to "*maintain system function*". But the empirical results highlight that 'undesirable change' (as is often used to describe maladaptation, see section 3.2.3) in a coastal context means different things to different people, and transformative interventions such as sandscaping and managed realignment will cause differential environmental 'harms'. Therefore, the resilience literature on transformation could be more explicit about the risk of maladaptation, and an environmental justice lens is found in this research to bring this contribution, which a resilience framing may overlook.

### 8.4 Thesis methodological contributions

This research demonstrates the value of an interdisciplinary investigation of resilience, highlights coastal which the close connection between geomorphological change and social perceptions, and differences in the timing and scale of building physical and social resilience. In contrast, despite longstanding and wide-ranging research on the concept of resilience, including a more recent proliferation of resilience studies in the social sciences, the resilience literature is typically associated with a particular discipline, with notable exceptions (for example Adger, 2000; Doorn et al., 2019; Nightingale et al., 2021). This study therefore furthers the resilience field by providing a broader perspective on resilience that bridges geomorphological and social science literature, going beyond the social sciences to include a geomorphological study of the coast. Secondly, as argued in Chapter 4, the methodological approach to the research, in terms of being both interdisciplinary and iterative in the sequential nature of research instruments, allows insights between the geomorphological analysis and social data to 'cross-pollinate' and influence the focus and future direction of the study. This allows for a deeper understanding of the dynamics within the case study area, and an opportunity to further explore emerging insights.

This thesis, therefore, provides an interdisciplinary assessment of the effectiveness of sandscaping in Chapter 5, that draws upon social perspectives as well as examining geomorphological performance. This is a novel contribution to the mega-nourishment literature and a broadening of scope of evaluating meganourishment schemes: research to date on the Bacton-Walcott sandscaping scheme, and the Zandmotor scheme in the Netherlands, have either applied exclusively quantitative methods to focus on ecological or geomorphological impacts (for example De Schipper et al., 2016; Hoonhout and de Vries, 2017; Luijendijk et al., 2017; Post, 2019; Pit et al., 2020; Roest et al., 2021), or separately examined specific social impacts such as recreational changes to using the coast (Taal et al., 2016; Huisman et al., 2021), implications for coastal governance (Bontje and Slinger, 2017; Vikolainen et al., 2017), or the added social value of implementing mega-nourishment schemes (Day et al., 2023; Vreugdenhil and Slinger, 2023), that are primarily qualitative studies (see also section 2.3.2). Therefore, this thesis is the first mixed-methods research to date on sandscaping that combines both qualitative and quantitative approaches, within a single study, to investigate the impacts of mega-nourishment across multiple dimensions (social and geomorphological). Figure 8.7 illustrates this methodological contribution to the research. Using sandscaping as a case study has provided broader insights on how we can devise evaluations of mega-nourishments schemes (or indeed more broadly, such as nature-based solutions), for a more holistic assessment of their effectiveness, and ultimately broader outcomes. In other words, effectiveness of coastal nature-based solutions should not just focus on how it reduces flood

280

and erosion risk over time (left-hand box in Figure 8.7) but how unintended social impacts and misunderstandings about the strategy can be avoided, alongside how positive social impacts can be maximised (right-hand box in Figure 8.7).



**Figure 8.7** Expanded scope for evaluating the effectiveness of meganourishment, based on findings from this thesis. This is a significant finding, because it allows negative social impacts and misunderstandings about future schemes to be avoided, alongside how positive social impacts can be maximised.

# 8.5 Limitations and areas of further research

The final section of this discussion and synthesis chapter considers limitations of

the research in this thesis and areas of future research. It builds on, but also goes

beyond the purely methodological focus of limitations to the research already

outlined in section 4.5.2, which included a review of the impact of the Covid-19 pandemic on the research.

# 8.5.1 Social impacts of sandscaping and coastal resilience

The most significant limitation, as highlighted already in this chapter, is that the research can only observe the impacts of sandscaping for a snapshot in time (during a 3.5 year PhD). Consequently, this thesis can only make an assessment on impacts to coastal resilience in the first four years of the sandscaping scheme (2019-2022). This is sufficient time to study direct geomorphological changes brought about by sandscaping, but implications to social resilience may unfold much later over the scheme's 15-20 year lifespan. The empirical findings demonstrate initial implications of sandscaping on reduced perceived relevance of long-term coastal risk, but this could change as the scheme evolves, as energy policy surrounding Bacton Gas Terminal develops, and community engagement on coastal adaptation continues, through CTAP or otherwise.

It would therefore be relevant to undertake further in-depth qualitative work in the later stages of sandscaping's lifetime, or longitudinal work throughout the lifetime of the scheme, to explore how Bacton and Walcott residents perceive and experience the sandscaping scheme, and coastal change post-sandscaping, over time. This would provide additional insights on social resilience, and links between transformation and maladaptation, over the full lifetime of the scheme. In 2023, The Crown Estate commissioned a study on the added value of the sandscaping scheme, in terms of wider socioeconomic benefits, (Day et al., 2023), but to date there has been no further research on how local resident perceptions of coastal change at Bacton and Walcott are evolving over time. Whilst ACM LiDAR data for the Norfolk coast are available from 2011, no such comparable social 'baseline' of

data exists, which limits the extent to which conclusions could be drawn on how social resilience to coastal change has been impacted by sandscaping. A future study or longitudinal work, following on from this research, would provide a more direct complement.

### 8.5.2 LiDAR availability and data extent

Limitations to the use of LiDAR data mainly relate to the methodological scope achievable from the LiDAR data available, and were initially outlined in section 4.5.2 but are returned to here in light of the empirical results. The secondary LiDAR data provided by RH offers a crucial snapshot of elevation changes below the foreshore, and extends to a greater depth (approximately -15m below sea level) than the LiDAR data available from the ACM programme. However, the data series from RH began after sandscaping was completed, in contrast to the ACM programme. Consequently, while Chapter 5 illustrates important changes in sediment in the lower sub-tidal zone with RH data, this trend can only be observed from the beginning of the sandscaping scheme. A beach profile to the depths of the lower sub-tidal before 2019 would have been advantageous as a comparison, to further identify the extent of geomorphological changes.

Secondly, the ACM LiDAR data series is annual, and the data is collected at different times of the year. While this was not found to significantly affect the findings in this study (although there is monthly variation in the date of the LiDAR campaigns, they are all conducted within a winter season and therefore all share a winter beach profile), some of the differences between years in Chapters 5 and 7 is possibly partly due to date of data collection. For example, data collection in some years may have occurred soon after a depositional or erosional event, which could explain some of the variation between years, i.e. there is natural variation in

the beach not caused by sandscaping. An expansion to the LiDAR data sets, such as biannual or quarterly LiDAR campaigns for the ACM data, would offer an opportunity for further research to examine seasonal geomorphological changes due to sandscaping, that were not possible to explore with the existing annual data series. Similarly to the qualitative findings of this thesis, continued analysis of the geomorphological impact of sandscaping would be useful. Furthermore, direct investigation of sediment movement (e.g. using sediment tracers), which was outside the scope of this study, would provide insight on where placed sediment is migrating to.

# 8.5.3 Sequential research methodology, and timing of policy developments

An iterative research design, where research instruments follow in succession of one other and inform the next step of the work, crucially has allowed findings of each stage of the research process to feed into the design of successive steps of the research. A challenge to this approach is comparing social perceptions across time, where one year elapsed between conducting the survey and conducting the policymaker interviews. This is particularly relevant in this case study context, where policy developments of CTAP emerged in 2022, after the survey data. To address this, the thesis is sensitive to these differences in timing when drawing conclusions between different perspectives in the resident survey and interview data. Ultimately, any data collection will always be reflecting a particular snapshot in time. Future research could explore undertaking social research methods at the same time, if researcher capacity allows.

### 8.5.4 Additional areas of future research

Finally, the empirical chapters identify specific areas of further research. This includes exploring the format, style, tone, and content for effective public communications on the deployment of innovative nature-based solutions such as sandscaping. Future research could investigate key questions raised from this study's empirical research; why different types of information, and forms of communication, are trusted or not, and how and why information is interpreted differently. Secondly, the empirical chapters highlight that sense of loss and ecoanxiety could be relevant to explore in a separate, specific in-depth study (i.e. longitudinal or qualitative work), particularly regarding their role as potential barriers to the development of community adaptation plans. This would provide insight into how challenges relating to sense of loss and eco-anxiety specifically can be addressed in community engagement on coastal issues, so that future public engagement can be best designed to support communities and overcome such challenges. Lastly, a further research study could compare this study's findings with other sandscaping sites across the UK or other countries. This would identify cross-cutting or context specific impacts of the use of sandscaping in coastal management.

# 8.6 Concluding remarks to chapter

This chapter has explored the contributions of the thesis, by returning to the research questions and conceptual framework. This research finds that sandscaping is physically transformative, but there is a risk of maladaptation on longer timescales if the intervention contributes to reduced prioritisation and perceived relevance of the risk of coastal change, for local residents. It demonstrates the value to incorporating environmental justice into adaptation policy focussed on building resilience, where a justice lens highlights the scale-sensitive nature of when, where and why resilience is built, makes explicit the risks

of maladaptation, and where a perceived just coastal policy may enhance community willingness to adapt to coastal change and acceptance of adaptation strategies. As more and more coastal settlements this century are at risk of losing 'hold the line' status and will require adapting to managed realignment, this thesis provides relevant learning on how a just approach to building resilience to future coastal change can be facilitated. The next and final chapter of the thesis provides a short concluding summary of the key findings of the study by research question, and a summary of the key empirical insights. These are also summarised in a short section for policymakers, practitioners, academics, and other key stakeholders working in a coastal management or adaptation context in the UK or internationally.

# **Chapter 9. Conclusions**

## 9.1 Introduction

This conclusions chapter builds on from the discussion of cross-cutting themes and the theoretical and methodological contributions of the thesis explored in Chapter 8, by providing a short summary of the key conclusions of the research in light of the research questions (section 9.2). Section 9.2.1 reflects on how sandscaping has transformed the physical resilience of the coastal system (research question 1), section 9.2.2 brings together the findings on residents perceptions and experiences of coastal change (research question 2), and section 9.2.3 considers implications for incorporating environmental justice into coastal adaptation policy that focuses on building resilience (research question 3). Secondly, a concluding summary of policy-orientated empirical findings is provided in section 9.3. In addition to this standalone conclusions chapter, a short summary of the research has been produced for local residents interviewed in the study (Appendix 8), and coastal policymakers (Appendix 9).

# 9.2 Conclusions by research question

<u>9.2.1 Research question 1. Has sandscaping transformed the physical resilience</u> of the coastal system? (Is the scheme working as expected, and what are the observed coastal changes in the first four years)?

Findings for research question 1 cut across the empirical chapters which examined changes to beach profile (Chapter 5), cliff rate of retreat (Chapter 6), and beach
sediment volumes (Chapter 7). From 2019 to 2022 (the first three and a half years of the sandscaping scheme), beach profile and beach sediment budgets have changed dramatically at Bacton and Walcott (RQ1a). Analysis in Chapter 5 revealed the width of the upper (dry) part of Bacton and Walcott beach increased by approximately 50m in 2019 (when sandscaping was implemented), as the foreshore had extended by approximately 50m offshore (Figure 5.1). The village beaches were initially several metres higher in 2019 than pre-sandscaping levels (2018), and although elevation of the upper beach generally saw a decline from 2019 onwards, elevation in 2022 still remains above pre-sandscaping levels. Furthermore, 2022 saw significant amounts of sediment returning (particularly at Bacton), with an increase in sediment on the upper beach compared to 2021. This demonstrates the dynamic nature of the sandscaping scheme.

The extent of the initial placed sediment, causing these changes in beach profile from 2019 onwards, is demonstrated in the beach volume calculations in Chapter 7. The village frontages saw a 50% (Bacton) and 77% (Walcott) increase in sediment in 2019 compared to 2018. Answering research question 1c (section 7.4.2), beach volume was found to be significantly greater after sandscaping (at a 95% confidence interval) at the terminal (p=0.007), the village of Bacton (p<0.001) and the village of Walcott (p=0.004), when comparing the four years before (2015-2018) to after sandscaping (2019-2022). By comparison, beach volume at Happisburgh North (p=0.57) and Happisburgh South (p=0.89) is not significantly different (at a 95% confidence interval), when comparing the years before (2015-2018) and after (2019-2022) sandscaping. This suggests that whilst changes in beach volume cannot only be attributed to sandscaping, the significant difference in response of beaches protected by the scheme (Bacton Gas Terminal, Bacton and Walcott) compared to Happisburgh, suggests that sandscaping was a dominant factor in determining beach volume. Moreover, average annual cliff retreat (between 2015-2022) at Happisburgh North (1.9m yr<sup>-1</sup>, +/- 1.4m) and Happisburgh South (1.1m yr<sup>-1</sup>, +/-1.0m) (Chapter 6) exceeds historical averages (approximately 0.4m yr<sup>-1</sup>, 1907-1950) (Dickson et al., 2007), highlighting the vulnerability of this village to coastal change. Cliff retreat maps in Chapter 6 show the considerable variation and localised extent of coastal retreat at Happisburgh, of significance to local residents (and linking to research question 2) in terms of uncertainty in where and by how much the local coast is changing.

The dramatic changes in beach profile and beach volume at Bacton and Walcott can be observed in the Winter 2019 LiDAR data. This indicates that the geomorphological changes occurred rapidly (i.e., within the first few months of the scheme), and therefore, that physical resilience has been built near instantaneously as a result of sandscaping. It is important to note however, that changes to beach profile and beach volume through addition of sediment are temporary, as placed sediment will gradually migrate. As outlined in Chapter 2, additional sediment through sandscaping provides a buffer during storm events and reduces the likelihood of scour at the base of cliffs or sea defences. Because of increases in beach height and width at Bacton and Walcott, waves are breaking further offshore than pre-sandscaping, also reducing the risk of overtopping existing sea defences. Whilst the analysis in Chapters 5 and 7 reveal that beach elevation and beach volume are fluctuating year to year, the observed decrease in beach volume in the first few years of the scheme was not the same order of magnitude as the addition of placed sediment in 2019, i.e. a significant amount of the placed sediment remained in the immediate beach system. Furthermore, the slight increase, particularly at Bacton, in the amount of sediment on the upper (dry) beach in 2022, was observed alongside an overall trend from 2019-2022 of increased sediment in the lower subtidal zone (Figure 5.1). Crucially, the engineers of the scheme (RH, 2020; 2022a) argue the presence of sediment in

the lower subtidal zone is still having a buffering effect against coastal storms, by impacting wave attenuation. It is likely that the sediment increase observable at this depth is from migrated placed sediment, but confirming this is outside the scope of this study. It does, however, suggest that placed sediment is remaining within the coastal system, and analysis by RH (2022a) demonstrates that sediment has not being lost offshore in significant quantities in the first few years of the scheme.

In conclusion, sandscaping has rapidly built physical resilience into the coastal system at Bacton and Walcott, in terms of dramatically increased beach width and beach volume from 2019-2022. The modelled expectations of the scheme (detailed in section 2.4.3) were for the nourished beaches to gradually lose sediment, and see sediment migrate cross-shore and alongshore, over time. Overall, the beach profile (Chapter 5) and beach volume (Chapter 7) estimations concur that the sandscaping scheme is working as expected in the first few years of the scheme, given the movement of sediment observable in this research's geomorphological analysis matches modelled expectations of the scheme. As waves are now breaking further offshore (Chapter 5), and beach volume has significantly increased across the protected area (Chapter 7), it is evident that sandscaping has improved physical resilience to coastal change within the timeframe studied, as was intended.

<u>9.2.2 Research question 2. What are resident perceptions and experiences of</u> <u>coastal change, and how do these compare to coastal policymakers, and between</u> <u>villages of differing proximity to sandscaping, and differing SMP designations?</u>

This research question qualitatively focuses on views and the lived experience of coastal change in the case study area, through three sub-questions, and the three empirical chapters (5-7) each focus on one sub-question:

## RQ2a. What are the social impacts and perceptions of sandscaping among Bacton and Walcott residents? (Chapter 5)

Chapter 5 reveals that sandscaping has brought wider socioeconomic benefits for Bacton and Walcott residents than solely coastal protection, although a uniquely positive experience is not reported by all. Increased tourism, trade, recreational opportunities and property prices, alongside reduced anxiety about coastal storms and the future, were the main reported benefits. There are strong differences of opinion on the scheme and how effective it is to protect from flooding and erosion, drawing upon different observations of the first few years of sandscaping, that likely reflect deeply held views, values, and levels of trust in local coastal management. A sizeable number of residents surveyed and interviewed expressed concern about the recent drop in placed sand on Bacton and Walcott beaches, and there is more widespread doubt that the scheme will last 20 years (even amongst residents that are very satisfied with the scheme). There is some degree of scepticism of sandscaping as a coastal management strategy, which likely reflects that it is novel (in contrast to hard defences), and consequently, local residents near unanimously expressed a desire for more information about sandscaping and whether the scheme is performing as expected.

## RQ2b. For Happisburgh residents, what are the perceptions and lived experiences of past, present, and future coastal change? (Chapter 6)

Chapter 6 highlights the numerous practical, financial and emotional impacts for Happisburgh residents living alongside an unprotected, eroding coastline, illuminated in the chapter as different types of environmental justice 'harms' (distribution, procedural, recognition). Residents describe the impact of blight, low property prices, and anxiety for the future of the village, and dismay of a coastal management approach, outlined by policymaker interviewees, that rules out further coastal defences or property compensation. The wide differences of opinion within the community, some strongly felt, on the use of coastal defences and the repurposing of at-risk land within the village, alongside prevailing distrust in how coastal risk has historically been managed, represent key barriers to CTAP and building a community consensus for coastal adaptation plans that coastal policymakers are now focussed on. There is concern amongst local residents that the rate of coastal erosion is progressing at a pace too fast to adapt to coastal change in a just manner, that protects the identity and future viability of the village.

RQ2c. For Bacton and Walcott residents, how do perceptions of sandscaping affect willingness to adapt and perceptions of coastal change beyond the lifetime of the scheme? (Chapter 7)

Lastly, Chapter 7 indicates that Bacton and Walcott residents have a sense of security and increased confidence that their frontage will remain protected postsandscaping, because of the perceived importance of Bacton Gas Terminal. At the time of the research, there was media speculation that Bacton Gas Terminal would repurpose as a site for renewable energy, which residents interpret as a sign that sandscaping may be repeated in the future. There is currently little evidence amongst local residents of perceived urgency or relevance of the need to prepare for coastal change after sandscaping has gone, in strong contrast to the perspectives of coastal policymaker interviewees. This suggests there is a risk that the implementation of sandscaping could inadvertently decrease perceptions of risks about coastal change on longer timescales (i.e. in 20 years' time). For many residents, the perceived return of risks associated with coastal change are now at a point beyond their lifetime. This is despite a strong desire from coastal policymakers to initiate discussions with local residents about managed realignment and future adaptation. It suggests that the future of sandscaping and Bacton Gas Terminal could distract from longer-term coastal adaptation planning for the villages.

## <u>9.2.3 Research question 3. What are the implications of the above findings on</u> incorporating environmental justice into coastal adaptation policy that focuses on building resilience?

This section summarises the key findings explored in depth in section 8.3.2. The findings from the empirical chapters demonstrate the value of an environmental justice perspective in coastal management. Firstly, an environmental justice lens draws attention to, and makes explicit, trade-offs in coastal decision-making. For example, the variety of perspectives on perceived injustices at different temporal and scalar levels, such as which locations receive coastal protection, shown in Chapter 6 by the disproportionate impacts Happisburgh residents' experience compared to Bacton and Walcott residents (at this current time), or trade-offs in national and local interests, shown in Chapter 7 by tensions in what is prioritised regarding the future of Bacton Gas Terminal. These trade-offs are not necessarily apparent from a coastal management approach that adopts a purely resilience framing, because a resilience lens focuses on changes at the system scale, rather than individuals or social groups at finer spatial scales within it. The differences in framing, scale, and epistemological standpoints of resilience and environmental justice are discussed further in section 8.3.2. This research's case study shows an intervention (sandscaping) designed to build geomorphological resilience may inadvertently create a risk of maladaptation, by reducing local residents' concern and perceived relevance of adapting to future coastal change e.g., managed realignment (post-sandscaping). Therefore, it could be argued that it is

advantageous for both resilience and environmental justice perspectives to be applied to inform coastal management policy, to consider issues in challenging circumstances, (i.e. where it is relevant to consider multiple spatial and temporal scales for a coastal management intervention).

Secondly, it is clear from the empirical findings of this study that an environmentally just approach to coastal management is needed, so that local communities perceive the process of coastal decision-making as one of meaningful consultation and engagement. The empirical chapters, particularly Chapter 6, present local residents' perspectives that past and current coastal management has/does not sufficiently incorporate local views or keep residents updated with information on managing coastal change. Moreover, Chapters 5 and 6 reveal how perceived injustices in coastal management affect an individual's trust in local authorities and willingness to engage in coastal decision-making. Therefore, incorporating justice principles (that reflect issues of distribution, participation and recognition) into the process of coastal management decisionmaking would enable (a) to identify how residents are differently impacted, (b) to consult and/or involve residents in decision-making in ways perceived as meaningful and fair, (c) to understand how residents would like to frame the wider societal narrative of where they live, and (d) to facilitate a bottom-up community governance approach to coastal adaptation. This could be adapting via managed realignment, tailoring with policymaker's aspirations of the CTAP programme (see Chapters 6 and 7), or a different model put forward by local communities themselves. The knowledge gained through an environmentally just approach would also help coastal policymakers to target support and resources in adaptation policies within communities.

Lastly, this thesis has shown that an interdisciplinary approach provides a deeper insight into the impacts of coastal policy. Chapter 5 revealed that focussing on

social perspectives as well as geomorphological performance adds greater insight into the effectiveness of sandscaping, a novel coastal management strategy. Meanwhile, Chapter 6 showed how sense of place and built cultural heritage, which are often more complex to define in economic terms, are nonetheless pivotal in shaping public perceptions of coastal management. Chapter 6 demonstrated there is currently a tension in what is valued: for instance in how coastal frontages are valued economically in current coastal policy, and how more pluralistic and non-monetary forms of value from the environmental justice field, such as peopleplace relations, may be considered. This is a key advantage of the environmental justice approach, in highlighting the importance of non-monetary forms of value in coastal decision-making. Combining an environmental justice approach with the current resilience approach to coastal management would therefore ensure more holistic decision-making on coastal areas beyond purely economic assessments. Moreover, recognising the social value of coastal areas and the tensions between perspectives at a variety of temporal and scalar levels, that emerges from an environmental justice approach, is vital information for coastal managers for facilitating relocation of community assets to new (inland) locations, ensuring social and cultural value attached to place is preserved in this transition.

#### 9.3 Empirical insights from the research

The empirical chapters reveal several practical findings with regards to coastal management, nature-based solutions, and adaptation policy more broadly. Such insights are summarised here for policymakers, practitioners, academics, and other key stakeholders working in a coastal management or adaptation context in the UK or internationally. Insights are grouped according to scale, in terms of national (9.3.1) and local (9.3.2) recommendations.

#### 9.3.1 National-level empirical insights

#### Funding and policy guidance on coastal adaptation

- Chapter 6 argues a designated funding stream for coastal adaptation is needed at a national level, so that local authorities have the finance and capacity to support communities in at-risk areas that are not defended by coastal protection schemes. This is echoed by local policymaker interviewees themselves in Chapters 6 and 7.
- A designated funding stream for coastal adaptation needs to be matched with clear, practical policy guidance for local authorities on how adaptation for managed realignment can be supported at a local level, and to remove current ambiguity and uncertainty that is evident in local policymaker reflections in Chapters 6 and 7.

#### Building social resilience to coastal change

- A key finding from the literature review (section 3.2.4) is that the current EA FCERM strategy (2020) focuses on increasing public risk perception, knowledge and awareness of coastal change ('capacity to prepare' factors) as a means to build social resilience to coastal change. Chapters 6 and 7 similarly find that policymaker interviewees discuss the aforementioned 'capacity to prepare' factors at greater length than other adaptive capacity factors reviewed in section 3.2.4. This suggests that policymakers, and policy guidance on coastal change, are currently focused on building social resilience to coastal change through increased knowledge and understanding of risk of erosion and flooding.
- As highlighted in Chapter 8, community cohesion, sense of place and building trust between residents and policymakers, which are 'capacity to

adapt' factors (Figure 3.3, section 3.2.4) are key barriers in this case study context for facilitating community engagement in coastal adaptation. Moreover, Chapters 5-7 reveal that Bacton, Walcott and Happisburgh residents are already highly aware of coastal change. Therefore, current coastal managed policy (EA, 2020) that focuses on building social resilience of at-risk communities through increasing knowledge and awareness of coastal risk is likely to have limited benefit, and overlooks the importance of building trust with local communities, community cohesion and addressing barriers associated with place attachment.

#### Policy announcements

Chapter 6 highlights initial scepticism of CTAP amongst local residents, due to the length of time between the national announcement of the policy and communication at a local level about the programme (Coastwise). National government departments could therefore delay new policy announcements at a national level until the required due diligence is completed at a local level, so that public communication of new programmes are not followed by a period of silence, as perceived by local residents, in terms of a lack of subsequent policy information. A lack of information on new policies for managing coastal change could contribute to scepticism and distrust from a local community perspective.

Public communication on coastal adaptation

 Language on coastal adaptation, such as use of the words 'transition' in CTAP, is perceived by residents in Chapter 6 as vague and unsubstantiated. Policymakers who are referring to coastal adaptation for an undefended coastline should provide greater detail on what specifically adaptation entails, to foster engagement and reduce scepticism in coastal management. Policymakers could also open up spaces for discussion within at-risk areas about what transition means to local communities, to help inform the programme.

#### 9.3.2 Local level empirical insights

#### Deploying future sandscaping schemes or other novel nature-based solutions

- Findings from Chapter 5 indicate the need to keep local residents updated on geomorphological changes of sandscaping or other novel nature-based solutions, by publicly disseminating environmental data postimplementation in an accessible, non-technical format on a regular basis. This could be through regular updates in a newsletter format or a dedicated website (that publishes citizen science data, see recommendation below). This provides local residents with information about how their coast or local environment is changing, avoids unnecessary concern, and keeps residents engaged in local coastal management.
- Any public communication on novel environmental strategies should explicitly provide details to the general public on the strengths, weaknesses and differences of nature-based solutions in relation to pre-existing strategies, so that local communities understand how nature-based solutions may behave differently to, for example, previous hard defences. Chapter 5 reveals there can be scepticism of nature-based solutions and this might be compounded by the different (i.e. more invisible) mechanism by which it offers coastal protection. It is transient, not static, and can operate on a much larger scale than hard defences.
- Chapter 5 recommends that environmental or citizen engagement projects
   explore the role of citizen science in monitoring the evolution of naturebased solutions. Creating citizen science schemes may offer several cobenefits; providing additional data to local councils or project organisers, as

well as keeping local communities engaged and informed, and **improving the** flow of information, communication and trust between residents and local authorities.

- As argued in Chapter 5, novel coastal management, nature-based solutions or other environmental projects which are implemented within a local community should continue with a dedicated community engagement officer in the immediate years following implementation. With tight local authority budgets (or tight budgets in general for non-policy actors), ringfence funding for post-implementation community engagement in the overall budget for the scheme.
- Evaluations of novel environmental strategies (Chapter 5) should draw upon multiple perspectives, such as qualitative data on the lived experience and perceptions of affected stakeholders, rather than solely focussing on geomorphological (or environmental) performance.

Insights for coastal communities adapting to coastal change under a managed realignment policy scenario

- Chapters 5-7 find, as a cross-cutting theme, that increasing the dissemination of information on coastal change, diversifying communication channels, and increasing trust with local communities appear relevant objectives for community engagement in coastal adaptation.
- Chapter 6 argues that policies designed to support communities transitioning from retreating coastlines should include emotional support as well as the needed financial and practical support. Community engagement needs to be mindful of, and consider, how a positive scenario and future vision can be created for settlements under managed realignment. Transition plans need to offer communities a narrative, that works with sense of loss and place disruption from coastal erosion and flooding. Chapter 6 highlights the

importance of communities defining the narrative of the future of their settlement.

- It is likely that practitioners, policymakers and other stakeholders working within a coastal adaptation context will encounter wide differences of opinion, and some residents who may not want to engage in conversations about future coastal management, for a variety of reasons (as evidenced in Chapters 5-7). Facilitating community engagement beyond traditional engagement routes (i.e. beyond pre-established community groups and parish councils), could assist with reaching and offering opportunities for involvement to a greater number of local residents.
- Local communities with current defences of residual life, or protected by schemes such as sandscaping, may perceive the need for transition, or relocating assets, as outside of their lifetime (a key finding in Chapter 7). It may be challenging to involve and persuade some residents under such contexts on the need for anticipatory adaptation. An intergenerational justice framing of coastal issues could be a useful framing for practitioners, policymakers, or other stakeholders seeking to initiate these discussions, where the importance of adaptation is framed from the perspective of future members of the community.

## Appendix 1. Survey questions and accompanying information sheet

## Two years of the Bacton-Walcott Sandscaping scheme Local community experiences

Please read the accompanying information sheet before completing this survey. This survey has a total of 24 questions and will take around 20 minutes to complete.

### Section 1. Experiences of the Sandscaping scheme

This section asks about your views and experiences of the Bacton-Walcott Sandscaping scheme

1. W	/hich village do you live in?
	Bacton
	Walcott
	Happisburgh
	Other (please name the area you live
in):_	

2. Have you heard of the Bacton-Walcott Sandscaping scheme?

- 🗌 Yes
- 🛛 No

If 'yes', what do you think the **purpose** of the Sandscaping scheme is?

3. Have there been any **positive impacts** of the Sandscaping scheme to you and/or the village you live in? If yes, please specify what these positive impacts are.

For me individually:

For my village:

4. Have there been any **negative impacts** of the Sandscaping scheme to you and/or the village you live in? If yes, please specify what these negative impacts are.

For me individually:

For my village:
5. Do you think the Sandscaping scheme has impacted people within your village similarly?
☐ Yes
□ No
Don't know
Please briefly explain your answer:

6. In the next 5-10 years, how do you think the Sandscaping scheme will impact you individually? Please tick the most appropriate box.

Largely negatively	Somewhat negatively	Neither positively or negatively	Somewhat positively	Largely positively	Don't know

7. In the next 5-10 years, how do you think the Sandscaping scheme will impact your village? Please tick the most appropriate box.

Largely negatively	Somewhat negatively	Neither positively or negatively	Somewhat positively	Largely positively	Don't know

8. Please explain your answers to questions 6 and 7 in your own words here (i.e. why you think the Sandscaping scheme will impact you and your village positively or negatively in the future).

For me individually:

For my	illage:
9. Has t	ne Sandscaping scheme changed your use of the coast at Bacton and/or Walcott?
∐ No If yes, p	: applicable ease explain how:

### Section 2. Managing coastal change

This section asks for your views on your local area, and managing coastal change in your village.

10. Has the Sandscaping scheme altered your views on how coastal change (i.e. coastal erosion and flooding) could be managed in your village?

🗌 Yes

🗆 No

Please briefly explain why:

11. What do you think causes coastal change in your village? Please tick all that apply, and write the most important cause in the comment line underneath.

	Natural coastal processes
	Water runoff/poor drainage
	Sea level rise and/or more intense or frequent storms
	Hard coastal defences
	Building/developments by the sea
□ spe	Other (please cify):
Mos	st important cause:
12. tick	Who do you think is/are responsible for managing coastal change on the Norfolk coast? Please all that apply.
	National government or institutions (e.g. Environment Agency, DEFRA, CEFAS)
	Local government (e.g. county or district councils)
	Local businesses
	Local communities

Other (please

specify):\_\_\_\_\_

13. Has your view of **who** is responsible for managing coastal erosion stayed the same, or changed, **since Sandscaping was introduced** in July 2019? (If it has changed, please briefly explain why)

□ No – my view hasn't changed

Don't know

 $\Box$  Yes – (please explain why)

14. How do you	think Sandscaping	will affect co	oastal change	in your <sup>.</sup>	village over	the next	15-20
years?							

- $\Box$  Sandscaping will slow the rate of coastal change in my village
- Sandscaping will increase the rate of coastal change in my village
- Sandscaping will stop coastal change from happening in my village
- Don't know

15. Do you think Sandscaping will affect the rate of coastal erosion in any **neighbouring** villages to you? (If yes, please state how, and where).

🛛 No
------

Don't know

Yes (please state how, and where):

16. What coastal management, if any, do you think should happen in your village **in 15-20 years**, which is after the projected lifetime of the Sandscaping scheme?

17. Other than Sandscaping, do you think **any actions could be taken now** to prepare for future coastal change in your village (and are you aware of anything happening in your village at present?)

Actions that could be taken now

Actions already happening

18. Please indicate the extent to which you agree or disagree with the following statements. Please tick the appropriate option for each statement.

	Strongly	Partially	Neither agree	Partially	Strongly
	disagree	disagree	nor disagree	agree	agree
I feel attached to my village					
My job/ income is attached to my village					
I would regret having to move to another					
village					
I feel part of a community in my village					
I trust that coastal change is managed					
appropriately in my village					

19. In your opinion, does your local area have sufficient access to local services and resources (such as healthcare, education and training, utilities) **that you are able to make use of?** 

	Yes
--	-----

🗋 No

Please explain your answer

20. How long would you be able to withstand financial loss due to coastal erosion or flooding? (Financial loss could be direct, e.g. property damage, or indirect, e.g. loss of revenue)

- Less than 1 month/this would impact me straight away
- □ 1-5 months
- 6 months 1 year
- Over 1 year
- Don't know
- □ Not applicable

21. In the event of a coastal flood or coastal erosion, do you have a friend or family member who could help you, if your property or village was impacted (e.g. damaged, left without utilities, services, transport)?

☐ Yes

🗌 No

Don't know

□ Not applicable

#### Section 3. About you

This final section asks for a few anonymised details, so we know how representative our survey is.

- 22. How long have you lived in Bacton, Walcott or Happisburgh?
- $\Box$  Less than 3 years
- □ 3-10 years
- □ 11-20 years
- Over 20 years

- 23. Is this your primary residence or a second home?
- Primary residence
- Second home (if so, roughly how much of the year do you stay here?) (e.g. 6 months)

24. What age group are you in?	
18-24	55 - 64
25 – 34	65 - 74
35 - 44	75+
45 - 54	

## Thank you for your time completing this survey.

As part of this research, we will also be interviewing residents of Bacton, Walcott, and Happisburgh covering similar topics to this survey.

If you would like to be further involved in this research, please leave a contact detail (email or phone number) below and we can send you more information.

This is entirely optional. The interview would not in any way be linked to your responses to the survey, which will be reported anonymously. Your contact details will be kept strictly confidential, and absolutely not passed to anyone else.

#### Contact details

## Survey information sheet

#### What is this survey about?

You are invited to complete this survey as a resident of Bacton, Walcott or Happisburgh. This study is part of a publicly funded research project (PhD) at the University of East Anglia (UEA) on community experiences of living with the Sandscaping scheme, which was implemented at Bacton and Walcott over two years ago in summer 2019. This research contributes to ongoing work about sandscaping. The survey asks about your views and experiences of the scheme, and your views on coastal change.

#### Why should I complete this survey?

This survey is completely voluntary. By completing this survey, you are contributing to our understanding of the impacts of sandscaping on local communities. North Norfolk is the first coastal area in the UK to use sandscaping, with several other areas interested in replicating the scheme. Understanding how communities are impacted can help inform future schemes. The research team (a PhD researcher and three supervisory staff members) are strictly independent of North Norfolk District Council, but will share research findings at the end of the project. This survey does not require any in-depth knowledge of the sandscaping scheme and we would like to get the views from as many residents as possible.

#### How to complete the survey

This survey has a total of 24 questions, and is expected to take around 20 minutes to complete.

Surveys will be collected over the doorstep on Monday 31<sup>st</sup> January and Tuesday 1<sup>st</sup> February. Please leave it outside your door or in a porch, in a plastic bag and weighed down by a stone or can of tinned food to avoid it being blown away in windy weather. This survey has been distributed using PPE (gloves and hand sanitizer) to protect from Covid-19.

Alternatively, this survey is available to complete online. If you would prefer to fill out this survey online, please visit: <u>https://www.surveymonkey.co.uk/r/MMQ2RXJ</u>

Please only complete this survey once. By completing this survey, you are consenting to take part in the study. It will not be possible to withdraw your answers after completing the survey.

#### How this information will be used and stored

This survey does not ask for information that would allow respondents to be personally identified, and has kept demographic questions (e.g. age) to a minimum. All respondents will remain anonymous, and answers confidential to the research team on a password protected computer, in accordance with the Data Protection Act (2018) and UK General Data Protection Regulation (UK GDPR), and the University of East Anglia's Research and Data Management Policy. Any results reported in research articles will be anonymized. This survey has received ethics approval from the UEA DEV ethics committee.

#### Where can I find out more information?

For any queries about this research or to discuss it further, please contact the researcher:

#### Isabel Cotton, PhD Researcher, University of East Anglia, i.cotton@uea.ac.uk

If you have any queries about the survey, please contact Dr Johanna Forster (j.forster@uea.ac.uk)

#### Thank you for taking the time to complete this survey and supporting this research

## Appendix 2. Interview questions and information sheet – Bacton and Walcott residents



## Conversation on coastal change with local residents

## Information sheet

#### What is this research about?

This study is part of a NERC-ESRC funded research project (PhD) at the University of East Anglia (UEA), researching local residents views and experiences of the Bacton-Walcott sandscaping scheme, and on managing flood and erosion risk. I am inviting local residents to take part in an interview, to gather people's views on this.

#### What will the interview be about?

I would like to talk to Bacton, Walcott and Happisburgh residents about the sandscaping scheme and how future coastal risk can be best managed. In particular, I am interested in what local residents think about sandscaping as a coastal management strategy, their experience of it (if any), and their views on community involvement in future coastal management decision-making. These conversations follow-on from a public survey I ran in January and February 2022. You can take part in an interview even if you weren't invited to complete the survey.

#### Why should I take part?

It is completely voluntary to take part in an interview. By taking part, you are contributing to our understanding of the impacts of sandscaping and coastal change on local communities. Bacton and Walcott is the first coastal area in the UK to use sandscaping, with a few other areas interested in replicating the scheme. Meanwhile, this year the Environment Agency launched a new programme on how coastal change can be best facilitated for at-risk coastal areas. Understanding community perspectives is important in the design of any future coastal management projects. This research project shared summary findings of the public survey with North Norfolk District Council, and will do so again at the end of the research. I can also share a summary of research findings with you at the end of the project, if you are interested in receiving this.

#### When and how will the interview take place?

The interview can take place in-person, over the phone, or online – whatever you prefer. I anticipate it would take 30 mins - 1 hour, but it can be shorter or longer than this, depending on your availability. The interview questions on page 3 are a rough guide, and we don't have to discuss all questions. We can stop the interview at any time.

#### How this information will be used and stored

I would like to use a dictaphone so I can record and listen back to our conversation, but it is entirely your choice if you are happy for the conversation to be recorded. No one else will have access to the recording and transcript apart from myself (the lead researcher, contact details overleaf), which I will delete at the end of my PhD. Any results reported in research articles will be anonymized. Notes or recordings of the interview will be stored on a password protected computer and kept no longer than necessary, in accordance with the Data Protection Act (2018) the UK General Data Protection Regulation (UK GDPR), and the University of East Anglia's Research and Data Management Policy.

This study has received ethics approval from the UEA DEV ethics committee. By completing this interview, you are consenting to take part in the study. It will not be possible to withdraw your participation after completing the interview. The research team (a PhD researcher and three supervisory staff members) are strictly independent of North Norfolk District Council.

#### Where can I find out more information?

For any queries about this research or to discuss it further, please contact myself, the lead researcher:

#### Isabel Cotton, PhD Researcher, University of East Anglia, i.cotton@uea.ac.uk

If you have any other queries about the research, please contact Dr Johanna Forster (j.forster@uea.ac.uk)

Thank you for your time supporting this research.

#### If you would like to take part in the interview, please sign and date below:

I have read the above information sheet and I am happy to take part in the research.

Name:

Date:

## Conversation on coastal change with local residents

### Question guide

Part 1. I would like to learn more about the local communities here, and any community involvement in coastal change issues:

What do you consider to be 'your local community' (i.e. where and whom does this encompass)?

Do you feel there is a strong sense of community there?

For the village you live in, how would you like coastal management decisions about at-risk areas to be made?

Do you think local residents have any role to play in managing coastal flood or erosion risk in your village?

If yes, in your opinion, who amongst the community should be involved?

Part 2. I would like to ask about your views on the Bacton-Walcott sandscaping scheme, and managing future coastal change in your village:

What official communication have you had about sandscaping, and what, if any, information would you like to receive about the sandscaping in the future?

Do you think that sandscaping is an 'effective' strategy? How do you think it compares to previous forms of flood and erosion risk reduction such as sea walls or timber revetments at Bacton/Walcott?

In the public survey I ran in January this year, some residents reported peace of mind from the sandscaping scheme. Has the sandscaping scheme made you think more/less/or no change, about the risk of coastal erosion or flooding after sandscaping has gone?

What would you like to see happen, at the end of the modelled lifetime of sandscaping (expected to be in about 15 years' time)?

What do you think the next steps for managing the coast in your village are now, given the scheme is expected to be in place for about 15-20 years?

Have you heard of the Coastal Transition Accelerator Programme, which was awarded to North Norfolk District Council this year? If yes, what do you hope it will achieve?

Is there anything else you'd like to mention or talk about?

## Appendix 3. Interview questions and information sheet – Happisburgh residents



### Conversation on coastal change with local residents

### Information sheet

#### What is this research about?

This study is part of a NERC-ESRC funded research project (PhD) at the University of East Anglia (UEA), researching local residents views of managing coastal flood and erosion risk, and the Bacton-Walcott sandscaping scheme. I am inviting local residents to take part in an interview, to gather people's views on this.

#### What will the interview be about?

I would like to talk to Bacton, Walcott and Happisburgh residents individually about the sandscaping scheme and how future coastal risk can be best managed. For Happisburgh residents, I am interested in what residents think about managing the risk of coastal change, and their views on community involvement in any decisions on Happisburgh's coast. These conversations follow-on from a public survey I ran in January and February 2022. You can take part in an interview even if you weren't invited to complete the survey.

#### Why should I take part?

It is completely voluntary to take part in an interview. By taking part, you are contributing to our understanding of the impacts of sandscaping and coastal change on local communities. Bacton is the first coastal area in the UK to use sandscaping, with a few other areas interested in replicating the scheme. Meanwhile, this year the Environment Agency launched a new programme on how coastal change can be best facilitated for at-risk coastal areas. Understanding community perspectives is important in the design of any future coastal management projects. This research project shared summary findings of the public survey with North Norfolk District Council, and will do so again at the end of the research. I can also share a summary of research findings with you at the end of the project, if you are interested in receiving this.

#### When and how will the interview take place?

The interview can take place in-person, over the phone, or online – whatever you prefer. I am in the area and around to meet in person from 11<sup>th</sup>-18<sup>th</sup> October. I anticipate it would take 30 mins - 1 hour, but it can be shorter or longer than this, depending on your availability. The interview questions on page 3 are a rough guide, and we don't have to discuss all questions. We can stop the interview at any time.

#### How this information will be used and stored

I would like to use a dictaphone so I can record and listen back to our conversation, but it is entirely your choice if you are happy for the conversation to be recorded. No one else will have access to the recording and transcript apart from myself (the lead researcher, contact details below), which I will delete at the end of my PhD. Any results reported in research articles will be anonymized. Notes or recordings of the interview will be stored on a password protected computer and kept no longer than necessary, in accordance with the Data Protection Act (2018) the UK General Data Protection Regulation (UK GDPR), and the University of East Anglia's Research and Data Management Policy.

This study has received ethics approval from the UEA DEV ethics committee. By completing this interview, you are consenting to take part in the study. It will not be possible to withdraw your participation after completing the interview. The research team (a PhD researcher and three supervisory staff members) are strictly independent of North Norfolk District Council.

#### Where can I find out more information?

For any queries about this research or to discuss it further, please contact myself, the lead researcher:

#### Isabel Cotton, PhD Researcher, University of East Anglia, i.cotton@uea.ac.uk

If you have any other queries about the research, please contact Dr Johanna Forster (<u>i.forster@uea.ac.uk</u>)

Thank you for your time supporting this research.

#### If you would like to take part in the interview, please sign and date below:

I have read the above information sheet and I am happy to take part in the research.

Name:

Date:

## Conversation on coastal change with residents

### Question ideas

Part 1. I would like to learn more about the local communities here, and any community involvement in coastal change issues:

What do you consider to be 'your local community' (i.e. where and whom does this encompass)?

Do you feel there is a strong sense of community there?

For the village you live in, how would you like coastal management decisions about at-risk areas to be made?

Do you think local residents have any role to play in managing coastal erosion risk in your village?

If yes, in your opinion, who amongst the community should be involved?

Part 2. I would like to ask about your views on the Bacton-Walcott sandscaping scheme, and managing future coastal change in Happisburgh:

Do you think that sandscaping in Bacton is an 'effective' strategy? How do you think it compares to previous forms of flood and erosion risk reduction such as sea walls or timber revetments at Bacton/Walcott?

How do you think uncertainty around coastal risk affects decision making about coastal management for your village?

What do you think the next steps for managing the coast in your village are now?

Have you heard of the Coastal Transition Accelerator Programme, which was awarded to North Norfolk District Council this year? If yes, what do you hope it will achieve?

Is there anything else you'd like to mention or talk about?

## Appendix 4. Interview questions and information sheet - policymakers



# PhD research project (2020-2024) on the Bacton-Walcott sandscaping scheme and managing coastal change

## Information sheet

#### What will the interview be about?

I would like to talk to local and national policymakers, either involved in managing coastal change in the area of North Norfolk I am researching (Bacton, Walcott and Happisburgh), or involved in coastal adaptation or resilience policy programmes at a national level. In particular, I am interested in policymakers perspectives on the Bacton-Walcott sandscaping scheme, and the current Coastal Transition Accelerator Programme. The interview questions are a rough guide; we don't have to discuss all questions, and can talk about other topics you feel are relevant.

#### Why should I take part?

It is completely voluntary to take part in an interview. This study is part of a publicly funded research project (PhD) at the University of East Anglia (UEA), looking at the impacts of sandscaping and coastal change on the communities of Bacton, Walcott and Happisburgh in North Norfolk. North Norfolk is the first coastal area in the UK to use sandscaping, with several other areas interested in replicating the scheme. The research team (a PhD researcher and three supervisory staff members) are strictly independent of North Norfolk District Council, but will share research findings at the end of the project.

#### When and how will the interview take place?

The interview will take place online, over Microsoft Teams, at a time convenient to you. I anticipate it could last 30 mins -1 hour, but it can be shorter or longer than this, depending on your availability.

#### How this information will be used and stored

I would like to record our meeting so I can transcribe and listen back to our conversation, but it is entirely your choice if you are happy for the conversation to be recorded. No one else will have access to the recording apart from myself (the lead researcher, contact details below), which I will delete at the end of my PhD. Any results reported in research articles will be anonymized. Notes or recordings of the interview will be stored on a password protected computer and kept no longer than necessary, in accordance with the Data Protection Act (2018) and UK General Data Protection Regulation (UK GDPR), and the University of East Anglia's Research and Data Management Policy. This study has received ethics approval from the UEA DEV ethics committee.

By completing this interview, you are consenting to take part in the study. It will not be possible to withdraw your participation after completing the interview.

#### Where can I find out more information?

For any queries about this research or to discuss it further, please contact the lead researcher:

#### Isabel Cotton, PhD Researcher, University of East Anglia, i.cotton@uea.ac.uk

If you have any other queries about the research, please contact Dr Johanna Forster (j.forster@uea.ac.uk)

If you would like to take part in the interview, please sign and date below:

I have read the above information sheet and I am happy to take part in the research.

Name:

Date:

Thank you for your time supporting this research.



# PhD research project (2020-2024) on the Bacton-Walcott sandscaping scheme and managing coastal change

## Interview guide

The below interview questions are a rough guide; some may not be applicable to you, we don't have to discuss all questions, and can talk about other topics you feel are relevant.

1. Can you tell me a bit about your role, and work in coastal management at a local or national level?

### Part 1. Bacton-Walcott sandscaping scheme

2. From your work, are you aware of the Bacton-Walcott sandscaping scheme, which was implemented in North Norfolk in 2019 and is the first scheme of its kind in the UK?

If yes:

3. Whilst sandscaping is in place, what do you envisage would happen for coastal management in Bacton and Walcott, and for Bacton Gas Terminal?

4 What, from your perspective, are the next steps for managing the coast at Bacton and Walcott after sandscaping?

5. Is there a possibility the sandscaping scheme will be repeated at the end of its lifetime, and what is this dependent on?

6. Are there any plans for follow-up engagement with local residents about the sandscaping scheme and how its evolving?

7. How do you perceive the level of engagement and trust between the villages of Bacton and Walcott, and nearby Happisburgh, and local authorities on coastal change issues? Has this changed over time?

8. Do you think anything else could be done/provided to better support coastal management at Bacton and Walcott?

9. Are you aware of any plans to introduce sandscaping elsewhere in the UK?

### Part 2. Coastal Transition Accelerator Programme (CTAP)

10. Will you or your organisation be involved in the Coastal Transition Accelerator Programme, and how?

11. What are your expectations for the Coastal Transition Accelerator Programme? What do you hope it will achieve?

12. What actions will be trialled as part of the programme?

13. How, if at all, will local communities be involved?

14. Do you envisage any updates or changes to the North Norfolk District Council EN12 policy *(Relocation and Replacement of Development Affected by Coastal Erosion Risk)* from the programme?

#### Part 3. Coastal change now and into the future

15. From your perspective, what are the key challenges, currently, to facilitating coastal adaptation?

16. Do you see any major challenges for the management of the Norfolk coast in the next 2-3 decades?

17. What information is used to inform the National Coastal Erosion Risk Mapping (or in Norfolk, the Coastal Erosion Constraint Area), and how regularly is this updated?

18. Is there anything else you'd like to mention or talk about?

## Appendix 5 – Survey results

#### Q1. Which village do you live in?

Bacton (n=48) Walcott (n=29) Happisburgh (n=23)

#### Q2. Have you heard of the Bacton-Walcott Sandscaping scheme?

98% 'yes'

If 'yes', what do you think the purpose of the sandscaping scheme is?

Protection, slowing down, alleviating, halting flooding & erosion – terminal, villages, coastal road, cliffs & beaches

48% answers chose to emphasise protecting gas terminal is sole or primary purpose

# Q3. Have there been any positive impacts of the sandscaping scheme to you and/or the village you live in? and Q4. Have there been any negative impacts of the sandscaping scheme to you and/or the village you live in?

Main theme (impact)	Codes			
No flooding/ erosion	-No flooding or erosion events and associated physical impacts (property damage, inundation, house shaking, sea spray, overtopping of sea wall, cliff collapses).			
Bigger/ sandier beach	-Restorative benefits of having a wider, bigger, sandier beach.			
	-More attractive beach and coastal scenery.			
Beerestienel	-Change (reverting back) to now beach used to look in the past.			
opportunities	-recreational benefits, with calmer sea for swithining, kayaking, supportional bird watching, and new shallow areas in the sea			
opportunities	-Cleaner beach /less rubbish washed up.			
Physical access/ safety	-Permanence of access for different parts of beach at all times of			
getting on/off beach	day (e.g. including during high tide).			
	-Now possible to walk between villages along the coast.			
	<ul> <li>Improved physical access and safety in getting on/off beach for</li> </ul>			
	wheelchair users/ users with reduced mobility.			
Reassurance/ peace of	-Mental health benefits of greater reassurance, peace of mind, and			
mind	reduced anxiety about flood or erosion risk and impacts.			
Increased property value	-Perceived financial benefits from increased property value and the			
	Village being a more desirable place to live.			
	-Not incurring infancial expense from flood of erosion property			
Coast road stays open	-The main road connecting the villages of Bacton and Walcott to			
ooust roug stays open	other parts of the coast does not flood, providing reliability for			
	transport and access.			
More visitors and trade	-More visitors to beach, more trade to shops, cafes, pubs. Village			
	thrives and has financial viability.			
More people using beach	Greater numbers of tourists and visitors to the beach.			
Impact of more visitors	Impact of more visitors:			
	<ul> <li>Cars (traffic, inconsiderate parking blocking houses and roads,</li> </ul>			
	visitors not using car parking provided, nowhere to park in village)			
	- Litter (more dog waste and other litter).			
	- Antisocial behaviour, petty crime.			

Wind-blown sand	<ul> <li>-Wind-blown sand into gardens, open windows, car screens, blocking gutters/ drains, damage to outdoor equipment, and depositing around the village.</li> <li>-Required locks to be taped over and financial cost and stress to clean-up houses and gardens.</li> </ul>
Change in physical access to beach	<ul> <li>Access to beach is harder due to slopes, the loss of concrete path along the beach by sea wall makes it harder to walk for some (prams, wheelchairs). Some areas now closed.</li> <li>Change in aesthetic to beach (appears scruffier)</li> <li>Safety issues of groynes partially or fully submerged by sandscaping.</li> </ul>

Q5. Do you think the Sandscaping scheme has impacted people within your village similarly? (n=97)



**Q6.** In the next 5-10 years, how do you think the Sandscaping scheme will impact you individually?



#### Bacton and Walcott, n=73

#### Happisburgh, n=22



## Q7. In the next 5-10 years, how do you think the Sandscaping scheme will impact your village?



#### Bacton and Walcott (n=73)

#### Happisburgh (n=22)



# Q8. Please explain your answers to questions 6 and 7 in your own words here (i.e. why you think the sandscaping scheme will impact you and your village positively or negatively in the future)

37% answers show doubt or negative perceptions (n=93).

Of this:

- 10% Don't knows/ too soon to tell
- 11% general scepticism of scheme
- 10% won't last/ sand gone/ call for sand to be topped up
- 4% feeling overwhelmed (H'bro)
- 2% makes no difference to me

## Q9. Has the Sandscaping scheme changed your use of the coast at Bacton and/or Walcott? (n=96)



If 'yes' please explain why' (typical examples include):

- Visit beach more often
- Better (and worse) access- changed route for walkers/ dog walkers
- Beach busier
- Better experience on beach e.g. waves
- Safer recreation swimming
- Don't do some activities anymore e.g. fossil hunting

## Q10. Has the Sandscaping scheme altered your views on how coastal change (i.e., coastal erosion and flooding) could be managed in your village? (n=90 for closed-text question, n=74 for open-text comment box)

49% Yes 51% No

Step 1: Initial coding	Step 2: focussed coding	% answers (step 2)	<u>Step 3</u> : theoretical coding
Already taking protective measures, aware of Lincshore, already aware of erosion issue	Already aware	4%	
not aware of sandscaping previously, didn't know would work on this scale, initially sceptical, changed opinion, feel less hopeless, increased knowledge	Raised awareness/ opinion change	14%	Indirect impacts of sandscaping
Sandscaping doesn't fully protect coast, stop cliff erosion, hard defences sturdier, sandscaping not fully effective, need further defences, not implemented sucessfully, partially works	Hard defences needed/ would have been better	22%	Perceptions of managing future coastal change
need to do better, no viable solution, can't work forever, putting off the inevitable, slows but doesn't solve issue	Just delaying inevitable	8%	Temporal impacts
Sand building up, it should continue, less flooding, altered beach, feel protected, not experienced any impacts, technology working, can monitor effectively, more positive, better than hard defences	Evidence / it works	39%	Perceptions of managing future coastal change
Need large amount funding, future funding, area not valued, needs future funds to keep going, needs topping up	Funding for future	12%	Perceptions of managing future coastal change
Lots of sand gone, won't last	Insufficient evidence will work in future	16%	Temporal impacts
--	---	-----	------------------------------------
Causes less impacts elsewhere, causes more impacts elsewhere, impacts further down coast, risk of impacts elsewhere, reduces unintended impacts	Unintended impacts	5%	Spatial and temporal impacts
Don't know, other		9%	

Q11. What do you think causes coastal change in your village? (tick all that apply) (n=94)



Q12. Who do you think is/are responsible for managing coastal change on the Norfolk coast? (tick all that apply) (n=96)



Q13. Has your view of who is responsible for managing coastal erosion stayed the same, or changed, since Sandscaping was introduced in July 2019? (n=95)



## Q14. How do you think Sandscaping will affect coastal change in your village over the next 15-20 years?



#### Bacton and Walcott, n=76





Q15. Do you think Sandscaping will affect the rate of coastal erosion in any neighbouring villages to you? If yes, please explain how/where (n=89)



If yes (open-text comment box, typical responses):

- It will help neighbouring villages, albeit temporarily (22 responses)
- It will make erosion worse in neighbouring villages (12 responses)

Q16. W	hat coastal	managemei	nt, if any,	do you th	ink should	happen i	n your vil	lage in
15-20 y	ears, which	is after the	projecte	d lifetime o	of the Sand	scaping	scheme?	(n=88)

Step 1: Initial coding	Step 2: focussed coding	% answers (step 2)	<u>Step 3</u> : theoretical coding
More sandscaping - (repeat, extend, top-up)	More sandscaping	41%	no mention of building community resilience, adaptive preference, perceptions of who manages coastal change,
Hard defences - new, existing, several, in addition to sandscaping, as alternative to sandscaping, removed, expanded. Reefs, sea wall, revetments, groynes, rocks, sea defences in general, sand grass/islands, breakwaters	Hard defences	30%	As above
Being overlooked/ insignificant, second in priority to terminal, more funding/ action needed	Being overlooked/ insignificant	14%	Recognitional injustice,

			distributional injustice
Issue not appropriately addressed, sandscaping not lasted/ waste of money, at high risk, more effective/ longer-term solution needed, too late	Scepticism/ of sandscaping	13%	Perceptions of coastal change, lack of trust/belief (barriers to community resilience)
If successful, if financially viable, if terminal still operating, if monitoring (that's positive)	Contingent on	16%	Uncertainty that comes with sandscaping
anything that works, unknown, not expert, after my time	'Something'/ unknown	13%	Perceptions of who manages coastal change

# Q17. Other than Sandscaping, do you think any actions could be taken now to prepare for future coastal change in your village (and are you aware of anything happening in your village at present?) (n=76)

<u>Step 1</u> : Initial coding	Step 2: focussed coding	% answers (step 2)	<u>Step 3</u> : theoretical coding
nothing else needed/ appropriate, no, none	None	11%	building social (community) resilience, managing future coastal change, temporal impacts
not known, unsure, don't know, not qualified to answer	Not known	9%	As above
new, reinstate, expand current area, maintain hard defences	More/ maintained hard defences	41%	Managing future coastal change
lack of government support, interest, information, money	Resources/ support	9%	Distributional impacts/ recognitional injustice
revert to previous defences/ approaches, wrongly abandoned	Revert to old approach	9%	
can't stop the inevitable/ tides, too big an issue, nothing will happen	inevitability	4%	Temporal impacts

keep coastline clean, litter free, monitoring, early intervention, ring fenced budget	Maintain current state	8%	Indirect impacts of sandscaping (managing coastal change)
improve drainage, reduce runoff	Improve drainage	7%	
something needs to be done, coast at high risk	Action urgently needed/ 'something'	9%	
soft engineering, community consultation, community preparation, climate change mitigation, early intervention, '/'	Other	9%	

Q18. Please indicate the extent to which you agree or disagree with the following statements (n=94/95)

Strongly agree Partially agree Neither agree nor disagree Partially disagree Strongly disagree

I feel attached to my village



I would regret having to move to another village



I feel part of a community in my village



I trust that coastal change is managed appropriately for my village



Q19. In your opinion, does your local area have sufficient access to local services and resources that you are able to make use of? (n=87)

39% Yes 61% No Open text comment box (n=73)

Step 1: Initial coding	Step 2: focussed coding	% answers (step 2)	Step 3: theoretical coding
can't drive, limited mobility, no car, ok as have car	Ok if have private transport	12%	Differing vulnerabilities, distributional justice

irregular, unreliable, Inadequate public insufficient for young transport people, isolated village		30%	Challenges to building community resilience to coastal change (when
no doctors/ dentist in village No local healthcare		11%	access basic services)
long wait times, rising demand, impacted by covid, no gym access, quality has declined, poor hospital access	Inadequate healthcare	16%	
More than one example given	Lack of multiple services	18%	
easy access, adequate no. services, as would expect for village, local or nearby	Sufficient services	23%	Differing vulnerabilities, distributional justice
no post office, no village shop, cinema	Provisions	8%	
lack of funding, rising demand with new developments, villages need investment	More funding/investment needed locally	7%	Recognitional justice
insufficient for young people, only need healthcare from list, don't know	Other	8%	

## Q20. How long would you be able to withstand financial loss due to coastal erosion or flooding? (n=94)







#### Q22. How long have you lived in Bacton, Walcott or Happisburgh? (n=95)



#### Q23. Is this your primary residence or a second home? (n=95)

95% ticked primary residence 5% ticked second home

#### Q24. What age group are you in? (n=102\*) (\*two surveys were completed by couples)



## Appendix 6. Coding framework for interview data

Parent code	Child codes	Code description
1. Perceived effectiveness	1.1 present day	Working well / created sand bar / breaking waves / sand will come back
		Loss of sand / groyne's resurfacing / needs maintaining/topping up to be effective
	1.2 future effectiveness	Not had big storms yet / untested / unsure it will last 20 years/ sand loss rate too fast to last 20 years
	1.3 hard defences	Perceived effectiveness of hard defences / need hard defences to support sandscaping
2. Knowledge (Capacity to	2.1 Lack of knowledge	Don't know how seas/tides work / not expert /
prepare/ adapt, Oriangi et al., 2020)	2.2 Different 'types' of knowledge	2.2.1 Sources (who/what) of knowledge: Experts, laymen, long-term/ lived experience / passed down knowledge
		2.2.2 Themes/subject of knowledge: Some not wanting 'laymens' to influence coastal decisions vs others not wanting 'experts' to ignore local knowledge. Links to 'community tensions/ divisions' code.
	2.3 Changing knowledge	Increased societal understanding of the risk of climate change (typically that erosion rates are faster than originally thought). Lifetime of houses/ erosion risk at time of buying has changed/ accelerated / not knowing house risk when bought it. Uncertainty around erosion science and risk. Increased knowledge of the need to adapt.
3. Communication (Capacity to prepare/ adapt, Oriangi et al., 2020)	3.1 Communication mis- understandings	3.1.1 General sandscaping misunderstandings (Beach access and safety, sand martins, wind blown sand) weren't told wind-blown sand would happen / not told when beach access closed / affects tourism/trade
		<u>3.1.2 About beach profile</u> Confusion/ surprise about how beach profile changed post- implementation in 2019, and what was expected. Lack of information on this.
	3.2 Lack of communication about future	Wanting clarity/ certainty of the future fate of one's village / status of village / want to be kept updated about sandscaping / sows doubt / unknown / blight / wanting early planning about future / Politically toxic
	3.3 Source of communication	3.3.1 Mode of communication

		How residents would like to receive information in the future (e.g. leaflets, social media, consultation).
		3.3.2 Independence of communication Importance of independence in information about sandscaping / not biased / evidence-based
		3.3.3 During sandscaping From whom/ sources of information about sandscaping while the scheme was developed
	3.4 Good communication	Anecdotes of good communication about sandscaping scheme or coastal change to residents
4. Trust in policymakers	4.1 Motive of policymakers	Scepticism of motive/ Vested interests/ official timeframe / actions / genuine words
	4.2 Strained relations with policymakers	Scepticism of sandscaping due to mistakes of past coastal interventions / strained relations / engagement going nowhere
	4.3 Good relationships with policymakers	Anecdotes of good relations/ experiences between residents and policymakers, good levels of trust
5. Justice (Different branches of justice, Coolsaet and Neron, 2020)	5.1 Procedural justice	5.1.1 Meaningful consultation Views not listened to / decision made before consultation / consultation pointless / not being valued / not approachable / not considering local voices / also good levels of consultation / good principles of consultation
		5.1.2 Limits of consultation Consultation of limited benefit/ value, local community opinions can be unhelpful / disagreements
		5.1.3 Everybody has a right to know A right to info on how local coast is changing, as it affects livelihoods / Both a right and a choice to be involved
	5.2 Distributional justice	5.2.1 Fairness in where gets protection Differences in levels of protection along coast, differing vulnerability, Happisburgh overlooked by sandscaping / fairness in where get's protection/ funding who pays / who experiences the impact
		5.2.2 Impact of no defences Blight/ low house prices / difficulty selling / community disappearing / cost to demolish homes / can't develop business
		5.2.3 SMP policy injustice/ impact of past policies/ past defences, the right to rollback and loss of right by SMPs, can't withdraw protection without compensation / policymaker comments on financial support 5.2.4 Unjust decision-making

		Lack of consideration / unequal impacts / overlooked / not prioritised / unfair decision-making
		5.2.5 Impact of policies on coastal risk impact of sandscaping on neighbouring villages/ impact of introducing defences e.g. reefs on neighbouring villages
		5.2.6 Poor implementation of policies / rollback homes not affordable / differences in levels of compensation
	5.3 Capabilities	5.3.1 Witnessing coastal change Speed of loss / unable to process loss or change to surroundings / surreal / sense of loss of buildings gone to sea
		5.3.2 Worry for future of village and livelihoods Despair/ impact of blight/ feeling abandoned / depression / affecting the right to 'a good life' / not being worthy enough of saving / exhaustion /
	5.4 Recognition	Awareness of plight of village by others / Happisburgh no longer being protected is an unknown injustice
	5.5 Inter- generational justice	Impact of coastal risk and climate change on future generations
6. Feeling secure (Adaptive preference, Bene and Doyen, 2018) (Maladaptation, Barnet and O'Neill, 2010)		Gas terminal too important to be abandoned in future / gas terminal will switch to hydrogen / sea walls will still protect / likely sandscaping will be topped up / villages will be protected / villages seen as 'safe place to buy' / expect houses to be compensated / expect to be helped to move / need more coastal defences, not retreat / hearsay of talk of replenishing sand
7. Perceived responsibility	7.1 Community responsibility	7.1.1 Individual level actionsBeach cleans & general maintenance, flood wardens, localknowledge, property alterations/ protections, monitoring,giving opinion on consultations, financial donations,Lack of power to influence decisions around coast in village
		7.1.2 Community activism Lobbying / engagement / CCAG / raising awareness / community meetings / pressure group / social media
		7.1.3 What could we do? Not asked before / out of our control / what could we do? (links to 'power' /
		7.1.4 Willingness hard to get locals involved in other local issues besides erosion
	7.2 Policymaker responsibility	The role local residents see policymakers do/ confusion over policymakers managing some aspects of coast but not others / half-in half-out / seen as responsible

8. Bacton Gas Terminal		Shell decides what happens next for coast / has the finance and resources/ the true purpose of sandscaping
equitable resiliend	;; · · · · · · · · · · · · · · · · · ·	
9. Community	9.1 Community relations	<u>9.1.1 Sandscaping disagreements</u> Differing opinions by 'old' vs 'new' residents on coastal management / different opinions on sandscaping's effectiveness /
		<u>9.1.2 Car park disagreements</u> different opinions on relocation of car park at Happisburgh / argument / differing opinions by 'old' vs 'new' residents on coastal management
		<u>9.1.3 General village tensions on coastal change</u> disagreement / tension between individuals / 'elephant in the room' / people don't want to talk about it
		9.1.4 Differences on opinion on future options What coastal management residents perceive should happen in Happisburgh
		9.1.5 Parish councils Arguments /tensions /disagreements between residents and the parish council
	9.2 Changing community	How communities change over time (e.g. rise in second homes, impact of sandscaping/ the pandemic, village becoming more upmarket) / locals vs visitors
	9.3 Sense of community	Perceptions of who / what / strength and sense of community
10. Relations of place	10.1 Cultural heritage and cultural value	What's not valued / what's worth saving / Number of listed/ unique buildings / ancient/modern historical significance of village
	10.2 Sense of place	Beauty/ uniqueness of village / village identity village identity under threat if 'emblem's' (i.e. H'bro lighthouse) fall into sea / not wanting to move / deep-rooted / whole village under threat
11. Inevitability		Can't stop the sea /Can't stop erosion entirely/ in long run / what can you do / nothing seems to hold back sea / the physical power of the sea
12. Scalar mismatches (Capacity to prepare, recover and adapt, Orianci et al	12.1 Jurisdictional	Local governance / short-termism / Lack of political priority / Tensions / mismatches central vs local government / different priorities/ need new governance structures / need local community group/ voice set up / more localised decision making / financial resources / adequate funding / the need to mainstream adaptation
2020) (Anticipatory vs reactive	12.2 Spatial	Wanting best for one's village even if it impacts other villages / connected fate of villages through longshore drift. Issue of coastal management nationwide

adaptation, Wolf et al., 2009)	12.3 Temporal	Far into future/ Long timescales/ not going to affect them directly / sandscaping will outlive them (or hope it will) / outside their lifetime / Need to proactively plan for retreat and/or adaptation now
13. Environment/ social tensions	13.1 Climate change	The impact of climate change on <i>accelerating</i> erosion risk / why sandscaping won't last it's expected lifetime
	13.2 Need to consider environment	Impact on sand martins / coastal ecosystems / important to preserve local environment
	13.3 Perceived 'naturalness' of sandscaping	More natural/ kinder to environment / nicer beach scenery / coastscape / should experiment more with nature-based solutions
<b>14. Infrastructural intervention</b> (Deliberate vs forced adaptation, Milhorance et al., 2021)		Reinforce hard defences / introduce new defences / expand/introduce/ top-up soft engineering (top-up sandscaping in '6 feeling secure')
15. Impact of sandscaping		See survey themes – recreational / restorative / mental health / commercial benefits
16. Uncertainty		Impact of not knowing the precise timeline of coastal erosion
17. CTAP		18.1 Language style         Clarity / no political spin / difficulty / vagueness of political language / intimidating to engage in it         18.2 Perceptions on purpose         Desidents desit want retreat (vet)
		rate of erosion
<b>18. Adaptation</b> (Capacity to prepare, recover and adapt, Oriangi et al., 2020)		Conversations about adaptation between policymakers and the community / the need to adapt / adaptive capacity / perceptions of adaptation

#### Appendix 7. Ethics confirmation

Isabel Cotton (DEV - Postgraduate Researcher) From: Ethics Monitor <no-reply@ethicsreview.uea.ac.uk> Sent: 02 December 2021 11:44 To: Isabel Cotton (DEV - Postgraduate Researcher) Subject: Decision - Ethics ETH2122-0192: Miss Isabel Cotton

Study title: Just and resilient adaptation? distributional trade-offs and co-benefits of the Bacton-Walcott sandscaping scheme, Norfolk Application ID: ETH2122-0192

Dear Isabel,

Your application was considered on 2nd December 2021 by the DEV S-REC (School of International Development Research Ethics Subcommittee).

The decision is: approved.

You are therefore able to start your project subject to any other necessary approvals being given.

This approval will expire on 30th September 2024.

Please note that your project is granted ethics approval only for the length of time identified above. Any extension to a project must obtain ethics approval by the DEV S-REC (School of International Development Research Ethics Subcommittee) before continuing.

It is a requirement of this ethics approval that you should report any adverse events which occur during your project to the DEV S-REC (School of International Development Research Ethics Subcommittee) as soon as possible. An adverse event is one which was not anticipated in the research design, and which could potentially cause risk or harm to the participants or the researcher, or which reveals potential risks in the treatment under evaluation. For research involving animals, it may be the unintended death of an animal after trapping or carrying out a procedure.

Any amendments to your submitted project in terms of design, sample, data collection, focus etc. should be notified to the DEV S-REC (School of International Development Research Ethics Subcommittee) in advance to ensure ethical compliance. If the amendments are substantial a new application may be required.

Approval by the DEV S-REC (School of International Development Research Ethics Subcommittee) should not be taken as evidence that your study is compliant with the UK General Data Protection Regulation (UK GDPR) and the Data Protection Act 2018. If you need guidance on how to make your study UK GDPR compliant, please contact the UEA Data Protection Officer (dataprotection@uea.ac.uk).

I would like to wish you every success with your project.

On behalf of the DEV S-REC (School of International Development Research Ethics Subcommittee) Yours sincerely, Teresa Armijos Burneo

#### Appendix 8. PhD summary (for resident interviewees)

## Adapting to coastal change

#### A summary of PhD research on the Bacton-Walcott sandscaping scheme

Isabel Cotton, University of East Anglia (UEA) <u>i.cotton@uea.ac.uk</u>

Supervisory team: Dr Johanna Forster, Prof Irene Lorenzoni, Dr Trevor Tolhurst (UEA)



Bacton beach (Image: I Cotton)

#### 1 About the research

This PhD project investigated the social and environmental impacts of the Bacton-Walcott sandscaping scheme, and public perceptions of coastal change. The project ran from 2020-2024, and involved:

- A household survey to approximately half of residences in Bacton, Walcott and Happisburgh in January 2022 (100 completed surveys, response rate 21%).
- Interviews with Bacton (15), Walcott (7) and Happisburgh (8) residents in Autumn 2022.
- Interviews with local (2), regional (2) and national (2) coastal policymakers, in early 2023.
- Analysis of environmental data, including changes to beach profile, beach volume and cliff rate of retreat at Bacton, Walcott and Happisburgh.

## 2 Main findings of the research

### How has sandscaping changed the coast?

Analysis of environmental data indicates a notable increase in 2019 of the volume of sand at Bacton Gas Terminal and at the villages of Bacton (a 50% increase from 2018) and Walcott (a 77% increase). Since 2019, the volume of sand has been fluctuating year to year. See Figure 1.



#### What are local residents' views of sandscaping?

Results from the household survey in January 2022 found <u>nearly a third</u> of respondents from Bacton and Walcott explicitly described the scheme as working effectively. But survey responses also showed other views about coastal management and the scheme:

- More than 1 in 4 expressed a preference for hard defences (such as sea walls), or suggested hard
  defences should also be used alongside sandscaping;
- 23% expressed doubt that sandscaping will last for 15-20 years;
- 16% expressed concern about loss of sand in the first 2-3 years of the scheme and 14% wrote that
  placed sediment needs 'topping up'.

In addition to the survey results, over half of Bacton and Walcott interviewees articulated some degree of concern that sandscaping would not last for 15-20 years (its official lifetime). This includes interviewees who have strongly positive views of the scheme. Some views were also similar to those in the survey, such as observing a drop in the volume of sand from Bacton and Walcott beaches since 2019.

#### What does the environmental data show?

Analysis of environmental data of the coast at Bacton and Walcott in the first two years of sandscaping (2020-2021, Figure 2 below) revealed several changes. Reflecting local residents' survey responses, there was a drop in sand on the upper (dry) beach at Bacton and Walcott in 2019-2022 (i.e., first 2-3 years of the scheme). This coincided with an increase in sand in the lower subtidal zone (i.e., below low tide). These changes would not be observable from land.



## What are the social impacts of sandscaping?

Overall, Bacton, Walcott and Happisburgh residents' survey responses reported more positive than negative impacts of the sandscaping scheme. A third of survey answers reported no negative outcomes (see Figure 3). The survey took place in January 2022, over two years since sandscaping was implemented in July 2019.



Coast road stays open

More visitors and trade

#### Reported negative outcomes



hand side) and negative (orange shading, right hand side) impacts of the sandscaping scheme from residents in the survey. Impacts were not mutually exclusive; respondents reported positive and/or negative impacts, or combinations of these.

#### Managing coastal risk after sandscaping

- Over three-quarters of Bacton and Walcott residents interviewed described feeling a sense of security from coastal erosion and flooding by living close to Bacton Gas Terminal. The terminal was described as a nationally-important energy infrastructure, that will be protected at any cost.
- At the time the interviews were conducted (2022), many residents perceived that the terminal will only increase in importance for national energy security, due to the cost of living crisis and the war in Ukraine.
- There has been frequent media coverage speculating that Bacton Gas Terminal will be repurposed into a renewable energy plant. At the time of the research, there was no official communication that sandscaping will be repeated in the future.
- All policymaker interviewees expressed the need to prepare now for a future managed realignment policy at Bacton and Walcott.

"I said 'Look. Look out of our window, you've got two aerials there, and you've got Bacton gas site. They are not going to let that drop into the sea, because it supplies a third of the country's gas'. So, as long as that is there, this little house will be fine" ~ Bacton resident

"It (sandscaping) was about buying time, and we kind of said it's like turning the clock back. But when we get, we'll get back to a situation where we're going to need to change and adapt. So we need to use this time wisely" ~ Local policymaker

SeNSS<sup>\*</sup> ARIES





### Views on managed realignment

Across all three villages, resident interviewees and survey respondents raised several concerns about a managed realignment coastal policy:

- Managed realignment is perceived by residents as an unjust coastal policy if it is without financial support for properties that are lost to erosion;
- All Happisburgh interviewees highlighted the cultural heritage or beauty of the village, with many listed buildings that are at risk from erosion;
- 78% of completed surveys from Happisburgh explicitly called for new or reinstated coastal defences in open-text survey answers;
- Even amongst Happisburgh interviewees who perceive management realignment in the long-term as inevitable nevertheless feel coastal defences are needed in the shortterm, because the rate of erosion is happening too quickly.

"It's frustrating when you're basically told that, you're not worth it, you're not important. There's not enough show. A church founded in 9 something (century) or other isn't important. The lighthouse, the iconic things in all of North Norfolk, no...it's all going to go in the sea" ~ Happisburgh resident

Page 4

"We need to buy some time, 30-40 years, so that the mechanism by which you move back an ancient village with old buildings, listed buildings, etc can be properly formulated" ~ Happisburgh resident

## <u>3 Recommendations</u>

This research project makes several recommendations, the full list of which are summarised in the PhD thesis, and in a PhD summary to coastal policymakers.

When implementing schemes like sandscaping in the future;

- Explicitly communicate the strengths and weaknesses of the scheme regarding pre-existing initiatives or strategies and how they may operate differently.
- Keep local residents updated by publishing monitoring data about the scheme in an accessible format on a regular basis (e.g. a newsletter or a dedicated website).
- Explore the role of citizen science in the monitoring of projects.
- Provide a dedicated community engagement officer in the years following implementation. Ringfence funding for community engagement in the overall budget of the scheme.

Facilitating adaptation at a local level;

- A designated funding stream for coastal adaptation is needed at a national level, so that local
  authorities have the finance and capacity to support communities in at-risk areas that are not
  defended by coastal protection schemes.
- A funding stream needs to be matched with clear, practical policy guidance on how coastal
  adaptation can be supported at a local level.
- Policies should include provision for emotional support as well as the needed financial and
  practical support. Community engagement needs to offer a compelling future vision for settlements,
  co-created by communities, that incorporates sense of loss from erosion and flooding impacts and
  looks forward to fostering resilience.
- Engaging with local stakeholders beyond traditional routes (for example pre-established community groups), could be advantageous to widen reach, given that some stakeholders may be reluctant to do so about coastal issues.





#### **4 Further reading**

This is a PhD summary of: I Cotton, 2024. *Examining just and resilient adaptation in light of the Bacton-Walcott sandscaping scheme*. UEA PhD thesis.

The thesis is expected to be available to read from 2025 and stored at the UEA digital repository, which can be accessed here: <u>https://ueaeprints.uea.ac.uk/</u>

Recently published articles from this research can also be accessed here:

- Cotton, I., Forster, J., Lorenzoni, I. and Tolhurst, T.J., 2024. Challenges to anticipatory coastal adaptation for transformative nature-based solutions. *Global Environmental Change*, 88, Article 102893. Available at: <u>doi.org/10.1016/j.gloenvcha.2024.102893</u>
- Cotton, I., Forster, J., Lorenzoni, I. and Tolhurst, T.J. 2022. Understanding perceived effectiveness of a novel coastal management project: The case of the Bacton-Walcott sandscaping scheme, UK. Frontiers in Marine Science, 9, doi.org/10.3389/fmars.2022.1028819

This PhD research was funded by the Economic and Social Research Council and the Natural Environment Research Council through a SeNSS-ARIES studentship awarded to I. Cotton. The authors would like to thank the Bacton, Walcott and Happisburgh residents who gave their time to contribute to this project by completing the study's public survey and interviews, alongside the coastal policymakers interviewed for this research.





#### Appendix 9. PhD summary (for policymakers)

## Adapting to coastal change

PhD research on the Bacton-Walcott sandscaping scheme

#### Summary for policymakers

Isabel Cotton, University of East Anglia (UEA) i.cotton@uea.ac.uk

Supervisory team: Dr Johanna Forster, Prof Irene Lorenzoni, Dr Trevor Tolhurst, UEA



Bacton beach (Image: I Cotton)

## About the research

This PhD project based at the University of East Anglia investigated the social and environmental impacts of the Bacton-Walcott sandscaping scheme, and public perceptions of coastal change. Drawing upon data gathered from the local communities of Bacton, Walcott and Happisburgh, alongside analysis of environmental data to investigate coastal changes, the research took place from 2020-2024.



Sand bar on Bacton beach following the sandscaping scheme, Autumn 2022 (Image: I Cotton).

#### The Bacton-Walcott sandscaping scheme

The Bacton-Walcott coastal management (sandscaping) scheme was implemented in July 2019 to protect the nationally important Bacton Gas Terminal and villages of Bacton and Walcott (in North Norfolk) from coastal erosion and flooding. Sandscaping is an innovative type of sand nourishment that placed 1.8 million m<sup>3</sup> sediment on beaches that will migrate across the coast over time. It is the first time that such large scale one-off sandscaping has been implemented in the UK.

## Data collection

- January 2022 Household survey delivered to approximately half of residences in Bacton, Walcott
  and Happisburgh (100 completed surveys, response rate 21%). The survey asked residents about
  the impacts of the sandscaping scheme and views of local coastal management strategies.
- <u>Autumn 2022</u> Interviews with Bacton (15), Walcott (7) and Happisburgh (8) residents. Interviews further explored community perceptions of sandscaping and managing coastal change.
- <u>Early 2023</u> Interviews with local (2), regional (2) and national (2) policymakers working on coastal change policy across the UK. Interviews discussed coastal adaptation and the Coastal Transition Accelerator Programme, now known locally in Norfolk as Coastwise.
- 2022-2024 GIS analysis of changes to beach profile, beach volume and cliff rate of retreat.





#### How has sandscaping changed the coast?

answers reported no

negative outcomes.

new recreational

coastal storms, and

latter was viewed

wind-blown sand.

unfavourably by some

residents in terms of

Figure 1 shows the dramatic increase in 2019 of the volume of sand at Bacton Gas Terminal and at the villages of Bacton (a 50% increase from 2018) and Walcott (a 77% increase). Since 2019, the volume of sand has been fluctuating year to year.



#### What are the social impacts of the scheme?

Open-text survey questions **Reported positive outcomes Reported negative outcomes** asking households about any impacts of sandscaping No flooding/ erosion More people using beach revealed mostly positive answers. A third of survey Bigger/ sandier beach Impact of more visitors (litter, parking provision) Recreational opportunities Wind-blown sand Wider social benefits to the 5. scheme beyond flood and erosion protection include Improved physical access/ Change in physical access opportunities and ways to safety getting on/off beach to beach access a bigger beach, reduced anxiety about Reassurance/ peace of Figure 2. mind increased tourism and Reported positive (blue shading, left trade for the villages. The hand side) and negative (orange Increased property value shading, right hand side) impacts of the sandscaping scheme from residents in the survey. Impacts were not mutually Coast road stays open increased litter, traffic and exclusive; respondents reported people using the beach, positive and/or negative impacts, or More visitors and trade alongside the effects of combinations of these.

344

Page 2

#### What are local residents' views of sandscaping?

Results from a household survey in January 2022 found <u>nearly a third</u> of respondents from Bacton and Walcott explicitly described the scheme as working effectively. But survey responses also showed other views about coastal management and the scheme:

- More than 1 in 4 respondents expressed a preference for hard defences or suggested sandscaping should be reinforced with hard defences to provide greater protection from erosion;
- 23% expressed doubt that sandscaping will last for 15-20 years;
- 16% expressed concern about loss of sand in the first 2-3 years of the scheme;
- 14% felt placed sediment needs 'topping up'.

Over half of Bacton and Walcott interviewees (in Autumn 2022) articulated some degree of <u>concern that</u> <u>sandscaping would not last for 15-20 years (its official lifetime)</u>. This includes interviewees who have strongly positive views of the scheme. Perceptions include that sandscaping is as of yet untested by extreme weather, and that large amounts of sand have gone from the upper (dry) beach in first few years. All interviewees would like to see more public information about how sandscaping is performing.



Figure 3. LiDAR (elevation) data of Walcott beach 2020-2021. Environmental data shows there was a drop in sand on the upper (dry) beach at Bacton and Walcott from 2019-2022 (first 2-3 years of the scheme), as local residents described. However, this coincided with <u>an</u> increase in sand in the lower subtidal <u>zone</u> (i.e. below low tide). These changes would not be observable from land.

(Data source: Royal HaskoningDHV @ This is copyright material and may not be copied in whole or in part without the express written permission of Royal HaskoningDHV).

#### **Policy recommendations:**

#### Introducing nature-based solutions

- Explicitly communicate the strengths and weaknesses of nature-based solutions over pre-existing strategies and how they may operate differently to previous policy approaches.
- Keep local residents updated by publishing monitoring data about the scheme in an accessible format on a regular basis (e.g. a newsletter or a dedicated website). For example, this could be content similar to Figures 1 and 3 in this document.
- Explore the role of citizen science in monitoring projects, which may offer several co-benefits; additional sources of data, community engagement, and increased communication between local authorities and the public.
- Continue with a dedicated community engagement officer in the years following implementation. Ring-fence funding for community engagement in the overall budget.
- Draw upon multiple perspectives in project evaluation, such as the social impacts or lived experience of affected stakeholders and communities.

#### Managing coastal risk after sandscaping

- Over three-quarters of Bacton and Walcott residents interviewed described feeling a sense of security from coastal erosion and flooding by living close to Bacton Gas Terminal. The terminal was described as a nationally-important energy infrastructure, that will be protected at any cost.
- To date, there has been no official communication that the sandscaping scheme will be repeated in the future. There has, however, been frequent media coverage speculating that Bacton Gas Terminal will be repurposed into a renewable energy plant.
- At the time the research was conducted (2022), many residents perceive that the terminal will only increase in importance for national energy security, due to the cost of living crisis and the war in Ukraine.

"I said 'Look. Look out of our window, you've got two aerials there, and you've got Bacton gas site. They are not going to let that drop into the sea, because it supplies a third of the country's gas'. So, as long as that is there, this little house will be fine" ~ Bacton resident

"It (sandscaping) was about buying time, and we kind of said it's like turning the clock back. But when we get, we'll get back to a situation where we're going to need to change and adapt. So we need to use this time wisely" ~ Local policymaker

"People were (saying) 'oh ok, the sand's here, the sea, it's going to last 20 years'...'well, that will see me out. I'd be happy with that" ~ Walcott resident

"Even if it wasn't replenished...then it's still probably another 10-15 years before the seawall would be breached" ~ Walcott resident All policymaker interviewees expressed the need to prepare now for a future managed realignment policy at Bacton and Walcott. However, interviews with Bacton and Walcott residents revealed:

A general perception amongst interviewees (across age groups) that any potential property rollback at Bacton and Walcott will be outside their lifetime, regardless of whether sandscaping is repeated;
A common view that erosion risk would not return immediately postsandscaping, with some residual life in pre-existing hard defences such as the sea wall. This was a view expressed both by interviewees inland and on the immediate coastal frontage.

#### Policy recommendations: Community engagement on coastal adaptation

- For policymakers, sandscaping has bought time to prepare for managed realignment, whereas for
  residents, sandscaping has bought time to postpone it.
- An intergenerational justice framing (i.e. the wellbeing of future generations of a community), could be useful for practitioners working in contexts where communities perceive the risk of coastal change as outside their lifetime.
- Facilitating engagement beyond traditional routes (for example pre-established community groups), could be advantageous to widen reach, given that some stakeholders may be reluctant to engage.

#### Views on managed realignment

Across all three villages, resident interviewees and survey respondents raised several concerns about a managed realignment coastal policy:

- Managed realignment is perceived by residents as an unjust coastal policy if it is without **financial support** for properties that are lost to erosion;
- All Happisburgh interviewees highlighted the cultural heritage or beauty of the village, with many listed buildings that are at risk from erosion;
- 78% of completed surveys from Happisburgh explicitly called for new or reinstated coastal defences in open-text survey answers;
- Even amongst Happisburgh interviewees who perceive management realignment in the long-term as inevitable nevertheless feel coastal defences are needed in the shortterm, because the rate of coastal erosion is happening too quickly.

"It's frustrating when you're basically told that, you're not worth it, you're not important. There's not enough show. A church founded in 9 something (century) or other isn't important. The lighthouse, the iconic things in all of North Norfolk, no...it's all going to go in the sea"~ Happisburgh resident

"We need to buy some time, 30-40 years, so that the mechanism by which you move back an ancient village with old buildings, listed buildings, etc can be properly formulated" ~ Happisburgh resident

This is a PhD summary of: I Cotton, 2024. *Examining just and resilient adaptation in light of the Bacton-Walcott sandscaping scheme*. This is a UEA PhD thesis, expected to be available from 2025 at the UEA digital repository (<u>https://ueaeprints.uea.ac.uk/</u>)

Further reading:

- Cotton, I., Forster, J., Lorenzoni, I. and Tolhurst, T.J., 2024. Challenges to anticipatory coastal adaptation for transformative nature-based solutions. *Global Environmental Change*, 88, Article 102893. Available at: <u>doi.org/10.1016/j.gloenvcha.2024.102893</u>
- Cotton, I., Forster, J., Lorenzoni, I. and Tolhurst, T.J. 2022. Understanding perceived effectiveness of a novel coastal management project: The case of the Bacton-Walcott sandscaping scheme, UK. Frontiers in Marine Science, 9, doi.org/10.3389/fmars.2022.1028819

This PhD research was funded by the Economic and Social Research Council and the Natural Environment Research Council through a SeNSS-ARIES studentship awarded to I Cotton. The authors would like to thank the Bacton, Walcott and Happisburgh residents who gave their time to contribute to this project by completing the study's public survey and interviews, alongside the coastal policymakers interviewed for this research.

### Policy recommendations:

#### Facilitating managed realignment at a local level

- A designated funding stream for coastal adaptation is needed at a national level, so that local
  authorities have the finance and capacity to support communities in at-risk areas that are not
  defended by coastal protection schemes.
- A funding stream needs to be matched with clear, practical policy guidance on how coastal adaptation can be supported at a local level.
- Policies should include provision for emotional support as well as the needed financial and
  practical support. Community engagement needs to offer a compelling future vision for settlements,
  co-created by communities, that works with sense of loss from erosion and flooding impacts, and
  looks forward to fostering resilience.

#### References

Adger, W.N., 2000. Social and Ecological Resilience: Are they related? *Progress in Human Geography*, **24**(3), pp. 347-364. doi.org/10.1191/030913200701540465

Adger, W.N., Hughes, T.P., Folke, C., Carpenter, S. and Rockstrom, J., 2005a. Social-Ecological Resilience to Coastal Disasters. *Science*, **309**(5737), pp. 1036-1039. <u>doi.org/10.1126/science.1112122</u>

Adger, W.N., Arnell, N.W. and Tompkins, E.L., 2005b. Successful adaptation to climate change across scales. *Global Environmental Change*, **15**(2), pp. 77-86. doi.org/10.1016/j.gloenvcha.2004.12.005

Adger, W.N., Dessai, S., Goulden, M., Hulme, M., Lorenzoni, I., Nelson, D., Naess, L., Wolf, J. and Wreford A., 2009. Are there social limits to adaptation to climate change?

Climatic Change, 93(2009), pp. 335-354. doi.org/10.1007/s10584-008-9520-z

Adger, W.N., Nicholson-Cole, S. and Arnold, D.G., 2011. Ethical dimensions of adapting to climate change-imposed risks. In Arnold, D.G., ed. *The Ethics of Global Climate Change*. Cambridge: Cambridge University Press, pp.255-271.

Adger, W.N., Quinn, Tara., Lorenzoni, I. and Murphy, C., 2016. Sharing the Pain: Perceptions of Fairness Affect Private and Public Response to Hazards. *Annals of the American Association of Geographers*. **106**(5), pp.1079-1096. doi.org/10.1080/24694452.2016.1182005

Adhikari, H., Heiskanen, J., Maeda, E.E. and Pellikka, P.K.E., 2016. The effect of topographic normalization on fractional tree cover mapping in tropical mountains: An assessment based on seasonal Landsat time series. *International Journal of Applied Earth Observation and Geoinformation*. **52**(October 2016), pp. 20-31. doi.org/10.1016/j.jag.2016.05.008

AECOM, 2012a. *Kelling to Lowestoft Ness Shoreline Management Plan.* Cheshire: AECOM.

AECOM, 2012b. Kelling to Lowestoft Ness Shoreline Management Plan – Appendix C: Baseline Process Understanding. Cheshire: AECOM.

Alizadeh, H. and Sharifi, A., 2021. Analysis of the state of social resilience among different socio-demographic groups during the COVID- 19 pandemic. *International Journal of Disaster Risk Reduction*, **64**(October 2021), Article 102514. <u>doi.org/10.1016/j.ijdrr.2021.102514</u>

Anderson C. C. and Renaud F. G., 2021. A review of public acceptance of nature-based solutions: The 'why', 'when', and 'how' of success for disaster risk reduction measures. *Ambio.* **50**(2021), pp.1552–1573. <u>doi.org/10.1007/s13280-021-01502-4</u>

Anglian Coastal Monitoring (ACM) Programme, 2022. *Programme background*. [Online]. Available at: <u>https://coastalmonitoring.org/anglian/</u> [Accessed 19 August 2022].

Anguelovski, I., Shi, L., Chu, E., Gallagher, D., Goh, K., Lamb, Z., Reeve, K. and Teichner, H., 2016. Equity Impacts of Urban Land Use Planning for Climate Change Adaptation: Critical Perspectives from the Global North and South. *Journal of Planning Education and Research*, **36**(3), pp.333-348. <u>doi.org/10.1177/0739456X16645166</u>

Anguelovski, I. and Corbera, E., 2023. Integrating justice in Nature-Based Solutions to avoid nature-enabled dispossession. *Ambio*, **52**(2023), pp.45-53. doi.org/10.1007/s13280-022-01771-7

Arias, J.P., Bronfman, N.C., Cisternas, P.C. and Repetto, P.B., 2017. Hazard proximity and risk perception of tsunamis in coastal cities: Are people able to identify their risk? *PLOS One*, **12**(10), Article e0186455. doi.org/10.1371/journal.pone.0186455.

Ariza, E., Lindeman, K.C., Mozumder, P. and Suman, D.O., 2014. Beach management in Florida: Assessing stakeholder perceptions on governance. *Ocean & Coastal Management*, **96**(August 2014), pp.82-93. doi.org/10.1016/j.ocecoaman.2014.04.033

Arnall, A. and Hilson, C., 2023. Climate change imaginaries: representing and contesting sea level rise in Fairbourne, North Wales. *Political Geography*, **102**(April 2023), Article 102839, <u>doi.org/10.1016/j.polgeo.2023.102839</u>

Arthur, J. and Shaw, W.H., 1978. *Justice and Economic Distribution*. Englewood Cliffs, NJ: Prentice-Hall.

Atteridge, A. and Remling, E., 2018. Is adaptation reducing vulnerability or redistributing it? *WIREs Climate Change*, **9**(1), Article e500. <u>doi.org/10.1002/wcc.500</u>

Bacton Energy Hub Special Interest Groups, 2022. *Bacton Energy Hub Business Opportunity Report prepared by the Bacton Energy Hub Special Interest Groups*. London: North Sea Transition Authority.

Bahadur, A.V., Ibrahim, M. and Tanner, T., 2011. *The resilience renaissance? Unpacking of resilience for tackling climate change and disasters. Strengthening Climate Resilience Discussion Paper 1.* Brighton: Institute of Development Studies.

Bahadur, A.V., Ibrahim, M. and Tanner, T., 2013. Characterising resilience: unpacking the concept for tackling climate change and development. *Climate and Development*, **5**(1), pp.55-65. <u>doi.org/10.1080/17565529.2012.762334</u>

Ballinger, R. C. and Dodds, W., 2020. Shoreline management plans in England and Wales: A scientific and transparent process? *Marine Policy*, **111**(January 2020), Article 102689. <u>doi.org/10.1016/j.marpol.2017.03.009</u>

Barnett, J. and O'Neil, S., 2010. Maladaptation. *Global Environmental Change*, **20**(2), pp.211-213. <u>doi.org/10.1016/j.gloenvcha.2009.11.004</u>

BBC, 2017. Bacton and Walcott sandscaping project key to protecting east coast. [Online]. Available at: <u>https://www.bbc.co.uk/news/av/uk-england-norfolk-49607888</u> [accessed 16 August 2022].

BBC, 2022. The UK 'climate refugees' who won't leave. [Online]. Available at: https://www.bbc.com/future/article/20220506-the-uk-climate-refugees-who-wontleave#:~:text=The%20council%20plans%20to%20%22decommission,or%20assi stance%20from%20the%20council [accessed 14 December 2023].

BBC, 2023a. *Hemsby: Fifth home on Norfolk cliff edge demolished*. [Online]. Available at: <u>https://www.bbc.co.uk/news/uk-england-norfolk-64987709</u> [accessed 14 December 2023].

BBC, 2023b. *Clifftop homeowner says it's time to 'up sticks'*. [Online]. Available at: <u>https://www.bbc.co.uk/news/articles/cjmp8jkndylo</u> [accessed 14 December 2023].

BBC, 2023c. *Bacton gas plant in Norfolk is hidden hero, says PM*. [Online]. Available at: <u>https://www.bbc.co.uk/news/uk-england-norfolk-67336227</u> [accessed 03 January 2024].

Bell, D. and Carrick, J., 2017. Procedural Environmental Justice. In Holifield, R., Chakraborty, J. and Walker, G., eds. *The Routledge Handbook of Environmental Justice*. First Edition. London: Routledge.

Bell, J. and Walters, S., 2014. *Doing your research project: a guide for first time researchers*. Maidenhead: McGraw Hill Education and Oxford University Press.

Bellamy, R., 2019. Social readiness of adaptation technologies. *WIREs Climate Change*, **10**(6), Article e623. <u>doi.org/10.1002/wcc.623</u>

Béné, C., Newsham, A. and Davies, M., 2013. *Making the most of resilience (IDS In Focus Policy Briefing 32)*. Brighton: Institute of Development Studies.

Béné, C., Frankenberger, T. and Nelson, S., 2015. *Design, monitoring and evaluation of resilience interventions: Conceptual and empirical considerations (IDS Working Paper No. 459)*. Brighton: Institute of Development Studies.

Béné, C. and Doyen, L., 2018. From Resistance to Transformation: a Generic Metric of Resilience Through Viability. *Earth's Future*, **6**(7), pp.979–996. doi.org/10.1002/2017EF000660

Bevacqua, A., Yu, D. and Zhang, Y., 2018. Coastal vulnerability: Evolving concepts in understanding vulnerable people and places. *Environmental Science and Policy*, **82**(2018), pp.19-29. <u>doi.org/10.1016/j.envsci.2018.01.006</u>

Bichard, E. and Kazmierczak, A., 2011. Are homeowners willing to adapt to and mitigate the effects of climate change? *Climatic Change*, **112**(2012), pp.633-654. doi.org/10.1007/s10584-011-0257-8

Bhaskar, R., 1975. A realist theory of science. New York: Routledge

Blaikie, P., Cannon, T., Davis, I. and Wisner, B., 2005. *At risk: natural hazards, people's vulnerability and disasters*. Routledge, Abingdon.

Blais, L. E., 1996. Environmental racism reconsidered. *North Carolina Law Review*, **75**(1), Article 4, pp.75-148.

Bertoldo, R., Guignard, S., Dias, P. and Schleyer-Lindenmann, A., 2021. Coastal inconsistencies: Living with and anticipating coastal flood risks in southern France. *International Journal of Disaster Risk Reduction*. **64**(October 2021), Article 102521. <u>doi.org/10.1016/j.ijdrr.2021.102521</u>

Bogna, F., Raineri, A., Dell, G., 2020. Critical realism and constructivism: merging research paradigms for a deeper qualitative study. *Qualitative Research in Organizations and Management: An International Journal*, **15**(4), pp.461-484. <u>doi.org/10.1108/QROM-06-2019-1778</u>

Bolle, A., Baelus, L., Kragiopoulou, E., Van Quickelborne, E. and Lanckriet, T., 2020. Comparing Hard and Soft Engineering Solutions for Marina Inlet Sedimentation and Coastal Erosion at Blankenberge and Wenduine, Belgium. In Hardiman, N. and Institution for Civil Engineers, eds. *Coastal Management 2019: Joining forces to shape our future coasts.* [online]. Available from: <u>https://www-icevirtuallibrary-com.uea.idm.oclc.org/doi/10.1680/cm.65147.605</u> [Accessed 07 June 2022].

Bontje, L. and Slinger, J., 2017. A narrative method for learning from innovative coastal projects – Biographies of the Sand Engine. *Ocean and Coastal Management*. **142**(June 2017), pp.186-197. doi.org/10.1016/j.ocecoaman.2017.03.008

Bontje, L.E., Gomes, S.L., Wang, Z. and Slinger, J.H., 2019. A narrative perspective on institutional work in environmental governance – insights from a beach nourishment case study in Sweden. *Journal of Environmental Planning and Management*, **62**(1), pp.30-50. <u>doi.org/10.1080/09640568.2018.1459512</u>

Bopp, J. and Bercht, A., 2021. Considering time in climate justice. *Geographica Helvetica*, **76**(1), pp. 29-46. doi.org/10.5194/gh-76-29-2021

Boret, S.P. and Gerster, J., 2021. Social lives of tsunami walls in Japan: Concrete culture, social innovation and coastal communities. *IOP Conference Series: Earth and Environmental Science*, **630**, Article 012029. <u>doi.org/10.1088/1755-1315/630/1/012029</u>

Boxshall, A. 2022., Perspectives on building climate resilience via marine and coastal management from the governance frontline in Victoria, Australia. *Journal of Ocean & Coastal Management*, **228**(September 2022), Article 106291, doi.org/10.1016/j.ocecoaman.2022.106291

Braun, V. and Clarke, V., 2006. Using thematic analysis in psychology. *Qualitative research in Psychology*, **3**(2), pp.77-101. <u>doi.org/10.1191/1478088706qp063oa</u>

Breil, M., Zandersen, M., Pishmisheva, P., Branth Pedersen, A., Romanovska, L., Coninx I., Rogger, M. and Johnson, K., 2021. *Leaving No One Behind' in Climate Resilience Policy and Practice in Europe. European Topic Centre on Climate Change impacts, Vulnerability and Adaptation (ETC/CCA) Technical Paper 2021/2.* Italy: European Environment Agency.

Brennan, R., 2007. The North Norfolk Coastline: A Complex Legacy. *Coastal Management*, **35**(5), pp. 587-599. <u>doi.org/10.1080/08920750701593428</u>

Briggs, D., Abellan, JJ. and Fecht, D., 2008. Environmental Inequity in England: Small area associations between socio-economic status and environmental pollution. *Social Science and Medicine*, **67**(10), pp.1612-1629. <u>doi.org/10.1016/j.socscimed.2008.06.040</u>

British Geological Survey (BGS), 2021. *Coastal erosion at Happisburgh, Norfolk*. [Online]. Available at: <u>https://www.bgs.ac.uk/case-studies/coastal-erosion-at-happisburgh-norfolk-landslide-case-study/</u> [accessed 17 December 2021].

Brooks, S.M. and Spencer, T., 2019. Long-term trends, short-term shocks and cliff responses for areas of critical coastal infrastructure. In Wang, P. and Rosati, J.D. and Vallee, M. eds. *Coastal Sediments 2019: International Conference on Coastal Sediments 2019*, Tampa/St. Petersburg, Florida, USA, 27 – 31 May 2019. World Scientific, pp. 1179-1187.

Brown, K., 2012. Policy discourses of resilience. In Pelling, M., Manuel-Navarrete, D., and Redclift, M., eds. *Climate Change and the Crisis of Capitalism*. Abingdon: Routledge, pp. 37–50.

Brown, K., 2014. Global environmental change I: A social turn for resilience? *Progress in Human Geography*, **38**(1), pp.107-117. <u>doi.org/10.1177/0309132513498837</u>

Brown, J.M., Prime, T., Phelps, J.C., Barkwith, A., Hurst, M.D., Ellis, M.A., Masselink, G. and Plater, A.J., 2016. Spatio-temporal Variability in the Tipping Points of a Coastal Defense. *Journal of Coastal Research*, **2**(2016), Article 10075, pp.1042-1046. <u>doi.org/10.2112/SI75-209.1</u>

Brown, S., Tompkins, E.L., Suckall, N., French, J., Haigh, I.D., Lazarus, E., Nicholls, R.J., Penning-Rowsell, E., Thompson, C.E.L., Townend, I. and Van der Plank, S., 2023. Transitions in modes of coastal adaptation: addressing blight, engagement and sustainability. *Frontiers in Marine Science*, **10**, Article 1153134. doi.org/10.3389/fmars.2023.1153134

Bruine de Bruin, W. and Dugan, A., 2022. On the differential correlates of climate change concerns and severe weather concerns: evidence from the world risk poll. *Climatic Change*, **171**(2022), Article 33, <u>doi.org/ 10.1007/s10584-022-03353-8</u>

Bryman, A., 2006. *Social research methods*. Fifth edition. Oxford: Oxford University Press.

Bullard, R,D., 1990. *Dumping in Dixie: Race, Class, and Environmental Quality*. Boulder: Westview Press.

Buser, M., 2020. Coastal Adaptation Planning in Fairbourne, Wales: lessons for Climate Change Adaptation. *Planning Practice & Research*, **35**(2), pp.127-147. doi.org/10.1080/02697459.2019.1696145

Byskov, M.F., Hyams, K., Satyal, P., Anguelovski, I., Benjamin, L., Blackburn, S., Borie, M., Caney, S., Chu, E., Edwards, G., Fourie, K., Fraser, A., Heyward, C., Jeans, H., McQuistan, C., Paavola, J., Page, E., Pelling, M., Priest, S., Swiderska, K., Tarazona, M., Thornton, T., Twigg, J. and Venn, A., 2019. An agenda for ethics and justice in adaptation to climate change. *Climate and Development*, **13**(1), pp.1-9. <u>doi.org/10.1080/17565529.2019.1700774</u>

Cabezas-Rabadán, C., Rodilla, M., Pardo-Pascual, J.E. and Herrera-Racionero, P., 2019. Assessing users' expectations and perceptions on different beach types and the need for diverse management frameworks along the Western Mediterranean. *Journal of Land Use Policy*, **81**(February 2019), pp.219-231. doi.org/10.1016/j.landusepol.2018.10.027

Capstick, S.B., Demski, C.C., Sposato, R.G., Pigeon, N., Spence, A. and Corner, A., 2015. *Public perceptions of climate change in Britain following the winter 2013/2014 flooding. Understanding Risk Research Group Working Paper 15-01.* Cardiff, UK: Cardiff University.

Chaigneau, T., and Brown, K., 2016. Challenging the win-win discourse on conservation and development: analyzing support for marine protected areas. *Ecology and Society*, **21**(1), 36. <u>doi.org/10.5751/ES-08204-210136</u>

Chaigneau, T., Coulthard, S., Brown, K., Daw, T.M. and Schulte-Herbrüggen, B., 2018. Incorporating basic needs to reconcile poverty and ecosystem services. *Conservation Biology*, **33**(3), pp.655-664. <u>doi.org/10.1111/cobi.13209</u>

Chapman, M.G. and Blockley, D.J., 2009. Engineering novel habitat on urban infrastructure to increase intertidal biodiversity. *Oecologia*, **161**(3), pp.625-35. doi.org/10.1007/s00442-009-1393-y

Charmaz, K., 2006. *Constructing Grounded Theory*. Second Edition. California: SAGE Publications Inc.

Chausson, A., Smith, A., Seddon, N., Coath, M. and Matheson, S., 2020. *The Role of Nature-based Solutions for Climate Change Adaptation in UK Policy*. University of Oxford, WWF-UK and RSPB: NbSI.

Choi, C., Berry, P. and Smith, A., 2021. The climate benefits, co-benefits, and trade-offs of green infrastructure: A systematic literature review. *Journal of Environmental Management*, **291**(1 August 2021), Article 112583. doi.org/10.1016/j.jenvman.2021.112583

Cinner, J.E. and Barnes, M.L., 2019. Social Dimensions of Resilience in Social-Ecological Systems. *One Earth*, **1**(1), pp.51-56. <u>doi.org/10.1016/j.oneear.2019.08.003</u>

Clarke, T., Foster, L., Sloan, L., Bryman, A. and Vacchelli, E. 2021. *Bryman's social research methods*. Sixth Edition. Oxford: Oxford University Press. Clarke, D., Murphy, I. and Lorenzoni, I., 2018. Place attachment, disruption, and transformative adaptation. *Journal of Environmental Psychology*, **55**(February 2018), pp.81-89. <u>doi.org/10.1016/j.jenvp.2017.12.006</u>

Clarke, D. and Murphy, I., 2023. Incremental adaptation when transformation fails: the importance of place-based values and trust in governance in avoiding maladaptation. *Journal of Environmental Psychology*, **88**(June 2023), Article 102037. <u>doi.org/10.1016/j.jenvp.2023.102037</u>

Clayton, S., 2020. Climate anxiety: Psychological responses to climate change. *Journal of Anxiety Disorders*, **74**(2020), Article 102263. doi.org/10.1016/j.janxdis.2020.102263

Clément V., Rey-Vallette H. and Rulleau B., 2015. Perceptions on equity and responsibility in coastal zone policies. *Ecological Economics*, **119**(November 2015), pp.284–291. <u>doi.org/10.1016/j.ecolecon.2015.09.005</u>

Climate ADAPT, 2019. Case Studies. Sand Motor - building with nature solution to improve coastal protection along Delfland coast (the Netherlands) (2019). [Online]. Available at: <u>https://climate-adapt.eea.europa.eu/metadata/case-</u> <u>studies/sand-motor-2013-building-with-nature-solution-to-improve-coastal-</u> <u>protection-along-delfland-coast-the-netherlands</u> [accessed 16 December 2020].

Clipsham V., Flikweert J., Adnitt C., Doygun G., Spaan G., Fletcher A., Goodliffe, R., Johnson, M., Tudor, D., Knaapen, M. and Burgess, K., 2018. Sandscaping: Making innovative coastal management work in the UK. In Burgess, K. ed. *Coasts, marine structures and breakwaters 2017.* London: ICE Publishing, (no page numbers).

Coast Protection Act, 1949, *c.* 74, 12, 13 and 14, Great Britain. [Online]. Available at: <u>https://www.legislation.gov.uk/ukpga/Geo6/12-13-14/74</u> [accessed 15/12/21].

Coastal Concern Action Group (Happisburgh Village Website), 2023a. CCAG – Objection to Coastal Management Document. [Online]. Available at: / https://happisburgh.org.uk/ccag-press/objection-to-coastal-management-document/ [accessed 07 August 2023].

Coastal Concern Action Group (Happisburgh Village Website), 2023b. *CCAG – Timeline*. [Online]. Available at: http://happisburgh.org.uk/ccag/timeline/ [accessed 07 August 2023].

Coastal Concern Action Group (Happisburgh Village Website), 2023c. CCAG – Press Articles. [Online]. Available at: <u>https://happisburgh.org.uk/ccag/press/</u>[accessed 05 June 2024].

Coastal Partnership East, 2019. *Written evidence submitted* by Bill Parker Head of Coastal Partnership East (FCC0010). [Online]. Available at:

http://data.parliament.uk/WrittenEvidence/CommitteeEvidence.svc/EvidenceDoc ument/Environment,%20Food%20and%20Rural%20Affairs/Coastal%20flooding %20and%20adaptation%20to%20climate%20change/written/101266.html [accessed 30 March 2021].

Coastal Partnership East, 2022. *Coastal Partnership East*. [Online]. Available at: <u>https://www.coasteast.org.uk/</u> [Accessed 16 August 2022].

Coffey, A. and Atkinson, P., 1996. *Making Sense of Qualitative Data: Complimentary Research Strategies (and Social Thought)*. California: SAGE Publications Inc.

Colten, C.E. and Sumpter, A.R. 2008. Social memory and resilience in New Orleans. *Natural Hazards*, **48**(2009), pp. 355-364. <u>doi.org/10.1007/s11069-008-9267-x</u>

Committee on Climate Change (CCC), 2017. UK climate change risk assessment 2017 synthesis report. London: CCC.

Committee on Climate Change (CCC), 2018. *Managing the coast in a changing climate.* London: CCC.

Committee on Climate Change (CCC), 2021. *Progress in adapting to climate change. 2021 Report to Parliament*. London: CCC.

Committee on Climate Change (CCC), 2023. *Progress in adapting to climate change. 2023 report to parliament.* London: CCC.

Coolsaet, B., 2020. Introduction. In Coolsaet, B., ed, *Environmental Justice: Key Issues*. Abingdon: Routledge, pp.1-5.

Coolsaet, B. and Neron, P.Y., 2020. Recognition and environmental justice. In Coolsaet, B., ed, *Environmental Justice: Key Issues*. Abingdon: Routledge, pp.52-63.

Cooper, J. A. G. and McKenna, J., 2008. Social justice in coastal erosion management: The temporal and spatial dimensions. *Geoforum*, **39**(1), pp. 294–306. <u>doi.org/10.1016/j.geoforum.2007.06.007</u>

Cooper, J.A.G. and Pile, J., 2014. The adaptation-resistance spectrum: A classification of contemporary adaptation approaches to climate-related coastal change. *Ocean & Coastal Management*, **94**(June 2014), pp. 90-98. doi.org/10.1016/j.ocecoaman.2013.09.006

Costas, S., Ferreira, O. and Martinez, G., 2015. Why do we decide to live with risk at the coast? *Ocean & Coastal Management*, **118**(Part A, December 2015), pp.1-11. <u>doi.org/10.1016/j.ocecoaman.2015.05.015</u>

Cote, M. and Nightingale, A.J., 2011. Resilience thinking meets social theory: Situating social change in socio-ecological systems (SES) research. *Progress in Human Geography*, **36**(4), pp. 475-489. <u>doi.org/10.1177/0309132511425708</u>

Cresswell, J.W., 2009. *Research design: qualitative, quantitative, and mixed method approaches*. Third edition. California: SAGE Publications Inc.

Cresswell, J.W., 2011. Controversies in mixed methods research. In Denzin, N.K. and Lincoln, Y.S., eds. *The Sage Handbook of Qualitative Research*. California: SAGE Publications Inc.

Creswell, J. W. and Plano Clark, V. L., 2011. *Designing and conducting mixed methods research*. Second edition. Thousand Oaks, California: Sage.

Cutter, S.L. and Corendea, C., 2013. *From social vulnerability to resilience: measuring progress toward disaster risk reduction*. Bonn: UNU-EHS.

Cutter, S., 2016. Resilience to What? Resilience for Whom? *The Geographic Journal*, **182**(2), pp.110-113. <u>doi.org/10.1111/geoj.12174</u>

D'Souza, M., Johnson, M.F. and Ives, C.D., 2021. Values influence public perceptions of flood management schemes. *Journal of Environmental Management*, **291**(1 August 2021), Article 112636. doi.org/10.1016/j.jenvman.2021.112636

Davidson-Hunt, I.J. and Berkes, F., 2009. Nature and society through the lens of resilience: toward a human-in-ecosystem perspective. In Berkes, F., Colding, J.

and Folke, eds. *Navigating Social-Ecological Systems. Building Resilience for Complexity and Change.* Cambridge: Cambridge University Press, pp.53-82.

Day, S.A., O'Riordan, T., Bryson, J., Frew, P. and Young, R., 2015. Many Stakeholders, Multiple Perspectives: Long-Term Planning for a Future Coast. In Nicholls, R.J., Dawson, R.J., and Day, S.A., eds. *Broad Scale Coastal Simulation*. Netherlands: Springer Netherlands, pp.299-323.

Day, S., 2020. *Delivering the Bacton to Walcott Sandscaping Scheme in North Norfolk: Lessons Learned*. Norwich: North Norfolk District Council.

Day, S.A., Cotton, I., Lorenzoni, I. and Bark, R., 2023. *Community Perspectives on the Outcomes of the Bacton to Walcott Sandscaping Scheme*. A report commissioned by Royal Haskoning DHV. Norwich: UEA.

De Schipper, M.A., de Vries, S., Ruessink, G., de Zeeuw, R.C., Rutten, J., van Gelder-Maas, C. and Stive, M.F., 2016. Initial spreading of a mega feeder nourishment: Observations of the Sand Engine pilot project. *Coastal Engineering*, **111**(May 2016), pp23-48. <u>doi.org/10.1016/j.coastaleng.2015.10.011</u>

De Schipper, M.A., Ludka, B.C., Raubenheimer, B., Luijendijk, A.P. and Schlacher, T.A., 2021. Beach nourishment has complex implications for the future of sandy shores. *Nature Reviews Earth and Environment*, **2**(2021), pp.70-84. doi.org/10.1038/s43017-020-00109-9

Demski, C., Capstick, S., Pidgeon, N., Sposato, R.G. and Spence, A., 2016. Experience of extreme weather affects climate change mitigation and adaptation responses. *Climatic Change*, **140**(2017), pp. 149-164. <u>doi.org/10.1007/s10584-016-1837-4</u>

Department for Environment, Food and Rural Affairs (Defra), 2005. *Making Space for Water*. London: Department for Environment, Food and Rural Affairs.

Department for Environment, Food and Rural Affairs (Defra), 2020. *How we are working to tackle coastal erosion*. [Online]. Available at: <u>https://deframedia.blog.gov.uk/2020/08/11/how-we-are-working-to-prevent-flooding-and-coastal-erosion/</u> [accessed 08 June 2024].

Department for Environment, Food and Rural Affairs (Defra), 2021. *Central Government Funding for Flood and Coastal Erosion Risk Management in England*. [Online]. Available at:

https://www.gov.uk/government/statistics/funding-for-flood-and-coastal-erosionrisk-management-in-england [accessed 15 December 2021].

Department for Environment, Food and Rural Affairs (Defra), 2022. *Communities to trial innovative ways of adapting to coastal erosion*. [Online]. Available at: <u>https://www.gov.uk/government/news/communities-to-trial-innovative-ways-of-adapting-to-coastal-</u>

erosion#:~:text=Known%20as%20the%20Coastal%20Transition,be%20defende d%20from%20coastal%20erosion. [accessed 03 June 2022].

de Schipper M. A., de Vries S., Ruessink G., de Zeeuw R. C., Rutten J., van Gelder-Maas C. and Stive, M., 2016. Initial spreading of a mega feeder nourishment: Observations of the sand engine pilot project. *Coastal* 

*Engineering,* **111**(May 2016), pp.23–48. doi.org/10.1016/j.coastaleng.2015.10.011

de Schipper M. A., Ludka B. C., Raubenheimer B., Luijendijk A. P. and Schlacher T. A., 2021. Beach nourishment has complex implications for the future of sandy shores. *Nature Reviews Earth and Environment.* **2**(2021), pp.70– 84. <u>doi.org/10.1038/s43017-020-00109-9</u>

Di Baldassarre, G. Viglione, A., Carr, G., Kuil, L., Yan, K., Brandimarte, L. and Bloschl, G., 2015. Debates—Perspectives on socio-hydrology: Capturing feedbacks between physical and social processes. *Water Resources Research*, **51**(6), pp.4770-4781. <u>doi.org/10.1002/2014WR016416</u>

Dickson, M.E., Walkden, M.J.A. and Hall, J.W., 2007. Systemic impacts of climate change on an eroding coastal region over the twenty-first century. *Climatic Change*, **84**(2007), pp.141–166. <u>doi.org/10.1007/s10584-006-9200-9</u>

Diep, L., Parikh, P., Dos Santos Duarte, B.P., Bourget, A.F., Dodman, D. and Martins, J.R.S., 2022. "It won't work here": Lessons for just nature-based stream restoration in the context of urban informality. *Environmental Science and Policy*, **136**(October 2022), pp.542-554. <u>doi.org/10.1016/j.envsci.2022.06.020</u>

Dietz, S. and Atkinson, G., 2007. Public perceptions of equity in environmental policy: Traffic emissions policy in an english urban area. *Local Environment*, **10**(4), pp. 445-449. <u>doi.org/10.1080/13549830500160982</u>

Dilling L., Prakash A., Zommers Z., Ahmad F., Singh N., de Wit, S., Nalau, J., Daly, M. and Bowman, K. 2019. Is adaptation success a flawed concept? *Nature Climate Change*, **9**(2019), pp.572–574. <u>doi.org/10.1038/s41558-019-0539-0</u>

Domingues, R.B., Costas, S., Neves de Jesus, S. and Ferreira, O., 2017. Sense of place, risk perceptions and preparedness of a coastal population at risk: the case of Faro Beach. *Journal of Spatial and Organisational Dynamics*, **5**(3), pp.163-175. <u>https://www.jsod-cieo.net/journal/index.php/jsod/article/view/93</u>

Domingues, R.B., de Jesus, S.N. and Ferreira, O., 2021. Place attachment, risk perception, and preparedness in a population exposed to coastal hazards: A case study in Faro Beach, southern Portugal. *International Journal of Disaster Risk Reduction*, **60**(June 2021), Article 102288. doi.org/10.1016/j.ijdrr.2021.102288

Doorn, N., Gardoni, P. and Murphy, C., 2019. A multidisciplinary definition and evaluation of resilience: the role of social justice in defining resilience. *Sustainable and Resilient Infrastructure*, **4**(3), pp.112-123. doi.org/10.1080/23789689.2018.1428162

Doswald N., Munroe, R., Roe, D., Giuliani, A., Castelli, I., Stephens, J., Möller, I., Spencer, T., Vira, B., and Reid, H., 2014. Effectiveness of ecosystem-based approaches for adaptation: review of the evidence-base. *Climate and Development.* **6**(2), pp.185–201. <u>doi.org/10.1080/17565529.2013.867247</u>

Dredging Today, 2023. *Boskalis wraps up coastal protection project in Benin*. [Online]. Available at: <u>https://www.dredgingtoday.com/2023/05/17/boskalis-</u> <u>wraps-up-coastal-protection-project-in-benin/</u> [accessed 10 August 2023]. Dryzek, J., 2013. *The Politics of the Earth. Environmental Discourses*. Oxford: Oxford University Press.

Dryzek, J. and Tanasoca, A., 2021. *Democratizing Global Justice. Deliberating Global Goals*. Cambridge: Cambridge University Press.

Eakin, H., Tompkins, E.L., Nelson, D.R. and Anderies, J.M., 2009. Hidden costs and disparate uncertainties: trade-offs in approaches to climate policy. In Adger, W.N., Lorenzoni, I. and O'Brien, K.L., eds. *Adapting to Climate Change (Thresholds, Values, Governance)*. Cambridge: Cambridge University Press, pp. 212-226.

Eastern Daily Press (EDP), 2022. *Cutting edge 'sandscaping' process could protect vulnerable coastline*. [Online]. Available at: <u>https://www.edp24.co.uk/news/local-council/22569714.cutting-edge-sandscaping-process-protect-vulnerable-coastline/</u> [accessed 07 July 2023].

Eastern Daily Press (EDP), 2023a. *Norfolk County Council plea for more cash to save coastline*. [Online]. Available at: <u>https://www.edp24.co.uk/news/23814703.norfolk-county-council-plea-cash-save-coastline/</u> [accessed 05 December 2023].

Eastern Daily Press (EDP), 2023b. *MP Duncan Baker welcomes carbon capture boost for Norfolk*. [Online]. Available at: <u>https://www.edp24.co.uk/news/23791671.mp-duncan-baker-welcomes-carbon-capture-boost-norfolk/</u> [accessed 24.10.23].

Eastern Daily Press (EDP), 2024. *Happisburgh car park to be replaced as erosion hits village*. [Online]. Available at: <u>https://www.edp24.co.uk/news/24045504.happisburgh-car-park-replaced-erosion-hits-village/</u> [accessed 21 February 2024].

Edwards, T., 2017. *Current and Future Impacts of Sea Level Rise on the UK*. London: Government Office for Science.

Edwards, G., 2020. Climate justice. In Coolsaet, B., ed, *Environmental Justice: Key Issues*. Abingdon: Routledge, pp.148-160.

Ensor, J.E., Mohan, T., Forrester, J., Khisa, U.K., Karim, T. and Howley, P., 2021. Opening space for equity and justice in resilience: A subjective approach to household resilience assessment. *Global Environmental Change*. **68**(May 2021), pp.1-12.

Environment Agency (EA) Anglian Region, 1996. Shoreline management plans in the Anglian Region. Peterborough: Environment Agency.

Environment Agency (EA), 2011. Coastal flood boundary conditions for the mainland UK coasts and islands. [Online]. Available at: <u>https://www.gov.uk/flood-and-coastal-erosion-risk-management-research-reports/coastal-flood-boundary-conditions-for-the-mainland-uk-coasts-and-islands</u> [accessed August 16 2022].

Environment Agency (EA), 2019. *Comparison of LIDAR and Ground-based Topographic Surveys for Coastal Monitoring - Anglian Coastal Monitoring Programme*. Bristol: Environment Agency. Environment Agency (EA), 2020. *National Flood and Coastal Erosion Risk Management Strategy for England*. Peterborough: Environment Agency.

Environment Agency (EA), 2021a. *Working together to adapt to a changing climate – flood and coast.* [Online]. Available at: <u>https://www.gov.uk/flood-and-coastal-erosion-risk-management-research-reports/working-together-to-adapt-to-a-changing-climate-flood-and-coast</u> [accessed 09/11/23].

Environment Agency (EA), 2021b. *A method for monetising the mental health costs of flooding*. [Online]. Available at: <u>https://www.gov.uk/flood-and-coastal-erosion-risk-management-research-reports/a-method-for-monetising-the-mental-health-costs-of-flooding</u> [accessed 19 September 2023].

Environment Agency (EA), 2023a. *Resilient Coasts - Great Yarmouth and East Suffolk*. [Online]. Available at:

https://engageenvironmentagency.uk.engagementhq.com/esf008-coastal. [accessed 02 August 2023].

Environment Agency (EA), 2023b. *Email from EA to Isabel Cotton (21 November 2023).* 

EFRA (Environment, Food and Rural Affairs) Committee, 2019. *Coastal Flooding, Erosion and Adaptation to Climate Change (Inquiry Report)*. [Online]. Available at: <u>https://committees.parliament.uk/work/2562/coastal-flooding-and-adaptation-to-climate-change-inquiry/publications/</u> [accessed 02 February 2024].

Esri, 2023. ArcGIS Pro. [Online]. Available at: <u>https://www.esri.com/en-us/arcgis/products/arcgis-pro/overview</u> [accessed 11 December 2023].

Esteves, L. and Thomas, K., 2014. Managed realignment in practice in the UK: results from two independent surveys. *Journal of Coastal Research*, **70**(sp1), pp.407-413. <u>doi.org/10.2112/SI70-069.1</u>

Fairbrother M., 2017. Environmental attitudes and the politics of distrust. *Sociology Compass* **11**(5), Article e12482. <u>doi.org/10.1111/soc4.12482</u>

Famuditi, T., Bray, M., Potts, J., Baily, B. and Inkpen, R., 2018. Adaptive management and community reaction: The activities of Coastal Action Groups (CAGs) within the shoreline management process in England. *Marine Policy*, **97**(November 2018), pp.270-277. <u>doi.org/10.1016/j.marpol.2018.06.025</u>

Fernandez-Bilbao, A., Zsamboky, M., Smith, D., Knight, J. and Allen, J., 2011. *Impacts of climate change on disadvantaged UK coastal communities*. York: Joseph Rowntree Foundation.

Ferreira, V., Barreira, A.P., Pinto, P. and Panagopoulos, T., 2022. Understanding attitudes towards the adoption of nature-based solutions and policy priorities shaped by stakeholders' awareness of climate change. *Environmental Science and Policy*, **131**(May 2022), pp. 149-159. <u>doi.org/10.1016/j.envsci.2022.02.007</u>

Few, R., Brown, K. and Tompkins, E.L., 2007. Public participation and climate change adaptation: avoiding the illusion of inclusion. *Climate Policy*, **7**(1), pp.46-59. <u>doi.org/10.1080/14693062.2007.9685637</u>
Fisher, D.M., Ragsdale, J.M. and Fisher, E.C.S., 2018. The Importance of Definitional and Temporal Issues in the Study of Resilience. *Applied Psychology*, **68**(4), pp.583-620. <u>doi.org/10.1111/apps.12162</u>

Fisher, K., Goodliffe, R., and Hardiman, N., 2020. Adaptive Coastal Management at Three Sites in East Anglia. In Hardiman, N., and the Institution of Coastal Engineers., eds. *Coastal Management 2019*. London: ICE Publishing

Flikweert, J., 2017. Sandscaping. Innovative coastal management, working with nature for multiple benefits. London: ICE Publishing.

Flood and Water Management Act, 2010. *c. 29, England and Wales*. [Online]. Available at: <u>https://www.legislation.gov.uk/ukpga/2010/29/contents</u> [accessed 15 December 2021].

FloodRe, 2023. *Flood Re insurance*. [Online]. Available at: <u>https://www.floodre.co.uk/</u> [accessed 16 November 2023].

Flyvbjerg, B., 2006. Five Misunderstandings About Case-Study Research. *Qualitative Inquiry*, **12**(2), pp.219-429. <u>doi.org/10.1177/1077800405284363</u>

Flyvbjerg, B., 2011. Case Study. In Denzin, N.K. and Lincoln, Y.S., eds., *The Sage Handbook of Qualitative Research*. Fourth edition. Thousand Oaks, California: Sage, pp. 301-316.

Foerster, A., Macintosh, A. and McDonald, J., 2015. Trade-Offs in Adaptation Planning: Protecting Public Interest Environmental Values. *Journal of Environmental Law*, **27**(3), pp. 459-487. <u>doi.org/10.1093/jel/eqv017</u>

Folke, C., Hahn, T., Olsson, P. and Norberg, J., 2005. Adaptive governance of social-ecological systems. *Annual Review of Environment and Resources*, **30**(November 2005), pp.441-473. doi.org/10.1146/annurev.energy.30.050504.144511

Folke, C., 2006. Resilience: The emergence of a perspective for social– ecological systems analyses. *Global Environmental Change*, **16**(3), pp.253-267. <u>doi.org/10.1016/j.gloenvcha.2006.04.002</u>

Folke, C., Carpenter, S.R., Walker, B., Scheffer, M., Chapin, T. and Rockstrom, J., 2010. Resilience thinking: integrating resilience, adaptability and transformability. *Ecology and Society*, **15**(4), Article 20. <u>istor.org/stable/26268226</u>

Folkes, L., 2022. Moving beyond 'shopping list' positionality: Using kitchen table reflexivity and in/visible tools to develop reflexive qualitative research. *Qualitative Research*, **23**(5), pp.1301-1308. <u>doi.org/10.1177/14687941221098922</u>

Fortnam, M., Chaigneau, T., Evans, L. and Bastian, L., 2023. Practitioner approaches to trade-off decision-making in marine conservation development. *People and Nature*, **5**(5), pp.1636-1648.

Fowler, F. J., 2009. *Survey research methods*. Fourth edition. California: SAGE Publications Inc.

Fraser, N. and Honneth, A., 2003. *Redistribution or Recognition? A Political-Philosophical Exchange*. London and New York: Verso.

French, P.W., 2004. The changing nature of, and approaches to, UK coastal management at the start of the twenty-first century. *The Geographical Journal*, **170**(2), pp.116-125. jstor.org/stable/3451588

French Government., 2022. Assurance et catastrophe naturelle (ou technologique) (Natural disaster insurance). [Online]. Available at: <u>https://www.service-</u>

public.fr/particuliers/vosdroits/F3076#:~:text=380%20%E2%82%AC%20pour%2 Oles%20habitations,%C3%A0%20une%20r%C3%A9hydratation%20du%20sol [accessed 20 January 2023].

Frew, P., 2012. Adapting to coastal change in North Norfolk, UK. *Maritime Engineering*, **165**(3), pp.131-138. <u>doi.org/10.1680/maen.2011.23</u>

Galende-Sánchez, E. and Sorman, A.H., 2021. From consultation toward coproduction in science and policy: A critical systematic review of participatory climate and energy initiatives. *Energy Research & Social Science*, **73**(2021), Article 101907, <u>doi.org/10.1016/j.erss.2020.101907</u>

Gibbs, M.T., 2016. Why is coastal retreat so hard to implement? Understanding the political risk of coastal adaptation pathways. *Ocean & Coastal Management*, **130**(October 2016), pp.107-114. <u>doi.org/10.1016/j.ocecoaman.2016.06.002</u>

Gillespie, B.J., Ruel, E. and Wagner, W.E., 2015. *The Practice of Survey Research: Theory and Applications*. California: SAGE Publications Inc.

Glaser, B.G. and Strauss, A.L., 1967. *The Discovery of Grounded Theory: Strategies for Qualitative Research*. Mill Valley, California: Sociology Press.

Gonçalves, G., Santos, S., Duarte, D. and Gomes, J., 2019. Monitoring Local Shoreline Changes by Integrating UASs, Airborne LiDAR, Historical Images and Orthophotos. In Grueau, C., Laurini, R. and Raggia, L., eds. 2019. *Proceedings of the 5th International Conference on Geographical Information Systems Theory, Applications and Management (GISTAM 2019),* pp.126-134. <u>doi.org/10.5220/0007744101260134</u>

Gopalakrishnan, S., McNamara, D., Smith, M. and Murray, A.B., 2018. Decentralized Management Hinders Coastal Climate Adaptation: The Spatialdynamics of Beach Nourishment. *Environmental and Resource Economics*, **67**(4), pp.761-787. <u>doi.org/10.1007/s10640-016-0004-8</u>

Graham, S., Barnett, J., Fincher, R., Hurlimann, A. and Mortreux, A., 2014. Local values for fairer adaptation to sea-level rise: A typology of residents and their lived values in Lakes Entrance, Australia. *Global Environmental Change*, **29**(2014), pp.41-52. <u>doi.org/10.1016/j.gloenvcha.2014.07.013</u>

Graham, J., van Proosdij, D., Ellis, K., Bowron, T., and Thomas, J., 2023. *Restoring Piping Plover Habitat and Building Coastal Resilience with Naturebased Solutions in Atlantic Canada*. EGU General Assembly 2023, Vienna, Austria, 24–28 Apr 202. EGU23-4673. <u>doi.org/10.5194/egusphere-egu23-4673</u>.

Gray J., O'Neill K. and Qiu Z., 2017. Coastal residents' perceptions of the function of and relationship between engineered and natural infrastructure for coastal hazard mitigation. *Journal of Ocean and Coastal* 

*Management,* **146**(September 2017), pp.144–156. doi.org/10.1016/j.ocecoaman.2017.07.005

Greg-Lloyd, M., Peel, D. and Duck, R.W., 2013. Towards a social-ecological resilience framework for coastal planning. *Land Use Policy*, **30**(1), pp.925-933. doi.org/10.1016/j.landusepol.2012.06.012

Groen, L., Alexander, M., King, J.P., Jager, N.W. and Huitema, D., 2022. Reexamining policy stability in climate adaptation through a lock-in perspective. *Journal of European Public Policy*, **30**(3), pp.488-512. <u>doi.org.uk/10.1080/13501763.2022.2064535</u>

Grothmann, T. and Reusswig, F., 2006. People at Risk of Flooding: Why Some Residents Take Precautionary Action While Others do not. *Natural Hazards*, **38**(2006), pp.101-120. <u>doi.org/10.1007/s11069-005-8604-6</u>

Gubrium, J. F. and Holstein, J. A., 2000. Analyzing interpretive practice. In Denzin, N.K. and Lincoln, Y.S., eds. *Handbook of qualitative research*. Second Edition. Thousand Oaks, California: SAGE Publications Inc, pp. 487-508.

Guest, G., Namey, E.E., Mitchell, M.L., 2013. Qualitative research: defining and designing. In Guest, G., Namey, E.E., Mitchell, M.L., eds. *Collecting Qualitative Data: A Field Manual for Applied Research*. Thousand Oaks, California: SAGE Publications Inc, pp.1-40.

Haigh, I.D., Nicholls, R.J., Penning-Rowsell, E.C., and Sayers, P., 2022. Impacts of climate change on coastal flooding, relevant to the coastal and marine environment around the UK. *MCCIP Science Review 2020*, pp.546–565. doi.org/10.14465/2020.arc23.cfl

Hall, J.W., Dawson, R.J. and Wu, X.Z., 2015. Analysing Flood and Erosion Risks and Coastal Management Strategies on the Norfolk Coast. In Nicholls, R.J., Dawson, R.J., and Day, S.A., eds. *Broad Scale Coastal Simulation*. Netherlands: Springer Netherlands, pp.233-254.

Hanley, M. E., Hoggart, S,P,G., Simmonds, D.J., Bichot, A., Colangelo, M.A., Bozzeda, F., Heurtefeux, H., Ondiviela, B., Ostrowski, R., Recio, M., Trude, R., Zawadzka-Kahlaug, E. and Thompson, R.C., 2014. Shifting sands? Coastal protection by sand banks, beaches and dunes. *Journal of Coastal Engineering*, **87**,(2014), pp.136–146. doi.org/10.1016/j.coastaleng.2013.10.020

Hanna, C., White, I. and Glavovic, B., 2020. The Uncertainty Contagion: Revealing the Interrelated, Cascading Uncertainties of Managed Retreat. *Sustainability*, **12**(2), Article 736. <u>doi.org/10.3390/su12020736</u>

Hanna, C., White, I. and Glavovic, B.C., 2021. Managed retreats by whom and how? Identifying and delineating governance modalities. *Climate Risk Management*, **31**(2021), Article 100278, <u>doi.org/10.1016/j.crm.2021.100278</u>

Hanson, H., Brampton, A., Capobianco, M., Dette, H.H., Hamm, L., Laustrup, C., Lechuga, A. and Spanhoff, R., 2002. Beach nourishment projects, practices, and objectives—a European overview. *Journal of Coastal Engineering*, **47**(2), pp.81-111. doi.org/10.1016/S0378-3839(02)00122-9

Harcourt, R. and Dessai, S., 2023. *Public Dialogue on Climate Adaptation: What does a well-adapted England look like? Literature review for Defra/ Sciencewise.* London: IPSOS

Harcourt, R., Dessai, S., Bruine de Bruin, W. and Taylor, A., 2023. A social science research agenda to accelerate public engagement in climate change adaptation. *Frontiers in Psychology*, **14**(2023), Article 1286525, doi.org/10.3389/fpsyg.2023.1286525

Harvatt, J., Petts, J. and Chilvers, J., 2011. Understanding householder responses to natural hazards: flooding and sea-level rise comparisons. *Journal of Risk Research*, **14**(1), pp. 63-83. <u>doi.org/10.1080/13669877.2010.503935</u>

Haasnoot, M., Winter, G., Brown, S., Dawson, R.J., Ward, P.J. and Eilander, D., 2021. Long-term sea-level rise necessitates a commitment to adaptation: A first order assessment. *Climate Risk Management*, **34**(2021), Article 100355. doi.org/10.1016/j.crm.2021.100355

Happisburgh Parish Council, 2020. *Minutes of an extraordinary meeting of Happisburgh Parish Council held on zoom on Tuesday 24<sup>th</sup> November 2020.* [Online]. Available at:

https://happisburghpc.norfolkparishes.gov.uk/2020/12/09/november-draftminutes-2020/202011-november-extraordinary-hb-minutes/ [accessed 21/11/23].

Hayward, B. M., 2013. Rethinking resilience: reflections on the earthquakes in Christchurch, New Zealand, 2010 and 2011. *Ecology and Society*, **18**(4), pp.1-6. jstor.org/stable/26269450

Hemmerling, S.A., Barra, M. and Bond, R.H., 2019. Adapting to a Smaller Coast: Restoration, Protection and Social Justice in Coastal Louisiana. In Laska, S., ed. *Louisiana's Response to Extreme Weather. Extreme Weather and Society.* Switzerland: Springer, pp.113-144.

Herman, P., 2019. The balance sheet of nature under water. In Luijendijk, A. and van Oudenhoven, A., eds. *The Sand Motor: A nature-based response to climate change.* Delft: Delft University Publishers. pp.93-96.

Hickman, C., Marks, L., Pihkala, P., Clayton, S., Lewandowski, R.E., Mayall, E.E., Wray, B., Mellor, C. and van Susteren, L., 2021. Climate anxiety in children and young people and their beliefs about government responses to climate change: a global survey. *The Lancet Planetary Health*, **5**(12), Article E863-E873. doi.org/10.1016/S2542-5196(21)00278-3

Himes, A. and Muraca, B., 2018. Relational values: the key to pluralistic valuation of ecosystem services. *Current Opinion in Environmental Sustainability*. **35**(December 2018), pp.1-7. <u>doi.org/10.1016/j.cosust.2018.09.005</u>

HM Government, 2018. *The Green Book (Central Government Guidance on Appraisal and Reevaluation)*. London: The Stationery Office.

HM Government, 2023a. *Flood and Coastal Innovation Programmes*. [Online]. Available at: <u>https://www.gov.uk/guidance/flood-and-coastal-resilience-innovation-programme</u>. [Accessed 02 August 2023].

HM Government, 2023b. *Hundreds of new North Sea oil and gas licences to boost British energy independence and grow the economy*. [Online]. Available at: <u>https://www.gov.uk/government/news/hundreds-of-new-north-sea-oil-and-gas-</u>

licences-to-boost-british-energy-independence-and-grow-the-economy-31-july-2023. [Accessed 03 January 2024].

Holifield, R., Porter, M. and Walker, G., 2009. Introduction - Spaces of environmental justice: frameworks for critical engagement. *Antipode*, **41**(4), pp.591-612. <u>doi.org/10.1111/j.1467-8330.2009.00690.x</u>

Holling, C.S., 1973. Resilience and Stability of Socio-ecological Systems. *Annual Review of Ecology and Systematics*, **4**, pp.1-23. jstor.org/stable/2096802

Holling, C.S., 1996. Engineering Resilience versus Ecological Resilience. In Schulze, P.E., ed. *Engineering within Ecological Constraints*. Washington: National Academy Press, pp.31-43.

Hoonhout, B. and de Vries, S., 2017. Field measurements on spatial variations in aeolian sediment availability at the Sand Motor mega nourishment. *Aeolian Research*, **24**(February 2017), pp.93-104. <u>doi.org/10.1016/j.aeolia.2016.12.003</u>

Howarth, C., Morse-Jones, S., Kythreotis, A., Brooks, K. and Lane, M., 2020. Informing UK governance of resilience to climate risks: improving the local evidence-base. *Climatic Change*, **163**, pp.499-520. <u>doi.org/10.1007/s10584-020-02821-3</u>

Huggel, C., Bouwer, L.M., Juhola, S., Mechler, R., Muccione, V., Orlove, B. and Wallimann-Helmer, I., 2022. The existential risk space of climate change. *Climatic Change*, **174**(8), <u>doi.org/10.1007/s10584-022-03430-y</u>

Huisman, B.J.A., Wijsman, J.W.M., Arens, S.M., Vertegaal, C.T.M., Van der Valk, L., Van Donk, S.C., Vreugdenhil, H.S.I. and Taal, M.D., 2021. *10-years evaluation of the Sand Motor* 

Results of the Monitoring- and Evaluation Program (MEP) for the period 2011 to 2021. [Online]. Available at: <u>https://dezandmotor.nl/en/research/</u> [accessed 03 August 2023].

Information Commisioner's Office (ICO), 2018. *Guide to the General Data Protection Regulations*. [Online]. Available at: <u>https://www.gov.uk/government/publications/guide-to-the-general-data-protection-regulation</u> [accessed 05 January 2021].

Intergovernmental Panel on Climate Change (IPCC), 2014. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. In Field, C.B., Barros, V.R., Dokken, D.J., Mach, K.J., Mastrandrea, M.D., Bilir, T.E., Chatterjee, M., Ebi, K.L., Estrada, Y.O., Genova, R.C., Girma, B., Kissel, E.S., Levy, A.N., MacCracken, S., Mastrandrea, P.R., and White, L.L., eds. *Climate Change 2014: Impacts, Adaptation, and Vulnerability.* Cambridge University Press, Cambridge, United Kingdom and New York, USA, pp.1-1132.

Intergovernmental Panel on Climate Change (IPCC), 2022. Climate Change 2022: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, USA: Cambridge University Press.

Ikeme, J., 2003. Equity, environmental justice and sustainability: incomplete approaches in climate change politics. *Global Environmental Change*, **13**(3), pp.195-206. <u>doi.org/10.1016/S0959-3780(03)00047-5</u>

Jacobs, 2018. Research to Assess the Economics of Coastal Change Management in England and to Determine Potential Pathways for a Sample of Exposed Communities. [Online]. Available at: <u>https://www.theccc.org.uk/publication/economics-of-coastal-change-</u> management-in-england-jacobs/ [accessed 30 May 2022].

Jacobs, 2019. Shoreline Management Plan Refresh. SMP Forum Pre-meeting briefing. [Online]. Available at: <u>https://scopac.org.uk/wp-content/uploads/2019/11/Paper-I-151119-SMP-Refresh-Update.pdf</u> [accessed 03 February 2021].

Jenkins, K., Ford, A., Robson, C. & Nicholls, R. J., 2022. Identifying adaptation 'on the ground': Development of a UK adaptation inventory. *Climate Risk Management*. **36**, Article 100430. <u>doi.org/10.1016/j.crm.2022.100430</u>

Johnson, C., Penning-Rowsell, E. and Parker, D., 2007. Natural and Imposed Injustices: The Challenges in implementing 'Fair' Flood Risk Management Policy in England. *The Geographical Journal*, **173**(4), pp.374-390. jstor.org/stable/30130632

Johnson, J.A., Baldos, U., Hertel, T., Liu, J., Nootenboom, C., Polasky, S. and Roxburgh, T., 2020a. *Global Futures: modelling the global economic impacts of environmental change to support policy-making. Technical Report, January 2020.* [Online]. Available at: <u>https://www.wwf.org.uk/globalfutures</u> [accessed 01 December 2021].

Johnson, M., Goodliffe, R., Doygun, G. and Flikweert, J., 2020b. The UK's First Sandscaping Project (Bacton, Norfolk) from Idea to Reality. In Hardiman, N. and Institution for Civil Engineers, eds. *Coastal Management 2019: Joining forces to shape our future coasts.* London: ICE Publishing, (no page numbers).

Jones, N. and Clark, J.R.A., 2014. Social capital and the public acceptability of climate change adaptation policies: a case study in Romney Marsh, UK. *Climatic Change*, **123**(2014), pp.133-145. <u>doi.org/10.1007/s10584-013-1049-0</u>

Joseph, L.I. and Humphries, A.T., 2018. Identifying social factors that undermine support for nature-based coastal management. *Journal of Environmental Management*, **212**(2018), pp.32-38. <u>doi.org/10.1016/j.jenvman.2018.01.085</u>

Kaswan, A., 2020. Distributive environmental justice. In Coolsaet, B., ed, *Environmental Justice: Key Issues*. Abingdon: Routledge, pp.21-36.

Kates, R.W., Travis, W.R. and Wilbanks, T.J., 2012. Transformational Adaptation When Incremental Adaptations to Climate Change Are Insufficient. *Proceedings of the National Academy of Sciences USA*, **109**(19), pp.7156–7161. doi.org/10.1073/pnas.1115521109

Kaufmann, M., Priest, S.J. and Leroy, P., 2018. The undebated issue of justice: silent discourses in Dutch flood risk management. *Regional Environmental Change*, **18**, pp.325-337. <u>doi.org/10.1007/s10113-016-1086-0</u>

Kendon, M., McCarthy, M., Jevrejeva, S., Matthews, A., Sparks, T. and Garforth, J., 2021. State of the UK Climate 2020. *International Journal of Climatology*. **41**(S2), pp.1-76. <u>doi.org/10.1002/joc.7285</u>

Kitchin, R. and Tate, N.J., 2000. *Conducting research in human geography. Theory, methodology, and practice*. London and New York: Routledge.

Kollmuss, A. and Agyeman, J., 2002. Mind the gap: why do people act environmentally and what are the barriers to pro-environmental behaviour?. *Environmental Education Research*, **8**(3), pp.239–260. <u>doi.org/10.1080/13504620220145401</u>

Lager, F., Coninx, I., Breil, M., Bakhtaoui, I., Branth Pedersen, A., Mattern, K., van den Berg, H., Sini, E., Gallucio, G., Klein, R., & Vierikko, K. 2023. Just Resilience for Europe: Towards measuring justice in climate change adaptation. Italy: European Environment Agency.

Laporte-Fauret, Q., Castelle, B., Michalet, R., Marieu, V., Bujan, S. and Rosebery, D., 2021. Morphological and ecological responses of a managed coastal sand dune to experimental notches. *Science of the Total Environment*, **782**(Aug 2001), Article 146813. <u>doi.org/10.1016/j.scitotenv.2021.146813</u>

Lau, J. D., Gurney, G.G. and Cinner, J., 2021. Environmental justice in coastal systems: perspectives from communities confronting change. *Global Environmental Change*, **66**(Jan 2021), Article 102208. <u>10.1016/j.gloenvcha.2020.102208</u>

Lavrakas, P.J., 2008. *Encyclopedia of Survey Research Methods*. California: SAGE Publications Inc.

Law, A., Huynh, J., Yerbury, J. and Goel, U., 2011. *Happisburgh Community A record of research from a Socio-Cultural Perspective, Unit 2.* [Online]. Available at: <u>https://happisburghcommunity.wordpress.com/</u> [accessed 07 August 2023].

Lawrence, J., Boston, J., Bell, R., Olufson, S., Kool, R., Hardcastle, M. and Stroombergen, A., 2020. Implementing Pre-Emptive Managed Retreat: Constraints and Novel Insights. *Current Climate Change Reports*, **6**(2020) pp.66–80. <u>doi.org/10.1007/s40641-020-00161-z</u>

Lazarus, E.D., McNamara, D.E., Smith, M.D., Gopalakrishnan, S. and Murray, A.B., 2011. Emergent behavior in a coupled economic and coastline model for beach nourishment. *Nonlinear Processes in Geophysics*, **18**(6), pp.989–999.

Leach, M., (ed) 2008. *Re-framing resilience: A symposium report. STEPS Working Paper 13.* Brighton: STEPS Centre.

Leeder, M., 2011. Sedimentology and Sedimentary Basins (From Turbulence to *Tectonics*). Second Edition. Chichester, UK: Wiley-Blackwell

Leichenko, R.L. and O'Brien, K.M., 2019. *Climate and society: transforming the future. Cambridge*. Medford, MA: Polity Press.

Lindlof, T.R., and Taylor, B.G., 2011. *Qualitative Communication Research Methods*. Third Edition. California: SAGE Publications Inc.

Logan, T.M., Guikema, S.D. and Bricker, J.D., 2018. Hard-adaptive measures can increase vulnerability to storm surge and tsunami hazards over time. *Nature sustainability*, **1**(2018), pp.526-530. <u>doi.org/10.1038/s41893-018-0137-6</u>

Loon-Steensma, J.M., and Vellinga, P., 2013. Trade-offs between biodiversity and flood protection services of coastal salt marshes. *Current Opinion in Environmental Sustainability*, **5**(3-4), pp.320-326. doi.org/10.1016/j.cosust.2013.07.007

Lorenzoni, I., Nicholson-Cole, S. and Whitmarsh, L., 2007. Barriers perceived to engaging with climate change among the UK public and their policy implications. *Global Environmental Change*, **17**(3–4), pp.445–459. doi.org/10.1016/j.gloenvcha.2007.01.004

Lozoya, J.P., Sardá, R. and Jiménez, J.A., 2014. Users expectations and the need for differential beach management frameworks along the Costa Brava: Urban vs. natural protected beaches. *Land Use Policy*, **38**(May 2014), pp.397-414.

Ludy, J. and Kondolf, G.M., 2012. Flood risk perception in lands "protected" by 100-year levees. *Natural Hazards*, **61**(2012), pp.829-842. doi.org/10.1007/s11069-011-0072-6

Luijendijk A. P., Ranasinghe R., de Schipper M. A., Huisman B. A., Swinkels C. M., Walstra D. J. and Stive, M.J.F., 2017. The initial morphological response of the sand engine: A process-based modelling study. *Coastal Engineering.* **119**(January 2017), pp.1–14. doi.org/10.1016/j.coastaleng.2016.09.005

Luijendijk, A., 2019. Physical System. In Luijendijk, A. and van Oudenhoven, A., eds. *The Sand Motor: A nature-based response to climate change. Delft, Delft University Publishers*. pp.52-53.

Luis, S., Pinho, L., Lima, M.L., Roseta-Palmer, C., Martins, F.C. and de Almeida, A.B. 2016. Is it all about awareness? The normalization of coastal risk. *Journal of Risk Research*, **19**(6), pp.810-826. <u>doi.org/10.1080/13669877.2015.1042507</u>

Lwin, K.K., Pal, I., Shrestha, S. and Warnitchai, P., 2020. Assessing social resilience of flood-vulnerable communities in Ayeyarwady Delta, Myanmar. *International Journal of Disaster Risk Reduction*. **51**(2020), Article 101745, pp.1-10. <u>doi.org/10.1016/j.ijdtr.2020.101745</u>

Lyon, C., Saupe, E.C., Smith, C.J., Hill, D.J., Beckerman, A.P., Stringer, L.C., Marchant, R., McKay, J., Burke, A., O'Higgins, P., Dunhill, A.M., Allen, B.J., Riel-Salvatore, J. and Aze, T., 2022. Climate change research and action must look beyond 2100. *Global Change Biology.* **28**(2), pp.349-361. <u>doi.org/10.1111/gcb.15871</u>

Maclean, K., Ross, H., Cuthill, M. and Witt, B., 2016. Converging disciplinary understandings of social aspects of resilience. *Journal of Environmental Planning and Management*, **60**(3), 1–19. doi.org/10.1080/09640568.2016.1162706

Magnan, A.K., Bell, R., Duvat, V.K.E., Ford, J.D., Garschagen, M., Haasnoot, M., Lacambra, C., Losada, I.J., Mach, K.J., Noblet, M., Parthasaranthy, D., Sano, M., Vincent, K., Anisimov, A., Hanson, S., Malmström, A., Nicholls, R.J. and Winter, G., 2023. Status of global coastal adaptation. *Nature Climate Change*, **13**, pp.1213-1221. <u>doi.org/10.1038/s41558-023-01834-x</u>

Makey, L., Parsons, M., Fisher, K., Te Huna, A., Henare, M., Miru, V., Ruka, M. and Miru, M., 2022. (Un)Heard Voices of Ecosystem Degradation: Stories from the Nexus of Settler-Colonialism and Slow Violence. *Sustainability*, **14**, Article 14672. <u>doi.org/10.3390/su142214672</u>

Maldonado, J., Wang, I.F.C., Eningowuk, F., Iaukea, L., Lascurain, A., Lazarus, H., Naquin, A., Naquin, J.R., Nogueras-Vidal, K.M., Peterson, K., Rivera-Collazo, I., Souza, M.K., Stege, M. and Thomas, B., 2021. Addressing the challenges of climate-driven community-led resettlement and site expansion: knowledge sharing, storytelling, healing, and collaborative coalition building. *Journal of Environmental Studies and Sciences*, **11**(2021), pp.294–304. doi.org/10.1007/s13412-021-00695-0

Marion Suiseeya, K.R., 2020. Procedural justice matters. In Coolsaet, B., eds, *Environmental Justice: Key Issues*. Abingdon: Routledge, pp.37-51.

Marin V., Palmisani F., Ivaldi R., Dursi R., and Fabiana M., 2009. Users' perception analysis for sustainable beach management in Italy. *Ocean & Coastal Management*, **52**(5), pp.268–277. doi.org.uk/10.1016/j.ocecoaman.2009.02.001

Martell R., Mendoza E., Mariño-Tapia I., Odériz I. and Silva, R., 2020. How effective were the beach nourishments at cancun? *Journal of Marine Science and Engineering*, **8**(2020), Article 388. <u>doi.org.uk/10.3390/jmse8060388</u>

Martinez G., Costas S. and Ferreira Ó., 2020. The role of culture for coastal disaster risk reduction measures: Empirical evidence from northern and southern Europe. *Advances in Climate Change Research*, **11**(4), pp.297–309. doi.org/10.1016/j.accre.2020.11.001

Mason, J., 2002. *Qualitative Researching*. Second Edition. London: SAGE Publications Ltd.

Matin, N., Forrester, J. and Ensor, J., 2018. What is equitable resilience? *World Development*, **109**(2018), pp.197-205. <u>doi.org/10.1016/j.worlddev.2018.04.020</u>

Matyas, D. and Pelling, M., 2015. Positioning resilience for 2015: the role of resistance, incremental adjustment and transformation in disaster risk management policy. *Disasters*, **39**(1), pp.1-18. <u>doi.org/10.1111/disa.12107</u>

Maxwell, J.A. and Mittapalli, K., 2010. Realism as a Stance for Mixed Methods Research. In Tashakkori, A., and Teddlie, C., eds. *SAGE Handbook of Mixed Methods in Social & Behavioral Research*. Second Edition. California: SAGE Publications Inc, pp.145-168.

MCCIP (Marine Climate Change Impacts Partnership), 2020. *REPORT CARD 2020*. [Online]. Available at: <u>https://www.mccip.org.uk/</u> [Accessed 25 January 2024].

McGinlay, J., Jones, N., Clark, J. and Maguire-Rajpaul, V.A., 2021. Retreating coastline, retreating government? Managing sea level rise in an age of austerity. *Ocean & Coastal Management*, **204**(April 2021), Article 105458. doi.org/10.1016/j.ocecoaman.2020.105458

McKinley E., Pages J. F., Ballinger R. C. and Beaumont N., 2020. Forgotten landscapes: Public attitudes and perceptions of coastal saltmarshes. *Ocean & Coastal Management*, **187**(1 April 2020). doi.org.uk/10.1016/j.ocecoaman.2020.105117

Meerow, S., Pajouhesh, P. and Miller, T.R., 2019. Social equity in urban resilience planning. *Local Environment (The International Journal of Justice and Sustainability)*, **24**(9), pp.793-808. <u>doi.org/10.1080/13549839.2019.1645103</u>

Meyer, L.H., 2017. Introduction. In Meyer, L.H., ed. Intergenerational Justice. First Edition. London: Routledge, xi-xxiv.

Mikkelsen, B., 2005. *Methods for development work and research: a new guide for practitioners*. New Delhi and London: Sage.

Mikulewicz, M., 2019. Thwarting adaptation's potential? A critique of resilience and climate-resilient development. *Geoforum*, **104**(2019), pp. 267-282. doi.org/10.1016/j.geoforum.2019.05.010

Milhorance, C., Le Coq, J-F., Sabourin, E., Andrieu, N., Mesquita, P., Cavalcante, L. and Nogueira, D., 2021. A policy mix approach for assessing rural household resilience to climate shocks: Insights from Northeast Brazil. *International Journal of Agricultural Sustainability*, **20**(4), pp.675-691.

Milligan, J. and O'Riordan, T., 2007. Governance for Sustainable Coastal Futures. *Coastal Management*. **35**(2007), pp.499-509. doi.org/10.1080/08920750701525800

Milligan, J., O'Riordan, T., Nicholson-Cole, S. and Watkinson, A., 2009. Nature conservation for future sustainable shorelines: Lessons from seeking to involve the public. *Land Use Policy*, **26**(2), pp.203-213. doi.org/10.1016/j.landusepol.2008.01.004

Milhorance, C., Le Coq, J-F., Sabourin, E., Andrieu, N., Mesquita, P., Cavalcante, L. and Nogueira, D., 2021. A policy mix approach for assessing rural household resilience to climate shocks: Insights from Northeast Brazil. *International Journal of Agricultural Sustainability*, **20**(4), pp.675-691. <u>doi.org/10.1080/14735903.2021.1968683</u>

Ministry for Housing, Communities and Local Government (MHCLG), 2019. *English Indices of Deprivation 2019*. [Online]. Available at: <u>https://www.gov.uk/government/statistics/english-indices-of-deprivation-2019</u> [accessed 17 December 2021].

Ministry for Housing, Communities and Local Government (MHCLG), 2020. *Land Use Statistics England 2018*. [Online]. Available at: <u>https://www.gov.uk/government/statistical-data-sets/live-tables-on-land-use</u> [accessed 17 December 2021].

Miriti, M.N., Rawson, A.J. and Mansfield, B., 2022. The history of natural history and race: Decolonizing human dimensions of ecology. *Ecological Applications*, 33(1), Article e2748. <u>doi.org/10.1002/eap.2748</u>

Mitchell, G., Norman, P. and Mullin, K., 2015. Who benefits from environmental policy? An environmental justice analysis of air quality change in Britain 2001-

2011. *Environmental Research Letters*, **10**(10), Article 105009. doi.org/10.1088/1748-9326/10/10/105009.

Moller, I., Kudella, M., Rupprecht, F., Spencer, T., Paul, M., van Wesenbeeck, B.K., Wolters, G., Jensen, K., Bouma, T.J., Miranda-Lange, M. and Schimmels, S., 2014. Wave attenuation over coastal salt marshes under storm surge conditions. *Nature Geoscience*, **7**, pp.727-731. <u>doi.org/10.1038/ngeo2251</u>

Möller, I., 2020. Towards Sustainable Coasts. The Challenge of Nature Based Coastal Protection. [Online]. Available at: <u>https://www.ccatproject.eu/event/coastal-communities-adapting-together-</u> <u>exchanging-knowledge-and-best-practice-across-borders/</u> [accessed 05 January 2021].

Moore, P., 2012. What does adaptation mean for coastal communities? *Maritime Engineering*, **165**(3), pp.147-154. <u>doi.org/10.1680/maen.2011.25</u>

Moraes, R.P.L., Reguero, B.G., Mazarrasa, I., Ricker, M. and Juanes, J.A., 2022. Nature-Based Solutions in Coastal and Estuarine Areas of Europe. *Frontiers in Environmental Science*. **10**(2022), <u>doi.org/10.3389/fenvs.2022.829526</u>

Moreno, L.J., and Muñoz-Perez, J.J., 2021. Beach Nourishment: A 21st Century Review. *Journal of Marine Science and Engineering*, **9**(5), pp. 499-453. doi.org/10.3390/jmse9050499

Morgan, D.L., 1998. Practical Strategies for Combining Qualitative and Quantitative Methods: Applications to Health Research. *Qualitative Health Research*, **8**(3), pp.301-484. <u>doi.org/10.1177/104973239800800307</u>

Mott MacDonald, 2014. *Bacton, Walcott and Ostend Coastal Management Study*. Norfolk: North Norfolk District Council.

Mott MacDonald, 2016. *Bacton to Walcott Sea Flooding Study*. Norfolk: North Norfolk District Council.

Muhl, E-K., Armitage, D., Anderson, K., Boyko, C., Busilacchi, S., Butler, J., Cvitanovic, C., Faulkner, L.A., Hall, J.A., Martynuik, G., Paul-Burke, K., Swerdfager, T., Thorpe, H. and van Putten, I.E., 2023. Transitioning toward "deep" knowledge co-production in coastal and marine systems: examining the interplay among governance, power, and knowledge. *Ecology and Society*, **28**(4), Article 17. <u>doi.org/10.5751/ES-14443-280417</u>

Mulder, J.P. and Tonnon, P.K., 2010. Sand engine: Background and design of a mega-nourishment pilot in The Netherlands. *Coastal Engineering Proceedings*, **1**(32), Article 35. <u>doi.org/10.9753/icce.v32.management.35</u>

Murdock, E.G., 2020. A history of environmental justice. In Coolsaet, B., eds, *Environmental Justice: Key Issues*. Abingdon: Routledge, pp.6-17.

Myatt, L.B., Scrimshaw, M.D. and Lester, J.N., 2003. Public perceptions and attitudes towards a forthcoming managed realignment scheme: Freiston Shore, Lincolnshire, UK. *Ocean & Coastal Management*, **46**(6-7), pp.565-582. doi.org/10.1016/S0964-5691(03)00035-8

Natural Resources Wales, 2021. *Shoreline Management Plans: Supplementary guidance for their ongoing maintenance and delivery – Wales.* [Online]. Available

at: <u>https://naturalresources.wales/flooding/managing-flood-risk/shoreline-management-plans/?lang=en</u> [accessed 21 November 2023].

Naylor, L.A., Brady, U., Quinn, T., Brown, K. and Anderies, J.M., 2019. A multiscale analysis of social-ecological system robustness and vulnerability in Cornwall, UK. *Regional Environmental Change*, **19**, pp.1835-1848. doi.org/10.1007/s10113-019-01530-7

NCCARF (National Climate Change Adaptation Research Facility), 2010. *The 2008 Floods in Queensland: A Case Study of Vulnerability, Resilience and Adaptive Capacity.* Australia: National Climate Change Adaptation Research Facility.

Nicholls, R.J., Townend, I.H., Bradbury, A.P., Ramsbottom, D. and Day, S.A., 2013. Planning for long-term coastal change: Experiences from England and Wales. *Ocean Engineering*, **71**(October 2013), pp.3-16. doi.org/10.1016/j.oceaneng.2013.01.025

Nicholson-Cole, S. and O'Riordan, T., 2009. Adaptive governance for a changing coastline: science, policy and publics in search of a sustainable future. In Adger, W.N., Lorenzoni, I., O'Brien, K.L., eds. *Adapting to Climate Change (Thresholds, Values, Governance).* Cambridge: Cambridge University Press, pp.368-383.

Nightingale, A.J. Gonda, N. and Eriksen, S.H., 2021. Affective adaptation = effective transformation? Shifting the politics of climate change adaptation and transformation from the status quo. *Wiley Interdisciplinary Reviews: Climate Change*. **13**(1), Article e740. <u>doi.org/10.1002/wcc.740</u>

NOAA, 2023. *What is lidar*? [Online]. Available at: <u>https://oceanservice.noaa.gov/facts/lidar.html</u> [accessed 07 June 2024].

Nobert, S. and Pelling, M., 2020. What can adaptation to climate-related hazards tell us about the politics of time making? Exploring durations and temporal disjunctures through the 2013 London heat wave. *Geoforum*, **85**(October 2017), pp.122-130. <u>doi.org/10.1016/j.geoforum.2017.07.010</u>

Nordstorm, K.F., Armaroli, C., Jackson, N.L. and Ciavolo, P., 2015. Opportunities and constraints for managed retreat on exposed sandy shores: Examples from Emilia-Romagna, Italy. *Ocean & Coastal Management*, **104**(February 2015), pp.11-21. <u>doi.org/10.1016/j.ocecoaman.2014.11.010</u>

North Norfolk District Council (NNDC), 2016. *Coastal Erosion Development Control Guidance.* [Online]. Available at: <u>https://www.north-norfolk.gov.uk/info/planning-policy/current-local-plan/coastal-erosion-development-control-guidance/</u> [accessed 15 March 2021].

North Norfolk District Council (NNDC), 2017. *Bacton to Walcott coastal management summary report (and appendices)*. [Online]. Available at: <u>https://www.north-norfolk.gov.uk/media/3694/bacton-to-walcott-coastal-management-scheme-summary-report-inc-appendices.pdf</u> [accessed 09 August 2022].

North Norfolk District Council (NNDC), 2019. *Bacton to Walcott Sandscaping Scheme, Information Letter 01 (distributed to residents)*. Cromer: North Norfolk District Council.

North Norfolk District Council (NNDC), 2021. Annual Monitoring Report 2019-2020. [Online]. Available at: <u>https://www.north-norfolk.gov.uk/tasks/planning-policy/annual-monitoring-report/</u> [accessed 16 August 2022].

North Norfolk District Council (NNDC), 2022a. *Coastal Loss Innovative Funding and Financing (CLIFF)*. [Online]. Available at: <u>https://www.north-norfolk.gov.uk/tasks/coastal-management/coastal-loss-innovative-funding-and-financing-cliff/</u> [accessed 20 January 2023].

North Norfolk District Council (NNDC), 2022b. *Bacton to Walcott coastal management.* [Online]. Available at: <u>https://www.north-norfolk.gov.uk/sandscaping</u> [accessed 06 December 2023].

North Norfolk District Council (NNDC), 2023a. *Policy EN12 Relocation and Replacement of Development Affected by Coastal Erosion Risk*. [Online]. Available at: <u>https://www.north-norfolk.gov.uk/info/planning-policy/current-local-plan/policies/policy-en12-relocation-and-replacement-of-development-affected-by-coastal-erosion-risk/</u> [accessed 07 August 2023].

North Norfolk District Council (NNDC), 2023b. Sandscaping: Frequently asked questions. [Online]. Available at: <u>https://www.north-norfolk.gov.uk/tasks/coastal-management/sandscaping-frequently-asked-questions/</u> [accessed 08 August 2023].

North Norfolk District Council (NNDC), 2023c. *Mundesley Coastal Management Scheme*. [Online]. Available at: <u>https://www.north-norfolk.gov.uk/tasks/coastal-management/mundesley-coastal-management-scheme/</u> [accessed 21 November 2023].

North Norfolk District Council (NNDC), 2023d. *Coastwise FAQs*. [Online]. Available at: <u>https://www.north-</u>

norfolk.gov.uk/tasks/projects/coastwise/coastwise-faqs/ [accessed 15 December 2023].

North Norfolk District Council (NNDC), 2023e. *Coastwise Cafes coming to North Norfolk*. [Online]. Available at: <u>https://www.north-</u>

norfolk.gov.uk/news/2024/january/coastwise-cafes-coming-to-north-norfolk/ [accessed 14 February 2024].

North Norfolk District Council (NNDC), 2024. *Policy EN12 Relocation and Replacement of Development Affected by Coastal Erosion Risk.* [Online]. Available at: <u>https://www.north-norfolk.gov.uk/info/planning-policy/current-local-plan/policies/policy-en12-relocation-and-replacement-of-development-affected-by-coastal-erosion-risk/</u> [accessed 08 May 2024].

NSTA, 2023. *Net zero boost as carbon storage licences accepted*. [Online]. Available at: <u>https://www.nstauthority.co.uk/news-publications/net-zero-boost-as-</u><u>carbon-storage-licences-accepted/</u> [accessed 24 October 2023].

Nunn, P.D., Klöck, C. and Duvat, V., 2021. Seawalls as maladaptations along island coasts. *Ocean and Coastal Management*, **205**(1 May 2021), Article 10554. doi.org/10.1016/j.ocecoaman.2021.105554

Nursey-Bray, M., Nicholls, R.J., Vince, J., Day, S. and Harvey, N., 2017. *Public participation, coastal management and climate change adaptation*. In Green, D.R. and Payne, J.L. eds. Marine and Coastal Management. 1<sup>st</sup> Edition. London: Routledge.

O'Brien, K., 2017. *Climate Change Adaptation and Social Transformation. The International Encyclopedia of Geography*. New Jersey: John Wiley & Sons,

O'Donnell, T., 2019. Don't get too attached: Property-place relations on contested coastlines. *Transactions of the Institute of British Geographers*, **45**(3), pp.559-574. <u>doi.org/10.1111/tran.12368</u>

Office for National Statistics (ONS), 2011. 2011 Census. [Online]. Available at: <u>https://www.ons.gov.uk/census/2011census/2011censusdata/2011censusdataca</u> talogue [accessed 17 December 2021].

Office for National Statistics (ONS), 2021. 2021 Census. [Online]. Available at: <u>https://www.ons.gov.uk/census/maps/choropleth?oa=E00135943</u> [accessed 03 August 2021].

Ogunbode, C., Liu, Y. and Tausch, N., 2017. The moderating role of political affiliation in the link between flooding experience and preparedness to reduce energy use. *Climatic Change*, **145**(2017), pp.445-458. <u>doi.org/10.1007/s10584-017-2089-7</u>

O'Riordan, T., Watkinson, A. and Milligan, J., 2006. *Living with a changing coastline: Exploring new forms of governance for sustainable coastal futures. Tyndall Centre Technical Report No. 49, July 2006.* University of East Anglia: Tyndall Centre for Climate Change Research.

O'Riordan, T., Gomes, C. and Schmidt, L., 2014. The Difficulties of Designing Future Coastlines in the Face of Climate Change. *Landscape Research*, **39**(6), pp. 613-630. <u>doi.org/10.1080/01426397.2014.975108</u>

Oriangi, G., Albrecht, F., Di Baldassarre, G., Bamutaze, Y., Mukwaya, P.I., Ardö, J. and Pilesjö, P., 2020. Household resilience to climate change hazards in Uganda. *International Journal of Climate Change Strategies and Management*. **12**(1), pp.59-73. <u>doi.org/10.1108/IJCCSM-10-2018-0069</u>

Owen, G., 2020. What makes climate change adaptation effective? a systematic review of the literature. *Global Environmental Change*, **62**(2020). Article 102071. doi.org/10.1016/j.gloenvcha.2020.102071

Palmer, C.G., Fry, A., Libala, N., Ralekhetla, M., Mtati, N., Weaver, M., Mtintsilana, Z. and Scherman, P-A., 2022. Engaging society and building participatory governance in a rural landscape restoration context. *Anthropocene*, **37**(March 2022), Article 100320. <u>doi.org/10.1016/j.ancene.2022.100320</u>

Parkinson, R.W. and Ogurcak, D.E., 2018. Beach nourishment is not a sustainable strategy to mitigate climate change. *Estuarine, Coastal and Shelf Science*, **212**(Nov 2018), pp.203-209. <u>doi.org/10.1016/j.ecss.2018.07.011</u>

Payo, A., Walkden, M., Ellis, M.A., Barkwith, A., Favis-Mortlock, D., Kessler, H., Wood, B., Burke, H. and Lee, J., 2018. A Quantitative Assessment of the Annual

Contribution of Platform Downwearing to Beach Sediment Budget: Happisburgh, England, UK. *Journal of Marine Science and Engineering*, **6**(4), Article 113. doi.org/10.3390/jmse6040113

Payo, A., French, J.R., Sutherland, J., Ellis, M.A. and Walkden, M., 2020. Communicating Simulation Outputs of Mesoscale Coastal Evolution to Specialist and Non-Specialist Audiences. *Journal of Marine Science and Engineering*, **8**(4), Article 235. <u>doi.org/10.3390/jmse8040235</u>.

Pellow, D., 2020. Critical environmental justice studies. In Coolsaet, B., eds. *Environmental Justice: Key Issues*. Abingdon: Routledge, pp.293-302.

Perez, B.F. and Tomaselli, A., 2021. Indigenous Peoples and climate-induced relocation in Latin America and the Caribbean: managed retreat as a tool or a threat? *Journal of Environmental Studies and Sciences*, **11**(2021), pp.352-364. doi.org/10.1007/s13412-021-00693-2

Peterson, C.H., Bishop, M.J., Johnson, G.A., D'Anna, L.M. and Manning, L.M., 2006. Exploiting beach filling as an unaffordable experiment: Benthic intertidal impacts propagating upwards to shorebirds. *Journal of Experimental Marine Biology and Ecology*, **338**(2), pp.205-221. <u>doi.org/10.1016/j.jembe.2006.06.021</u>

Pit, I.R., Wassen, M.J., Kooijman, A.M., Dekker, S.C., Griffioen, J., Arens, S.M. and Van Dijk, J., 2020. Can sand nourishment material affect dune vegetation through nutrient addition? *Science of the Total Environment*, **725**(July 2020), Article 138233. <u>doi.org/10.1016/j.scitotenv.2020.138233</u>

Pitt Review, 2008. *Learning lessons from the 2007 floods*. [Online]. Available at: <u>https://webarchive.nationalarchives.gov.uk/ukgwa/20100702215619/http://archive.cabinetoffice.gov.uk/pittreview/thepittreview/final\_report.html</u> [accessed 23 August 2022].

Plumper, T., Quiroz Flores, A. and Neumayer, E., 2017. The double-edged sword of learning from disasters: Mortality in the Tohoku tsunami. *Global Environmental Change*, **44**(May 2017), pp.49-56. <u>doi.org/10.1016/j.gloenvcha.2017.03.002</u>

Porter, J.J., Dessai, S. and Tompkins, E.L., 2014. What do we know about UK household adaptation to climate change? A systematic review. *Climatic Change*, **127**, pp.371-379. <u>doi.org/10.1007/s10584-014-1252-7</u>

Post, M., 2019. Flatfish Growing up in a sandpit. In Luijendijk, A. and van Oudenhoven, A., eds. 2019. *The Sand Motor: A nature-based response to climate change.* Delft: Delft University Publishers, pp.93-96.

POST (The Parliamentary Office of Science and Technology), 2021. *Coastal Management, No 647 (UK Parliament POST Note).* [Online]. Available at: <u>https://post.parliament.uk/type/postnote/</u> [accessed 15 December 2021].

Prati, G., Albanesi, C., Pietrantoni, L. and Airoldi, L., 2016. Public perceptions of beach nourishment and conflict management strategies: A case study of Portonovo Bay in the Adriatic Italian Coast. *Land Use Policy*, **50**(January 2016), pp.422-428. <u>doi.org/10.1016/j.landusepol.2015.06.033</u>

QSR, 2022. *NVivo qualitative data analysis software*. Available at: <u>https://www.qsrinternational.com/nvivo-qualitative-data-analysis-software/home</u> [accessed 26 August 2022].

Quinn, T., Lorenzoni, I. and Adger, W.N., 2015. Place Attachment, Identity, and Adaptation. In O'Brien, K. and Selboe, E., eds. *The Adaptive Challenge of Climate Change*. Cambridge: Cambridge University Press, pp.160-170.

Papadimitriou, L., Holman, I.P., Dunford, R. and Harrison, P.A., 2019. Trade-offs are unavoidable in multi-objective adaptation even in a post-Paris Agreement world. *Science of the Total Environment*, **696**(15 December 2019), Article 134027. <u>doi.org/10.1016/j.scitotenv.2019.134027</u>

Poulton, C.V., Lee, J., Hobbs, P., Jones, L. and Hall, M., 2006. Preliminary investigation into monitoring coastal erosion using terrestrial laser scanning: Case study at Happisburgh, Norfolk. *The Bulletin of the Geological Society of Norfolk*, **56**, pp.45–64. <u>nora.nerc.ac.uk/id/eprint/214</u>

Preston, I., Banks, N., Hargreaves, K., Kazmierczak, A., Lucas, K., Mayne, R., Downing, C. and Street, R., 2014. *Climate change and social justice: an evidence review.* York: Joseph Rowntree Foundation.

Radermacher, M., 2018. *Impact of sand nourishments on hydrodynamics and swimmer safety.* The Netherlands: Delft University of Technology. [Online]. Available from: doi.org/10.4233/uuid:0816cbe5-4e42-4fd3-a328-4775c5ccb633 [Accessed 08 June 2022].

Rafliana, I., Jalayer, F., Cerase, A., Cugliari, L., Baiguera, M., Salmanidou, D., Necmioglu, O., Aguirre Ayerbe, I., Lorito, S., Fraser, S., Lovholt, F., Babeyko, A., Salgado-Galvez, M.A., Selva, J., De Risi, R., Sorensen, M.B., Behrens, J., Aniel-Quiroga, I., Del Zoppo, M., Belliazzi, S. and Hancilar, U., 2022. Tsunami risk communication and management: Contemporary gaps and challenges. *International Journal of Disaster Risk Reduction*, **70**(15 February 2022), Article 102771. doi.org/10.1016/j.ijdrr.2021.102771

Rawls, J., 1971. A theory of justice. Boston: Belknap Press.

Reyes-García, V., Fernández-Llamazares, A., McElwee, P., Molnár, Z., Öllerer, K., Wilson, S.J. and Brondizio, E., 2019. The contributions of Indigenous Peoples and local communities to ecological restoration. *Restoration Ecology*, **27**(1), pp.3-8. doi.org/10.1111/rec.12894

Rizzo, A., Vandelli, V., Buhagiar, G., Micallef, A. and Soldati, M., 2020. Coastal Vulnerability Assessment along the North-Eastern Sector of Gozo Island(Malta, Mediterranean Sea). *Water*, **12**(5), Article 1405. pp.1-26. doi.org/10.3390/w12051405

Roca E. and Villares M., 2012. Public perceptions of managed realignment strategies: The case study of the ebro delta in the Mediterranean basin. *Ocean & Coastal Management.* **60**(May 2012), pp.38–47. doi.org/10.1016/j.ocecoaman.2012.01.002

Rodriguez, I. and Inturias, M.L., 2018. Conflict transformation in indigenous peoples' territories: doing environmental justice with a 'decolonial turn'.

*Development Studies Research*, **5**(1), pp.90-105. <u>doi.org/10.1080/21665095.2018.1486220</u>

Roest, B., de Vries, S., de Schipper, M. and Aarninkhof, S., 2021. Observed Changes of a Mega Feeder Nourishment in a Coastal Cell: Five Years of Sand Engine Morphodynamics. *Journal of Marine Science and Engineering*, **9**(1), Article 37. <u>doi.org/10.3390/jmse9010037</u>

Rousseau, J-J., 1998. *The Social Contract.* (Fifth Edition). United Kingdom: Wordsworth Editions.

Royal Haskoning DHV, no date. *Digital twin provides resilience to coastal change*. [Online]. Available at: <u>https://www.royalhaskoningdhv.com/en/projects/digital-twin-provides-resilience-to-coastal-change</u> [accessed 24 October 2023].

Royal Haskoning DHV, 2017. *Norfolk Vanguard Coastal Erosion Study*. Peterborough: Haskoning DHV UK LTD.

Royal Haskoning DHV, 2018a. *Bacton Gas Terminal Coastal Defence Scheme: Scoping Report*. Peterborough: Haskoning DHV UK LTD.

Royal HaskoningDHV, 2018b. *Fairbourne Preliminary Coastal Adaptation Masterplan*. [Online]. Available at: <u>http://fairbourne.info/</u> [accessed 21 November 2023].

Royal Haskoning DHV, 2020. *Short Term Morphological Analysis*. Peterborough: Haskoning DHV UK LTD.

Royal Haskoning DHV, 2022a. *Morphological Analysis Period 3: Nov 2020 – September 2021.* Internal report to North Norfolk District Council. Peterborough: Haskoning DHV UK LTD.

Royal Haskoning DHV, 2022b. *Short Term Morphological Analysis: Updrift Coast Oct 2019 – Nov 2020.* Internal report to North Norfolk District Council. Peterborough: Haskoning DHV UK LTD.

Royal Haskoning DHV (2022c) *Royal Haskoning DHV*. Available at: <u>https://global.royalhaskoningdhv.com/</u> [accessed 08 August 2022].

Rozer, V. and Surminski, S., 2020. *New build homes, flood resilience and environmental justice – current and future trends under climate change across England and Wales.* Centre for Climate Change Economics and Policy Working Paper 381/Grantham Research Institute on Climate Change and the Environment Working Paper 353. London: London School of Economics and Political Science.

RSPB, 2023. *Wallasea Island*. [Online]. Available at: <u>https://www.rspb.org.uk/days-out/reserves/wallasea-island</u> [accessed 06 December 2023].

Rumson, A.G., Hallett, S.H. and Brewer, T.R., 2019. The application of data innovations to geomorphological impact analyses in coastal areas: An East Anglia, UK, case study. *Ocean and Coastal Management*, **181**(2019), pp.1-35. doi.org/10.1016/j.ocecoaman.2019.104875

Santoro, S., Pluchinotta, I., Pagano, A., Pengal, P., Cokan, B., Giordano, R. 2019. Assessing stakeholders' risk perception to promote Nature Based Solutions as flood protection strategies: The case of the Glinščica river (Slovenia). *Science of the Total Environment*, **665**(March 2019), pp.188-201. doi.org/10.1016/j.scitotenv.2018.11.116

Save Happisburgh Action Group, 2023. Save Happisburgh Action Group (SHAG) Facebook Page. [Online]. Available at:

https://www.facebook.com/groups/savehappisburghactiongroup/ [accessed 07 August 2023].

Sayers, P.B., Horritt, M.S., Penning-Rowsell, E. and McKenzie, A., 2017. *Climate Change Risk Assessment 2017 Projections of Future Flood Risk in the UK.* Watlington: Sayers and Partners.

Sayers, P.B., Penning-Rowsell, E. and Horritt, M., 2018. Flood vulnerability, risk, and social disadvantage: current and future patterns in the UK. *Regional Environmental Change*, **18**, pp.339–352. <u>doi.org/10.1007/s10113-017-1252-z</u>

Sayers, P., Moss, C., Carr, S. and Payo, A., 2022. Responding to climate change around England's coast – the scale of the transformational challenge. *Ocean and Coastal Management*, **225**(June 2022), Article 106187. doi.org/10.1016/j.ocecoaman.2022.106187

Schernewski G., Schumacher J., Weisner E. and Donges, L., 2018. A combined coastal protection, realignment and wetland restoration scheme in the southern Baltic: planning process, public information and participation. *Journal of Coastal Conservation*. **22**, pp.533–547. <u>doi.org/10.1007/s11852-017-0542-4</u>

Schlosberg, D., 2013. Theorising environmental justice: the expanding sphere of a discourse. *Environmental Politics*, **22**(1), pp.37-55. doi.org/10.1080/09644016.2013.755387

Schlosberg, D., 2020. Sustainable materialism and environmental justice. In Coolsaet, B., eds, *Environmental Justice: Key Issues*. Abingdon: Routledge, pp.303-315.

Secor, A., 2010. Chapter 12. Social Surveys, Interviews and Focus Groups. In Gomez, B. and Jones, J.P, eds. *Research Methods in Geography*. West Sussex, United Kingdom: Blackwell Publishing, pp.194-205.

Seddon, N., Chausson, A., Berry, P., Girardin, C.A.J., Smith, A. and Turner, B., 2020. Understanding the value and limits of nature-based solutions to climate change and other global challenges. *Philosophical Transactions of the Royal Society B.* **375**, Article 20190120. <u>doi.org/10.1098/rstb.2019.0120</u>

Seddon, N., Smith, A., Smith, P., Key, I., Chausson, A., Girardin, C., House, J., Srivastava, S. and Turner, B., 2021. Getting the message right on nature-based solutions to climate change. *Global Change Biology*, **27**(8), pp.1518-1546. doi.org/10.1111/gcb.15513

Sen, A.K., 2009. The idea of justice. Cambridge, Harvard University Press.

Sesana, E., Gagnon, A.S., Ciantelli, C., Cassar, J. and Hughes, J.J., 2021. Climate change impacts on cultural heritage: A literature review. *WIREs Climate Change*, **12**(4), Article e710. <u>doi.org/10.1002/wcc.710</u>

Sharifi, A., 2020. Trade-offs and conflicts between urban climate change mitigation and adaptation measures: A literature review. *Journal of Cleaner Production*, **276**(10 December 2013), Article 122813. doi.org/10.1016/j.jclepro.2020.122813

Shell, 2021. *About Bacton Gas Plant*. [Online]. Available at: <u>https://www.shell.co.uk/about-us/what-we-do/bacton-gas-plant/about-bacton-gas-plant.html</u> [accessed 16 August 2021].

Siders, A.R. and Ajibade, I., 2021. Introduction: Managed retreat and environmental justice in a changing climate. *Journal of Environmental Studies and Sciences*, **11**(2021), pp.287-293. <u>doi.org/10.1007/s13412-021-00700-6</u>

Sieber, J.E. and Tolich, M.B., 2012. *Planning Ethically Responsible Research*. California: SAGE Publications Inc.

Singh, C., Iyer, S., New, M. G., Few, R., Kuchimanchi, M., Segnon A. C. and Morchain, D., 2021. Interrogating 'effectiveness' in climate change adaptation: 11 guiding principles for adaptation research and practice. *Climate and Development*, **14**(7), pp.650–664. <u>doi.org/10.1080/17565529.2021.1964937</u>

Sjöberg L., 1999. Risk Perception by the Public and by Experts: A Dilemma in Risk Management. *Human Ecology Review.* **6**(2), pp.1–9. <u>doi.org/10.1027/1016-9040.3.1.1</u>

Sjöberg, L., 2000. Factors in risk perception. *Risk Analysis*, **20**(1), pp.1–11. doi.org/10.1111/0272-4332.00001

Slobodan, M. and Walvin, S., 2015. A comparative study of beach nourishment methods in selected areas of the coasts of the United Kingdom and The Netherlands. *Coastal Cities*, **148**(2015), pp. 85-96. <u>doi.org/10.2495/CC150081</u>

Smith, J., Blevins, B., Werse, N.R. and Talbert, S., 2021. Researcher Positionality in the Dissertation in Practice. In Throne, R., ed. *Practice-Based and Practice-Led Research for Dissertation Development*. Pennsylvania, United States: IGI Global, pp.43-63.

Soane, E., Schubert, I., Challenor, P., Lunn, R., Narendran, S. and Pollard, S., 2010. Flood Perception and Mitigation: The Role of Severity, Agency, and Experience in the Purchase of Flood Protection, and the Communication of Flood Information. *Environment and Planning A*, **42**, pp.3023-3038. doi.org/10.1068/a43238

Spalding, M.D., McIvor, A.L., Beck, M.W., Koch, E.W., Moller, I., Reed, D.J., Rubinoff, P., Spencer, T., Tolhurst, T., Wamsley, T.V., van Wesenbeeck, B., Wolanski, E. and Woodroffe, C.D., 2013. Coastal Ecosystems: A Critical Element of Risk Reduction. *Conservation Letters*, **7**(3), pp.293-301. <u>doi.org/10.1111/conl.12074</u>

Spence, A., Poortinga, W. and Pidgeon, N., 2012. The psychological distance of climate

change. *Risk Analysis*, **32**(6), pp.957–972. <u>doi.org/10.1111/j.1539-</u> 6924.2011.01695.x

Stallworthy, M., 2006. Sustainability, Coastal Erosion and Climate Change: An Environmental Justice Analysis. *Journal of Environmental Law*, **18**(3), pp. 357-373. <u>doi.org/10.1093/jel/eql017</u>

Stive, M., De Schipper, M., Luijendijk, A., Aarninkhof, S., van Gelder-Maas, C., van Thiel de Vries, J., de Vries, S., Henriquez, M., Marx, S. and Ranasinghe, R., 2013. A new alternative to saving our beaches from sea-level rise: The sand engine. *Journal of Coastal Research*, **29**(5), pp.1001-1008. doi.org/10.2112/JCOASTRES-D-13-00070.1

Sumner, A. and Tribe, M.A., 2008. *International development studies: theories and methods in research and practice*. London and California: SAGE Publications Inc.

Swanborn, P., 2010. *Case Study Research: What, Why and How?* London: SAGE Publications Inc.

Taal, M.D., Löffler, M., Vertegaal, C., Wijsman, J., van der Valk, L. and Tonnon, P., (Deltares) 2016. *Development of the Sand Motor. Concise report describing the first four years of the monitoring and evaluation programme*. Delft: Deltares.

Taylor, C., 1994. The politics of recognition. In Gutmannm A., ed, *Multiculturalism: Examining the Politics of Recognition*. Princeton: Princeton University Press, pp.24-73.

Tebboth, M., 2014. Understanding intractable environmental policy conflicts: the case of the village that would not fall quietly into the sea. *The Geographical Journal*, **180**(3), pp.224–235. <u>doi.org/10.1111/geoj.12040</u>

Thaler, T., Fuchs, S., Priest, S. and Doorn, N., 2017. Social justice in the context of adaptation to climate change—reflecting on different policy approaches to distribute and allocate flood risk management. *Regional Environmental Change*, **18**(2018), pp.305-309. <u>doi.org/10.1007/s10113-017-1272-8</u>

The Guardian, 2023. *Experience: I lost my home to the sea*. [Online]. Available at: <u>https://www.theguardian.com/lifeandstyle/2023/sep/01/experience-i-lost-my-home-to-the-sea</u> [accessed 14 December 2023].

The Maritime Executive, 2022. *Boskalis to Construct Sand Engine Concept for Togo and Benin Coastline*. [Online] Available at: <u>https://maritime-executive.com/corporate/boskalis-to-construct-sand-engine-concept-for-togo-and-benin-coastline</u> [accessed 06 December 2023].

Thomalla, F. and Vincent, C.E., 2003. Beach response to shore-parallel breakwaters at Sea Palling, Norfolk, UK. *Estuarine, Coastal and Shelf Science*, **56**(2), pp.203-212. <u>doi.org/10.1016/S0272-7714(02)00157-9</u>

Todd D. J. and Bowa, K. 2016. Development of beach health index for the gold coast, Australia. *Journal of Coastal Research*, **75**, Article 10075, pp.710–714. doi.org/10.2112/SI75-142.1

Tompkins, E. L. and W. N. Adger., 2004. Does adaptive management of natural resources enhance resilience to climate change? *Ecology and Society*, **9**(2), Article 10. <u>doi.org/10.5751/ES-00667-090210</u>.

Tompkins, E.L., Few, R. and Brown, K., 2008. Scenario-based stakeholder engagement: Incorporating stakeholders preferences into coastal planning for climate change. *Journal of Environmental Management*, **88**(4), pp.1580-1592. doi.org/10.1016/j.jenvman.2007.07.025

Tompkins, E.L., Adger, W.N., Boyd, E., Nicholson-Cole, S., Weatherhead, K. and Arnell, N., 2010. Observed adaptation to climate change: UK evidence of transition to a well-adapting society. *Global Environmental Change*, **20**(4), pp.627-635. <u>doi.org/10.1016/j.gloenvcha.2010.05.001</u>

Townend, I.H., French, J.R., Nicholls, R.J., Brown, S., Carpenter, S., Haigh, I.D., Hill, C.T., Lazarus, E., Penning-Rowsell, E.C., Thompson, C.E.L. and Tompkins, E.L., 2021. Operationalising Coastal Resilience to Flood and Erosion Hazard: A Demonstration for England. *Science of The Total Environment*, **783**, Article 146880, pp. 1-16. <u>doi.org/10.1016/j.scitotenv.2021.146880</u>

Tubridy, F., Lennon, M. and Scott, M., 2022. Managed retreat and coastal climate change adaptation: The environmental justice implications and value of a coproduction approach. *Land Use Policy*, **114**, (2022), Article 105960, <u>doi.org/10.1016/j.landusepol.2021.105960</u>

Turnhout, E., Metze, T., Wyborn, C., Klenk, N. and Louder, E., 2020. The politics of co-production: participation, power, and transformation. *Current Opinion in Environmental Sustainability*, **42**(February 2020), pp.15-21. doi.org/10.1016/j.cosust.2019.11.009

Twigger-Ross, C., Brooks, K., Papadopoulou, L., Orr, P., Sadauskis, R., Coke, A., Simcock, N., Stirling, A. and Walker, G., 2015. *Community resilience to climate change: an evidence review*. York: Joseph Rowntree Foundation.

Twigger-Ross, C., Sadauskis, R., Orr, P., Jones, R., McCarthy, S., Parker, D., Priest, S. and Simms, J. 2021. *Flood and coastal erosion risk management research and development framework: working with communities*. [Online]. Available at: <u>https://www.gov.uk/flood-and-coastal-erosion-risk-management-research-reports/flood-and-coastal-erosion-risk-management-research-and-development-framework-working-with-communities</u> [accessed 21 November 2023].

Tyler, T.R., 2000. Social Justice: Outcome and Procedure: *International Journal of Psychology*, **35**(2), pp.117-125. <u>doi.org/10.1080/002075900399411</u>

UK National Ecosystem Assessment, 2021. *Ecosystem Services*. [Online]. Available at: <u>http://uknea.unep-</u> wcmc.org/EcosystemAssessmentConcepts/EcosystemServices/tabid/103/Defaul

wcmc.org/EcosystemAssessmentConcepts/EcosystemServices/tabid/103/Defaul t.aspx [accessed 11 August 2021].

United Nations, 2015. *The Paris agreement*. [Online]. Available at: <u>https://unfccc.int/process-and-meetings/the-paris-agreement/the-paris-agreement</u> [accessed 08 August 2022].

Usher, L.E., 2021. Virginia and North Carolina surfers' perceptions of beach nourishment. *Ocean and Coastal Management*, **203**(1 April 2021), Article 105471. <u>doi.org/10.1016/j.ocecoaman.2020.105471</u>

Van Bergen, J. and Nijhuis, S., 2020. ShoreScape: Nature-Based Design for Urban Coastal Zones. In Hardiman, N. and Institution for Civil Engineers, eds.

*Coastal Management 2019: Joining forces to shape our future coasts.* [online]. Available from: <u>https://www-icevirtuallibrary-</u>com.uea.idm.oclc.org/doi/10.1680/cm.65147.605 [Accessed 07 June 2022].

Van der Plank, S., Brown, S., Nicholls, R.J., Tompkins, E. and Hardiman, N., 2020. Stakeholder expectations of the public in local coastal flood risk management in England. *Coastal Management 2019*. January 2020, pp. 605-618. doi.org/10.1680/cm.65147.605

Van der Plank, S., Brown, S., Tompkins, E.L. and Nicholls, R.J. 2022. A typology of responsibility for coastal flood risk adaptation. *Frontiers in Marine Science*. **9**(2022), doi.org/10.3389/fmars.2022.954950

Van Puijenbroek., 2019. Embryo dune development. In Luijendijk, A. and van Oudenhoven, A., eds. 2019. *The Sand Motor: A nature-based response to climate change*. Delft: Delft University Publishers. pp.93-96.

Vandebroek, E., Lindenbergh, R., Van Leijen, F., De Schipper, M., De Vries, S. and Hanssen, R., 2017. Semi-Automated Monitoring of a Mega-Scale Beach Nourishment Using High-Resolution TerraSAR-X Satellite Data. *Remote Sensing in Geology, Geomorphology and Hydrology*, **9**(7), Article 653. doi.org/10.3390/rs9070653

Vasseur, L., 2021. How Ecosystem-Based Adaptation to Climate Change Can Help Coastal Communities through a Participatory Approach. *Sustainability*, **13**(4), Article 2344. <u>doi.org/10.3390/su13042344</u>

Vecco, M. 2010. A definition of cultural heritage: From the tangible to the intangible. *Journal of Cultural Heritage*, **11**(3), pp.321-324. doi.org/10.1016/j.culher.2010.01.006

Vermaat, J., Bouwer, L., Turner, K. and Salomons, W., 2005. *Managing European coasts—Past, present and future*, Berlin: Springer.

Vikolainen, V., Filkweert, J., Bressers, H. and Lulofs, K., 2017. Governance context for coastal innovations in England: The case of sandscaping in North Norfolk. *Ocean and Coastal Management*, **145**(October), pp.82-93. doi.org/10.1016/j.ocecoaman.2017.05.012

Virapongse, A., Brooks, S., Metcalf, E.C., Zedalis, M., Gosz, J., Kliskey, A. and Alessa, L., 2016. A social-ecological systems approach for environmental management. *Journal of Environmental Management*, **178**(1 August 2016), pp.83-91. <u>doi.org/10.1016/j.jenvman.2016.02.028</u>

Vreugdenhil, H. and Slinger, J., 2023. Where space is created societal values are generated – the case of the Sand Engine. *International Journal of Water Governance*, **10**. doi.org/10.59490/ijwg.10.2023.6691

Walkden, M., Dickson, M., Thomas, J. and Hall, J.W., 2015. Simulating the Shore and Cliffs of North Norfolk. In Nicholls, R.J., Dawson, R.J., and Day, S.A., eds. *Broad Scale Coastal Simulation*. Dordrecht: Springer, pp.187-211.

Walker, B.H., Holling, C.S., Carpenter, S.R. and Kinzig, A., 2004. Resilience, adaptability and transformability in social–ecological systems. *Ecology and Society*, **9**(2), Article 5. <u>ecologyandsociety.org/vol9/iss2/art5/</u>

Walker, G., Mitchell, G., Fairburn, J. and Smith, G., 2005. Industrial pollution and social deprivation: Evidence and complexity in evaluating and responding to environmental inequality. *Local Environment*, **10**(4), pp.361-377. doi.org/10.1080/13549830500160842

Walker, G., 2009. Environmental Justice and Normative Thinking. *Antipode*, **41**(1), pp.203-205. <u>doi.org/10.1111/j.1467-8330.2008.00663.x</u>

Walker, G. and Burningham, K., 2011. Flood risk, vulnerability and environmental justice: Evidence and evaluation of inequality in a UK context. *Critical Social Policy*, **31**(2), pp.216-240. <u>doi.org/10.1177/0261018310396149</u>

Wamsler, C. 2017. Stakeholder involvement in strategic adaptation planning: Transdisciplinarity and co-production at stake? *Environmental Science and Policy*, **75**(2017), pp.148-157. <u>https://doi.org/10.1016/j.envsci.2017.03.016</u>

Warner B,P. and Kuzdas C., 2016. Manufactured global-change risk pathways in industrial-based agrarian development. *Climate and development*, **8**(5), pp.385-396. doi.org/10.1080/17565529.2015.1085359

Whitmarsh, L., 2008. Are flood victims more concerned about climate change than other people? The role of direct experience in risk perception and behavioural response. *Journal of Risk Research*, **11**(3), pp.351-374. doi.org/10.1080/13669870701552235

Wiltshire, G., 2018. A case for critical realism in the pursuit of interdisciplinarity and impact. *Qualitative Research in Sport, Exercise and Health*, **10**(5), pp.525-542. doi.org/10.1080/2159676X.2018.1467482

Wingfield, R., and Evans, C., 1998. The significance of the shoreface ramp for coastal development: Holderness, Eastern England, UK. *Proc Littoral '98 conference*. Spain, September 1998. Suport Serveis: Barcelona.

Wolf, J., Lorenzoni, I., Few, R., Abrahamson, V. and Raine, R., 2009. Conceptual and practical barriers to adaptation: vulnerability and responses to heat waves in the UK. In Adger, W.N., Lorenzoni, I., and O'Brien, K., eds. *Adapting to Climate Change. Thresholds, Values, Governance.* Cambridge: Cambridge University Press, pp.181-196.

Wolf, J., Allice, I. and Bell, T., 2013. Values, climate change, and implications for adaptation: evidence from two communities in Labrador, Canada. *Global Environmental Change*, **23**(2013), pp. 548-562. doi.org/10.1016/j.gloenvcha.2012.11.007

Woodworth, P.L., Teferle, R.M., Bingley, R.M., Shennan I. and Williams, S.D.P., 2009. Trends in UK mean sea level revisited. *Geophysics Journal International*, **176**(1), pp.19-30. <u>doi.org/10.1111/j.1365-246X.2008.03942.x</u>

Wong, P.P., Losada, I.J., Gattuso, J.P., Hinkel, J., Khattabi, A., McInnes, K.L., Saito, Y. and Sallenger, A., 2014. Coastal systems and low-lying areas. In Field, C.B., Barros, V.R., Dokken, D.J., Mach, K.J., Mastrandrea, M.D., Bilir, T.E., Chatterjee, M., Ebi, K.L., Estrada, Y.O., Genova, R.C., Girma, B., Kissel, E.S., Levy, A.N., MacCracken, S., Mastrandrea, P.R. and White L.L, eds. *Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and*  Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, USA, Cambridge University Press, pp. 361-409.

Young, A.W. 2018. How to retreat: The necessary transition from buyouts to leasing. *Coastal Management*, **46**(5), pp.527-535. doi.org/10.1080/08920753.2018.1498716

Zaalberg, R., Midden, C., Meijnders, A. and McCalley, T., 2009. Prevention, adaptation, and threat denial: Flooding experiences in the Netherlands. *Risk Analysis.* **29**(12), pp.1759–78. <u>doi.org/10.1111/j.1539-6924.2009.01316.x</u>

Zhou, G. and Xie, M., 2009. Coastal 3-D Morphological Change Analysis Using LiDAR Series Data: A Case Study of Assateague Island National Seashore. *Journal of Coastal Research*, **25**(2), pp.435-447. <u>jstor.org/stable/27698335</u>